

## CONTAINING ALL NOTICES PUBLISHED ONLINE ON 31 AUGUST 2023

PRINTED ON 1 SEPTEMBER 2023 | NUMBER 64160 PUBLISHED BY AUTHORITY | ESTABLISHED 1665 WWW.THEGAZETTE.CO.UK

### Contents

### State/17458\*

Royal family/ Parliament Assemblies & Government/ Honours & Awards/ Church/

Environment & infrastructure/17459\* Health & medicine/

Other Notices/17465\*

Money/

Companies/17485\* People/17545\*

Terms & Conditions/17569\*

\* Containing all notices published online on 31 August 2023

## STATE

#### STATE APPOINTMENTS

#### THREE HUNDREDS OF CHILTERN

The Chancellor of the Exchequer has this day appointed Nadine Vanessa Dorries to be Steward and Bailiff of the Three Hundreds of Chiltern.

29 August 2023

(4431856)

# ENVIRONMENT & INFRASTRUCTURE

## Planning

#### **TOWN PLANNING**

#### DEPARTMENT FOR TRANSPORT TOWN AND COUNTRY PLANNING ACT 1990

The Secretary of State gives notice of the proposal to make an Order under section 247 of the above Act to authorise the stopping up of two northern part widths of Park Grove at Keighley, in the Metropolitan Borough of Bradford.

If made, the Order would authorise the stopping up only to enable development as permitted by City of Bradford Metropolitan District Council, under reference 19/03696/FUL.

Copies of the draft Order and relevant plan will be available for inspection during normal opening hours at Keighley Civic Centre, North Street, Keighley, BD21 3RZ, in the 28 days commencing on 31 August 2023, and may be obtained, free of charge, from the addresses stated below quoting NATTRAN/Y&H/S247/5473.

Any person may object to the making of the proposed order by stating their reasons in writing to the Secretary of State at nationalcasework@dft.gov.uk or National Transport Casework Team, Tyneside House, Skinnerburn Road, Newcastle upon Tyne NE4 7AR, quoting the above reference. Objections should be received by midnight on **28 September 2023**. You are advised that your personal data and correspondence will be passed to the applicant/agent to enable your objection to be considered. If you do not wish your personal data to be forwarded, please state your reasons when submitting your objection.

S Zamenzadeh, Casework Manager

(4431858)

#### DEPARTMENT FOR TRANSPORT TOWN AND COUNTRY PLANNING ACT 1990

The Secretary of State gives notice of the proposal to make an Order under section 247 of the above Act to authorise the stopping up of a length of Keckwick Lane at Daresbury in the Borough of Halton.

If made, the Order would authorise the stopping up only to enable development as permitted by Halton Borough Council, under reference 17/00406/FULEIA.

Copies of the draft Order and relevant plan will be available for inspection during normal opening hours at Runcorn Library, Granville Street, Runcorn, Cheshire, WA7 1NE, in the 28 days commencing on 31 August 2023, and may be obtained, free of charge, from the addresses stated below quoting NATTRAN/NW/S247/5489.

Any person may object to the making of the proposed order by stating their reasons in writing to the Secretary of State at nationalcasework@dft.gov.uk or National Transport Casework Team, Tyneside House, Skinnerburn Road, Newcastle upon Tyne NE4 7AR, quoting the above reference. Objections should be received by midnight on **28 September 2023**. You are advised that your personal data and correspondence will be passed to the applicant/agent to enable your objection to be considered. If you do not wish your personal data to be forwarded, please state your reasons when submitting your objection.

C Newton, Casework Manager

(4431860)

#### DEPARTMENT FOR TRANSPORT TOWN AND COUNTRY PLANNING ACT 1990

The Secretary of State gives notice of the proposal to make an Order under section 247 of the above Act to authorise the stopping up of an area of unnamed highway which lies off Henley Road at Hampton-On-The-Hill in the District of Warwick.

If made, the Order would authorise the stopping up only to enable development as permitted by Warwick District Council, under reference W/23/0519.

Copies of the draft Order and relevant plan will be available for inspection during normal opening hours at Warwick Library, Shire Hall, Warwick, CV34 4RL in the 28 days commencing on 31 August 2023, and may be obtained, free of charge, from the addresses stated below quoting NATTRAN/WM/S247/5524.

Any person may object to the making of the proposed order by stating their reasons in writing to the Secretary of State at nationalcasework@dft.gov.uk or National Transport Casework Team, Tyneside House, Skinnerburn Road, Newcastle upon Tyne NE4 7AR, quoting the above reference. Objections should be received by midnight on **28 September 2023.** You are advised that your personal data and correspondence will be passed to the applicant/agent to enable your objection to be considered. If you do not wish your personal data to be forwarded, please state your reasons when submitting your objection.

S Zamenzadeh, Casework Manager

(4431861)

## **Property & land**

#### PROPERTY DISCLAIMERS

T S Ref: BV22207107/4/MPC

## NOTICE OF DISCLAIMER UNDER SECTION 1013 OF THE COMPANIES ACT 2006

**DISCLAIMER OF WHOLE OF THE PROPERTY** 1. In this Notice the following shall apply:

Company Name: MADSTON DEVELOPMENTS LIMITED Company Number: 04872685

Property: Any interest in the rent charge created by a Transfer dated 24 March 2009 in respect of the property at 132 Church Street, Gainsborough DN21 2JU registered under title number LL306227 and any other right or interest whatsoever arising in or from the said Transfer dated 24 March 2009

The Solicitor for the Affairs of His Majesty's Treasury of HMRC Croydon Regional Centre, 1 Ruskin Square, Croydon CR0 2WF (DX 325801 Croydon 51).

2. In pursuance of the powers granted by s.1013 of the Companies Act 2006 the Treasury Solicitor as nominee for the Crown (in whom the property and rights of the company vested when the Company was dissolved) hereby disclaims the Crown's title (if any) in the Property the vesting of the Property having come to her notice on 27 January 2023.

Assistant Treasury Solicitor

(Section 3 Treasury Solicitor Act 1876) Dated 23 August 2023

(4431885)

T S Ref: BV22207107/4/MPC

#### NOTICE OF DISCLAIMER UNDER SECTION 1013 OF THE COMPANIES ACT 2006 DISCLAIMER OF WHOLE OF THE PROPERTY

1. In this Notice the following shall apply:

Company Name: MADSTON DEVELOPMENTS LIMITED Company Number: 04872685

Property: Any interest in the Restriction in favour of Madston Developments Limited at entry B3 of the register of title (edition date 26.05.2021) of the property at 132 Church Street, Gainsborough DN21 2JU registered under title number LL306227

Treasury Solicitor: The Solicitor for the Affairs of His Majesty's Treasury of HMRC Croydon Regional Centre, 1 Ruskin Square, Croydon CR0 2WF (DX 325801 Croydon 51).

2. In pursuance of the powers granted by s.1013 of the Companies Act 2006 the Treasury Solicitor as nominee for the Crown (in whom the property and rights of the company vested when the Company was dissolved) hereby disclaims the Crown's title (if any) in the Property the vesting of the Property having come to her notice on 27 January 2023.

Assistant Treasury Solicitor

(Section 3 Treasury Solicitor Act 1876) Dated 23 August 2023

(4431886)

### **Roads & highways**

#### **ROAD RESTRICTIONS**

#### **BURY COUNCIL**

METROPOLITAN BOROUGH OF BURY ROAD TRAFFIC REGULATION ACT 1984, SECTION 14(1) THE METROPOLITAN BOROUGH OF BURY (NUTTALL LANE/ NUTTALL ROAD, RAMSBOTTOM) TEMPORARY CLOSURE ORDER 2023

NOTICE IS HEREBY GIVEN that the Council of the Metropolitan Borough of Bury have made an Order under Section 14(1) of the Road Traffic Regulation Act 1984 and of all other enabling powers, the effect of which will be to prohibit any vehicles, pedestrians, cyclists and horse riders from proceeding along:-

Nuttall Lane, Ramsbottom - between Whittingham Drive & Nuttall Road

Nuttall Road, Ramsbottom – full length.

The reason for the proposed order is to facilitate proposed works including highway modifications associated with a stormwater storage scheme.

There is no suitable alternative route for vehicles although emergency service access will be maintained.

The alternative route for pedestrians will be via Whittingham Drive, Falstone Avenue, Footpath 83 Ramsbottom & vice versa.

Via Whittingham Drive, Falstone Avenue, Footpath 83, Ramsbottom, informal path between Footpath 83 & Footpath 100 & the informal path adjacent to the river & vice versa.

The Order comes into operation on 04/09/2023. The Council is satisfied that the execution of the works will take longer than 18 months. The anticipated duration of the works is until January 2026. The temporary order will be revoked on completion of the works.

Dated 31/08/2023

Council Solicitor

TOWN HALL, BURY, LANCASHIRE BL9 0SW Tel: 0161-253-5000

(4431937)

#### CROYDON COUNCIL INTRODUCTION OF WAITING RESTRICTIONS IN VARIOUS LOCATIONS IN CROYDON

THE CROYDON (PROHIBITION AND RESTRICTION OF STOPPING, LOADING AND WAITING) AND (FREE PARKING PLACES) (NO.18) ORDER 2023

## THE CROYDON (ON-STREET CHARGED-FOR PARKING PLACES) (NO.18) ORDER 2023

1. NOTICE IS HEREBY GIVEN that Croydon Council, on 29 August 2023, made the above Traffic Orders under the relevant sections of the Road Traffic Regulation Act 1984, as amended, and all other enabling powers.

2. The general effect of the Orders will be:

a) To introduce/amend existing "at any time" double yellow line waiting restrictions in the lengths of streets specified in Schedule 1 to this Notice.

b) To remove a section of "at any time" loading restrictions in the length of street specified in Schedule 2 to this Notice.

c) To update the map-based tiles to the Schedule attached to the above Orders to accurately represent the new arrangements as they exist on site.

3. Copies of the Orders, which will come into operation on 04 September 2023, and of all associated Orders can be inspected by appointment with the Parking Design Team from 9am to 4pm on Mondays to Fridays inclusive until the end of six weeks from the date on which the Orders were made, at the "Access Croydon" Facility, 8 Mint Walk, Croydon, CR0 1EA (telephone 020 8726 6000 and ask for Parking Design).

4. Alternatively, copies of the Orders can be requested via email or post from the Parking Design Team. Email: parking.design@croydon.gov.uk. Postal address: Parking Design Team, 3rd Floor, Zone C, Bernard Weatherill House, 8 Mint Walk, Croydon CR0 1EA. 5. Any person desiring to question the validity of the Orders or of any provision contained therein on the ground that it is not within the relevant powers of the Road Traffic Regulation Act, 1984 or that any of the relevant requirements thereof or of any relevant regulations made there under have not been complied with in relation to the Orders may, within six weeks of the date on which the Orders were made, make application for the purpose to the High Court.

Dated 31 August 2023

Hilary Dickson Road Space Manager

Sustainable Communities, Regeneration & Economic Recovery

Directorate

SCHEDULE 1

## (Introducing new/amending existing double yellow line waiting restrictions "at any time")

#### Abingdon Road, Norbury

The west side, either side of its junction with Turle Road for 7 metres. Aveling Close, Purley

The north-west side, from a point opposite the north-eastern boundary of No. 80 Aveling Close for 15 metres north-eastwards (measured in a straight line). The restrictions will follow the quadrant shaped kerb-lines of the turning head.

#### Callow Field, Purley

Both sides, from its junction with Higher Drive for 10 metres southwestwards.

#### **Ecclesbourne Road, Thornton Heath**

The north-west side:

• From the north-eastern boundary of Bensham Manor School for 17 metres south-westwards.

• From the south-western boundary of Bensham Manor School for 17 metres north-eastwards.

#### Furze Road, Thornton Heath

Both sides, from its junction with Heath Road for 8 metres northeastwards.

#### Galpin's Road, Thornton Heath

The north-west side, for 7 metres either side of its junction with Lonsdale Gardens.

#### Heath Road, Thornton Heath

The north-west side, for 8 metres either side of its junction with Furze Road.

## Hermitage Road, Upper Norwood (The arm fronting property Nos. 77 to 97)

From the north-western extremity of the road to a point opposite the common boundary of Nos. 85 and 87 (on the south-west side) and a point opposite the north-western flank wall of No. 85 (on the north-east side) including the turning head.

#### Higher Drive, Purley

The south-west side, for 10 metres either side of its junction with Callow Field.

#### Kitchener Road, Thornton Heath

The north-west side, either side of its junction with Milner Road for 10 metres.

#### Little Roke Road, Kenley

• Both sides, from its junction with Godstone Road for 10 metres south-westwards.

• The north-west side, either side of its junction with Lower Road for 7 metres.

#### Lonsdale Gardens, Thornton Heath

Both sides, from its junction with Galpin's Road for 7 metres north-westwards.

#### Lower Road, Kenley

Both sides, from its junction with Little Roke Road for 10 metres north-westwards.

#### Milner Road, Thornton Heath

Both sides, from its junction with Kitchener Road for 10 metres north-westwards.

#### Montpelier Road, Purley

• The north-west, north-east, and south-east side, following the bend from a point opposite the common boundary of Nos. 96 and 98 Montpelier Road, to a point 9 metres north-east of its junction with Kingsdown Avenue (the section fronting Nos. 145-75 Kingsdown Avenue).

• The south-east side (the arm fronting property Nos. 10 to 134) reducing the length of existing double yellow lines by 15.5 metres at the south-western end, so that they extend for 40.5 metres south-westwards from the common boundary of Nos. 84 and 86 Montpelier Road.

• The north-west and south-west side, on the bend adjacent to No. 73 Montpelier Road, reducing the existing restrictions by 15.5 metres at their south-western end and by 4.5 metres at their north-western end, so that they extend for 18.4 metres in total (or 9.2 metres either side of the apex).

#### Norman Road, Thornton Heath

Both sides, from its junction with Penshurst Road for 7 metres southeastwards.

#### Osterley Gardens, Thornton Heath

• The north-eastern extremity of the road (adjacent to the rear boundaries of Nos. 52 to 56 Springfield Road) from a point 3.8 metres north-east of the common boundary of Nos. 18 and 20 Osterley Gardens to a point 1.6 metres north-east of the common boundary of Nos. 17 and 19 Osterley Gardens.

• The carriageway surrounding the island site opposite No. 15 Osterley Gardens, excluding the existing disabled parking bay.

#### Parkside Gardens, Coulsdon

Both sides, from its junction with Rickman Hill for 10 metres northwestwards.

#### Penshurst Road, Thornton Heath

• Both sides, from its junction with Brigstock Road for 10 metres south-eastwards.

• The south-east side, either side of its junction with Norman Road for 7 metres.

• The south-west and north-west side, on the corner adjacent to No. 39, for 5 metres either side of the apex.

• The north-east and south-east side, on the corner adjacent to No. 80, for 5 metres either side of the apex.

#### **Raymead Avenue, Thornton Heath**

The west and north-west side, on the bend adjacent to No. 62, from the common boundary of Nos. 60 and 62 to a point opposite the common boundary of Nos. 65 and 67.

#### **Richmond Road, Coulsdon**

Both sides, from its junction with St. Andrew's Road for 10 metres north-westwards.

#### **Rickman Hill, Coulsdon**

The north-west side, either side of its junction with Parkside Gardens for 10 metres.

#### St. Andrew's Road, Coulsdon

The north-west side, either side of its junction with Richmond Road for 10 metres.

#### Station Approach Road, Coulsdon

The north-east side, extending the existing double yellow lines at its junction with Brighton Road for 55 metres south-eastwards.

#### Trojan Way, Croydon

Both sides, amending the existing single yellow lines to double yellow lines.

#### **Turle Road, Norbury**

Both sides, from its junction with Abingdon Road for 7 metres westwards.

#### **SCHEDULE 2**

(Removal of existing "at any time" loading restrictions) **Dingwall Avenue, Croydon** 

The south-east side, removing a section of loading restrictions extending from the south-western kerb-line of the vehicular entrance to Electric House, No. 3 Wellesley Road, for 15 metres south-(4431936) westwards.

#### LONDON BOROUGH OF MERTON

#### PROPOSED SINUSOIDAL HUMPS - ST JAMES ROAD, MITCHAM

1. Notice is hereby given that the Council of the London Borough of Merton, under powers conferred by section 90A-F of the Highways Act 1980 and after consultation with the Commissioner of Police of the Metropolis, intend to introduce 6 sinusoidal humps, each 3.7 metres in length, with a maximum height of 75mm high across the full width of St James Road, Mitcham at the locations specified in the Schedule to this notice.

2. A plan of the area showing the proposed sinusoidal humps can be inspected during normal office hours on Mondays to Fridays at Merton Link, Merton Civic Centre, London Road, Morden, Surrey.

3. Any person desiring to comment on the proposal should send a statement in writing of their representations or objections and the grounds thereof in writing to the Environment and Regeneration Department, Merton Civic Centre, London Road, Morden, Surrey, SM4 5DX or alternatively bv email to trafficandhighways@merton.gov.uk quoting reference ES/RH/ StJames 2023 no later than 22 September 2023.

Dated 31 August 2023 Paul McGarry

#### Head of futureMerton

SCHEDULE

1. Outside Nos. 102/104 St James Road

2. Outside No. 82 St James Road

3. Outside No. 64 St James Road

- 4. Outside No. 46 St James Road
- 5. Outside No. 30 St James Road 6. Outside No. 10 St James Road

(4431857)

#### LONDON BOROUGH OF MERTON PROPOSED SINUSOIDAL HUMPS - FRAMFIELD ROAD, МІТСНАМ

1. Notice is hereby given that the Council of the London Borough of Merton, under powers conferred by section 90A-F of the Highways Act 1980 and after consultation with the Commissioner of Police of the Metropolis, intend to introduce 5 sinusoidal humps, each 3.7 metres in length, with a maximum height of 75mm high across the full width of Framfield Road, Mitcham at the locations specified in the Schedule to this notice.

2. A plan of the area showing the proposed sinusoidal humps can be inspected during normal office hours on Mondays to Fridays at Merton Link, Merton Civic Centre, London Road, Morden, Surrey.

3. Any person desiring to comment on the proposal should send a statement in writing of their representations or objections and the grounds thereof in writing to the Environment and Regeneration Department, Merton Civic Centre, London Road, Morden, Surrey, SM4 5DX or alternatively by email trafficandhighways@merton.gov.uk quoting reference ES/RH/

Framfield 2023 no later than 22 September 2023. Dated 31 August 2023

Dated 31 August 2023

Paul McGarry

Head of futureMerton SCHEDULE

- 1. Outside No. 86 Framfield Road
- 2. Outside No. 64 Framfield Road

3. Outside No. 40 Framfield Road

4. Outside No. 16 Framfield Road

5. Framfield Road, 21.5 metres north-west of the north-western kerbline of Streatham Road (4431859)

#### LONDON BOROUGH OF MERTON

AMENDMENTS TO CONTROLLED PARKING ZONES 5F, GC1, M2, P2(S), RP AND VSW

PROPOSED CONVERSION OF PARKING PLACES TO WAITING RESTRICTIONS IN VARIOUS ROADS

THE MERTON (PARKING PLACES) (GC1) (AMENDMENT NO. \*) **ORDER 202\*** 

THE MERTON (PARKING PLACES) (M1 AND M2) (AMENDMENT NO. \*) ORDER 202\*

THE MERTON (PARKING PLACES) (RAYNES PARK)

(AMENDMENT NO. \*) ORDER 202\*

THE MERTON (PARKING PLACES) (VSW) (AMENDMENT NO. \*) **ORDER 202\*** 

THE MERTON (PARKING PLACES) (WIMBLEDON PARK) (AMENDMENT NO. \*) ORDER 202\*

THE MERTON (PARKING PLACES) (W5 AND 5F) (AMENDMENT NO. \*) ORDER 202\*

1. Further to a notice dated 24 August 2023, NOTICE IS HEREBY GIVEN that the Council of the London Borough of Merton under the powers conferred by sections 45, 46, 49 and 124 of and Part IV of Schedule 9 to the Road Traffic Regulation Act 1984, as amended, propose to make the above-mentioned Orders.

2. The general effect of the Orders would be to convert sections of existing parking places in lengths of road generally fronting the locations referred to in the Schedule to this notice, to single yellow line waiting restrictions matching the operational times of the controlled parking zone operating at each of those locations, to facilitate proposed vehicle crossovers.

3. A copy of each of the proposed Orders and other documents giving more detailed particulars of the Orders, including plans which indicate the roads or lengths of roads to which the Orders relate, can be inspected Mondays to Fridays during normal office hours at Merton Link, Merton Civic Centre, London Road, Morden. Documents are also available for inspection at https://www.merton.gov.uk/streetsparking-transport/traffic-management/consultations.

4. Any person desiring to comment on the proposed Orders should send a statement in writing of their representations or objections and the grounds thereof, to the Environment and Regeneration Department at the Merton Civic Centre, London Road, Morden, Surrev. SM4 5DX. or alternatively email by to trafficandhighways@merton.gov.uk quoting reference ES/ CROSSOVER28 no later than 22 September 2023.

Dated 31 August 2023 Paul McGarry Head of futureMerton London Borough of Merton Merton Civic Centre, London Road Morden SM4 5DX SCHEDULE

Road, SM4 (CPZ M2) O/s No. 43 Bayham Road, SM4 (CPZ M2) O/s No. 28 Buckfast Road, SM4 (CPZ M2) O/s No. 41 Camberley Avenue, SW20 (CPZ RP) O/S No. 8 Chester Road, SW19 (CPZ VSW)

Outside (o/s) No. 175 Abbotsbury O/s No. 15 Hazelwood Avenue, SM4 (CPZ M2) O/s No. 42 Hazelwood Avenue, SM4 (CPZ M2) O/s No. 82 Kenilworth Avenue, SW19 (CPZ P2(s)) O/s No. 42 Kingswood Road, SW19 (CPZ 5F) O/s No. 21B Rural Way, SW16 (CPZ GC1)

(4431940)

#### LONDON BOROUGH OF MERTON

PARKING PLACES FOR THE LAMP POST CHARGING OF **ELECTRIC VEHICLES** 

THE MERTON (PARKING PLACES) (ELECTRIC VEHICLE

CHARGING POINTS) (NO. 5) ORDER 2023

THE MERTON (PARKING PLACES) (3E) (AMENDMENT NO. 2) **ORDER 2023** 

THE MERTON (PARKING PLACES) (CW) (AMENDMENT NO. 2) **ORDER 2023** 

THE MERTON (PARKING PLACES) (CW1) (AMENDMENT NO. 1) **ORDER 2023** 

THE MERTON (PARKING PLACES) (CW2) (AMENDMENT NO. 1) **ORDER 2023** 

THE MERTON (PARKING PLACES) (CW3) (AMENDMENT NO. 2) **ORDER 2023** 

1. NOTICE IS HEREBY GIVEN that the Council of the London Borough of Merton, on 30 August 2023, made the above-mentioned Orders under sections 6, 45, 46, 47 and 124 of, and Part IV of Schedule 9 to, the Road Traffic Regulation Act 1984, as amended and all other enabling powers.

2. The general effect of the 'Electric Vehicle Charging Points' Order will be to: -

(a) designate parking places for electric vehicle charging at the locations listed in the Schedule to this notice:

(b) provide that the parking places will operate at all times, unless otherwise stated;

(c) specify that the parking places in controlled parking zones referred to in the Schedule to this notice will be available for electric or plug-in hybrid electric vehicles only, for up to 12 hours, provided that during the controlled parking zone operational hours the applicable charge for the relevant controlled parking zone permit has been paid with respect to the vehicle and that any electric vehicle is connected via a charging lead to a lamp column upgraded to enable electric charging at all times while it is left in that parking place. Electric vehicles having left a parking place for electric vehicle charging will be able to return after 4 hours have elapsed.

3. The general effect of the 'Parking Places Amendment' Orders will be to remove an existing length of parking place in the lengths of road listed in the Schedule to this notice to accommodate the introduction of a parking place designated for electric vehicle charging.

4. A copy of each of the Orders, which will come into force on 11 September 2023, and other documents giving more detailed particulars of the Orders, including plans which indicate the roads or lengths of road to which the Orders relate, can be inspected during normal office hours Mondays to Fridays, until the end of six weeks from the date on which the Orders were made, at Merton Link, Merton Civic Centre, London Road, Morden.

5. Any person desiring to question the validity of the Orders or of any provision contained therein on the grounds that it is not within the relevant powers of the Road Traffic Regulation Act 1984, or that any of the relevant requirements thereof, or of any relevant regulations made thereunder has not been complied with in relation to the Orders may, within six weeks from the date on which the Orders were made, make application for the purpose to the High Court.

Dated 30 August 2023

Paul McGarry

Head of futureMerton

London Borough of Merton,

Merton Civic Centre,

London Road, Morden SM4 5DX

SCHEDULE

#### Electric Vehicle Charging Parking Places in Controlled Parking Zones - CPZ Permit Required

CPZ 3E

ASHLEY ROAD, SW19; the west side, from a point 11.8 metres north of the northern kerb-line of Florence Road northward for a distance of 6 metres

CLARENCE ROAD, SW19; the north side, from a point 6 metres east of the eastern kerb-line of Ashley Road eastward for a distance of 5.5 metres

CRAVEN GARDENS, SW19; the east side, from a point 7 metres south of the southern kerb-line of Anchorage Close southward for a distance of 6 metres.

CRAVEN GARDENS, SW19; the east side, from a point 1 metre north of the common boundary of Nos. 89 and 90 Craven Gardens northward for a distance of 4.9 metres.

FARADAY ROAD, SW19; the south side, from a point 0.4 metres east of a point in line with the eastern flank wall of No. 85 Faraday Road westward for a distance of 5.6 metres.

#### **CPZ CW**

BYEGROVE ROAD, SW19; the south-west side, from a point 3 metres north-west of the common boundary of Wandle Park and No. 23 Byegrove Road north-westward for a distance of 5 metres.

CAVENDISH ROAD, SW19: the north side, from a point 3.5 metres east of a point perpendicular to the common boundary of Nos. 1a and 1c Park Road eastward for a distance of 5.5 metres.

COLLEGE ROAD, SW19; the east side, from a point 6.9 metres south of the southern kerb-line of Denison Road southward for a distance of 5.5 metres.

DENISON ROAD, SW19; the north-west side, form a point 4.5 metres south-west of the common boundary of No. 1A Miller Road and No. 26 Denison Road south-westward for a distance of 6 metres.

DEVONSHIRE ROAD, SW19; the north-west side, from a point 3.7 metres south-west of the common boundary of Nos. 72 and 74 Devonshire Road north-eastward for a distance of 6 metres.

NORFOLK ROAD, SW19; the north-west side, from a point 8 metres south-west of the south-western kerb-line of Robinson Road southwestward for a distance of 5.5 metres.

ROBINSON ROAD, SW17; the south-west side, from a point 1.4 metres north-west of the common boundary of Nos. 112 and 114 Robinson Road south-eastward for a distance of 6 metres.

SINGLETON CLOSE, SW17; the south-west to north-east arm, the north-west side, from a point 2.1 metres north-east of the southwestern kerb-line of Singleton Close, the south-east to north-west arm, north-eastward for a distance of 5.5 metres.

#### CPZ CW1

BYEGROVE ROAD, SW19; the north side, from a point 2 metres east of a point in line with the westernmost flank wall of Kendall Court, Byegrove Road eastward for a distance of 5.5 metres. CPZ CW2

VALLEY GARDENS, SW19; the west side, from a point 8.1 metres north of the northern kerb-line of South Gardens northward for a distance of 5.5 metres.

#### CPZ CW3

MYRNA CLOSE, SW19; the south side, from a point 4 metres west of the common boundary of Nos. 5 and 7 Myrna Close westward for a distance of 6 metres. (4431942)

#### LONDON BOROUGH OF SOUTHWARK WESTMORELAND ROAD – PERMANENT ROAD CLOSURE THE LONDON BOROUGH OF SOUTHWARK (PRESCRIBED ROUTES) (WESTMORELAND ROAD) TRAFFIC ORDER 2023

1. Southwark Council hereby GIVES NOTICE that on 31 August 2023 it has made the above Order under sections 6 and 124 of and Part IV of Schedule 9 to the Road Traffic Regulation Act 1984, as amended.

2. The effects of the Order, the intention of which is to make permanent the trialled provision of prescribed routes in the Westmoreland Road area to alleviate traffic congestion, are:-

(a) to provide a 'no motor vehicles' point road closure prohibiting the movement of motor vehicles, which will be indicated by traffic signs and enforced by ANPR cameras, in WESTMORELAND ROAD southeast of its junction with Queen's Row;

(b) to introduce a compulsory left turn for all motor vehicles in QUEEN'S ROW at its junction with Westmoreland Road; and

(c) to remove an existing point road closure in PHELP STREET (northwest of its junction with Westmoreland Road) and the prescribed routes associated with this point closure in Westmoreland Road.

NOTES: (1) These Orders supersede similar provisions at the above locations implemented by way of experimental Orders made under section 9 of the 1984 Act (which are revoked). (2) Exemption to the restrictions described in (a) and (b) preceding are provided for emergency services vehicles and for vehicles used for road maintenance, refuse collection or cleansing purposes; and exceptions are provided for postal service, taxis and (on a discretionary basis) disabled persons vehicles (registered to Southwark resident 'blue badge' holders), rapid response healthcare workers vehicles and SEND vehicles.

3. Copies of the Order, which will come into force on 4 September 2023, and of all other relevant documents are available for inspection at Highways, Southwark Council, Environment, Neighbourhoods and Growth, 3rd floor hub 2, 160 Tooley Street, London SE1 2QH, by appointment only. E-mail traffic.orders@southwark.gov.uk or call 020 7525 3497 for booking details.

4. Any person desiring to question the validity of the Order or of any provision contained therein on the grounds that it is not within the relevant powers of the Road Traffic Regulation Act 1984 or that any of the relevant requirements thereof or of any relevant regulations made thereunder has not been complied with in relation to the Order may, within six weeks of the date on which the Order was made, make application for the purpose to the High Court.

Dated 31 August 2023

Dale Foden - Head of Service, Highways

(4431938)

#### NATIONAL HIGHWAYS LIMITED ROAD TRAFFIC REGULATION ACT 1984 THE A5 TRUNK ROAD (HEATHENCOTE, WEST NORTHAMPTONSHIRE) (PROHIBITION OF RIGHT TURNS) ORDER 2023

National Highways Limited (Company no.9346363) hereby gives notice that it has make an Order under sections 1(1), 2(1) and (2) of the Road Traffic Regulation Act 1984 on the A5 Trunk Road at Heathencote in the County of West Northamptonshire.

The effect of the Order, which comes into force on Monday 4 September 2023, is to prohibit vehicles from making a right turn from the unnamed road leading to and from Shutlanger onto the northbound carriageway of the A5, at Heathencote.

A copy of the Order, together with a plan and a Statement of the reasons for making the Order, may be inspected during opening hours at Towcester Library, Moat Lane, Towcester NN12 6AD; West Northamptonshire Council, The Forum, Moat Lane, Towcester NN12 6AD, or may be may be obtained by emailing ordersbirmingham@nationalhighways.co.uk. Any person wishing to question the validity of, or of any provision contained in, the Order, on the ground that it is not within the powers with respect to the Order conferred by the above Act, or on the ground that any requirement of, or of any instrument made under, any provision of that Act has not been complied with in relation to the Order may, within 6 weeks of 1 September 2023, apply to the High Court for the suspension or quashing of the Order or of any provision contained therein.

For further information contact Richard Cullis on 07523 941 546.

Karen Eustace, an Officer in National Highways Limited

National Highways Limited (Company no. 9346363) registered office: Bridge House, Walnut Tree Close, Guildford, GU1 4LZ. A Company registered in England and Wales. (4431941)

#### **CROYDON COUNCIL**

#### WAITING RESTRICTIONS IN VARIOUS LOCATIONS IN CROYDON THE CROYDON (PROHIBITION AND RESTRICTION OF STOPPING, LOADING AND WAITING) AND (FREE PARKING PLACES) (NO.19) ORDER 2023 THE CROYDON (ON-STREET CHARGED-FOR PARKING PLACES)

## (NO.19) ORDER 2023

1. NOTICE IS HEREBY GIVEN that Croydon Council on 29 August 2023 made the above Traffic Orders under the relevant sections of the Road Traffic Regulation Act 1984, as amended, and all other enabling powers.

2. The general effect of the Orders will be:

a) To introduce "at any time" double yellow line waiting restrictions in the lengths of streets specified in the Schedule to this Notice.

b) To update the map-based tiles to the Schedule attached to the above Orders to accurately represent the new arrangements as they exist on site.

3. Copies of the Orders, which will come into operation on 04 September 2023, and of all associated Orders can be inspected by appointment with the Parking Design Team from 9am to 4pm on Mondays to Fridays inclusive until the end of six weeks from the date on which the Orders were made, at the "Access Croydon" Facility, 8 Mint Walk, Croydon, CR0 1EA (telephone 020 8726 6000 and ask for Parking Design).

4. Alternatively, copies of the Orders can be requested via email or post from the Parking Design Team. Email: parking.design@croydon.gov.uk. Postal address: Parking Design Team, 3rd Floor, Zone C, Bernard Weatherill House, 8 Mint Walk, Croydon CR0 1EA.

5. Any person desiring to question the validity of the Orders or of any provision contained therein on the ground that it is not within the relevant powers of the Road Traffic Regulation Act, 1984 or that any of the relevant requirements thereof or of any relevant regulations made there under have not been complied with in relation to the Orders may, within six weeks of the date on which the Orders were made, make application for the purpose to the High Court.

Dated 31 August 2023

Hilary Dickson Road Space Manager

Sustainable Comr

Sustainable Communities, Regeneration & Economic Recovery Directorate

#### SCHEDULE

#### (Introducing double yellow line waiting restrictions "at any time") Bartlett Street, South Croydon

The north side, amending the existing single yellow line to double yellow lines along the entire length of that side of the road.

#### Lodge Lane, Croydon

Both sides, extending from the turning head at the north-western extremity of Lodge Lane (fronting property Nos. 267, 277 and 287) for 13 metres south-eastwards on the south-west side, and up to the existing disabled bay on the north-east side. Including the turning head itself. (4431935)

#### LONDON BOROUGH OF SOUTHWARK

#### SCHOOL STREETS PROGRAMME: DULWICH WOOD AREA THE LONDON BOROUGH OF SOUTHWARK (PEDESTRIAN AND CYCLE ZONES) (SCHOOL STREETS DULWICH WOOD AREA) TRAFFIC ORDER 2023

1. Southwark Council hereby GIVES NOTICE that on 31 August 2023 it has made the above Order under sections 6 and 124 of, and Part IV of Schedule 9 to, the Road Traffic Regulation Act 1984, as amended.

2. The effects of the Order are to introduce a 'School Streets' pedestrian and cycle zone in the entirety of BOWEN DRIVE, LYALL AVENUE, and SEELEY DRIVE in so much of the carriageway as is public highway, in the vicinity of Dulwich Wood Primary School and Dulwich Wood Nursery School. All motor vehicles, except permit holders, would be prohibited from entering or proceeding in the designated 'School Streets' pedestrian and cycle zone, as described above, between 8.00 am and 9.30 am and between 2.30 pm and 4.00  $\,$ pm on Mondays to Fridays, during school term time (as determined by Dulwich Wood Primary School and Dulwich Wood Nursery School). Permits allowing vehicles to enter the 'School Streets' pedestrian and cycle zone described above would be available to residents of premises located within the zone, to registered carers or residents of premises located within the zone, to disabled person's blue badge holders who are school staff, carers or dropping off and/or collecting pupils to the school and to SEND vehicles, by application to the Council. Exemptions would also apply in respect of emergency service vehicles.

3. Copies of the Order, which will come into force on 4 September 2023, and of all other relevant documents are available for inspection at Highways, Southwark Council, Environment, Neighbourhoods and Growth, 3rd floor hub 2, 160 Tooley Street, London SE1 2QH, by appointment only. E-mail traffic.orders@southwark.gov.uk or call 020 7525 3497 for booking details.

4. Any person desiring to question the validity of the Order or of any provision contained therein on the grounds that it is not within the relevant powers of the Road Traffic Regulation Act 1984 or that any of the relevant requirements thereof or of any relevant regulations made thereunder has not been complied with in relation to the Order may, within six weeks of the date on which the Order was made, make application for the purpose to the High Court.

Dated 31 August 2023

Dale Foden - Head of Service, Highways (4431939)

## **OTHER NOTICES**

#### COMPANY LAW SUPPLEMENT

The Company Law Supplement details information notified to, or by, the Registrar of Companies. The Company Law Supplement to *The London, Belfast and Edinburgh Gazette* is published weekly on a Tuesday.

These supplements are available to view at https:// www.thegazette.co.uk/browse-publications.

Alternatively use the search and filter feature which can be found here https://www.thegazette.co.uk/all-notices on the company number and/or name. (4253148)

#### NOTICE OF CLAIM TO NON-PARTIES

In the matter of The Director of the Serious Fraud Office –v-Abdallah Ali Jammal and Jammal Trust Bank S.A.L (A company incorporated in Lebanon in Liquidation KB-2022-004409).

TO: ALL NON-PARTIES TO THE ABOVE-MENTIONED PROCEEDINGS TAKE NOTICE THAT:

1. Civil Recovery Proceedings have been issued by the Director of the Serious Fraud Office in the Kings Bench Division of the High Court of Justice of England & Wales under Claim No KB-2022-004409 (the "Claim") pursuant to the Proceeds of Crime Act 2002 ("POCA 2002").

2. Without prejudice to any application that may be made by a third party under section 281 POCA 2002, the High Court has determined that the following is recoverable property within the meaning of ss.304-310 of POCA 2002:

a) The balance (including any interest accrued) in an HSBC sterling account (Account Number: 52085089, Sort code: 40- 47-17) in the name of Abdallah Ali Jammal and;

b) The funds (including any interest accrued) held in the Court Funds Office with reference number CJANO88OF2004.

2. Further information in relation to these civil recovery proceedings and/or in relation to how third parties may take steps to make an application under section 281 POCA 2002 can be obtained by contacting the Serious Fraud Office at 2-4 Cockspur Street, London, WC2Y 5BS, email: AAJ81.Litigation@sfo.gov.uk, (using reference AAJ81) before 31st October 2023. (4431884)

IN THE HIGH COURT OF JUSTICE Claim No. CR-2023-001437 BUSINESS AND PROPERTY COURTS OF ENGLAND AND WALES INSOLVENCY AND COMPANIES LIST (ChD) BEFORE: Mr Justice Trower DATED: 25 August 2023 IN THE MATTER OF CIMOLAI S.P.A. AND IN THE MATTER OF LUIGI CIMOLAI HOLDING S.P.A. AND IN THE MATTER OF THE COMPANIES ACT 2006

ORDER

**UPON THE APPLICATION** made by Cimolai S.p.A. ("**Cimolai**") and Luigi Cimolai Holding S.p.A. ("**LCH**" and, together, the "**Plan Companies**") by the Part 8 Claim Form dated 16 March 2023 (the "**Claim Form**").

AND UPON READING the Claim Form and evidence filed in support thereof, including the draft restructuring plan between Cimolai and the Cimolai Plan Creditors (the "Cimolai Plan Document") and the draft restructuring plan between LCH and the LCH Plan Creditors (the "LCH Plan Document" and together the "Plan Documents") promulgated by the Plan Companies pursuant to Part 26A of the Companies Act 2006 (the "Act") and the Explanatory Statement dated 17 July 2023 distributed pursuant to section 901D of the Act (the "Explanatory Statement").

AND UPON HEARING Adam Al-Attar and Lottie Pyper for the Plan Companies.

**AND UPON** the Court having adopted in this Order (save where terms are otherwise expressly defined) the abbreviations, words, definitions and phrases and rules of interpretation contained in the Explanatory Statement and the Plan Documents.

**AND UPON** the sole shareholder of LCH, Mr Luigi Cimolai, having undertaken by a deed of undertaking dated 24 July 2023 to, among other things, agree to provide the Shareholder Commitment. **IT IS ORDERED THAT:** 

1. The Court hereby sanctions the Cimolai Restructuring Plan between Cimolai and the Cimolai Plan Creditors as set out in the First Schedule hereto.

2. The Court hereby sanctions the LCH Restructuring Plan between LCH and the LCH Plan Creditors as set out in the Second Schedule hereto.

3. The Plan Companies or their solicitors shall deliver, as soon as reasonably practicable, a copy of this Order to the Registrar of Companies for England and Wales and to The Gazette for publication. **SERVICE** 

The Court has provided a sealed copy of this Order to the serving party:

Kirkland & Ellis International LLP, 30 St. Mary Axe, London, EC3A 8AF DATED **25 August 2023** 

FIRST SCHEDULE FIRST SCHEDULE The Cimolai Plan Document THE RESTRUCTURING PLAN Claim No. CR-2023-001437 IN THE HIGH COURT OF JUSTICE BUSINESS AND PROPERTY COURTS OF ENGLAND AND WALES INSOLVENCY AND COMPANIES LIST (ChD) IN THE MATTER OF CIMOLAI S.P.A.

and IN THE MATTER OF THE COMPANIES ACT 2006 RESTRUCTURING PLAN

(UNDER PART 26A OF THE COMPANIES ACT 2006) BETWEEN

CIMOLAI S.P.A. AND

EACH OF ITS PLAN CREDITORS (AS DEFINED HEREIN) 1 Preliminary

Recitals

1.1 Cimolai S.p.A. is a joint stock company incorporated under the laws of Italy, having its registered office at Rome (RM), Viale Pasteur No. 49 and operating office in Porcia (PN), Corso Lino Zanussi no. 26, and registration number 01507200937 with the Companies Register of Rome (the "**Company**").

1.2 The Company filed the Concordato Proposal on 20 February 2023. The Restructuring Plan is a parallel process to the Concordato Proposal.

#### The purpose and effect of the Restructuring Plan

1.3 The purpose of the Restructuring Plan is to implement the Concordato Proposal in its entirety as a matter of English law. As a result, the Restructuring Plan will effect a compromise and arrangement between the Company and its Plan Creditors in accordance with the terms of the Concordato Proposal.

1.4 To effect the compromises and arrangements that form the Concordato Proposal, each Plan Creditor will be bound by the applicable Restructuring Documents (including for the avoidance of doubt the terms of the Concordato Proposal Document).

#### 2 Definitions and Interpretation

2.1 Definitions

In this Restructuring Plan, unless defined otherwise, the following terms shall have the following meaning:

"Act" means the Companies Act 2006 (as amended from time to time) (UK).

"Al Bayt Disputes" means ongoing disputes and proceedings relating to the Al Bayt Joint Venture. "**Al Bayt Joint Venture**" means the GSIC JV entered into by Cimolai pursuant to a joint venture agreement dated 18 November 2013 (as subsequently amended and supplemented) (with Cimolai's share being 20%) together with Galfar Al Misnad Engineering and Contracting WLL (with a share of 40%) and Webuild S.p.A. (formerly Salini Impregilo, with a share of 40%).

"Assessment Claims" means the Plan Claims assessed as at the Assessment Date for the purposes of the Concordato Proposal.

"Assessment Date" means 19 October 2022.

"Business Day" means a day (other than a Saturday or Sunday) on which banking institutions in London and Italy are open for general business.

**"Chair"** means the chairman (as such term is used in the Act) of the Plan Meetings appointed pursuant to the Convening Order.

"Claim" means any claim or claims (including any contingent or disputed claims) in respect of any Liability of the Company to a Plan Creditor or any other claim or claims in respect of any Liability of the Company, including in each case (without limitation) any Assessment Claims, any claim or claims in relation to any Liability of the Company in respect of loss or damage suffered or incurred, whether directly or indirectly, as a result of or in connection with such Liability (including, for the avoidance of doubt, where relevant any interest accruing on, or accretions arising in respect of, such claims).

**"Concordato Proposal"** means the Company's *proposta di concordato preventive* (concordato proposal for composition with creditors in business continuity) pursuant to Article 84 of **Legislative** Decree no. 14 of 12 January 2019 (Italy), the terms of which are set out in the Concordato Proposal Document, which was filed with the Court of Trieste in Italy on 20 February 2023, as subsequently amended on 29 April 2023 and 30 June 2023.

**"Concordato Proposal Document"** means the plan document setting out the terms of the Concordato Proposal originally dated 20 February 2023 (the "**Original Concordato Proposal Document**"), as supplemented and amended by the First Supplemental Concordato Proposal Document and Second Supplemental Concordato Proposal Document.

**"Convening Order"** means the order dated on or around 14 July 2023 made by the Court in the matter of the Company and in the matter of the Companies Act 2006 directing that the Plan Meetings be convened for the purposes of considering and, if thought fit, approving (with or without modification) the Restructuring Plan.

"Court" means the High Court of Justice of England and Wales.

**"Demoted Unsecured Claim"** means any Claim against the Company held by a Demoted Unsecured Creditor.

"Demoted Unsecured Creditor" means:

(a) the Mortgage Creditors in respect of the Unsecured Portion of their Claim with respect to the Mortgage Loans;

(b) FRIE Financing Creditors in respect of the Unsecured Portion of their Claim with respect to the FRIE Financing;

(c) SACE in respect of the Unsecured Portion of its Claim with respect to the SACE Financing;

(d) tax authorities which are owed VAT accrued by foreign branches (France, Luxembourg and Switzerland) in respect of the Unsecured Portion of their Plan Claims (as described in the Concordato Proposal Document):

(e) creditors relating to VAT refunds payable to suppliers in respect of the Unsecured Portion of their Plan Claims (as described in the Concordato Proposal Document); and

(f) creditors relating to municipal and provincial taxes, duties and levies in respect of the Unsecured Portion of their Plan Claims (as described in the Concordato Proposal Document).

**"Disputed Unsecured Claims"** means any Claim against the Company held by a Disputed Unsecured Creditor.

"Disputed Unsecured Creditor" means:

(g) Ballinger & Co Ltd;

(h) Ebury Partners Belgium SA/NV;

(i) GPS Capital Markets Ltd;

(j) Macquarie Bank Europe DAC;

(k) Mediobanca Banca di Credito Finanziario S.p.A.;

(I) Natwest Markets Plc;

(m) Natixis;

(n) Deutsche Bank AG;

(o) Morgan Stanley & Co International Plc;

(p) Global Reach Markets B.V.;

(q) Cornèr Banca;

(r) JB Drax; and

(s) the Al Bayt Joint Venture.

**"English Law Proceedings"** means the Court proceedings brought against the Company by the Disputed Unsecured Creditors (excluding the Al Bayt Joint Venture, Global Reach and Corner Bank) in connection with their respective derivative contracts entered into with the Company.

"Excluded Liabilities" means each of the following:

(t) any Liabilities incurred after the Assessment Date;

(u) Liabilities which are preferred liabilities as a matter of Italian law, including payables to employees, VAT payments to suppliers and certain other debts, which will be paid in full as a result of their preferential status as a matter of Italian law; and

(v) Liabilities in respect of subordinated debt owed to affiliated companies which will be released in full on or promptly following the Plan Effective Time.

**"Existing Documents"** means any document evidencing or giving rise to the Liabilities owed by the Company to any Plan Creditor (in its capacity as such).

**"Explanatory Statement"** means the explanatory statement to the Restructuring Plan dated on or about 14 July 2023 required to be provided to the Plan Creditors pursuant to section 901D of the Act.

**"First Supplemental Concordato Proposal Document"** means the supplemental plan document dated 30 May 2023 (in respect of the amendment to the Concordato Proposal on 29 April 2023) supplementing and amending the terms of the Original Concordato Proposal Document.

**"Foreign Law Proceedings"** means the Court proceedings brought against the Company by Global Reach and Corner Bank (as Disputed Unsecured Creditors) in connection with their respective derivative contracts entered into with the Company.

"FRIE" means the Fund for Economic Initiatives in Italy.

"FRIE Financing" means loans entered into by the Company which relate to public financing administered by FRIE entered into by the Company.

"FRIE Financing Creditors" means creditors in respect of the FRIE Financing.

"Group" means the Shareholder, the Company and its subsidiaries (from time to time).

**"JB Drax"** means JB Drax Honorè (DIFC) Limited as counterparty in respect of the JB Drax Derivative Contract entered into by the Company as (guarantor).

**"JB Drax Derivative Contract"** means the derivative contract entered into by the Shareholder (as primary debtor) and the Company (as guarantor) with JB Drax.

**"JB Drax Proceedings"** means the Court proceedings brought by JB Drax against the Shareholder and the Company in respect of the JB Drax Derivative Contract.

"Judicial Commissioner" means Alberto Cimolai as the judicial commissioner appointed by the Court of Trieste in relation to the Concordato Proposal.

**"LCH Concordato Proposal"** means the Shareholder's *proposta di concordato preventive* (concordato proposal for composition with creditors in business continuity) pursuant to Article 84 of **Legislative** Decree no. 14 of 12 January 2019 (Italy), the terms of which are set out in the LCH Concordato Proposal Document, which was filed with the Court of Trieste in Italy on 20 February 2023, as subsequently amended on 9 May 2023 and 30 June 2023.

**"LCH Concordato Proposal Document"** means the plan document setting out the terms of the LCH Concordato Proposal originally dated 20 February 2023, as supplemented and amended on 30 May 2023 and 30 June 2023.

"Liability" or "Liabilities" means any present or future obligation, liability, claim, debt, claims for specific performance, damages or restitution, counterclaims, suits, rights of action, or rights whatsoever or howsoever arising, including, without limitation, for the payment of money or the performance of an act or obligation or any failure to perform any obligation or any omission, whether for negligence, breach of duty, breach of trust or misrepresentation or otherwise, whether in respect of principal, interest or otherwise, whether actual or contingent, whether fixed or undetermined, whether admitted or disputed, whether known or unknown, whether owed jointly or severally and whether owed as principal, surety or in any capacity whatsoever and whether it arises at common law, in equity or by statute, in England and Wales or in any other jurisdiction under whatsoever.

"Mortgage Creditors" means creditors in respect of the secured portion of Liabilities owed by the Company in relation to the Mortgage Loans.

"Mortgage Loans" means loans entered into by the Company which are guaranteed by mortgages on real estate owned by the Company and by special voluntary lien on systems and machinery of the Company.

"Plan Attorney" has the meaning given to that term in Clause 4.1.

**"Plan Claims"** means the Secured Claims, the Demoted Unsecured Claims, the Unsecured Claims and the Disputed Unsecured Claims but does not include any Claims in respect of Excluded Liabilities.

**"Plan Consideration"** means the consideration to be allocated to each of the Plan Creditors, as applicable, pursuant to the terms of and as set out in the Concordato Proposal Document.

"Plan Effective Time" means the time at which the Concordato Proposal is irrevocably and definitively sanctioned by the Court of Trieste.

"Plan Portal" has the meaning given to such term in the Explanatory Statement.

**"Plan Meetings"** means each meeting of the relevant Plan Creditors to vote on the Restructuring Plan convening pursuant to the Convening Order (and any adjournment of such meeting).

"Proceedings" means any process, suit, action, legal or other proceeding including without limitation any arbitration, mediation, alternative dispute resolution, judicial review, adjudication, demand, execution, distraint, restraint, forfeiture, re-entry, seizure, lien, enforcement of judgment or enforcement of any security.

**"Registrar of Companies"** means the registrar of companies in England and Wales within the meaning of the Act.

"Released Party" means each of the following:

(w) the Company;

(x) the Shareholder; and

(y) any other member of the Group.

**"Restructuring"** means the financial restructuring of the Group in accordance with and as implemented through the Restructuring Plan, the Concordato Proposal Document and the other Restructuring Documents, and any and all compromises, arrangements, and/or agreements with persons that are not parties to the Restructuring Plan, as described in the Explanatory Statement.

#### "Restructuring Documents" means:

(z) this Restructuring Plan;

(aa) the Concordato Proposal Document; and

(bb) all other documents, agreements and instruments contemplated by the Restructuring Plan and/or the Concordato Proposal Document as necessary or desirable (as determined by the Company, acting reasonably) to implement or consummate the Restructuring.

**"Restructuring Completion Time"** means the time at which the Restructuring is fully implemented and completed in accordance with the terms of the Concordato Proposal Document and in, any event, on or before 31 December 2029.

**"Restructuring Plan"** means, the restructuring plan proposed by the Company under Part 26A of the Act in its present form or with or subject to any modifications, additions or conditions approved or imposed by the Court or approved in accordance with the terms of the Restructuring Plan.

**"SACE"** means SACE S.p.A., the Italian credit finance agency which issued Garanzia Italia pursuant to Law Decree no 23/2020.

**"SACE Financing"** means loan agreements entered into by the Company with financial institutions, as described in the Concordato Proposal Document, in each case guaranteed by SACE (pursuant to the Garanzia Italia) covering 90% of the total financed amount.

**"SACE Financing Creditors"** means the creditors in respect of the SACE Financing.

**"Sanction Hearing"** means the hearing of the Court on whether or not to sanction the Restructuring Plan, which took place on 25 August 2023.

**"Sanction Order"** means the order of the Court sanctioning the Restructuring Plan under section 901F of the Act.

**"Second Supplemental Concordato Proposal Document"** means the supplemental plan document dated 30 June 2023 (in respect of the amendment to the Concordato Proposal on 30 June 2023) supplementing and amending the terms of the Original Concordato Proposal Document (as supplemented and amended by the First Supplemental Concordato Proposal Document).

**"Secured Claim"** means the Secured Portion of any Claim against the Company held by a Secured Creditor.

**"Secured Creditor"** means (a) Mortgage Creditors in respect of the Secured Portion of their Claim relating to the Mortgage Loans; (b) FRIE Financing Creditors in respect of the Secured Portion of their Claim relating to the FRIE Financing and (c) SACE in respect of the Secured Portion of its Claims relating to the SACE Financing.

**"Secured Portion"** means, in respect of a Secured Creditor, the portion of that Secured Creditor's Claims, which represents the value, as at the Assessment Date, of the property of the Company that is subject to a security interest granted in favour, or for the benefit, of that Secured Creditor.

**"Shareholder"** means Luigi Cimolai Holding S.p.A., a joint stock company incorporated under the laws of Italy, its registered office in Rome (RM), Viale Pasteur no. 49 and operating office in Porcia (PN), Corso Lino Zanussi no. 26, and registration number 02396610301 with the Companies Register of Rome.

**"Termination Date"** means the earlier of: (i) the Restructuring Completion Time; (ii) the date on which the Concordato Proposal is rejected by the Court of Trieste; or (iii) the date on which the Concordato Proposal is validly terminated in accordance with applicable Italian bankruptcy law (other than by reason of limbs (i) and (ii) above).

"Unsecured Claim" means any Claim against the Company held by an Unsecured Creditor.

**"Unsecured Creditor"** means all unsecured creditors of the Company including but not limited to the following creditors (but excluding the Disputed Unsecured Creditors and creditors in respect of Excluded Liabilities):

(cc) suppliers of the Company, as described in the Concordato Proposal Document;

(dd) smaller companies in respect of certain payables (including receivables claimed by smaller companies), as described in the Concordato Proposal Document;

(ee) Argentex Group PLC, Alpha FX Europe Limited (Malta), Hamilton Court Foreign Exchange Limited, Western Union International Bank GmbH and AFEX Corpay Markets Europe LTD as counterparties in respect of derivative contracts entered into by the Company that are not disputed;

(ff) SACE Financing Creditors in respect of the SACE Financing not guaranteed by the SACE-issued Garanzia Italia and certain of the remaining SACE Financing;

(gg) bondholders in respect of €40,000,000 2.4% amortising notes due 7 February 2027 issued by the Company on 7 February 2020;

(hh) Group companies in respect of intercompany commercial receivables, as described in the Concordato Proposal Document;

(ii) directors in respect of fees payable, as described in the Concordato Proposal Document; and

(jj) banks and other lenders in respect of other debts, as described in the Concordato Proposal Document.

**"Unsecured Portion"** means, in respect of a Secured Creditor, the total amount of all its Claims against the Company as at the Assessment Date; less the Secured Portion of such Claims.

#### 2.2 Interpretation

In this Restructuring Plan, unless the context otherwise requires or otherwise expressly provides:

(a) references to Clauses and Schedules are references to the Clauses of, and Schedules to, the Restructuring Plan;

(b) references to a person shall be construed so as to include its successors in title, permitted assigns and permitted transferees;

(c) references to an enactment, a statute or statutory provision or any subordinate legislation include the same as subsequently modified, amended, supplemented or re-enacted from time to time and to any instrument or order made from time to time under such enactment, statute, statutory provision or subordinate legislation;

(d) references to an agreement, deed or document shall be deemed also to refer to such agreement, deed or document as amended, supplemented, restated, varied, replaced and/or novated (in whole or in part) from time to time and to any agreement, deed or document executed pursuant thereto;

(e) references to an agreement, deed or document shall include any schedules, annexes and appendices to such agreement, deed or document;

(f) references to (or to any specified provision of) the Restructuring Plan shall be construed as references to the Restructuring Plan as in force for the time being;

(g) the singular includes the plural and vice versa and words importing one gender shall include all genders;

(h) headings to Clauses are for ease of reference only and shall not affect the interpretation of the Restructuring Plan;

(i) unless otherwise stated, all references in the Restructuring Plan to times are to London time;

(j) the term "including" is without limitation;

2.3 "€", "EUR", and "euro" denote the currency of those member states of the European Union that adopt or have adopted, and in each case continues to adopt, the euro as their lawful currency in accordance with the legislation of the European Community relating to Economic and Monetary Union.

#### 3 Effectiveness of the Restructuring Plan

**3.1** This Restructuring Plan provides for a compromise and arrangement between the Company and each of its Plan Creditors in respect of all the Plan Claims such that they shall receive equivalent treatment to their proposed treatment under the Concordato Proposal.

3.2 This Restructuring Plan shall implement the Concordato Proposal in its entirety and the terms of the Concordato Proposal (as set out in the Concordato Proposal Document) shall be incorporated into this Restructuring Plan such that (without limitation):

(a) the Plan Claims of the Plan Creditors shall be compromised in accordance with paragraph 11 of the Original Concordato Proposal Document, as supplemented by paragraph 3 of the Second Supplemental Concordato Proposal Document (and as reflected and described in the Concordato Proposal);

(b) the Company shall waive its right to commence, take or continue any English Law Proceedings or any Foreign Law Proceedings in respect of any Disputed Unsecured Creditor (other than JB Drax and the Al Bavt Joint Venture) which:

(i) votes in favour of the Concordato Proposal and Restructuring Plan; and

(ii) submits to the jurisdiction of the Court of Trieste in respect of the Concordato Proposal: and

(c) the Company shall not waive its right to commence, take or continue the JB Drax Proceedings or the Al Bayt Disputes irrespective of whether JB Drax or the Al Bayt Joint Venture:

(i) votes in favour of the Concordato Proposal and Restructuring Plan; and/or

(ii) submits to the jurisdiction of the Court of Trieste in respect of the Concordato Proposal.

3.3 The terms of the Restructuring Plan shall:

(a) apply to all Plan Claims and bind all Plan Creditors and their respective permitted successors and assigns and the Company; and

(b) apply to each other person that has undertaken to be bound by the terms of the Restructuring Plan and bind their respective successors and assigns.

3.4 Subject to Clause 11 (Effect of Termination) below, the Restructuring Plan shall become effective on and from the Plan Effective Time in accordance with its terms.

3.5 The Company shall promptly notify the Plan Creditors in writing of the effectiveness of the Restructuring Plan as soon as reasonably practicable following the occurrence of the Plan Effective Time.

3.6 Each of the Plan Creditors acknowledges and agrees that if the Concordato Proposal Document terminates in accordance with applicable Italian bankruptcy law (other than as a result of the Restructuring Completion Time having occurred), then the Restructuring Plan shall terminate and the execution, delivery or release of any deed, document or agreement in accordance with, or pursuant to, the Restructuring Plan shall be rescinded (insofar as legally possible) and deemed never to have become effective, and each relevant Plan Creditor and member of the Group, to the extent legally and practically possible, shall be put back into the position it was in prior to the date on which such deed, document or agreement was executed, delivered or released (as applicable), and each Plan Creditor and member of the Group shall, and shall use all reasonable efforts to procure that any necessary other party shall, execute such documents and perform such acts and things as may be required in order to do so.

#### 4 Implementation of the Restructuring

4.1 With effect on and from the Plan Effective Time, in consideration of the rights provided to the Plan Creditors under this Restructuring Plan and notwithstanding any term of any relevant document, each Plan Creditor (including any person to whom a Plan Creditor has transferred its rights in respect of its Plan Claims after the Assessment Date):

(a) authorises, appoints, instructs, and empowers, and shall for all purposes be treated as having authorised, appointed, instructed, and empowered, the Company, as its lawful attorney and agent (the "Attorney") of that Plan Creditor (or any person to whom a Plan Creditor has transferred its rights in respect of its Plan Claims after the Assessment Date) in respect of, represented by any duly authorised representative, to:

(i) enter into, sign, execute, notarise and, release (or authorise the release) and deliver (whether as a deed or otherwise), on behalf of each Plan Creditor (including any person to whom a Plan Creditor has transferred its rights in respect of its Plan Claim, to the extent applicable), the Restructuring Documents to which that Plan Creditor will be a party, for and on behalf of that Plan Creditor, such that each Plan Creditor will become a party to and be bound by those Restructuring Documents;

(ii) sign, execute and deliver all such deeds, documents, agreements, instruments, notices, transfers or instructions scheduled to, referred to in and/or contemplated by the Restructuring, including any Restructuring Documents, in each case which are required to be executed and/or delivered by or on its behalf, as the Company considers reasonably necessary or desirable to implement the Restructuring;

(iii) take any such other action as may be reasonably necessary or desirable to give effect to the terms of the Restructuring Plan and the Restructuring Documents to which they will be a party, to the extent required in order to give effect to the terms of the Restructuring Plan and the Restructuring Documents;

(iv) agree on their behalf any amendments to the Restructuring Documents to which they will be a party (and any amendments to the order of the steps contemplated by the Concordato Proposal) which the Attorney may deem necessary or desirable in order to ensure that: (A) they reflect the terms of the Restructuring Plan, the Concordato Proposal Document and the transactions intended to be entered into in order to effect the Restructuring;

(B) any blanks are completed (including, without limitation, any dates, times, bank account details, notice provisions or legal entity names), lists of parties and/or signature blocks;

(C) the information and categories of information contained, or referred to, in any formula, schedule, annex or similar, signature blocks, parties provisions, notice details or blanks in those Restructuring Documents (including details of any exposures, commitments or principal claims) reflect the relevant information and categories of information as of the applicable date;

(D) the Restructuring Documents take into account any modification of, or addition to, the Restructuring Plan and/or the Restructuring Documents approved or imposed by the Court in accordance with Clause 10.1 (Modifications to the Restructuring Plan);

(E) those Restructuring Documents may be duly executed and delivered: and/or

(F) those Restructuring Documents are legal, valid, binding and enforceable upon the parties to them;

(v) grant powers of attorney (whether governed by English law or the laws of any other jurisdiction) with the same scope as those authorisations and directions provided by the Plan Creditors pursuant to this Clause 4 for the benefit of any specified person in order to sign, execute, notarise, apostille, and/or deliver all such deeds, documents, agreements, instruments, notices, transfers or instructions scheduled to, referred to in and/or contemplated by the Restructuring, including any Restructuring Documents, in each case which are required to be executed and/or delivered by or on its behalf, as the Company considers reasonably necessary or desirable to implement the Restructuring; and

(vi) carry out any related or ancillary actions that it considers necessary or desirable for the purposes of implementing the Restructuring, provided that such related or ancillary actions are in accordance with the Restructuring Plan;

(b) ratifies and confirms any actions that:

(i) the Attorney (acting by its director(s), officer(s), or other duly appointed representative as may be authorised by the Attorney from time to time for this purpose) may have taken or may take under the power of attorney granted pursuant to Clause 4.1(a); and

(ii) the Chair may have taken or may take in connection with the conduct of the Plan Meetings (including any exercise of discretion in accordance with the Explanatory Statement) provided that such actions are in accordance with the terms of the Restructuring Plan;

(c) undertakes to the Company to take all such actions as may be reasonably required to implement the Restructuring; and

(d) agrees that save in the event of fraud, gross negligence and/or wilful misconduct, not to take any proceedings against any member of the Group (and its respective directors and officers) in respect of actions such member of the Group may have taken or may take in connection with the implementation of the Restructuring.

4.2 The authority granted under Clause 4.1(a) in favour of the Attorney shall be treated for all purposes whatsoever and without limitation as having been granted by deed.

4.3 Any transfer forming part of the Restructuring Plan shall be effected by means of the Concordato Proposal Document or by means of executing a form of transfer or other instrument or instruction of transfer falling within Clause 4.1(a)(ii) above.

5 Implementation of the Concordato Proposal

The Restructuring shall be implemented in accordance with the terms of the Concordato Proposal Document and shall become effective from the Plan Effective Time.

#### **6 Calculation of Entitlements**

Plan Creditors' entitlements to Plan Consideration shall be calculated in accordance with the Concordato Proposal Document and the claims confirmation process under the Concordato Proposal as supervised by the Judicial Commissioner.

#### 7 Registration of Sanction Order

As soon as reasonably practicable following the granting of the Sanction Order by the Court, the Company shall deliver the Sanction Order to the Registrar of Companies in accordance with Part 26A of the Act or advertise the same in the Gazette as may be required.

#### 8 Releases

8.1 With effect from the Plan Effective Time, each Plan Creditor shall, irrevocably and unconditionally, fully and finally, waive, release and discharge forever to the fullest extent permitted by applicable law, any and all Liabilities, in each case that were ever owed, may be owing to or may hereafter become owing to it or any of its Related Parties, in any capacity whatsoever, by or from any and all Released Parties whatsoever, whensoever or howsoever arising, in connection with or by reason of or resulting directly or indirectly from a Released Party's Assessment Claims or participation in the preparation, discussions, negotiation, sanction, execution or implementation of the Restructuring, this Restructuring Plan, and/or the other Restructuring Documents and all and any documentation ancillary or related thereto (including, without limitation, any steps, acts, or omissions for the purposes of achieving a restructuring transaction with respect to the Group and participation in any discussions and negotiations with stakeholders of the Group in any capacity, the execution of any Restructuring Document and the carrying out of the steps and transactions contemplated therein and in accordance with their terms).

8.2 Each release, waiver, and discharge effected by the terms of this Clause 8 shall not extend to any Liability:

(a) arising or resulting from gross negligence, wilful misconduct or fraud;

(b) against any Adviser arising under, or relating to, a duty of care owed to such Adviser's client or arising under a duty of care to another person which has been specifically and expressly accepted or acknowledged in writing by that Adviser:

(c) arising under and expressly contemplated by the Restructuring Documents; or

(d) for the avoidance of doubt, (other than in accordance with the terms of the Restructuring Plan or any Restructuring Document) owed to a Disputed Unsecured Creditor.

8.3 A Released Party shall be entitled to enforce and enjoy the benefit of and rely upon this Clause 8 whether or not it is a party to this Restructuring Plan. The parties to this Restructuring Plan shall not be entitled to rescind or vary any term of this Clause 8 in a manner prejudicial to a Released Party without the consent of the relevant Released Party.

8.4 On and from the Plan Effective Time, each Plan Creditor shall not commence or continue, or instruct, direct or authorise any other person to commence or continue any Proceedings in respect of or arising from any of the Claims released pursuant to this Clause 8.

8.5 On and from the Plan Effective Time, any and all of JB Drax's Claims against the Shareholder (as primary debtor) in respect of the JB Drax Derivative Contract shall be compromised, and related Liabilities released, in the same manner as provided for under the LCH Concordato Proposal.

#### 9 Stay of Proceedings

9.1 On and from the Plan Effective Time up to and including the Termination Date (other than by reason of the Restructuring Completion Time occurring, in which case the following undertakings shall continue past the Termination Date), each Plan Creditor (in each case, on behalf of itself and each of its successors and assignees) irrevocably and unconditionally undertakes that it will not:

(a) exercise any rights, remedies, powers or discretions (including any action to enforce any guarantee) under or in respect of any Existing Documents (or instruct any other person to do the same);

(b) commence, take or continue, or support any person commencing, taking or continuing, or instruct any person to commence, take or continue any Proceeding or other judicial, quasi-judicial, administrative or regulatory process in any jurisdiction whatsoever against any Released Party; and/or

(c) seize, attach and/or enforce or execute liens or judgments against the Company's or any other member of the Group's property or transfer, encumber, or otherwise dispose of or interfere with the Company or any other member of the Group's assets or agreements to the extent such actions affect the Company's or any other member of the Group's assets in any jurisdiction (or instruct any other person to do the same), in each case arising out of or in connection with:

(i) the negotiation, preparation, implementation and/or consummation of the Restructuring Plan and the Restructuring Documents;

(ii) the execution of the Restructuring Documents; and/or

(iii) the execution of the Restructuring Plan or the taking of any steps or actions necessary or desirable to implement the transactions contemplated in the Restructuring Plan and the Restructuring Documents and the carrying out of the actions, steps and transactions contemplated by the Restructuring Plan.

9.2 However, this Clause 9 shall not:

(a) in any way impair or prejudice any rights of any Plan Creditor arising under any Restructuring Document (including as a consequence of non-compliance with the terms of any Restructuring Document, other than if such non-compliance or the consequences thereof have been expressly waived by the relevant Plan Creditor) or any remedy in respect of such right;

(b) apply to any claim or liability in respect of gross negligence, fraud or wilful misconduct by any Released Party; or

(c) in any way impair or prejudice any rights of a Plan Creditor to seek directions or an adjudication of the court in relation to the terms of the Restructuring Plan,

in each case, other than as contemplated by, and in accordance with, the terms of the Restructuring Plan or any Restructuring Document or with the intent of giving effect to the Restructuring and the Restructuring Documents, which for the avoidance of doubt includes the continuation by the Company of (i) the JB Drax Proceedings, (ii) the Al Bayt Disputes and/or (iii) the English Law Proceedings or Foreign Law Proceedings in respect of any other Disputed Unsecured Creditor which does not vote in favour of the Concordato Proposal and Restructuring Plan or submit to the jurisdiction of the Court of Trieste in respect of the Concordato Proposal.

#### 10 General Restructuring Plan Provisions

10.1 Modifications of the Restructuring Plan

The Company may, at the Sanction Hearing, consent on behalf of itself and all of its respective Plan Creditors to any modification of, or addition to, this Restructuring Plan and / or any of the other Restructuring Documents or to any terms or conditions that the Court may think fit to approve or impose and which are otherwise necessary for, or assist with, the purpose of implementing the Restructuring, provided that such modification, addition, term or condition could not reasonably be expected directly or indirectly to change any right or obligation of or impose an additional obligation (by reference to such rights or obligations as are contemplated at the date of the Explanatory Statement) on any Plan Creditor or have a material adverse effect on the interests of any Plan Creditor under the Restructuring Plan. However, if such modifications could reasonably be expected directly or indirectly to change any right or obligation of or impose an additional obligation (by reference to such rights or obligations as are contemplated at the date of the Explanatory Statement) on any Plan Creditor or have a material adverse effect on the interests of a Plan Creditor, then a Company may not give such consent without the prior written consent of that Plan Creditor.

#### 10.2 Assignments or transfers

(a) All Plan Claims shall be determined as at the Assessment Date. Neither Company shall be under any obligation to recognise any assignment or transfer of rights, benefits or interests in any Plan Claim after the Assessment Date for the purposes of calculating Plan Claims and neither Company has any obligation hereunder to any person other than the applicable Plan Creditors, provided that, where the relevant Company has received from the relevant parties notice in writing of such assignment or transfer prior to the Plan Effective Time, such Company may, in its sole discretion and subject to the production of such other evidence in relation to such assignment or transfer as it may reasonably require and to any other terms and conditions which such Company may consider necessary or desirable, agree to recognise such assignment or transfer for the purposes of calculating Plan Claims.

(b) Any assignee or transferee of a Plan Creditor shall be bound by the terms of the Restructuring Plan as a Plan Creditor.

(c) Each Plan Creditor acknowledges and agrees that, in the event that any Plan Creditor transfers or assigns its Plan Claim, the Company may, on or prior to the Plan Effective Time, make such amendments to the Restructuring Documents, in each case to the extent necessary solely to reflect the change in ownership of a Plan Claim.

(d) Each Plan Creditor authorises the relevant Company to consent to and enter into any amendments to the Restructuring Documents which are made in accordance with this Clause 10.

#### 10.3 Obligations on days other than a Business Day

If any obligation is to be performed under the terms of the Restructuring Plan on a date other than a Business Day, the relevant obligation shall be performed on the next Business Day.

#### **11 Effect of Termination**

The Restructuring Plan will terminate if the Termination Date occurs (other than by reason of the Restructuring Completion Date occurring), upon which:

(a) the terms of and the obligations on the parties under or pursuant to the Restructuring Plan shall lapse and the execution, delivery or release of any deed, document, agreement, compromise and arrangement provided by the Restructuring Plan and any other documents relating to the Restructuring shall be of no effect and the rights of each Plan Creditor and the Company will not be affected and will be reinstated and remain in full force and effect, such that each Plan Creditor and the Company will be put back into the position it was in prior to the date on which such deed, document, agreement comprise or arrangement was executed delivered or released (as applicable); and

(b) the Plan Attorney shall immediately cease to be the agent and attorney of any Plan Creditor.

#### 12 Notice

12.1 Any notice or other written communication to be given under or in relation to the Restructuring Plan shall be given in the English language in writing and shall be deemed to have been duly given if it is delivered by hand, email, posted on the Plan Portal, fax, pre-paid recorded delivery or international courier to the address or email address as set out below (or as may be notified by notice to Plan Creditors from time to time).

12.2 The addresses for notices are as follows:

(a) in the case of the Company, to Kirkland & Ellis International LLP, at email address: project\_c@kirkland.com; and

(b) in the case of any other person, any address set forth for that person in any agreement entered into in connection with the Restructuring Plan.

12.3 Any notice or other written communication to be given under the Restructuring Plan shall be deemed to have been served:

(a) at the time of delivery if delivered personally;

(b) at the time of transmission if sent by email;

(c) 2 Business Days after the time and date of posting if sent by prepaid recorded delivery;

(d) 3 Business Days after the time and date of posting if sent by international courier; or (

e) when posted on the Plan Portal.

12.4 The accidental omission to send any notice, written communication or other document in accordance with Clauses 12.1 to 12.3 (inclusive), or the non-receipt of any such notice by any Plan Creditor, shall not affect the provisions of the Restructuring Plan.

#### 13 Governing law and jurisdiction

13.1 The Restructuring Plan and any non-contractual obligations arising out of or in connection with them shall be governed by, and construed in accordance with, the laws of England and Wales and each Plan Creditor hereby agrees that the Court shall have exclusive jurisdiction to hear and determine any suit, action or proceeding and to settle any dispute which may arise out of or in connection with the Explanatory Statement or any provision of the Restructuring Plan, or out of any action taken or omitted to be taken under the Restructuring Plan or any non-contractual obligations arising out of or in connection with the Restructuring Plan. For such purposes each Plan Creditors irrevocably submit to the jurisdiction of the Court, provided, however, that nothing in this Clause 13.1 shall affect the validity of other provisions determining governing law and jurisdiction as between the Company and the Plan Creditors, whether contained in any contract or otherwise.

13.2 The Restructuring Plan shall take effect subject to any prohibition or condition imposed by law.

DATED 25 August 2023 SECOND SCHEDULE The LCH Plan Document THE RESTRUCTURING PLAN Claim No. CR-2023-001437 IN THE HIGH COURT OF JUSTICE BUSINESS AND PROPERTY COURTS OF ENGLAND AND WALES INSOLVENCY AND COMPANIES LIST (ChD) IN THE MATTER OF LUIGI CIMOLAI HOLDING S.P.A. and

IN THE MATTER OF THE COMPANIES ACT 2006 RESTRUCTURING PLAN (UNDER PART 26A OF THE COMPANIES ACT 2006) BETWEEN LUIGI CIMOLAI HOLDING S.P.A. AND

#### EACH OF ITS PLAN CREDITORS (AS DEFINED HEREIN) 1 Preliminary

Recitals

1.1 Luigi Cimolai Holding S.p.A., is a joint stock company incorporated under the laws of Italy, its registered office in Rome (RM), Viale Pasteur no. 49 and operating office in Porcia (PN), Corso Lino Zanussi no. 26, and registration number 02396610301 with the Companies Register of Rome (the "**Company**").

1.2 The Company filed the Concordato Proposal on 20 February 2023. The Restructuring Plan is a parallel process to the Concordato Proposal.

#### The purpose and effect of the Restructuring Plan

1.3 The purpose of the Restructuring Plan is to implement the Concordato Proposal in its entirety as a matter of English law. As a result, the Restructuring Plan will effect a compromise and arrangement between the Company and its Plan Creditors in accordance with the terms of the Concordato Proposal.

1.4 To effect the compromises and arrangements that form the Concordato Proposal, each Plan Creditor will be bound by the applicable Restructuring Documents (including for the avoidance of doubt the terms of the Concordato Proposal Document).

#### 2 Definitions and Interpretation

2.1 Definitions

In this Restructuring Plan, unless defined otherwise, the following terms shall have the following meaning:

"**Act**" means the Companies Act 2006 (as amended from time to time) (UK).

"Assessment Claims" means the Plan Claims assessed as at the Assessment Date for the purposes of the Concordato Proposal.

"Assessment Date" means 19 October 2022.

"Business Day" means a day (other than a Saturday or Sunday) on which banking institutions in London and Italy are open for general business.

"Chair" means the chairman (as such term is used in the Act) of the Plan Meetings appointed pursuant to the Convening Order.

"Claim" means any claim or claims (including any contingent or disputed claims) in respect of any Liability of the Company to a Plan Creditor or any other claim or claims in respect of any Liability of the Company, including in each case (without limitation) any Assessment Claims, any claim or claims in relation to any Liability of the Company in respect of loss or damage suffered or incurred, whether directly or indirectly, as a result of or in connection with such Liability (including, for the avoidance of doubt, where relevant any interest accruing on, or accretions arising in respect of, such claims).

"Cimolai" means Cimolai S.p.A, a joint stock company incorporated under the laws of Italy, having its registered office at Rome (RM), Viale Pasteur No. 49 and operating office in Porcia (PN), Corso Lino Zanussi no. 26, and registration number 01507200937 with the Companies Register of Rome.

"Cimolai Concordato Proposal" means Cimolai's *proposta di concordato preventive* (concordato proposal for composition with creditors in business continuity) pursuant to Article 84 of Legislative Decree no. 14 of 12 January 2019 (Italy), the terms of which are set out in the Cimolai Concordato Proposal Document, which was filed with the Court of Trieste in Italy on 20 February 2023, as subsequently amended on 29 April 2023 and 30 June 2023.

"Cimolai Concordato Proposal Document" means the plan document setting out the terms of the Cimolai Concordato Proposal originally dated 20 February 2023, as supplemented and amended on 30 May 2023 and 30 June 2023. "Concordato Proposal" means the Company's proposta di concordato preventive (concordato proposal for composition with creditors in business continuity) pursuant to Article 84 of **Legislative** Decree Number no. 14 of 12 January 2019 (Italy), the terms of which are set out in the Concordato Proposal Document, which was filed with the Court of Trieste in Italy on 20 February 2023, as subsequently amended on 29 April 2023 and 30 June 2023.

"Concordato Proposal Document" means the plan document setting out the terms of the Concordato Proposal originally dated 20 February 2023 (the "Original Concordato Proposal Document"), as supplemented and amended by the First Supplemental Concordato Proposal Document and Second Supplemental Concordato Proposal Document.

"**Convening Order**" means the order dated on or around 14 July 2023 made by the Court in the matter of the Company and in the matter of the Companies Act 2006 directing that the Plan Meetings be convened for the purposes of considering and, if thought fit, approving (with or without modification) the Restructuring Plan.

"Court" means the High Court of Justice of England and Wales.

"Disputed Unsecured Claims" means any Claim against the Company held by the Disputed Unsecured Creditor.

"Disputed Unsecured Creditor" means JB Drax.

"Excluded Liabilities" means each of the following:

(a) any Liabilities incurred after the Assessment Date;

(b) Liabilities which are preferred liabilities as a matter of Italian law, including payables to employees, VAT payments to suppliers and certain other debts, which will be paid in full as a result of their preferential status as a matter of Italian law; and

(c) Liabilities in respect of subordinated debt owed to affiliated companies which will be released in full on or promptly following the Plan Effective Time.

"Existing Documents" means any document evidencing or giving rise to the Liabilities owed by the Company to any Plan Creditor (in its capacity as such).

**"Explanatory Statement"** means the explanatory statement to the Restructuring Plan dated on or about 14 July 2023 required to be provided to the Plan Creditors pursuant to section 901D of the Act.

"First Supplemental Concordato Proposal Document" means the supplemental plan document dated 30 May 2023 (in respect of the amendment to the Concordato Proposal on 29 April 2023) supplementing and amending the terms of the Original Concordato Proposal Document.

"Group" means the Company and its subsidiaries (from time to time).

"JB Drax" means JB Drax Honorè (DIFC) Limited as counterparty in respect of the JB Drax Derivative Contract entered into by the Company (as primary debtor).

"JB Drax Derivative Contract" means the derivative contract entered into by the Company (as primary debtor) and Cimolai (as guarantor) with JB Drax.

"JB Drax Proceedings" means the Court proceedings brought by JB Drax against the Company and Cimolai in respect of the JB Drax Derivative Contract.

"JB Drax SFP" means the new quasi-equity instrument to be issued by the Company to Cimolai and certain other Group members pursuant to the terms of the Restructuring.

"Judicial Commissioner" means Alberto Cimolai as the judicial commissioner appointed by the Court of Trieste in relation to the Concordato Proposal.

"Liability" or "Liabilities" means any present or future obligation, liability, claim, debt, claims for specific performance, damages or restitution, counterclaims, suits, rights of action, or rights whatsoever or howsoever arising, including, without limitation, for the payment of money or the performance of an act or obligation or any failure to perform any obligation or any omission, whether for negligence, breach of duty, breach of trust or misrepresentation or otherwise, whether in respect of principal, interest or otherwise, whether actual or contingent, whether fixed or undetermined, whether admitted or disputed, whether known or unknown, whether owed jointly or severally and whether owed as principal, surety or in any capacity whatsoever and whether it arises at common law, in equity or by statute, in England and Wales or in any other jurisdiction under whatever applicable law, under any legal theory, and in any manner whatsoever.

"Plan Attorney" has the meaning given to that term in Clause 4.1.

"Plan Claims" means the Unsecured Claims and the Disputed Unsecured Claims but does not include any Claims in respect of Excluded Liabilities. "**Plan Consideration**" means the consideration to be allocated to each of the Plan Creditors, as applicable, pursuant to the terms of and as set out in the Concordato Proposal Document.

"Plan Effective Time" means the time at which the Concordato Proposal is irrevocably and definitively sanctioned by the Court of Trieste.

"Plan Portal" has the meaning given to such term in the Explanatory Statement.

"**Plan Meetings**" means each meeting of the relevant Plan Creditors to vote on the Restructuring Plan convening pursuant to the Convening Order (and any adjournment of such meeting).

"**Proceedings**" means any process, suit, action, legal or other proceeding including without limitation any arbitration, mediation, alternative dispute resolution, judicial review, adjudication, demand, execution, distraint, restraint, forfeiture, re-entry, seizure, lien, enforcement of judgment or enforcement of any security.

**"Registrar of Companies**" means the registrar of companies in England and Wales within the meaning of the Act.

"Released Party" means each of the following:

(a) the Company;

(b) Cimolai; and

(c) any other member of the Group.

**"Restructuring"** means the financial restructuring of the Group in accordance with and as implemented through the Restructuring Plan, the Concordato Proposal Document and the other Restructuring Documents, and any and all compromises, arrangements, and/or agreements with persons that are not parties to the Restructuring Plan, as described in the Explanatory Statement.

"Restructuring Documents" means:

(d) this Restructuring Plan;

(d) the Concordato Proposal Document; and

(e) all other documents, agreements and instruments contemplated by the Restructuring Plan and/or the Concordato Proposal Document as necessary or desirable (as determined by the Company, acting reasonably) to implement or consummate the Restructuring.

"Restructuring Completion Time" means the time at which the Restructuring is fully implemented and completed in accordance with the terms of the Concordato Proposal Document and, in any event, on or before 31 December 2029.

"**Restructuring Plan**" means, the restructuring plan proposed by the Company under Part 26A of the Act in its present form or with or subject to any modifications, additions or conditions approved or imposed by the Court or approved in accordance with the terms of the Restructuring Plan.

"Sanction Hearing" means the hearing of the Court on whether or not to sanction the Restructuring Plan, which took place on 25 August 2023.

"Sanction Order" means the order of the Court sanctioning the Restructuring Plan under section 901F of the Act.

"Second Supplemental Concordato Proposal Document" means the supplemental plan document dated 30 June 2023 (in respect of the amendment to the Concordato Proposal on 30 June 2023) supplementing and amending the terms of the Original Concordato Proposal Document (as supplemented and amended by the First Supplemental Concordato Proposal Document).

"Termination Date" means the earlier of: (i) the Restructuring Completion Time; (ii) the date on which the Concordato Proposal is rejected by the Court of Trieste; or (iii) the date on which the Concordato Proposal is validly terminated in accordance with applicable Italian bankruptcy law (other than by reasons of limbs (i) and (ii) above).

"**Unsecured Claim**" means any Claim against the Company held by an Unsecured Creditor.

"**Unsecured Creditor**" means all unsecured creditors of the Company including but not limited to the following creditors (but excluding the Disputed Unsecured Creditors and creditors in respect of Excluded Liabilities):

(e) creditors in respect of certain unsecured loans of the Company, as described in the Concordato Proposal Document;

(f) suppliers of the Company, as described in the Concordato Proposal Document; and

(f) the sole director of the Company relating to accrued compensation, as described in the Concordato Proposal Document.

#### 2.2 Interpretation

In this Restructuring Plan, unless the context otherwise requires or otherwise expressly provides:

(a) references to Clauses and Schedules are references to the Clauses of, and Schedules to, the Restructuring Plan;

(b) references to a person shall be construed so as to include its successors in title, permitted assigns and permitted transferees;

(c) references to an enactment, a statute or statutory provision or any subordinate legislation include the same as subsequently modified, amended, supplemented or re-enacted from time to time and to any instrument or order made from time to time under such enactment, statute, statutory provision or subordinate legislation;

(d) references to an agreement, deed or document shall be deemed also to refer to such agreement, deed or document as amended, supplemented, restated, varied, replaced and/or novated (in whole or in part) from time to time and to any agreement, deed or document executed pursuant thereto;

(e) references to an agreement, deed or document shall include any schedules, annexes and appendices to such agreement, deed or document;

(f) references to (or to any specified provision of) the Restructuring Plan shall be construed as references to the Restructuring Plan as in force for the time being;

(g) the singular includes the plural and vice versa and words importing one gender shall include all genders;

(h) headings to Clauses are for ease of reference only and shall not affect the interpretation of the Restructuring Plan;

(i) unless otherwise stated, all references in the Restructuring Plan to times are to London time;

(j) the term "including" is without limitation;

2.3 "€", "EUR", and "euro" denote the currency of those member states of the European Union that adopt or have adopted, and in each case continues to adopt, the euro as their lawful currency in accordance with the legislation of the European Community relating to Economic and Monetary Union.

#### 3 Effectiveness of the Restructuring Plan

3.1 This Restructuring Plan provides for a compromise and arrangement between the Company and each of its Plan Creditors in respect of all the Plan Claims such that they shall receive equivalent treatment to their proposed treatment under the Concordato Proposal.

3.2 This Restructuring Plan shall implement the Concordato Proposal in its entirety and the terms of the Concordato Proposal (as set out in the Concordato Proposal Document) shall be incorporated into this Restructuring Plan such that (without limitation):

(a) the Plan Claims of the Plan Creditors shall be compromised in accordance with paragraph 9 of the Original Concordato Proposal Document, as supplemented by paragraph 3 of the Second Supplemental Concordato Proposal Document (and as reflected and described in the Concordato Proposal); and

(b) the Company shall not waive its right to commence, take or continue the JB Drax Proceedings irrespective of whether JB Drax:

(i) votes in favour of the Concordato Proposal and Restructuring Plan; and/or

(ii) submits to the jurisdiction of the Court of Trieste in respect of the Concordato Proposal.

3.3 The terms of the Restructuring Plan shall:

(a) apply to all Plan Claims and bind all Plan Creditors and their respective permitted successors and assigns and the Company; and(b) apply to each other person that has undertaken to be bound by the terms of the Restructuring Plan and bind their respective

successors and assigns. 3.4 Subject to Clause 11 (*Effect of Termination*) below, the Restructuring Plan shall become effective on and from the Plan Effective Time in accordance with its terms.

3.5 The Company shall promptly notify the Plan Creditors in writing of the effectiveness of the Restructuring Plan as soon as reasonably practicable following the occurrence of the Plan Effective Time.

3.6 Each of the Plan Creditors acknowledges and agrees that if the Concordato Proposal Document terminates in accordance with applicable Italian bankruptcy law (other than as a result of the Restructuring Completion Time having occurred), then the Restructuring Plan shall terminate and the execution, delivery or release of any deed, document or agreement in accordance with, or pursuant to, the Restructuring Plan shall be rescinded (insofar as legally possible) and deemed never to have become effective, and each relevant Plan Creditor and member of the Group, to the extent legally and practically possible, shall be put back into the position it was in prior to the date on which such deed, document or agreement

was executed, delivered or released (as applicable), and each Plan Creditor and member of the Group shall, and shall use all reasonable efforts to procure that any necessary other party shall, execute such documents and perform such acts and things as may be required in order to do so.

#### 4 Implementation of the Restructuring

4.1 With effect on and from the Plan Effective Time, in consideration of the rights provided to the Plan Creditors under this Restructuring Plan and notwithstanding any term of any relevant document, each Plan Creditor (including any person to whom a Plan Creditor has transferred its rights in respect of its Plan Claims after the Assessment Date):

(a) authorises, appoints, instructs, and empowers, and shall for all purposes be treated as having authorised, appointed, instructed, and empowered, the Company, as its lawful attorney and agent (the "Attorney") of that Plan Creditor (or any person to whom a Plan Creditor has transferred its rights in respect of its Plan Claims after the Assessment Date) in respect of, represented by any duly authorised representative, to:

(i) enter into, sign, execute, notarise and, release (or authorise the release) and deliver (whether as a deed or otherwise), on behalf of each Plan Creditor (including any person to whom a Plan Creditor has transferred its rights in respect of its Plan Claim, to the extent applicable), the Restructuring Documents to which that Plan Creditor will be a party, for and on behalf of that Plan Creditor, such that each Plan Creditor will become a party to and be bound by those Restructuring Documents;

(ii) sign, execute and deliver all such deeds, documents, agreements, instruments, notices, transfers or instructions scheduled to, referred to in and/or contemplated by the Restructuring, including any Restructuring Documents, in each case which are required to be executed and/or delivered by or on its behalf, as the Company considers reasonably necessary or desirable to implement the Restructuring;

(iii) take any such other action as may be reasonably necessary or desirable to give effect to the terms of the Restructuring Plan and the Restructuring Documents to which they will be a party, to the extent required in order to give effect to the terms of the Restructuring Plan and the Restructuring Documents;

 (iv) agree on their behalf any amendments to the Restructuring Documents to which they will be a party (and any amendments to the order of the steps contemplated by the Concordato Proposal) which the Attorney may deem necessary or desirable in order to ensure that:
 (A) they reflect the terms of the Restructuring Plan, the Concordato Proposal Document and the transactions intended to be entered into in order to effect the Restructuring;

(B) any blanks are completed (including, without limitation, any dates, times, bank account details, notice provisions or legal entity names), lists of parties and/or signature blocks;

(C) the information and categories of information contained, or referred to, in any formula, schedule, annex or similar, signature blocks, parties provisions, notice details or blank in those Restructuring Documents (including details of any exposures, commitments or principal claims) reflect the relevant information and categories of information as of the applicable date;

(D) the Restructuring Documents take into account any modification of, or addition to, the Restructuring Plan and/or the Restructuring Documents approved or imposed by the Court in accordance with Clause 10.1 (Modifications to the Restructuring Plan);

(E) those Restructuring Documents may be duly executed and delivered; and/or

(F) those Restructuring Documents are legal, valid, binding and enforceable upon the parties to them;

(v) grant powers of attorney (whether governed by English law or the laws of any other jurisdiction) with the same scope as those authorisations and directions provided by the Plan Creditors pursuant to this Clause 4 for the benefit of any specified person in order to sign, execute, notarise, apostille, and/or deliver all such deeds, documents, agreements, instruments, notices, transfers or instructions scheduled to, referred to in and/or contemplated by the Restructuring, including any Restructuring Documents, in each case which are required to be executed and/or delivered by or on its behalf, as the Company considers reasonably necessary or desirable to implement the Restructuring; and (vi) carry out any related or ancillary actions that it considers necessary or desirable for the purposes of implementing the Restructuring, provided that such related or ancillary actions are in accordance with the Restructuring Plan;

(b) ratifies and confirms any actions that:

(i) the Attorney (acting by its director(s), officer(s), or other duly appointed representative as may be authorised by the Attorney from time to time for this purpose) may have taken or may take under the power of attorney granted pursuant to Clause 4.1(a); and

(ii) the Chair may have taken or may take in connection with the conduct of the Plan Meetings (including any exercise of discretion in accordance with the Explanatory Statement) provided that such actions are in accordance with the terms of the Restructuring Plan;

(c) undertakes to the Company to take all such actions as may be reasonably required to implement the Restructuring; and

(d) agrees that save in the event of fraud, gross negligence and/or wilful misconduct, not to take any proceedings against any member of the Group (and its respective directors and officers) in respect of actions such member of the Group may have taken or may take in connection with the implementation of the Restructuring.

4.2 The authority granted under Clause 4.1(a) in favour of the Attorney shall be treated for all purposes whatsoever and without limitation as having been granted by deed.

4.3 Any transfer forming part of the Restructuring Plan shall be effected by means of the Concordato Proposal Document or by means of executing a form of transfer or other instrument or instruction of transfer falling within Clause 4.1(a)(ii) above.

#### 5 Implementation of the Concordato Proposal

5.1 The Restructuring shall be implemented in accordance with the terms of the Concordato Proposal Document and shall become effective from the Plan Effective Time.

#### 6 Calculation of Entitlements

Plan Creditors' entitlements to Plan Consideration shall be calculated in accordance with the Concordato Proposal Document and the claims confirmation process under the Concordato Proposal as supervised by the Judicial Commissioner.

#### 7 Registration of Sanction Order

As soon as reasonably practicable following the granting of the Sanction Order by the Court, the Company shall deliver the Sanction Order to the Registrar of Companies in accordance with Part 26A of the Act or advertise the same in the Gazette as may be required.

#### 8 Releases

8.1 With effect from the Plan Effective Time, each Plan Creditor shall, irrevocably and unconditionally, fully and finally, waive, release and discharge forever to the fullest extent permitted by applicable law, any and all Liabilities, in each case that were ever owed, may be owing to or may hereafter become owing to it or any of its Related Parties, in any capacity whatsoever, by or from any and all Released Parties whatsoever, whensoever or howsoever arising, in connection with or by reason of or resulting directly or indirectly from a Released Party's Assessment Claims or participation in the preparation, discussions, negotiation, sanction, execution or implementation of the Restructuring, this Restructuring Plan, and/or the other Restructuring Documents and all and any documentation ancillary or related thereto (including, without limitation, any steps, acts, or omissions for the purposes of achieving a restructuring transaction with respect to the Group and participation in any discussions and negotiations with stakeholders of the Group in any capacity, the execution of any Restructuring Document and the carrying out of the steps and transactions contemplated therein and in accordance with their terms).

8.2 Each release, waiver, and discharge effected by the terms of this Clause 8 shall not extend to any Liability:

(a) arising or resulting from gross negligence, wilful misconduct or fraud;

(b) against any Adviser arising under, or relating to, a duty of care owed to such Adviser's client or arising under a duty of care to another person which has been specifically and expressly accepted or acknowledged in writing by that Adviser;

(c) arising under and expressly contemplated by the Restructuring Documents; or

(d) for the avoidance of doubt, (other than in accordance with the terms of the Restructuring Plan or any Restructuring Document) owed to the Disputed Unsecured Creditor.

8.3 A Released Party shall be entitled to enforce and enjoy the benefit of and rely upon this Clause 8 whether or not it is a party to this Restructuring Plan. The parties to this Restructuring Plan shall not be entitled to rescind or vary any term of this Clause 8 in a manner prejudicial to a Released Party without the consent of the relevant Released Party.

8.4 On and from the Plan Effective Time, each Plan Creditor shall not commence or continue, or instruct, direct or authorise any other person to commence or continue any Proceedings in respect of or arising from any of the Claims released pursuant to this Clause 8.

8.5 On and from the Plan Effective Time, any and all of JB Drax's Claims against Cimolai (as guarantor) in respect of the JB Drax Derivative Contract shall be compromised, and related Liabilities released, in the same manner as provided for under the Cimolai Concordato Proposal.

#### 9 Stay of Proceedings

9.1 On and from the Plan Effective Time up to and including the Termination Date (other than by reason of the Restructuring Completion Time occurring, in which case the following undertakings shall continue past the Termination Date), each Plan Creditor (in each case, on behalf of itself and each of its successors and assignees) irrevocably and unconditionally undertakes that it will not:

(a) exercise any rights, remedies, powers or discretions (including any action to enforce any guarantee) under or in respect of any Existing Documents (or instruct any other person to do the same);

(b) commence, take or continue, or support any person commencing, taking or continuing, or instruct any person to commence, take or continue any Proceeding or other judicial, quasi-judicial, administrative or regulatory process in any jurisdiction whatsoever against any Released Party; and/or

(c) seize, attach and/or enforce or execute liens or judgments against the Company's or any other member of the Group's property or transfer, encumber, or otherwise dispose of or interfere with the Company or any other member of the Group's assets or agreements to the extent such actions affect the Company's or any other member of the Group's assets in any jurisdiction (or instruct any other person to do the same), in each case arising out of or in connection with:

(i) the negotiation, preparation, implementation and/or consummation of the Restructuring Plan and the Restructuring Documents;

(ii) the execution of the Restructuring Documents; and/or

(iii) the execution of the Restructuring Plan or the taking of any steps or actions necessary or desirable to implement the transactions contemplated in the Restructuring Plan and the Restructuring Documents and the carrying out of the actions, steps and transactions contemplated by the Restructuring Plan.

#### 9.2 However, this Clause 9 shall not:

(a) in any way impair or prejudice any rights of any Plan Creditor arising under any Restructuring Document (including as a consequence of non-compliance with the terms of any Restructuring Document, other than if such non-compliance or the consequences thereof have been expressly waived by the relevant Plan Creditor) or any remedy in respect of such right;

(b) apply to any claim or liability in respect of gross negligence, fraud or wilful misconduct by any Released Party; or

(c) in any way impair or prejudice any rights of a Plan Creditor to seek directions or an adjudication of the court in relation to the terms of the Restructuring Plan,

in each case, other than as contemplated by, and in accordance with, the terms of the Restructuring Plan or any Restructuring Document or with the intent of giving effect to the Restructuring and the Restructuring Documents, which for the avoidance of doubt includes the continuation by the Company of the JB Drax Proceedings.

#### 10 General Restructuring Plan Provisions

10.1 Modifications of the Restructuring Plan The Company may, at the Sanction Hearing, consent on behalf of itself and all of its respective Plan Creditors to any modification of, or addition to, this Restructuring Plan and / or any of the other Restructuring Documents or to any terms or conditions that the Court may think fit to approve or impose and which are otherwise necessary for, or assist with, the purpose of implementing the Restructuring, *provided* that such modification, addition, term or condition could not reasonably be expected directly or indirectly to change any right or obligation of or impose an additional obligation (by reference to such rights or obligations as are contemplated at the date of the Explanatory Statement) on any Plan Creditor under the Restructuring Plan. However, if such modifications could reasonably be expected directly or indirectly to change any spected directly or the restructuring Plan.

right or obligation of or impose an additional obligation (by reference to such rights or obligations as are contemplated at the date of the Explanatory Statement) on any Plan Creditor or have a material adverse effect on the interests of a Plan Creditor, then a Company may not give such consent without the prior written consent of that Plan Creditor.

#### 10.2 Assignments or transfers

(a) All Plan Claims shall be determined as at the Assessment Date. Neither Company shall be under any obligation to recognise any assignment or transfer of rights, benefits or interests in any Plan Claim after the Assessment Date for the purposes of calculating Plan Claims and neither Company has any obligation hereunder to any person other than the applicable Plan Creditors, provided that, where the relevant Company has received from the relevant parties notice in writing of such assignment or transfer prior to the Plan Effective Time, such Company may, in its sole discretion and subject to the production of such other evidence in relation to such assignment or transfer as it may reasonably require and to any other terms and conditions which such Company may consider necessary or desirable, agree to recognise such assignment or transfer for the purposes of calculating Plan Claims.

(b) Any assignee or transferee of a Plan Creditor shall be bound by the terms of the Restructuring Plan as a Plan Creditor.

(c) Each Plan Creditor acknowledges and agrees that, in the event that any Plan Creditor transfers or assigns its Plan Claim, the Company may, on or prior to the Plan Effective Time, make such amendments to the Restructuring Documents, in each case to the extent necessary solely to reflect the change in ownership of a Plan Claim.

(d) Each Plan Creditor authorises the relevant Company to consent to and enter into any amendments to the Restructuring Documents which are made in accordance with this Clause 10.

#### 10.3 Obligations on days other than a Business Day

If any obligation is to be performed under the terms of the Restructuring Plan on a date other than a Business Day, the relevant obligation shall be performed on the next Business Day.

#### **11 Effect of Termination**

The Restructuring Plan will terminate if the Termination Date occurs other than by reason of the Restructuring Completion Date occurring, upon which:

(c) the terms of and the obligations on the parties under or pursuant to the Restructuring Plan shall lapse and the execution, delivery or release of any deed, document, agreement, compromise and arrangement provided by the Restructuring Plan and any other documents relating to the Restructuring shall be of no effect and the rights of each Plan Creditor and the Company will not be affected and will be reinstated and remain in full force and effect, such that each Plan Creditor and the Company will be put back into the position it was in prior to the date on which such deed, document, agreement comprise or arrangement was executed delivered or released (as applicable); and

(d) the Plan Attorney shall immediately cease to be the agent and attorney of any Plan Creditor.

#### 12 Notice

12.1 Any notice or other written communication to be given under or in relation to the Restructuring Plan shall be given in the English language in writing and shall be deemed to have been duly given if it is delivered by hand, email, posted on the Plan Portal, fax, pre-paid recorded delivery or international courier to the address or email address as set out below (or as may be notified by notice to Plan Creditors from time to time).

12.2 The addresses for notices are as follows:

(a) in the case of the Company, to Kirkland & Ellis International LLP, at email address: project\_c@kirkland.com; and

(b) in the case of any other person, any address set forth for that person in any agreement entered into in connection with the Restructuring Plan.

12.3 Any notice or other written communication to be given under the Restructuring Plan shall be deemed to have been served:

(a) at the time of delivery if delivered personally;

(b) at the time of transmission if sent by email;

(c) 2 Business Days after the time and date of posting if sent by prepaid recorded delivery;

(d) 3 Business Days after the time and date of posting if sent by international courier: or

(e) when posted on the Plan Portal.

12.4 The accidental omission to send any notice, written communication or other document in accordance with Clauses 12.1 to 12.3 (inclusive), or the non-receipt of any such notice by any Plan Creditor, shall not affect the provisions of the Restructuring Plan.

#### 13 Governing law and jurisdiction

13.1 The Restructuring Plan and any non-contractual obligations arising out of or in connection with them shall be governed by, and construed in accordance with, the laws of England and Wales and each Plan Creditor hereby agrees that the Court shall have exclusive jurisdiction to hear and determine any suit, action or proceeding and to settle any dispute which may arise out of or in connection with the Explanatory Statement or any provision of the Restructuring Plan, or out of any action taken or omitted to be taken under the Restructuring Plan or any non-contractual obligations arising out of or in connection with the Restructuring Plan. For such purposes each Plan Creditors irrevocably submit to the jurisdiction of the Court, provided, however, that nothing in this Clause 13.1 shall affect the validity of other provisions determining governing law and jurisdiction as between the Company and the Plan Creditors, whether contained in any contract or otherwise.

13.2 The Restructuring Plan shall take effect subject to any prohibition or condition imposed by law. DATED 25 August 2023

(4431854)

#### IN THE HIGH COURT OF JUSTICE CR-2023-004013 BUSINESS AND PROPERTY COURTS OF ENGLAND AND WALES

**INSOLVENCY AND COMPANIES LIST (ChD)** IN THE MATTER OF YUNNENG WIND POWER CO., LTD. AND IN THE MATTER OF THE COMPANIES ACT 2006 Before the Honourable Mr Justice Michael Green 22 August 2023 ORDER

UPON THE APPLICATION by a Part 8 Claim Form dated 24 July 2023 of Yunneng Wind Power Co., Ltd. (the "Company"), whose registered office is Level 14, No. 35, Lane 11, Guangfu North Road, Songshan District, Taipei 10560, Taiwan, for an Order sanctioning the restructuring plan pursuant to Part 26A of the Companies Act 2006 (the "Restructuring Plan")

AND UPON READING the evidence filed in support of the application for the sanctioning of the Restructuring Plan (comprising the first witness statement of Illia Vyshenskyi dated 23 July 2023, the first witness statement of Lars Muck dated 24 July 2023, the first witness statement of Ian Wallace dated 23 July 2023, the second witness statement of Illia Vyshenskyi dated 17 August 2023, the second witness statement of Lars Muck dated 17 August 2023, the first witness statement of Max Krasner dated 17 August 2023, the second witness statement of Ian Wallace dated 17 August 2023, the third witness statement of Lars Muck dated 18 2 August 2023 and the third witness statement of Ian Wallace dated 18 August 2023 (and the exhibits thereto))

AND UPON the parties referred to in paragraphs 3.22 and 3.23 of the second witness statement of Lars Muck dated 17 August 2023 having entered into the deeds of undertaking referred to therein

AND UPON HEARING Mark Arnold K.C. (leading Charlotte Cooke and Peter Burgess) for the Company

#### IT IS ORDERED AND DIRECTED THAT:

1 Pursuant to section 901F of the Companies Act 2006, the Court hereby sanctions the Restructuring Plan in the form set out in the Schedule hereto. For the avoidance of doubt, this form includes the amendments to the Restructuring Plan set out in the evidence, including those made: (a) following the convening hearing on 26 July 2023 and before the creditors' meetings on 15 August 2023; and (b) following the creditors' meetings and before the sanction hearing on 22 August 2023.

2 The Company or their solicitors shall deliver, as soon as reasonably practicable, an office copy of this Order to the Registrar of Companies for England and Wales.

#### SERVICE OF THE ORDER:

3 The Court has provided a sealed copy of this order to the serving party: Linklaters LLP at One Silk Street, London EC2Y 8HQ (ref: Conway/Wloch/Byrne).

DATED THIS 22nd DAY OF AUGUST 2023 Schedule 1 The Restructuring Plan IN THE HIGH COURT OF JUSTICE

Claim No. CR-2023-004013 BUSINESS AND PROPERTY COURTS OF ENGLAND AND WALES INSOLVENCY AND COMPANIES LIST (ChD) IN THE MATTER OF YUNNENG WIND POWER CO., LTD. - and -

Dage

#### IN THE MATTER OF THE COMPANIES ACT 2006 RESTRUCTURING PLAN

(under Part 26A of the Companies Act 2006)

#### between YUNNENG WIND POWER CO., LTD.

and

#### THE PLAN CREDITORS

(as hereinafter defined) Linklaters LLP One Silk Street London EC2Y 8HQ

Telephone (+44) 20 7456 2000 Bef I -308671

#### Contents

Contents	
Clause	

Clause		Page
1	Definitions	1
2	Interpretation	7
3	Plan Effective Date and Plan Claims	9
4	Plan Undertaking Parties	9
5	Grants of authority and instructions to execute the Restructuring Documents	10
6	Instructions to the Existing Agents	12
7	Waivers	13
8	Plan Steps	13
9	Termination	14
10	Determinations	15
11	Assignments and transfers	15
12	Future insolvency	15
13	Exclusion of liability	15
14	Modification	17
15	Application to the Court for directions	17
16	Exercise of discretion	17
17	Performance of obligations on dates other than a Business Day	17
18	Notices	17
19	Delegation	19
20	Governing law and jurisdiction	19
Schedule 1 Restructuring Implementation 20 Deed		
Schedule 2 Plan Under	taking Parties	21
Schedule 3 Existing Agents		
Schedule 4 Allocation Methodology		

#### Between:

## (1) YUNNENG WIND POWER CO., LTD. (the "Plan Company"); and (2) THE PLAN CREDITORS (as hereinafter defined). Recitals:

# (A) The Plan Company is a company incorporated in Taiwan with company no. 72976520 and whose registered office is at Level 3, No. 35, Lane 11, Guangfu North Road, Songshan District, Taipei 10560, Taiwan.

(B) The Plan Company is party to the Existing Debt Documents (as defined below).

(C) This Plan is being proposed by the Plan Company in order to facilitate the implementation of a wider restructuring of the financial indebtedness of the Plan Company and the Parent (as defined below) as described in more detail in the Explanatory Statement.

(D) Each of the Plan Undertaking Parties, upon the sanctioning of this Plan by the Court, has agreed to be bound by and comply with the obligations expressed to apply to it under this Plan and the Restructuring Documents to which it is a party pursuant to the provisions of such person's Deed of Undertaking.

#### 1 Definitions

In this Plan, words and expressions used but not defined shall have the meaning given to them in the Common Terms Agreement (as defined below). In addition, the following terms shall, unless the context otherwise requires, have the following meanings:

"Act" means the United Kingdom Companies Act 2006 (as amended); "Advisers" means (as applicable):

(a) Houlihan Lokey EMEA, LLP., financial advisers to the Plan Company;

(b) the Plan Company's Counsel (and counsel instructed by Linklaters LLP to advise on, and to appear before the Court in relation to, the Plan);

(c) the Plan Creditors' Counsel (and counsel instructed by White & Case LLP to advise on, and to appear before the Court in relation to, the Plan);

(d) PJT Partners, financial advisers to certain Coverage Providers;

(e) FTI Consulting (Australia) Pty Ltd, financial advisers to the Finance Parties;

(f) Norton Rose Fulbright LLP, legal advisers to TotalEnergies Renewables SAS;

(g) Freshfields Bruckhaus Deringer LLP, legal advisers to Taiwan Offshore Holding GmbH;

(h) Hunton Andrews Kurth LLP, legal advisers to Greenwing Energy B.V.;

(i) Norton Rose Fulbright Australia and Hogan Lovells International LLP, legal advisers Starwind Offshore GmbH; and

 (j) any successor professional advisers to the foregoing, the foregoing's partners, employees and affiliated partnerships and the partners and employees of such affiliated partnerships and their respective subsidiaries and holding companies and any local counsel engaged;

"Affiliate" means, with respect to any person or entity:

(a) a Subsidiary of that person or entity or a Holding Company of that person or entity or any other Subsidiary of such a Holding Company;

(b) any Affiliated Entities of any of the persons or entities referred to in paragraph (a) above; and

(c) any other person who, directly or indirectly, is in control of, or controlled by, or is under common control with, such person and, for the purposes of this definition, "control" shall mean the power, direct or indirect, to (a) vote on more than 50 per cent. of the securities having ordinary voting power for the election of directors of such person, or (b) direct or cause the direction of the management and policies of such person whether by contract or otherwise;

"Affiliated Entities" means, in respect of any person or entity (the "First Entity"):

(a) any entity (or its Affiliates) that is the investment manager or investment adviser of the First Entity (or its Affiliates);

(b) any entity (or its Affiliates) in respect of which the First Entity (or its Affiliates or its or its Affiliates' associates) is the investment manager or investment adviser;

(c) any entity (or its Affiliates) that is managed or advised by the same investment manager or investment adviser as the First Entity (or its Affiliates); and

(d) any entity (or its Affiliates) whose investment manager or investment adviser is an associate of the investment manager or investment adviser of the First Entity (or its Affiliates);

"Allocation Methodology" means the methodology set out in Schedule 4 (Allocation Methodology);

"Amended and Restated Atradius Base Facility Agreement" means the Atradius Base Facility Agreement as amended and restated pursuant to the terms of the Restructuring Implementation Deed (once executed) and on terms substantially consistent with the Debt Term Sheet;

"Amended and Restated Base and Standby Facilities" means the Amended and Restated Base Facility Agreements and the Amended and Restated Commercial Standby Facility Agreement; "Amended and Restated Base and Standby Facility Commitments" means the commitments of each Base and Standby Plan Creditor under the relevant Amended and Restated Base and Standby Facilities as calculated in accordance with the Allocation Methodology;

"Amended and Restated Base Facility Agreements" means the Amended and Restated Atradius Base Facility Agreement, the Amended and Restated EKF Base Facility Agreement, the Amended and Restated Euler Hermes Base Facility Agreement, the Amended and Restated Euro Commercial Base Facility Agreement, and the Amended and Restated KfW Base Facility Agreement;

"Amended and Restated Commercial Standby Facility Agreement" means the Commercial Standby Facility Agreement as amended and restated pursuant to the terms of the Restructuring Implementation Deed (once executed) and on terms substantially consistent with the Debt Term Sheet;

"Amended and Restated EKF Base Facility Agreement" means the EKF Base Facility Agreement as amended and restated pursuant to the terms of the Restructuring Implementation Deed (once executed) and on terms substantially consistent with the Debt Term Sheet;

"Amended and Restated Euler Hermes Base Facility Agreement" means the Euler Hermes Base Facility Agreement as amended and restated pursuant to the terms of the Restructuring Implementation Deed (once executed) and on terms substantially consistent with the Debt Term Sheet;

"Amended and Restated Euro Commercial Base Facility Agreement" means the Euro Commercial Base Facility Agreement as amended and restated pursuant to the terms of the Restructuring Implementation Deed (once executed) and on terms substantially consistent with the Debt Term Sheet;

"Amended and Restated Intercreditor Agreement" means the Intercreditor Agreement as amended and restated pursuant to the terms of the Restructuring Implementation Deed (once executed) and on terms substantially consistent with the Debt Term Sheet;

"Amended and Restated KfW Base Facility Agreement" means the KfW Base Facility Agreement as amended and restated pursuant to the terms of the Restructuring Implementation Deed (once executed) and on terms substantially consistent with the Debt Term Sheet;

"Authorised Person" means each of the:

(a) directors of the Plan Company;

(b) managing directors of the Parent; and

(c) the following employees and members of Linklaters LLP: Johannes Gottwald, Thomas Schulz, Sarah Louise Mook, James Matthew Warboys, Max Benjamin Krasner, and Hannah Barrett Brown, in their year estimated in dividual experience.

in their respective individual capacities;

**"Base and Standby Plan Creditor"** means a Plan Creditor who is a Senior Lender under the Base Facilities or a Senior Lender under the Commercial Standby Facility Agreement (or a Senior Lender under both the Base Facilities (or any of them) and the Commercial Standby Facility Agreement);

"Base Facilities" means the Atradius Base Facility Agreement, EKF Base Facility Agreement, Euler Hermes Base Facility Agreement, KfW Base Facility Agreement, NTD Commercial Base Facility Agreement and Euro Commercial Base Facility Agreement (each a "Base Facility");

"Business Day" means a day (other than a Saturday or Sunday) on which banks are open for general business in London, Hamburg, Frankfurt, and Taipei;

"Common Terms Agreement" means the common terms agreement dated 24 May 2019 as amended and restated on 30 May 2021 and on 6 April 2022, made between, among others, the Plan Company, the Parent, the Existing Agents and the Plan Creditors;

"Court" means the High Court of Justice of England and Wales;

"Debt Term Sheet" means the form of debt term sheet agreed between the Plan Company and members of the working group of Plan Creditors advised by the Plan Creditors' Counsel and made available to the Plan Creditors on the Plan Website;

"Deeds of Undertaking" means the deeds of undertaking in the form agreed between the Plan Company and the respective Plan Undertaking Parties before the date of the Sanction Hearing, pursuant to which each Plan Undertaking Party has agreed to be bound by and comply with the obligations expressed to apply to it under this Plan and to execute and deliver the Restructuring Documents to which it is party (each a "Deed of Undertaking");

**"Excluded Persons"** has the meaning given to that term in Clause 13.1; "Existing Agents" means the parties listed in Schedule 3 (*Existing Agents*);

"Existing Debt Documents" has the meaning given to the term "Finance Documents" in the Common Terms Agreement;

**"Explanatory Statement"** means the explanatory statement to this Plan dated on or around [26] July 2023 required to be provided to the Plan Creditors pursuant to section 901D of the Act;

"FFS Debt Facilities" means:

(a) the NTD FFS Debt Facility Agreement;

(b) the EUR FFS Debt Facility Agreement;

(c) the Atradius NTD FFS Debt Facility Agreement; (d) the Atradius EUR FFS Debt Facility Agreement;

(e) the Euler Hermes NTD FFS Debt Facility Agreement; and

(f) the Euler Hermes EUR FFS Debt Facility Agreement;

"Final Allocation Schedule" has the meaning given to that term in Clause 8.3.2;

"German Plan Creditor" means a Plan Creditor which is incorporated under the laws of the Federal Republic of Germany;

"Information Agent" means Kroll Issuer Services Limited in its capacity as information agent in connection with the Plan;

"Intercreditor Agent" means Crédit Agricole Corporate and Investment Bank, Hong Kong, Branch, as intercreditor agent under the Intercreditor Agreement and the Common Terms Agreement;

"Intercreditor Agreement" means the intercreditor agreement originally dated 24 May 2019 between, among others, the Plan Company, the Parent, the Intercreditor Agent and the Senior Lenders, as amended and restated on 30 May 2021;

"Liability" means any debt, liability or obligation whatsoever whether it is present, future, prospective or contingent, whether or not its amount is fixed or undetermined, whether or not it involves the payment of money or the performance of an act or obligation and whether it arises at common law, in equity or by statute, in England and Wales or in any other jurisdiction, or in any other manner whatsoever;

"Lock-Up Agreement" means a lock-up agreement to be entered into in connection with the Restructuring between the Plan Company, the Parent, the Sponsors and certain of the Plan Creditors and, if entered into prior to the Plan Effective Date, in the form uploaded to the Plan Website;

"Longstop Time" means 5 p.m. on 14 September 2023 or such later time as may be agreed in writing (whether pursuant to a single extension or multiple extensions) by:

(a) in the case of any extension (or extensions, as the case may be) to a date falling on or before 30 September 2023, the Parent, the Majority Sponsors and the Intercreditor Agent; and

(b) in the case of any extension (or extensions, as the case may be) to a date falling on or after 1 October 2023, the Plan Company, the Parent, each Sponsor and the Intercreditor Agent,

provided that such date may not be later than 31 December 2023;

"Majority Secured Creditors" means, at any time, those Secured Creditors whose Exposures at such time represent at least 50.1% of the aggregate Exposures of all Secured Creditors at such time;

"Majority Sponsors" means, at any time, at least three (out of four) Sponsors:

"**Parent**" means Yunlin Holding GmbH as the parent company of the Plan Company, incorporated on 15 September 2016 organised under the laws of the Federal Republic of Germany as a private limited company and entered in the commercial register of Bremen under HRB 31722 HB on 26 September 2016, and whose registered office is at Stephanitorsbollwerk 3, 28217 Bremen, Germany;

"Plan" means this restructuring plan under Part 26A of the Act between the Plan Company and the Plan Creditors with, or subject to, any modification, addition or condition which the Court may think fit to approve or impose, as appropriate;

"Plan Claims" means any claim or claims in respect of any Liability of the Plan Company to a Plan Creditor arising directly or indirectly in relation to, or arising out of or in connection with, the Existing Debt Documents, but excluding any claim or claims in respect of any Liability of the Plan Company to a Plan Creditor arising directly or indirectly in relation to, or arising out of or in connection with, a failure to comply with the Plan;

"Plan Company" has the meaning given to that term in the "Parties" clause above;

"Plan Company's Counsel" means Linklaters LLP (in respect of English, German and Luxembourg law) and Lee & Li, Attorneys-at-Law (in respect of Taiwanese law), (and/or any of its Affiliates and/or any of its affiliated partnerships), legal counsel to the Plan Company, or any successor legal counsel to the Plan Company;

"Plan Creditors" means (i) the Senior Lenders other than the Guarantee Facility Lenders in their capacity as such; and (ii) the Hedging Banks;

"Plan Creditors' Counsel" means White & Case LLP (in respect of English and German law) and Tsar & Tsai Law Firm (in respect of Taiwanese law), and any of its or their Affiliates and/or any of its or their affiliated partnerships, or any successor legal counsel to the Plan Creditors;

"Plan Effective Date" means the time and date at which an office copy of the Sanction Order has been delivered to the Registrar of Companies for registration in respect of the Plan Company;

"Plan Party" means each of the Plan Company, each Plan Creditor and each Plan Undertaking Party;

"Plan Steps" means the steps, transactions or actions set out in Clause 8 (*Plan Steps*);

"Plan Undertaking Finance Parties" means each entity listed in Part I of Schedule 2 (*Plan Undertaking Parties*);

"Plan Undertaking Parties" means the Plan Undertaking Finance Parties, the Plan Undertaking Sponsor Parties and the Information Agent;

"Plan Undertaking Sponsor Parties" means each entity listed in Part II of Schedule 2 (*Plan Undertaking Parties*) who has signed a Deed of Undertaking in favour of the Court in connection with this Plan;

"Plan Website" means https://deals.is.kroll.com/yunneng;

"Project" means the design, development, construction, commissioning, ownership, financing, operation, management and maintenance of the proposed 640MW Yunlin offshore windfarm, located off the coast of Sihu Township and Kouhu Township in the Yunlin County, Western Taiwan, and activities reasonably incidental thereto;

"Proxy and Voting Form" means a proxy and voting form to be delivered to the Information Agent substantially in the form as set out in the Explanatory Statement;

"Record Time" means 5 p.m. (London time) on 8 August 2023;

"Registrar of Companies" means the registrar of companies within the meaning of the Act;

"Replacement Covered Super Senior Lender" means, in respect of an Eligible Senior Lender which is a Covered Base Facility Lender (and in its capacity as such), either (i) any third party lender which is to provide lending pursuant to that Eligible Senior Lender's Relevant Covered Super Senior Facility; or (ii) the Relevant Coverage Provider itself;

"Restructuring" means the financial, debt and corporate restructuring of the Plan Company and the Parent contemplated by this Plan, the Restructuring Implementation Deed, the Explanatory Statement, the Debt Term Sheet and any and all connected compromises, arrangements and/or agreements with persons that are not parties to this Plan;

"Restructuring Documents" has the meaning given to that term in the Restructuring Implementation Deed;

"Restructuring Effective Date" has the meaning given to that term in the Restructuring Implementation Deed;

**"Restructuring Implementation Deed"** means the restructuring implementation deed substantially in the form scheduled hereto at Schedule 1 (*Restructuring Implementation Deed*) with any amendments or modifications as may reasonably be required to give effect to the terms of the Restructuring;

"Sanction Hearing" means the hearing for the purpose of obtaining the Sanction Order;

**"Sanction Order"** means the order of the Court sanctioning the Plan under section 901F of the Act;

"Sponsors" means Taiwan Offshore Holding GmbH, TotalEnergies Renewables SAS, Starwind Offshore GmbH and Greenwing Energy B.V.;

#### "Super Senior Agents" means:

(a) Crédit Agricole Corporate and Investment Bank, Hong Kong Branch in its capacity as Super Senior EKF Facility Agent, Super Senior KfW Facility Agent and Super Senior NTD Commercial Facility Agent, in each case as defined in the Amended and Restated Intercreditor Agreement;

(b) SMBC Bank International PLC (formerly known as Sumitomo Mitsui Banking Corporation Europe Limited) in its capacity as Super Senior Atradius ECA Agent, as defined in the Amended and Restated Intercreditor Agreement;

(c) Sumitomo Mitsui Banking Corporation, Düsseldorf Branch in its capacity as Super Senior Euler Hermes ECA Agent, as defined in the Amended and Restated Intercreditor Agreement;

(d) Taipei Fubon Commercial Bank Co., Ltd. in its capacity as Super Senior Euler Hermes Facility Agent and Super Senior Atradius Facility Agent, as defined in the Amended and Restated Intercreditor Agreement;

"Super Senior Election Deadline" means 8 a.m. (London time) on 22 August 2023;

"Super Senior Election Form" means a super senior election form to be delivered to the Information Agent substantially in the form as set out in the Explanatory Statement;

**"Super Senior Facilities"** means the up to NTD 16,750,000,000 (or its EUR equivalent) super senior loan facilities to be advanced to the Plan Company pursuant to the Super Senior Facility Agreements;

"Super Senior Facility Agreements" means the facility agreements to be entered into between, among others, the Plan Company, the Super Senior Facility Lenders and the Super Senior Agents pursuant to the Restructuring Implementation Deed and in substantially the form scheduled to the Explanatory Statement;

"Super Senior Facility Commitments" means the Commitments under and as defined in the Super Senior Facility Agreements;

"Super Senior Facility Commitments Schedule" has the meaning given to it in Clause 8.2 (*Plan Steps*);

"Super Senior Facility Lenders" has the meaning given to it in Clause 8.2 (*Plan Steps*);

#### 2 Interpretation

**2.1** In this Plan, with effect from the Plan Effective Date or such other date as may be necessary to construe this Plan, unless the context otherwise requires or otherwise expressly provides for:

2.1.1 references to this Plan shall include the Schedules to this Plan; 2.1.2 references to "Clauses" and "Schedules" are references to the clauses and schedules respectively of this Plan;

2.1.3 references to "paragraphs" when used in a Schedule, unless indicated otherwise, shall be to the paragraphs of that Schedule;

2.1.4 references to a "person" include references to an individual, firm, partnership company, corporation, unincorporated body of persons or any state or state agency and shall be construed so as to include its successors in title, permitted assigns and permitted transferees;

2.1.5 all references to "€" or "EUR" are references to euro, the single currency of the member states that adopt or have adopted, and in each case continues to adopt, the euro as its lawful currency in accordance with legislation of the European Union relating to Economic and Monetary Union and "NT\$", "NTD", and "New Taiwanese Dollars" is to the lawful currency of Taiwan;

2.1.6 references to the "NTD equivalent amount" in EUR and vice versa shall be calculated at the Fixed FX Rate;

2.1.7 references to a statute or statutory provision include the same as subsequently modified, amended or re-enacted from time to time;

2.1.8 references to a document include the same as subsequently supplemented, amended and/or restated, novated, varied or replaced from time to time;

2.1.9 references to a time of the day are references to the time in London, United Kingdom;

2.1.10 a reference to a party, Plan Party or any other person includes its successors in title, permitted assigns and permitted transferees;

2.1.11 any references to a Plan Creditor shall be references to a Plan Creditor in its capacity as a Senior Lender or as a Hedging Bank (as applicable) and a Plan Creditor shall be required to take all steps and actions required in this Plan to be taken by a Plan Creditor in any such capacity or capacities (as applicable):

2.1.12 a reference to "validly completed" shall have the meaning given to it in the Proxy and Voting Form;

2.1.13 where this Plan refers to the Plan Company delivering any notice and dating, completing, releasing and, if applicable, delivering a Restructuring Document, such obligation may be satisfied by the Plan Company's Counsel delivering any such notice and dating, completing, releasing and, if applicable, delivering any such Restructuring Document on the Plan Company's behalf;

2.1.14 where this Plan refers to a document, notice or confirmation being delivered to a Plan Party or the Plan Parties, each Plan Party agrees that it will be sufficient (if applicable) for the relevant document, notice or confirmation to be sent by way of email to: (i) in respect of the Plan Company, the Plan Company's Counsel;

(ii) in respect of the Plan Creditors, the Plan Creditors' Counsel or the Information Agent (copied to the Plan Creditors' Counsel) (for onward distribution to the Plan Creditors); and (iii) in respect of any Plan Undertaking Party, any other respective adviser or legal adviser (other than the Plan Creditors' Counsel) as may be notified by that Plan Undertaking Party to the Plan Company or as set out in the Restructuring Implementation Deed;

2.1.15 the singular includes the plural and vice versa and words importing one gender shall include all genders;

2.1.16 the term "including" means "including, without limitation"; and 2.1.17 headings are for ease of reference only and shall not affect the interpretation of this Plan.

2.2 References to any figures or amounts in this Plan are subject to:

2.2.1 potential modifications or adjustments under Clause 14 (Modification) or otherwise as a result of any rounding or other calculation or allocation procedures set out herein;

2.2.2 such minor or technical amendments or any amendments to correct manifest errors as may be agreed by or on behalf of (i) the Plan Company (or the Plan Company's Counsel); and (ii) the Plan Creditors' Counsel; and

2.2.3 such other amendments, adjustments or changes (including in respect of their levels or order of funds flow, to the extent applicable) as may be agreed by the Plan Company and the Plan Creditors' Counsel to the extent that such amendments, adjustments or changes do not, taken as a whole, result in any material adverse effect on any Plan Party without its prior written consent.

#### 3 Plan Effective Date and Plan Claims

**3.1** As soon as reasonably practicable following the date upon which the Court grants the Sanction Order, the Plan Company shall deliver an office copy of the Sanction Order to the Registrar of Companies.

**3.2** The compromise and arrangement effected by the Plan and the Restructuring Implementation Deed shall bind all Plan Creditors in respect of their Plan Claims, their successors, assigns, and the Plan Company. The amendments, variations and restatements of the relevant Existing Debt Documents and the transactions and steps in respect of the Plan Creditors contemplated by the Restructuring Implementation Deed and the Restructuring Documents pursuant to the authorisations and instructions granted to the Plan Company and the Existing Agents under this Plan shall be final and shall apply for all purposes including, without limitation, for the purpose of amending, varying and restating claims against (and obligations and liabilities of) the Plan Creditor shall be prohibited from asserting any right or taking any action against the Plan Company or the Parent which would be inconsistent therewith.

**3.3** The Plan shall become effective and legally binding on each Plan Party in accordance with its terms on the Plan Effective Date.

**3.4** The Plan Company shall promptly notify the Plan Parties following the occurrence of the Plan Effective Date and the Information Agent shall make such notice available on the Plan Website.

#### 4 Plan Undertaking Parties

**4.1** Each Plan Undertaking Party has severally agreed to be bound by and comply with the obligations expressed to apply to it under this Plan and the Restructuring Documents to which it is a party, pursuant to the provisions of its respective Deed of Undertaking.

**4.2** Promptly following the occurrence of the Plan Effective Date, each Plan Undertaking Party shall (to the extent it has not already done so): 4.2.1 sign the Restructuring Implementation Deed (if it is a party thereto) and release its signed signature page to the Plan Company; and

4.2.2 sign but leave undated (and, where applicable, undelivered) each Restructuring Document (other than the Restructuring Implementation Deed) to which it is a party.

## 5 Grants of authority and instructions to execute the Restructuring Documents

**5.1** On and from the Plan Effective Date, in consideration of the rights provided to the Plan Creditors under this Plan and notwithstanding any term of any other document, (a) each Plan Creditor (other than a German Plan Creditor) hereby appoints, and shall for all purposes be treated as having appointed, the Plan Company; and (b) each German Plan Creditor hereby appoints, and shall for all purposes be treated as having appointed, the Authorised Persons, in each case as its attorney and agent and irrevocably authorises, directs, instructs and empowers respectively the Plan Company (represented by any authorised representative) and the Authorised Persons (as applicable) to do the following:

5.1.1 enter into, execute and deliver (whether as a deed or otherwise), for and on behalf of such Plan Creditor the Restructuring Implementation Deed and each Restructuring Document to which such Plan Creditor, or any of them, are named as a party and, if applicable, complete, date and release such Restructuring Documents in accordance with the terms of the Restructuring Implementation Deed;

5.1.2 grant powers of attorney with the same scope as those authorisations and directions provided by each Plan Creditor pursuant to this Clause 5.1 for the benefit of a specified person in order to execute, notarise and/or register (if applicable) any Restructuring Document;

5.1.3 carry out any actions that the Plan Company considers necessary or desirable to give effect to the terms of the Plan, the Restructuring Implementation Deed and the Restructuring Documents, including without limitation giving instruction to any Existing Agent to the extent required to give effect to the Plan, the Restructuring Implementation Deed and the Restructuring Documents.

5.1.4 subject to Clause 5.1.5, in respect of the Restructuring Documents, agree on behalf of each Plan Creditor to:

(i) any amendments to the Restructuring Documents (including the Restructuring Implementation Deed) which the Plan Company (acting reasonably) considers necessary in order to ensure that they reflect the terms of this Plan or the transactions intended to be entered into in order to effect the Restructuring;

 (ii) insert the calculation and completion of any commitments, participations or allocations to any Plan Creditor or any other party under the Restructuring Documents (as determined in accordance with the terms of this Plan and/or the Restructuring Implementation Deed);

(iii) insert information into any blanks (including without limitation, any bank account details, notice provisions or legal entity names), lists of parties and/or signature blocks;

(iv) at or prior to the Sanction Hearing, schedule the Restructuring Implementation Deed (as amended or modified in accordance with this Clause 5.1.5) to this Plan;

(v) make any mechanical amendments required by the Plan Company;(vi) make any other minor or technical amendments;

(vii) ensure that they are legal, valid, binding and enforceable upon the parties to them in accordance with this Plan and the Restructuring Implementation Deed;

(viii) any amendment to take into account any modification of, or addition to, this Plan and/or the Restructuring Documents approved or imposed by the Court in accordance with Clause 14 (Modification),

in each case, subject to consultation with the Plan Creditors' Counsel and provided that if any such amendment or insertion could reasonably be expected directly or indirectly to have an adverse or disproportionate effect on the interests of a Plan Creditor, then such amendment will require the prior written consent of that Plan Creditor; and

5.1.5 in respect of the Restructuring Documents, agree on behalf of each Plan Creditor any amendments to the Restructuring Documents (including the Restructuring Implementation Deed) as may be agreed between the Plan Company and the Intercreditor Agent (acting on the instructions of Majority Secured Creditors) in order to finalise any Restructuring Document provided that such amendments are in order to give effect to, and are not inconsistent with, the provisions of the Debt Term Sheet.

5.1.6 do all things to give effect to, agree and/or approve the amendments to the Existing Debt Documents and to undertake such steps as it considers necessary or desirable for it to take for the purposes of facilitating the implementation of the Restructuring, including (without limitation) entering into and executing in its respective capacity the Restructuring Documents to which it is a party and any other document that it reasonably considers necessary or advisable to implement this Plan.

**5.2** The authority and power granted and conferred on the Plan Company and/or the Authorised Persons (as applicable) under this Clause 5 (*Grants of authority and instructions to execute the Restructuring Documents*) shall be treated, for all purposes whatsoever and without limitation, as having been granted and conferred by deed and the Plan Company and/or Authorised Persons (as applicable) shall be entitled to delegate the authority granted and conferred by this Clause 5 (*Grants of authority and instructions to execute the Restructuring Documents*) to any duly authorised officer or agent of the Plan Company or Authorised Person as necessary.

5.3 The Plan Company and each Plan Creditor agrees that each Existing Agent and Authorised Person shall be entitled to enforce, enjoy the benefit of and rely upon the provisions of this Clause 5 (*Grants of authority and instructions to execute the Restructuring Documents*).

5.4 Notwithstanding any other provision of this Plan, each Plan Creditor hereby confirms that it agrees to, shall be bound by, and shall comply with, and shall for all purposes be treated as having agreed to and be bound by the Restructuring Implementation Deed after it has been executed by the Plan Company or Authorised Person (as applicable) on its behalf in accordance with this Clause 5 (*Grants of authority and instructions to execute the Restructuring Documents*). Each Plan Creditor hereby irrevocably and unconditionally ratifies and confirms everything which the Plan Company (including its respective authorised signatories) and the Authorised Persons may lawfully do or cause to be done in accordance with any authority conferred by this Plan.

5.5 Unless barred by its articles of association or similar internal rules from granting such exemption, each Plan Creditor hereby irrevocably releases to the extent permitted by law and under its articles of association or similar internal rules the Plan Company (and each of its respective authorised signatories) or, as applicable, each Authorised Person from the restrictions set out in section 181 of the German Civil Code (Bürgerliches Gesetzbuch) and any similar restrictions under any applicable law in respect of any action undertaken in accordance with this Clause 5. To the extent released by the Plan Creditors from the restrictions set out in section 181 of the German Civil Code (Bürgerliches Gesetzbuch) and any similar restrictions under any applicable law, the Plan Company and each Authorised Person shall be authorised to release any person to whom it has delegated authority pursuant to this Clause 5 from the restrictions set out in section 181 of the German Civil Code (Bürgerliches Gesetzbuch) and any similar restrictions under any applicable law to the extent permitted by law and under the articles of association or similar internal rules of the Plan Company or Authorised Person (if applicable).

**5.6** Once the Restructuring Implementation Deed has been fully executed, released, dated and delivered, the authority granted by each Plan Creditor to the Plan Company and/or the Authorised Persons under this Clause 5 (*Grants of authority and instructions to execute the Restructuring Documents*) shall expire automatically in respect of the Restructuring Implementation Deed at that time and, thereafter, the Restructuring Implementation Deed may only be amended in accordance with its terms. Any remaining authorities granted by the Plan Creditors to the Plan Company and/or the Authorised Persons under this Clause 5 (*Grants of authority and instructions to execute the Restructuring Documents*) shall terminate two Business Days following the occurrence of the Restructuring Effective Date.

#### 6 Instructions to the Existing Agents

**6.1** Without prejudice to the authorities conferred on the Plan Company pursuant to Clause 5.1, with effect from the Plan Effective Date the Plan Creditors (pursuant to this Plan) and the relevant Plan Undertaking Finance Parties (in accordance with their respective Deeds of Undertaking) irrevocably authorise and instruct each applicable Existing Agent (and, to the extent such steps have already been completed, confirm, agree and ratify any steps taken by each applicable Existing Agent) to:

6.1.1 enter into, execute and deliver (whether as a deed or otherwise), for and on behalf of such Plan Creditor and Plan Undertaking Finance Party (as applicable) the Restructuring Implementation Deed and each Restructuring Document to which such Existing Agent is named as a party (including any Deed of Undertaking); and

6.1.2 execute, deliver (if applicable) and perform its obligations under any agreement, letter or other document, and do all such acts or things as may be necessary or desirable to be executed or done by it for the purposes of facilitating the implementation of, and giving effect to, the terms of this Plan, the Restructuring Implementation Deed, the Restructuring Documents and the Restructuring.

**6.2** Each of the Existing Agents shall be entitled to enforce and enjoy the benefit of, and rely upon each term in, this Plan.

**6.3** Where this Plan confers any discretion or approval right upon the Intercreditor Agent, the Intercreditor Agent shall act in accordance with the instructions of the Majority Consenting Creditors (as defined in the Lock-up Agreement) or, if the Lock-up Agreement is not effective at the relevant time, the Majority Secured Creditors. **7 Waivers** 

With effect from on and from the Plan Effective Date, each Plan Creditor shall unconditionally and irrevocably waive any termination right, Default or Event of Default (howsoever defined or described) that may occur or have occurred as a result of or in connection with the preparation, proposal and implementation of the Plan and the Restructuring.

#### 8 Plan Steps

**8.1** Promptly following the Plan Effective Date, the Plan Company shall circulate the fully compiled execution versions of the Restructuring Implementation Deed and each other Restructuring Document to the parties thereto for execution (to the extent not already circulated by it or by an Adviser).

8.2 Each Base and Standby Plan Creditor that has elected to participate in the Super Senior Facilities by submitting a Super Senior Election Form to the Plan Company prior to the Super Senior Election Deadline (or any Replacement Covered Super Senior Lender in respect of a Covered Base Facility Lender who has submitted a Super Senior Election Form to the Plan Company prior to the Super Senior Election Deadline) (a "Super Senior Facility Lender") shall be allocated Super Senior Facilities Commitments in accordance with the Allocation Methodology and as reflected in the schedule of commitments in relation to each Super Senior Facility Agreement to be confirmed by the Intercreditor Agent (the "Super Senior Facility Commitments Schedule"). No Base and Standby Plan Creditor may cancel, amend or withdraw a Super Senior Election Form after the Super Senior Election Deadline unless such amendment is to increase its commitment in which case the Plan Company may in its sole discretion accept such amendment after the Super Senior Election Deadline but before 10 a.m. on the date of the Sanction Hearing.

**8.3** The following steps shall be taken within one Business Day of the Plan Effective Date:

8.3.1 the Information Agent shall promptly provide the Plan Company, the Intercreditor Agent and the Advisers with copies of each validly completed Super Senior Election Form received by it by the Super Senior Election Deadline (and any subsequently amended Super Senior Election Form in accordance with Clause 8.2):

8.3.2 the Information Agent in consultation with the Plan Company, the respective Advisers to the Plan Company (where applicable) and the Finance Parties, and the Intercreditor Agent, shall:

(i) calculate the Amended and Restated Base and Standby Facility Commitments of each Base and Standby Plan Creditor under the Amended and Restated Base Facility Agreements in accordance with the Allocation Methodology; and

(ii) confirm the Super Senior Facility Commitments of each Super Senior Facility Lender under the Super Senior Facility Agreements by reference to the Super Senior Facility Commitments Schedule;

and shall promptly provide the Plan Company, the Intercreditor Agent and the Advisers with a schedule containing the results of those calculations (the "**Final Allocation Schedule**"); and

8.3.3 promptly following the completion of the step set out in Clause 8.3.2 the Information Agent shall upload the Final Allocation Schedule to the Plan Website.

**8.4** Promptly following the Plan Effective Date:

8.4.1 the Plan Company (and, if applicable, the Authorised Person(s)) shall, as soon as reasonably practicable, execute but not date the Restructuring Implementation Deed and each other Restructuring Document to which it is a party on its own behalf and on behalf of each Plan Creditor;

8.4.2 each Plan Undertaking Party shall, as soon as reasonably practicable, and to the extent it has not already done so, execute but not date the Restructuring Implementation Deed and each other Restructuring Document on its own behalf and deliver a copy of each such executed document to the Plan Company and the Plan Company's Counsel in accordance with Clause 18 (*Notices*);

8.4.3 promptly following receipt of all signatures to the Restructuring Implementation Deed, the Plan Company or the Plan Company's Counsel on its behalf shall, and are hereby irrevocably authorised to, date, complete, release and deliver the Restructuring Implementation Deed (without the consent of or any further notice to any party thereto); and

8.4.4 the Plan Company (on behalf of itself and the Plan Creditors) and each Plan Undertaking Party (to the extent any action is required by it in connection with the satisfaction of any condition precedent to any step specified in the Restructuring Implementation Deed) shall use reasonable endeavours to procure that any condition precedent to any step specified in the Restructuring Implementation Deed

(including, for the avoidance of doubt, any conditions precedent to the effectiveness of any Restructuring Document) shall be satisfied or be capable of satisfaction in accordance with the terms of the Restructuring Implementation Deed (unless waived in accordance with the terms of the Restructuring Implementation Deed).

#### 9 Termination

#### 9.1 Failure of all Plan Steps to occur before the Longstop Time

If any of the Plan Steps do not occur before the Longstop Time or any termination of this Plan in accordance with Clause 9.2.1 occurs, Clause 3 (Plan Effective Date and Plan Claims) shall not apply and, to the extent permitted by applicable law, all Plan Steps will not or will be deemed not to have occurred and any actions taken under or pursuant to Clause 8 (Plan Steps) shall have no valid or binding effect. To the extent permitted by law, all relevant parties agree to take such steps as are necessary and/or desirable to reverse any such Plan Steps that have already occurred in order to restore the parties to the position they were in before the Plan Steps occurred, provided that no party shall be required to incur any material out-of- pocket costs or expenses.

#### 9.2 Termination of this Plan

9.2.1 If the Restructuring Effective Date has not occurred by the Longstop Time, this Plan shall terminate and shall be construed as if it had never become effective and the rights and obligations of the Plan Creditors under the Existing Debt Documents shall not be affected and shall remain in full force (and any default or Event of Default (however described) continuing, subsisting or occurring in connection with the terms of this Plan under and as defined in the relevant Existing Debt Documents shall be deemed not to have been waived and any grace period that expired during the duration of this Plan shall remain expired following the termination of this Plan).

9.2.2 Clauses 1 (Definitions), 9.1 (*Failure of all Plan Steps to occur before the Longstop Time*), 9.2.1, 13 (*Exclusion of liability*), 18 (*Notices*), 19 (*Delegation*) and 20 (*Governing law and jurisdiction*) shall survive any such termination of this Plan.

#### **10 Determinations**

All Plan Claims shall be determined by the Information Agent, on behalf of the Plan Company as at the Record Time in accordance with the mechanism set out in paragraph 2.1 of Part 1 (*Valuation of Plan Claims for Voting Purposes*) of the Explanatory Statement.

#### 11 Assignments and transfers

The Plan Company shall be under no obligation to recognise any assignment or transfer of Plan Claims by a Plan Creditor after the Record Time until (and including) the Restructuring Effective Date provided that, where the Plan Company has received from the relevant parties notice in writing of such assignment or transfer, the Plan Company may, in its sole discretion and subject to the production of such other evidence as it may require, agree to recognise such assignment or transfer. It shall be a term of such recognised by the Plan Company shall be bound by the terms of this Plan, and for the purposes of this Plan shall be a Plan Creditor to the extent of the assigned or transferred Plan Claim.

#### 12 Future insolvency

With effect from the Restructuring Effective Date, this Plan shall continue according to its terms to the fullest extent permitted by law or unless otherwise determined by the Court, notwithstanding any administration, reorganisation, liquidation or other insolvency proceeding in respect of the Plan Company.

#### 13 Exclusion of liability

**13.1** To the extent permitted by law, no Plan Creditor shall be entitled to challenge the validity of any act done or omitted to be done in good faith by any of the Advisers, the Plan Company, each Authorised Person, each of the Existing Agents, the Information Agent or the Super Senior Agents (or any of their respective authorised signatories, agents, employees and/or delegates) (the "**Excluded Persons**") in connection with their actions or omissions pursuant to the provisions of this Plan or the exercise by any of the Excluded Persons in good faith of any power conferred upon them for the purposes of this Plan if exercised in accordance with the provisions of this Plan.

**13.2** To the extent permitted by law, the Plan Company shall not be entitled to challenge the validity of any act done or omitted to be done in good faith by the Excluded Persons in accordance, or in connection, with the provisions of this Plan or the exercise by any of the Excluded Persons in good faith of any power conferred upon it for the purposes of this Plan if exercised in accordance with the provisions of this Plan.

**13.3** No Excluded Person shall be liable for any cost, loss or liability in connection with this Plan unless such loss is attributable to its gross negligence, wilful misconduct or fraud.

**13.4** The Plan Company shall hold each Existing Agent, each Authorised Person and the Information Agent and each of their respective directors, officers, employees, agents and advisers harmless from, and shall indemnify each Existing Agent, each Authorised Person and the Information Agent from and against, any claims, damages, charges, losses, liabilities, costs and expenses which may be incurred by, or asserted or awarded against it in taking any of the steps contemplated by this Plan and executing any releases of security pursuant to the Restructuring Implementation Deed except to the extent that the same arises from the gross negligence, wilful misconduct or fraud of each Existing Agent, each Authorised Person and/or the Information Agent.

With respect to the Plan Creditors or any other person affected or bound by this Plan, each of the Existing Agents undertakes to perform or to observe only such of its covenants or obligations as are specifically set forth in the Intercreditor Agreement and this Plan.

**13.5** Nothing in the Explanatory Statement or this Plan shall impose any obligation on any Existing Agent, Authorised Person or Super Senior Agent to expend its own funds or pay any amount out of its personal assets with respect to any claims made by a Plan Creditor as a result of the each Existing Agent, Authorised Person or Super Senior Agent taking any of the steps contemplated by this Plan and executing any releases of security pursuant to the Restructuring Implementation Deed except to the extent that the same arises from the gross negligence, wilful misconduct or fraud of the relevant Existing Agent, Authorised Person or Super Senior Agent.

**13.6** None of the Existing Agents, the Authorised Persons, the Information Agent or the Super Senior Agents and/or their respective directors, officers, employees, agents and advisers shall be personally responsible or accountable in damages or otherwise to any Plan Creditor or any other person affected or bound by this Plan for any loss, damage or claim incurred by reason of any act or omission performed or omitted by each Existing Agent, each Authorised Person, the Information Agent or the Super Senior Agents in good faith in accordance with this Plan that it reasonably believes to be within the scope of the authority conferred on it by the Intercreditor Agreement or the Amended and Restated Intercreditor Agreement (as applicable) and this Plan.

**13.7** None of the Existing Agents, the Authorised Persons, the Information Agent or the Super Senior Agents shall be personally liable for or on account of any of the statements, representations, warranties, covenants or obligations stated to be those of any Plan Creditors or any other person affected or bound by this Plan, with all such liability, if any, being expressly waived by any such persons claiming by, through or under any of the foregoing.

**13.8** Each Existing Agent, each Authorised Person, the Information Agent and the Super Senior Agents shall at all times be entitled to and may rely on any document notice, consent, order, opinion or certificate given, issued or granted by any person or court that it (acting in good faith) believes to be genuine and correct pursuant to any Restructuring Document and this Plan without being under any obligation to enquire or otherwise determine whether any such notice, consent, order, opinion or certificate is adequate, accurate and/or complete and has been given or granted in accordance with applicable laws or any contractually binding obligation and without being under any responsibility or being under any obligation to validate the legality, effectiveness, completeness, adequacy or enforceability of the Restructuring that is to be implemented as a consequence of this Plan.

#### 14 Modification

The Plan Company may, at the Sanction Hearing, consent on behalf of the Plan Creditors to any modification of or addition to the Plan that the Court may think fit to approve or impose for the purpose of implementing and/or consummating the Plan, and which would not directly or indirectly have a material adverse effect on the rights and interests of any Plan Creditor under this Plan, the Restructuring Implementation Deed or any Restructuring Document.

#### 15 Application to the Court for directions

Without prejudice to any rights that the Plan Company might otherwise have in connection with this Plan or any aspect of it, the Plan Company shall be entitled to make an application to the Court for directions at any time in connection with any matter arising under or in relation to this Plan.

16 Exercise of discretion

Where, under or pursuant to any provision of this Plan, a matter is to be determined by the Plan Company, it shall be determined by the board of directors of the Plan Company, in its discretion and in such manner as they may consider fair and reasonable and after consultation with the Plan Creditors' Counsel. If any difficulty shall arise in determining any such matter either generally or in any particular case or in ensuring the result described above, it shall be resolved by consultation between the Plan Company's Counsel and the Plan Creditors' Counsel, in such manner as the board of directors of the Plan Company shall consider to be fair and reasonable and its decision shall, insofar as permitted by law, be final and binding on all concerned.

## 17 Performance of obligations on dates other than a Business Day

If any obligation is to be performed under the terms of this Plan on a date other than a Business Day and is not capable of being performed on such date, the relevant obligation shall be performed on the next Business Day.

#### 18 Notices

**18.1** Any notice or other written communication to be given under or in relation to this Plan shall be given in the English language in writing and shall be deemed to have been duly given if it is delivered by hand, email, pre-paid recorded delivery or international courier to the address or email address as set out below (or as may be notified by notice to Plan Parties from time to time).

18.2 The addresses for notices are as follows:

18.2.1 in the case of the Plan Company:

by courier or post to: Linklaters LLP, One Silk Street, London, EC2Y  $\rm 8HQ$ 

by email to dlprojectyunlin@linklaters.com;

18.2.2 in the case of a Plan Creditor:

by email to the Information Agent copying in White & Case LLP, as follows:

Attention: Illia Vyshenskyi/ Mu-yen Lo

Email: yunneng@is.kroll.com

In copy: White & Case LLP WCYunlinLendersLegalCounsel@whitecase.com,

and by the Information Agent uploading such notice to the Plan Website;

18.2.3 in the case of the Intercreditor Agent:

by courier or post to: 27th Floor, Two Pacific Place, 88 Queensway, Hong Kong, marked for the attention of Thomas Delmarre / Elissa Sin / Harry He / Boyi Zhang

by email to: thomas.delmarre@ca-cib.com; elissa.sin@ca-cib.com; harry.he@ca-cib.com; boyi.zhang@ca-cib.com (Attention: Thomas Delmarre, Elissa Sin, Harry He and Boyi Zhang);

18.2.4 in the case of the Information Agent:

Attention : Illia Vyshenskyi/ Mu-yen Lo

Email: yunneng@is.kroll.com;

18.2.5 in the case of any other person, any address set forth for that person in any agreement entered into in connection with this Plan including the Restructuring Implementation Deed.

**18.3** Any notice or other written communication to be given under this Plan shall be deemed to have been served:

18.3.1 at the time of delivery if delivered personally;

18.3.2 at the time of transmission if sent by email (provided such written communication is in leaible form):

18.3.3 two (2) Business Days after the time and date of posting if sent by pre-paid recorded delivery if the recipient is in the country of dispatch, otherwise seven (7) Business Days after posting;

18.3.4 seven (7) Business Days after the time and date of posting if sent by international courier; or

18.3.5 when the recipient received (or, having regard to Clause 18.3.2, is deemed to have received) an email notification or other written communication that such notice or other written communication is available through access of the Plan Website.

**18.4** The accidental omission to send any notice, written communication or other document in accordance with this Clause 18, or the non-receipt of any such notice by any Plan Creditor shall not affect the operation of the provisions of this Plan or the validity of anything done in accordance with its terms.

**18.5** The Plan Company shall not be responsible for any loss or delay in the transmission of any notices, other documents or payments posted by or to any Plan Creditors, which shall be posted at the risk of such Plan Creditors.

19 Delegation

**19.1** The Plan Company may perform its rights, powers, duties, discretions and/or obligations through such one or more authorised signatories, acting jointly or severally, as it may appoint from time to time.

**19.2** The Plan Company may also delegate its rights, powers, duties, discretions and/or obligations (including the execution and delivery of any document or instrument) to any person it deems appropriate, in its sole discretion.

#### 20 Governing law and jurisdiction

**20.1** This Plan and any non-contractual obligations arising out of or in connection with it shall be governed by, and construed in accordance with, the laws of England and Wales.

**20.2** Each of the Plan Parties hereby agrees that the Court shall have exclusive jurisdiction to hear and determine any suit, action or proceeding and to settle any dispute which may arise out of the Explanatory Statement or any provision of this Plan, or out of any action taken or omitted to be taken under this Plan or in connection with the administration of this Plan or any non-contractual obligations arising out of or in connection with this Plan and, for such purposes, each of the Plan Parties irrevocably submit to the jurisdiction of the Court, provided, however, that nothing in this Clause 20 shall:

20.2.1 affect the validity of other provisions determining governing law and jurisdiction as between the Plan Company and any of the Plan Parties, whether contained in any contract (including any Restructuring Document) or otherwise; or

20.2.2 prevent the Plan Company from relying upon the provisions of this Plan in any foreign court.

Schedule 1

**Restructuring Implementation Deed** 

Schedule 2

Plan Undertaking Parties

Part I

Plan Undertaking Finance Parties The Plan Undertaking Finance Parties

Atradius Dutch State Business N.V. in its capacity as Coverage Provider in relation to the Atradius Insurance Policy, the Atradius EUR FFS Debt Facility Insurance Policy and the Atradius NTD FFS Debt Facility Insurance Policy

Cathay United Bank Co., Ltd. in its capacity as EUR FFS Debt Facility Agent, Euro Commercial Facility Agent, Guarantee Facilities Agent, Onshore Security Agent, Euro Commercial Super Senior Facility Agent and LC Issuing Bank

Commerzbank Aktiengesellschaft in its capacity as Original Performance LC Facility Lender

Crédit Agricole Corporate and Investment Bank, Hong Kong Branch in its capacity as Intercreditor Agent and Super Senior Agent

Crédit Agricole Corporate and Investment Bank, Taipei Branch in its capacity as Commercial Standby Facility Agent, EKF Facility Agent, KfW Facility Agent, NTD Commercial Facility Agent, NTD FFS Debt Facility Agent, VAT Facility Agent, Working Capital Facility Agent, EKF NTD Super Senior Facility Agent and Commercial NTD Super Senior Facility Agent

EKF Denmark's Credit Support Agency in its capacity as EKF in relation to the EKF Guarantee

KfW IPEX-Bank GmbH in its capacity as KfW, Original Decommissioning LC Facility Lender and Coverage Provider in relation to the KfW Guarantee

Siemens Bank GmbH Singapore Branch in its capacity as EKF Counter Guarantor

SMBC Bank International PLC (formerly known as Sumitomo Mitsui Banking Corporation Europe Limited) in its capacity as Atradius EUR FFS Debt ECA Agent, Atradius Facility ECA Agent, Atradius NTD FFS Debt ECA Agent and Atradius NTD Super Senior Facility ECA Agent

Sumitomo Mitsui Banking Corporation, Düsseldorf Branch in its capacity as Euler Hermes EUR FFS Debt ECA Agent, Euler Hermes Facility ECA Agent, Euler Hermes NTD FFS Debt ECA Agent and Euler Hermes NTD Super Senior Facility ECA Agent

Sumitomo Mitsui Banking Corporation, Taipei Branch in its capacity as Account Bank

Sumitomo Mitsui Banking Corporation, Singapore Branch in its capacity as Offshore Security Agent

Taipei Fubon Commercial Bank Co., Ltd. in its capacity as Atradius EUR FFS Debt Facility Agent, Atradius Facility Agent, Atradius NTD FFS Debt Facility Agent, Euler Hermes EUR FFS Debt Facility Agent, Euler Hermes Facility Agent, Euler Hermes NTD FFS Debt Facility Agent, Euler Hermes NTD Super Senior Facility Agent and Atradius NTD Super Senior Facility Agent

#### Part II

#### **Plan Undertaking Sponsor Parties** The Plan Undertaking Sponsor Parties

Greenwing Energy B.V. in its capacity as Sponsor Starwind Offshore GmbH in its capacity as Sponsor Taiwan Offshore Holding GmbH in its capacity as Sponsor TotalEnergies Renewables SAS in its capacity as Sponsor Skyborn Taiwan Offshore S.á.r.l. in its capacity as a new sponsor Mirai Power Europe Limited in its capacity as new sponsor Yunlin Holding GmbH in its capacity as Parent

#### Schedule 3

#### **Existing Agents**

#### The Existing Agents

Cathay United Bank Co., Ltd. in its capacity as Onshore Security Agent, EUR FFS Debt Facility Agent, Guarantee Facilities Agent and Euro Commercial Facility Agent

Crédit Agricole Corporate and Investment Bank, Hong Kong Branch in its capacity as Intercreditor Agent

Crédit Agricole Corporate and Investment Bank, Taipei Branch in its capacity as Commercial Standby Facility Agent, EKF Facility Agent, KfW Facility Agent, NTD Commercial Facility Agent, NTD FFS Debt Facility Agent, VAT Facility Agent and Working Capital Facility Agent

SMBC Bank International PLC (formerly known as Sumitomo Mitsui Banking Corporation Europe Limited) in its capacity as Atradius Base Facility ECA Agent, Atradius EUR FFS Debt ECA Agent, and Atradius NTD FFS Debt ECA Agent

Sumitomo Mitsui Banking Corporation, Düsseldorf Branch in its capacity as Euler Hermes Base Facility ECA Agent, Euler Hermes EUR FFS Debt Facility ECA Agent and Euler Hermes NTD FFS Debt Facility ECA Agent

Sumitomo Mitsui Banking Corporation, Singapore Branch in its capacity as Offshore Security Agent

Taipei Fubon Commercial Bank Co., Ltd. in its capacity as Atradius Facility Agent, Atradius EUR FFS Debt Facility Agent, Atradius NTD FFS Debt Facility Agent, Euler Hermes Facility Agent, Euler Hermes EUR FFS Debt Facility Agent, Euler Hermes NTD FFS Debt Facility Agent

#### Schedule 4

Allocation Methodology

1 Definitions

1.1 In this Schedule 4:

"Above Pro Rata Commitment" has the meaning given to it in paragraph 3.3 of Part 1;

"Above Pro Rata Election" means the amount (if any) by which an Eligible Senior Lender's Super Senior Election Amount exceeds its Notional Pro Rata Amount;

"Available Tranche A Percentage" means NTD 52,000,000,000 less the aggregate of:

(a) the Total FFS Debt Facility Commitments;

(b) the aggregate commitments under the Euler Hermes Base Facility Tranche A1 (as calculated pursuant to paragraph 2.1.1); and

(c) the aggregate commitments under the Commercial Standby Facility Tranche A1 (as calculated pursuant to paragraph 2.1.1),

as a percentage of the total Commitments under the Base and Standby Facilities less any amounts allocated to Euler Hermes Base Facility Tranche A1 and Commercial Standby Facility Tranche A1 pursuant paragraph 2.1.1:

"Available Tranche B Percentage" means 100% less the Available Tranche A Percentage;

"Base and Standby Facilities" means the Base Facilities and the Commercial Standby Facility Agreement;

"Commercial Standby Facility Tranche A1" has the meaning given to it under paragraph 2.2.2 of Part 2;

"EKF Counter-Guaranteed Commitment" has the meaning given to the term "Tranche B Commitment" in the EKF Base Facility Aareement:

"EKF Counter-Guaranteed Lender Notional Pro Rata Amount" means the Notional Pro Rata Amount attributed to an Eligible Senior Lender which is an EKF Base Facility Lender in respect of its EKF Counter-Guaranteed Commitment in accordance with paragraph 2.3 of Part 1;

"EKF Counter-Guaranteed Retranched Commitments" has the meaning given to the term "Tranche B Commitment" in the Amended and Restated EKF Base Facility Agreement;

"EKF Covered Commitment" has the meaning given to the term "Tranche A Commitment" in the EKF Base Facility Agreement;

"EKF Covered Lender Notional Pro Rata Amount" means the Notional Pro Rata Amount attributed to an Eligible Senior Lender which is an EKF Base Facility Lender in respect of its EKF Covered Commitment in accordance with paragraph 2.3 of Part 1;

"EKF Covered Retranched Commitments" has the meaning given to the term "Tranche A Commitment" in the Amended and Restated EKF Base Facility Agreement;

"Eligible Senior Lenders" means the Base and Standby Plan Creditors:

"Euler Hermes Base Facility Tranche A1" has the meaning given to it under paragraph 2.2.1 of Part 2;

"Fixed FX Rate" means a currency conversion rate of 32.77 NTD to 1 EUR:

"Lower Pro Rata Assumption" has the meaning given to it in paragraph 2.1.1 of Part 1;

"New Tranches" has the meaning given to it under paragraph 1.1 of Part 2:

"Notional Pro Rata Amount" has the meaning given to it in paragraph 2.1.6 of Part 1

"Pro Rata Adjustment" has the meaning given to it in paragraph 2.1.4 of Part 1:

"Pro Rata Calculation Commitment" has the meaning given to it in paragraph 2.1.5 of Part 1:

"Reallocated Commitments" means the total amount of Retranched Drawn Commitments allocated to Tranche B1 and Tranche B2 pursuant to paragraphs 2.2.3(ii) and 2.2.4(ii)(b) of Part 2;

"Relevant Coverage Provider" means, in respect of an Eligible Senior Lender which is a Covered Base Facility Lender under a Covered Base Facility, the relevant Coverage Provider in respect of that Covered Base Facility;

"Relevant Covered Super Senior Facility" means, in respect of an Eligible Senior Lender which is a Covered Base Facility Lender, the Super Senior Facility which is to be covered (or alternatively in whole or in part provided directly) by that Eligible Senior Lender's Relevant Coverage Provider;

"Replacement Covered Super Senior Lender" means, in respect of an Eligible Senior Lender which is a Covered Base Facility Lender (and in its capacity as such), either (i) any third party lender which is to provide lending pursuant to that Eligible Senior Lender's Relevant Covered Super Senior Facility; or (ii) the Relevant Coverage Provider itself:

"Retranched Commitments" means, in respect of each the Eligible Senior Lender, its Commitments under the Amended and Restated Base and Standby Facilities;

"Retranched Drawn Commitments" has the meaning given to it under paragraph 2.2.2 of Part 2:

"Super Senior Election Amount" means the amount which an Eligible Senior Lender has elected to participate (or, in the case of an Eligible Senior Lender which is a Covered Base Facility Lender (and in its capacity as such), the amount which any Replacement Covered Super Senior Lender shall participate) in the Super Senior Facilities as stated in its validly completed Super Senior Election Form submitted in accordance with the procedures set out in the Explanatory Statement:

"Total Above Pro Rata Commitments" means NTD 16,750,000,000 less the aggregate of all Super Senior Election Amounts but excluding the amount by which any Super Senior Election Amount exceeds the relevant Eligible Senior Lender's Notional Pro Rata Amount;

"Total Above Pro Rata Election" means the aggregate of all Above Pro Rata Elections;

"Unadjusted Pro Rata Calculation Commitment" has the meaning given to it under paragraph 2.1.3 of Part 1;

"Undrawn Commercial Standby Facility Amount" means the amount of the undrawn Commitments under the Commercial Standby Facility as at 10 July 2023, being NTD 2,023,494,874;

"Undrawn Euler Hermes Base Facility Amount" means the amount of the undrawn Commitments under the Euler Hermes Base Facility as at 11 July 2023, being NTD 2,917,675,158; and

"Upper Pro Rata Assumption" has the meaning given to it under paragraph 2.1.2 of Part 1.

1.2 In this Schedule 4:

1.2.1 references in clauses and in formulae to NTD 16,750,000,000 as the amount of the Super Senior Facilities may be adjusted in de minimis amounts (following consultation between the Information Agent, the Advisers, and as confirmed by or on behalf of the Intercreditor Agent and uploaded to the Plan Website) to reflect the total amount of the Super Senior Facilities required to reflect the NTD equivalent of EUR 500,000,000, plus amounts in respect of the ECA insurance premia; and

1.2.2 Eligible Senior Lenders that are, in relation to each other, either (i) branches of a single legal entity; or (ii) members of the same corporate group (each an "**Eligible Senior Lender Institution**") shall be treated as a single Eligible Senior Lender such that for the purposes of Part 1, a single Notional Pro Rata Amount shall be calculated in respect of such Eligible Senior Lender Institutions and references in this Schedule 4 to an individual Eligible Senior Lender shall be construed as the Eligible Senior Lender Institutions together shall be construed as the Eligible Senior Lender Institutions together save that, for the purposes of paragraph 1 of Part 1, either or both Eligible Senior Lender Institutions that Super Senior Lender in the Super Senior Facilities.

#### Part 1 – Super Senior Facilities Allocation Methodology

#### 1 Participations in the Super Senior Facility

Each Eligible Senior Lender shall be entitled to participate as a Super Senior Facility Lender in the Super Senior Facilities.

#### 2 Calculation of Notional Pro Rata Amounts

**2.1** For the purposes of this Plan, each Eligible Senior Lender's Notional Pro Rata Amount shall be calculated (by reference to the Commitments of the Eligible Senior Lenders in the Base and Standby Facilities as at the Record Time) as follows:

2.1.1 *First*, a "**Lower Pro Rata Assumption**" for each Eligible Senior Lender shall be calculated as follows:

### Lower Pro Rata Assumption = (NTD 16,750m) $x \frac{A}{R}$ )

#### where:

(i) A = such Eligible Senior Lender's Commitments under the Base and Standby Facilities; and

(ii) B = the aggregate of the Commitments of all Eligible Senior Lenders in the Base and Standby Facilities.

2.1.2 Second, an "Upper Pro Rata Assumption" for each Eligible Senior Lender shall be calculated as follows:

Upper Pro Rata Assumption = ((NTD 16,750m + D) x 
$$\frac{A}{B}$$
) less C

#### where

(i) A and B are as per paragraph 2.1.1 above;

(ii) C = such Eligible Senior Lender's Commitments in the FFS Debt Facilities (if any); and

(iii) D = the Total FFS Debt Facility Commitments.

2.1.3 Third, an **"Unadjusted Pro Rata Calculation Commitment**" for each Eligible Senior Lender shall be calculated as follows:

Unadjusted Pro Rata Calculation Commitment = (50% of (E+F)) where:

(i) E = such Eligible Senior Lender's Lower Pro Rata Assumption;

(ii) F = such Eligible Senior Lender's Upper Pro Rata Assumption; and 2.1.4 Fourth, a "**Pro Rata Adjustment**" for each Eligible Senior Lender shall be calculated as follows:

Pro Rata Adjustment = 
$$G \times \frac{H}{I}$$

where:

(i) G = the total Unadjusted Pro Rata Calculation Commitments of any Eligible Senior Lenders who hold FFS Debt Facility Commitments but who do not hold any other Base Facility Commitments or Commercial Standby Facility Commitments;

(ii) H = such Eligible Senior Lender's Unadjusted Pro Rata Calculation Commitment; and

(iii) I = the total Unadjusted Pro Rata Calculation Commitments of all Eligible Senior Lenders other than any Eligible Senior Lenders whose Unadjusted Pro Rata Calculation Commitments are included in "G" in this paragraph 2.1.4.

2.1.5 Fifth, a "**Pro Rata Calculation Commitment**" for each Eligible Senior Lender shall be calculated by deducting its Pro Rata Adjustment from its Unadjusted Pro Rata Calculation Commitment.

2.1.6 Sixth, a "**Notional Pro Rata Amount**" for each Eligible Senior Lender shall be calculated (in EUR and converted to NTD at the Fixed FX Rate for the purposes of the Final Allocation Schedule) as follows:

## Notional Pro Rata Amount = NTD 16,750m $x \frac{J}{r}$

#### where:

(i) J = such Eligible Senior Lender's total Pro Rata Calculation Commitments; and

(ii)  $\mathsf{K}$  = the total of all Eligible Senior Lenders' Pro Rata Calculation Commitments,

provided that, to the extent that any Eligible Senior Lender's Notional Pro Rata Amount is, following such calculation, equal to or less than EUR 1,500,000 (at the Fixed FX Rate), such Eligible Senior Lender's Notional Pro Rata Amount shall be zero.

**2.2** Subject to paragraph 2.3 below, each Eligible Senior Lender which is a Covered Base Facility Lender shall have a separate Notional Pro Rata Amount in its capacities as (i) a Covered Base Facility Lender under each individual Covered Base Facility (excluding any FFS Debt Facility) and (ii) as a Senior Lender under any other Base and Standby Facilities.

**2.3** Each Eligible Senior Lender which is an EKF Base Facility Lender shall also have a separate Notional Pro Rata Amount in respect of its (i) EKF Covered Commitments; and (ii) EKF Counter-Guaranteed Commitments.

## 3 Allocation of Super Senior Facility Commitments Subject to paragraph 4 below:

**3.1** Each Eligible Senior Lender whose Super Senior Election Amount is equal to its Notional Pro Rata Amount shall be allocated (or its applicable Replacement Covered Super Senior Lender shall be allocated) Super Senior Facility Commitments which are equal to that Eligible Senior Lender's Super Senior Election Amount.

**3.2** Each Eligible Senior Lender whose Super Senior Election Amount is greater than its Notional Pro Rata Amount shall be allocated (or its applicable Replacement Covered Super Senior Lender shall be allocated) Super Senior Facility Commitments which are equal to the sum of:

3.2.1 that Eligible Senior Lender's Notional Pro Rata Amount; and

3.2.2 that Eligible Senior Lender's Above Pro Rata Commitment.

**3.3** The "Above Pro Rata Commitment" of each Eligible Senior Lender shall be calculated as follows:

## Above Pro Rata Commitment = $L \times \frac{M}{N}$

where:

3.3.1 L = the Total Above Pro Rata Commitments;

3.3.2 M = such Eligible Senior Lender's Above Pro Rata Elections; and 3.3.3 N = the Total Above Pro Rata Elections.

**3.4** Each Eligible Senior Lender's allocation of Super Senior Facility Commitments pursuant to paragraphs 3.1 and 3.2 above shall be allocated either directly to such Eligible Senior Lender or, if an Eligible Senior Lender is a Covered Base Facility Lender and if so nominated by that Eligible Senior Lender in its Super Senior Election Form, to a Replacement Covered Super Senior Lender as set out in the Super Senior Facility Commitments Schedule.

## 4 EKF and Euler Hermes specific provisions in relation to the Super Senior Facilities

**4.1** The Above Pro Rata Election for each Eligible Senior Lender under the EKF Base Facility shall be equal to its pro rata share (by reference to its EKF Covered Commitments only) of the NTD equivalent of EUR 18,000,000.

**4.2** The Above Pro Rata Election for each Eligible Senior Lender under the Euler Hermes Base Facility shall be equal to its pro rata share of the NTD equivalent of EUR 32,000,000.

**4.3** For the purposes of the calculations under paragraph 3 above, the EKF Covered Lender Notional Pro Rata Amount of each Eligible Senior Lender under the EKF Base Facility shall be amended pro rata such that the aggregate of the EKF Covered Lender Notional Pro Rata Amounts shall be NTD 3.332.785.391.82.

#### Part 2 – Retranched Commitments Allocation Methodology 1 Retranching of the Base and Standby Facilities

**1.1** Pursuant to the terms of the Restructuring Implementation Deed (once executed), each Base and Standby Facility shall be amended and restated so as to introduce a Tranche A2, Tranche A3, Tranche B1 and Tranche B2 and, in respect of the Euler Hermes Base Facility Agreement and the Commercial Standby Facility Agreement only, a Tranche A1 (the "**New Tranches**").

**1.2** The respective rights and obligations attaching to the New Tranches and the Retranched Drawn Commitments shall be governed by the relevant Amended and Restated Base and Standby Facilities, the Amended and Restated Intercreditor Agreement, the Amended and Restated Common Terms Agreement and any other Existing Debt Document as amended or amended and restated pursuant to the terms of the Restructuring Implementation Deed.

#### 2 Allocation of the Retranched Commitments

2.1 For the purposes of this paragraph 2 of Part 2 of Schedule 4:

2.1.1 references to an Eligible Senior Lender being allocated Retranched Drawn Commitments under the New Tranches shall be to the relevant New Tranches under the Amended and Restated Base and Standby Facility Agreements under which it is a Senior Lender;

2.1.2 where an Eligible Senior Lender is a Senior Lender under more than one of the Base and Standby Facilities, its allocation of the Retranched Commitments in respect of such Amended and Restated Base and Standby Facility Agreements (pursuant to paragraph 2.2) shall be in the same proportions as its Commitments prior to the amendment and restatement of such Amended and Restated Base and Standby Facility Agreements pursuant to the Restructuring Implementation Deed;

2.1.3 in respect of the allocation of Retranched Drawn Commitments under the EKF Base Facility Agreement only:

(i) each EKF Base Facility Lender shall have a separate Super Senior Election Amount in respect of its EKF Covered Commitments and its EKF Counter-Guaranteed Commitments (as stated in its validly completed Super Senior Election Form submitted in accordance with the procedures set out in the Explanatory Statement);

(ii) references to an EKF Base Facility Lender's Notional Pro Rata Amount shall be construed as its EKF Counter-Guaranteed Lender Notional Pro Rata Amount and its EKF Covered Lender Notional Pro Rata Amount separately,

such that for the purposes of the allocation of Retranched Drawn Commitments under the EKF Base Facility Agreement, the EKF Covered Retranched Commitments and the EKF Counter-Guaranteed Retranched Commitments shall be treated as Commitments under separate Amended and Restated Base and Standby Facility Agreements.

**2.2** For the purposes of this Plan, each Eligible Senior Lender's Retranched Commitments shall be allocated to the New Tranches under the Base and Standby Facilities to which it is party as follows: 2.2.1 *First*:

(i) each Euler Hermes Base Facility Lender whose Super Senior Election Amount (if any) is not less than its Notional Pro Rata Amount shall be allocated Retranched Commitments under "Tranche A1" of the Amended and Restated Euler Hermes Base Facility Agreement in an amount equal to its Undrawn Euler Hermes Base Facility Amount (the aggregate of such Tranche A1 Commitments being the "**Euler Hermes Base Facility Tranche A1**");

(ii) each Commercial Standby Facility Lender whose Super Senior Election Amount (if any) is not less than its Notional Pro Rata Amount shall be allocated Retranched Commitments under "Tranche A1" of the Amended and Restated Commercial Standby Facility Agreement in an amount equal to its Undrawn Commercial Standby Facility Amount (the aggregate of such Tranche A1 Commitments being the "Commercial Standby Facility Tranche A1").

2.2.2 Second, each Base and Standby Facility Lender whose Super Senior Election Amount is equal to or greater than its Notional Pro Rata Amount (or whose Notional Pro Rata Share is zero) shall be allocated Retranched Commitments (excluding any Retranched Commitments under the Euler Hermes Base Facility Tranche A1 and the Commercial Standby Facility Tranche A1) (the "**Retranched Drawn Commitments**"):

(i) in the Available Tranche A Percentage to "Tranche A3"; and

(ii) in the Available Tranche B Percentage to "Tranche B1";

2.2.3 *Third*, each Base and Standby Facility Lender whose Super Senior Election Amount is less than 75% of its Notional Pro Rata Amount shall be allocated Retranched Drawn Commitments as follows:

(i) the percentage of its Retranched Drawn Commitments that is equal to its Super Senior Election Amount as a percentage of its Notional Pro Rata Amount (the "**Taken Up Pro Rata Percentage**") shall be allocated:

(a) in the Available Tranche A Percentage to "Tranche A3"; and

(b) in the Available Tranche B Percentage to "Tranche B2";

(ii) the percentage of its Retranched Drawn Commitments that is equal to 100% less its Taken Up Pro Rata Percentage (the "**Untaken Pro Rata Percentage**") shall be allocated as follows:

(a) 10% of the Untaken Pro Rata Percentage of its Retranched Drawn Commitments shall be allocated to "Tranche A3"; and

(b) the remainder of the Untaken Pro Rata Percentage of its Retranched Drawn Commitments shall be allocated to "Tranche B2";

2.2.4 *Fourth*, each Base and Standby Facility Lender whose Super Senior Election Amount is greater than 75% and less than 100% of its Notional Pro Rata Amounts shall be allocated Retranched Drawn Commitments as follows:

(i) the Taken Up Pro Rata Percentage of its Retranched Drawn Commitments shall be allocated:

(a) in the Available Tranche A Percentage to "Tranche A3"; and

(b) in the Available Tranche B Percentage to "Tranche B1";

(ii) the Untaken Pro Rata Percentage of its Retranched Drawn Commitments shall be allocated:

(a) 10% of the Untaken Pro Rata Percentage of its Retranched Drawn Commitments shall be allocated to "Tranche A3";

(b) 25% of the Untaken Pro Rata Percentage of the Retranched Drawn Commitments shall be allocated to "Tranche B1"; and

(c) the remainder of the Untaken Pro Rata Percentage of its Retranched Drawn Commitments shall be allocated:

(I) in the Available Tranche A Percentage to "Tranche A3"; and

(II) in the Available Tranche B Percentage to "Tranche B1".

2.2.5 *Fifth*, each Eligible Senior Lender whose Super Senior Election Amount exceeds its Notional Pro Rata Amount, shall have its Retranched Drawn Commitments that have been allocated to "Tranche B1" pursuant to paragraph 2.2.2 above elevated to "Tranche A2" (the "**Elevated Commitments**") in accordance with the following formula:

## Elevated Commitments = $O \ge \frac{P}{O}$

where:

O = the total Reallocated Commitments

- P = such Eligible Senior Lender's Above Pro Rata Commitments
- Q = the Total Above Pro Rata Commitments. (4431950)

## **COMPANIES**

#### COMPANIES HOUSE DOCUMENTS

#### **COMPANIES ACT 2006- UNREGISTERED COMPANIES**

Notice is hereby given, pursuant to Section 1064 and Section 1077 of the Companies Act 2006 as applied to unregistered companies by the Unregistered Companies Regulations 2009, that in respect of the undermentioned companies, documents of the following description were received by me on the dates indicated.

Company Number	Company Name	Document Type	Date of Receipt
ZC000159	MESSRS HOARE TRUSTEES	CS01	20/07/2023
ZC000065	NATIONAL PROVIDENT INSTITUTION	CS01	03/08/2023
ZC000147	LANDS IMPROVEMEN T COMPANY	AA	04/08/2023
Louise Smyth			

Chief Executive & Registrar of Companies Companies House, Crown Way, Cardiff CF14 3UZ

## **Corporate insolvency**

#### NOTICES OF DIVIDENDS

#### BALTHAZAR CYBER SOLUTIONS LTD

#### 10629746

Registered office: Gill House 140 Holyhead Road, Birmingham, B21 0AF

Principal trading address: Flat 5, Gables Lodge, Cresent East, Barnet, Hertfordshire, EN4 0EL

Notice is hereby given, pursuant to Rule 11.2(1A) of the Insolvency Rules 1986 (as amended), that the Liquidator intends to declare a first dividend to secondary preferential creditors of the Company within two months of the last date for proving specified below.Creditors who have not yet done so must prove their debts by sending their full names and addresses, particulars of their debts or claims, and the names and addresses of their solicitors (if any), to the Liquidator at Gill House, 140 Holyhead Road, Birmingham, B21 0AF by no later than 22 September 2023 (the last date for proving).Creditors who have not proved their debt by the last date for proving may be excluded from the benefit of this dividend or any other dividend declared before their debt is proved.

Date of appointment: 25 February 2021

Office Holder Details: Laura Anne Walshe (IP No. 26250) of Keywood Group Limited, Gill House, 140 Holyhead Road, Birmingham, West Midlands, B21 0AF

Contact information for liquidators: team@keywoodgroup.co.uk Alternative Contact: Steph Jolly

Laura Anne Walshe, Liquidator

29 August 2023

Ag FJ52967

(4430103)

#### DG TRADING LIMITED

#### 06883812

Trading Name: Classic Crockery Event Hire

Registered office: 1580 Parkway, Solent Business Park, Whiteley, Fareham, PO15 7AG

Principal trading address: Unit 1, Felthambrook Industrial Estate, Feltham, TW16 5HY

Notice is hereby given, pursuant to Rule 14.28 of the INSOLVENCY (ENGLAND AND WALES) RULES 2016, that the Joint Liquidators intend to declare a first and final dividend to unsecured creditors of the Company within the period of two months from the last date for proving specified below.

Creditors who have not yet done so must prove their debts by delivering their proofs (in the format specified in Rule 14.4) to the Joint Liquidators at 1580 Parkway, Solent Business Park, Whiteley, Fareham, Hampshire, PO15 7AG by no later than 19 September 2023 (the last date for proving).

Creditors who have not proved their debt by the last date for proving may be excluded from the benefit of this dividend or any other dividend declared before their debt is proved.

In accordance with the provisions of Part 14 of the Insolvency (England and Wales) Rules 2016, a creditor whose debt is a 'small debt' not exceeding £1,000 is deemed to have proved for the purposes of determination and payment of this dividend and is not required to prove their debt as described by this notice unless advised otherwise by the Joint Liquidators.

Date of Appointment: 20 January 2021

Office Holder Details: David Smithson (IP No. 9317) and Andrew Beckingham (IP No. 8683) both of Leonard Curtis, 1580 Parkway, Solent Business Park, Whiteley, Fareham, Hampshire, PO15 7AG

Further details contact: The Joint Liquidators, Tel: 01489 5500440. Alternative contact: Joseph Clayden

David Smithson, Joint Liquidator 30 August 2023

(4430102)

#### **GAMBINO FISH LIMITED**

08645435

(4431889)

Ag FJ53000

In Creditors Voluntary Liquidation

Trading Name: Quality Fish

Registered office: 136 Hertford Road, Enfield, Middlesex, EN3 5AX

Principal trading address: AML Maybrook House, 97 Godstone Road, Caterham, Surrey, CR3 6RE

NOTICE IS HE REBY GIVEN by the Liquidator, Engin Faik of Cornerstone Business Turnaround and Recovery Limited, 136 Hertford Road, Enfield, Middlesex EN3 5AX, pursuant to Rule 14.29 of the Insolvency (England & Wales) Rules 2016 that the Liquidator intends to declare a first and final dividend to the unsecured creditors of the Company within 2 months of the last date for proving set out below.

The unsecured creditors of the Company are required, on or before 29 September 2023 ("the last date for proving"), to prove their debts by sending to the undersigned, Engin Faik of Cornerstone Business Turnaround and Recovery Limited, 136 Hertford Road, Enfield, Middlesex, EN3 5AX, the Liquidator of the Company, written statements of the amounts they claim to be due to them from the Company and, if so requested, to provide such further details or produce such documentary evidence as may appear to the liquidator to be necessary.

A creditor who has not proved this debt before the declaration of any dividend is not entitled to disturb, by reason that he has not participated in it, the distribution of that dividend or any other dividend declared before his debt was proved.

Liquidator: Engin Faik (IP number 9635) of Cornerstone Business Turnaround and Recovery Limited, 136 Hertford Road, Enfield, Middlesex EN3 5AX .

Date of Appointment: 28 July 2016

For further details contact Gulsun Faik on 020 3793 3338 or at gfaik@cornerstonerecovery.co.uk (4431949)

#### MITCHELLS SOUTHERN LIMITED

#### 08638825

Registered office: 1580 Parkway, Solent Business Park, Whiteley, Fareham, PO15 7AG

Principal trading address: Mariners Quay, Shore Road, Warsash, SO31 9FR

Notice is hereby given, pursuant to Rule 14.28 of the INSOLVENCY (ENGLAND AND WALES) RULES 2016, that the Joint Liquidators intend to declare a first and final dividend to unsecured creditors of the Company within the period of two months from the last date for proving specified below.

Creditors who have not yet done so must prove their debts by delivering their proofs (in the format specified in Rule 14.4) to the Joint Liquidators at 1580 Parkway, Solent Business Park, Whiteley, Fareham, Hampshire, PO15 7AG by no later than 19 September 2023 (the last date for proving).

Creditors who have not proved their debt by the last date for proving may be excluded from the benefit of this dividend or any other dividend declared before their debt is proved.

In accordance with the provisions of Part 14 of the Insolvency (England and Wales) Rules 2016, a creditor whose debt is a 'small debt' not exceeding £1,000 is deemed to have proved for the purposes of determination and payment of this dividend and is not required to prove their debt as described by this notice unless advised otherwise by the Joint Liquidators.

Date of Appointment: 13 January 2020

Office Holder Details: David Smithson (IP No. 9317) and Andrew Beckingham (IP No. 8683) both of Leonard Curtis, 1580 Parkway, Solent Business Park, Whiteley, Fareham, Hampshire, PO15 7AG

Further details contact: The Joint Liquidators, Tel: 01489 550440. Alternative contact: Joseph Clayden.

David Smithson, Joint Liquidator

30 August 2023 Ag FJ53015

(4430104)

#### **R PLASTICS LIMITED**

04081363

Previous Name of Company: Rotalac Plastics Limited

Registered office: Baird House, Seebeck Place, Milton Keynes, Buckinghamshire, MK5 8FR

Principal trading address: Southmoor Road, Roundthorn Ind Estate, Manchester, M23 9DS

Notice is hereby given, pursuant to Rule 14.28 of the Insolvency (England and Wales) Rules 2016, that the Joint Liquidators intend to declare a First and final dividend to creditors of the Company within the period of two months from the last date for proving specified below.

Creditors who have not yet done so must prove their debts by delivering their proofs (in the format specified in Rule 14.4) to the Joint Liquidators at FTS Recovery Limited, Baird House, Seebeck Place, Milton Keynes, Buckinghamshire, MK5 8FR by no later than 21 September 2023 (the last date for proving).

Creditors who have not proved their debt by the last date for proving may be excluded from the benefit of this dividend or any other dividend declared before their debt is proved.

Date of Appointment: 11 October 2016

Office Holder Details: Alan Coleman (IP No. 9402) and Marco Piacquadio (IP No. 19910) both of FTS Recovery Limited, Ground Floor, Baird House, Seebeck Place, Knowlhill, Milton Keynes, MK5 8FR

Further details contact: The Joint Liquidators, Tel: 0161 938 0240. Alternative contact: David Appleby, Email: david.appleby@ftsrecovery.co.uk

Alan Coleman, Joint Liquidator

30 August 2023

Ag FJ53024

#### VALDE CONSTRUCTION LIMITED

07879117

In Creditors Voluntary Liquidation

Trading Name: Valde Construction

Registered office: c/o Connect Insolvency Limited, 30/32 Aston House, Redburn Road, Westerhope, Newcastle upon Tyne, NE5 1NB & Swarland East House, Swarland, Morpeth, Northumberland, NE65 9HZ

NOTICE IS HEREBY GIVEN by the Liquidator, Simon Blakey of Connect Insolvency Limited, 30/32 Aston House, Redburn Road, Westerhope, Newcastle upon Tyne, NE5 1NB, pursuant to Rule 14.28 of the Insolvency (England & Wales) Rules 2016 that I intend to declare a first and final dividend to the creditors of the Company within 2 months of the last date for proving set out below.

The creditors of the Company are required, on or before 29 September 2023 ("the last date for proving"), to prove their debts by sending to the undersigned, Simon Blakey of Connect Insolvency Limited, 30/32 Aston House, Redburn Road, Westerhope, Newcastle upon Tyne, NE5 1NB, the Liquidator of the Company, written statements of the amounts they claim to be due to them from the Company and, if so requested, to provide such further details or produce such documentary evidence as may appear to the Liquidator to be necessary.

A creditor who has not proved this debt before the declaration of any dividend is not entitled to disturb, by reason that he has not participated in it, the distribution of that dividend or any other dividend declared before his debt was proved.

Date of Appointment: 20 August 2019

For further details contact Simon Blakey on 0191 816 1416 or at simon@connectinsolvency.com Dated: 30 August 2023

(4431862)

#### **OTHER CORPORATE INSOLVENCY NOTICES**

In the High Court of Justice, Business and Property Courts of England and Wales Insolvency and Companies List (ChD)

In the matter of Viola Money (Europe) Ltd (in Special Administration) Case No: CR-2021-002402

And in the matter of the Payment and Electronic Money Institution Insolvency Regulations 2021

And in the matter of the Payment and Electronic Money Institution Insolvency (England and Wales) Rules 2021

And in the matter of the Insolvency Act 1986

UPON the application of Edward George Boyle, of Interpath Ltd, 10 Fleet Place. London EC4M 7RB.

as joint administrator of Viola Money (Europe) Ltd (in special administration) (the "Company") dated

10 August 2023 that Robert Thomas Spence, of Interpath Ltd, 10 Fleet Place, London EC4M 7RB, be appointed as joint administrator under the Payment and Electronic Money Institution Insolvency

(England and Wales) Rules 2021 (the "Rules") and the Insolvency Act 1986 (as applied by regulation

37 of the Payment and Electronic Money Institution Insolvency Regulations 2021) (the "Act").

AND UPON James Robert Bennett having delivered notice of his intention to resign as joint administrator of the Company with effect from 14 August 2023.

AND UPON READING the evidence filed.

NOTICE IS HEREBY GIVEN that pursuant to an order of Deputy ICC Judge Schaffer dated 14 August 2023 (Order) that:

1. Mr Spence is appointed as joint administrator of the Company pursuant to paragraph 91(1) of Schedule B1 to the Act and Rule 177;

2. Mr Spence shall manage the affairs, business and property of the Company during the period of the special administration, together with Edward George Boyle;

3. Any act required or authorised under any enactment to be done by either or all of the joint administrators may be done by any one or more of the persons for the time being holding that office;

4. The appointment of Mr Spence as joint administrator of the Company be gazetted pursuant to Rules 17 and 178 and notification be given to Companies House pursuant to Rule 180; and

5 This order takes effect on 14 August 2023 (being the date of Mr Bennett's resignation as joint administrator of the Company).

(4431376)

#### **RE-USE OF A PROHIBITED NAME**

NOTICE TO THE CREDITORS OF AN INSOLVENT COMPANY **REGARDING THE RE-USE OF A PROHIBITED NAME - RULE 22.4** OF THE INSOLVENCY (ENGLAND AND WALES) RULES 2016 1111 HOLDINGS LTD

12326279

(4430105)

Registered office: 28 Kansas Avenue, Salford, M50 2GL

Trading Name: 'Eleven Eleven' and '11:11'

On 31 August 2023 the above-named company will enter into insolvent liquidation.

I, Mark Gerard Mann of 98 Park Road, Prestwich, M25 0DY am a director of the above-named company during the 12 months ending with the day before it goes into liquidation.

I give notice that it is my intention to act in all or any of the ways specified in section 216(3) of the Insolvency Act 1986 in connection with, or for the purposes of, the carrying on of the whole or substantially the whole of the business of the insolvent company under the following name:

Eleven Eleven UK Group Limited trading as '1111', '11.11' and 'Eleven Eleven'

Rule 22.5 - Statement as to the effect of the notice under rule 22.4(2):

"Section 216(3) of the Insolvency Act 1986 lists the activities that a director of a company that has gone into insolvent liquidation may not undertake unless the court gives permission or there is an exception in the Insolvency Rules made under the Insolvency Act 1986. (This includes the exceptions in Part 22 of the Insolvency (England and Wales) Rules 2016). These activities are-

(a) acting as a director of another company that is known by a name which is either the same as a name used by the company in insolvent liquidation in the 12 months before it entered liquidation or is so similar as to suggest an association with that company;

(b) directly or indirectly being concerned or taking part in the promotion, formation or management of any such company; or

(c) directly or indirectly being concerned in the carrying on of a business otherwise than through a company under a name of the kind mentioned in (a) above.

This notice is given under rule 22.4 of the Insolvency (England and Wales) Rules 2016 where the business of a company which is in, or may go into, insolvent liquidation is, or is to be, carried on otherwise than by the company in liquidation with the involvement of a director of that company and under the same or a similar name to that of that company.

The purpose of the giving of this notice is to permit the director to act in these circumstances where the company enters (or has entered) insolvent liquidation without the director committing a criminal offence and in the case of the carrying on of the business through another company, being personally liable for that company's debts.

Notice may be given where the person giving the notice is already the director of a company which proposes to adopt a prohibited name". (4431894)

NOTICE TO THE CREDITORS OF AN INSOLVENT COMPANY REGARDING THE RE-USE OF A PROHIBITED NAME - RULE 22.4 OF THE INSOLVENCY (ENGLAND AND WALES) RULES 2016 1111 HOLDINGS LTD 12326279

Registered office: 28 Kansas Avenue, Salford, M50 2GL

Trading Name: 'Eleven Eleven' and '11:11'

On 31 August 2023 the above-named company will enter into insolvent liquidation.

I, Paula Jayne Thomas of 9 Green Ash, Denshaw Road, Delph OL3 5LF am a director of the above-named company during the 12 months ending with the day before it goes into liquidation.

I give notice that it is my intention to act in all or any of the ways specified in section 216(3) of the Insolvency Act 1986 in connection with, or for the purposes of, the carrying on of the whole or substantially the whole of the business of the insolvent company under the following name:

Eleven UK Group Limited trading as '1111', '11.11' and 'Eleven Eleven'

## Rule 22.5 - Statement as to the effect of the notice under rule 22.4(2):

"Section 216(3) of the Insolvency Act 1986 lists the activities that a director of a company that has gone into insolvent liquidation may not undertake unless the court gives permission or there is an exception in the Insolvency Rules made under the Insolvency Act 1986. (This includes the exceptions in Part 22 of the Insolvency (England and Wales) Rules 2016). These activities are-

(a) acting as a director of another company that is known by a name which is either the same as a name used by the company in insolvent liquidation in the 12 months before it entered liquidation or is so similar as to suggest an association with that company;

(b) directly or indirectly being concerned or taking part in the promotion, formation or management of any such company; or

(c) directly or indirectly being concerned in the carrying on of a business otherwise than through a company under a name of the kind mentioned in (a) above.

This notice is given under rule 22.4 of the Insolvency (England and Wales) Rules 2016 where the business of a company which is in, or may go into, insolvent liquidation is, or is to be, carried on otherwise than by the company in liquidation with the involvement of a director of that company and under the same or a similar name to that of that company.

The purpose of the giving of this notice is to permit the director to act in these circumstances where the company enters (or has entered) insolvent liquidation without the director committing a criminal offence and in the case of the carrying on of the business through another company, being personally liable for that company's debts. Notice may be given where the person giving the notice is already the director of a company which proposes to adopt a prohibited name". (4431895)

#### NOTICE TO THE CREDITORS OF AN INSOLVENT COMPANY REGARDING THE RE-USE OF A PROHIBITED NAME - RULE 22.4 OF THE INSOLVENCY (ENGLAND AND WALES) RULES 2016 HI-CALIBRE PERSONNEL LTD

02256651

Registered office: 11 CASTLE HILL, MAIDENHEAD, BERKSHIRE, SL6 4AA

Principal trading address: Trading address BRINSMEAD, SAND HILLS, KNOWL HILL, READING, RG10 9XX

On 29 AUGUST 2023, a voluntary arrangement under Part 1 of the Insolvency Act 1986 was approved in respect of the company ("CVA").

I, PAULINE COX of BRINSMEAD, SAND HILLS, KNOWL HILL, READING, BERKS, RG10 9XX was a director of the above-named company on the day the CVA under Part 1 of the Insolvency Act 1986 was approved.

I give notice that

It is my intention to act in all or any of the ways specified in section 216(3) of the Insolvency Act 1986 if the above-named company were to go into insolvent liquidation in connection with, or for the purposes of, the carrying on of the whole or substantially the whole of the business of the above-named company under the following name:

HI-CALIBRE INTERNATIONAL

On 29 TH AUGUST 2023 the above-named company went into insolvent liquidation.

I, PAULINE COX of BRINSMEAD, SAND HILLS, KNOWL HILL, READING, BERKS, RG10 9XX was a director of the above-named company during the 12 months ending with the day before it went into liquidation.

I give notice that it is my intention to act in all or any of the ways specified in section 216(3) of the Insolvency Act 1986 in connection with, or for the purposes of, the carrying on of the whole or substantially the whole of the business of the insolvent company under the following name:

HI-CALIBRE INTERNATIONAL

Rule 22.5 - Statement as to the effect of the notice under rule 22.4(2):

"Section 216(3) of the Insolvency Act 1986 lists the activities that a director of a company that has gone into insolvent liquidation may not undertake unless the court gives permission or there is an exception in the Insolvency Rules made under the Insolvency Act 1986. (This includes the exceptions in Part 22 of the Insolvency (England and Wales) Rules 2016). These activities are-

(a) acting as a director of another company that is known by a name which is either the same as a name used by the company in insolvent liquidation in the 12 months before it entered liquidation or is so similar as to suggest an association with that company;

(b) directly or indirectly being concerned or taking part in the promotion, formation or management of any such company; or

(c) directly or indirectly being concerned in the carrying on of a business otherwise than through a company under a name of the kind mentioned in (a) above.

This notice is given under rule 22.4 of the Insolvency (England and Wales) Rules 2016 where the business of a company which is in, or may go into, insolvent liquidation is, or is to be, carried on otherwise than by the company in liquidation with the involvement of a director of that company and under the same or a similar name to that of that company.

The purpose of the giving of this notice is to permit the director to act in these circumstances where the company enters (or has entered) insolvent liquidation without the director committing a criminal offence and in the case of the carrying on of the business through another company, being personally liable for that company's debts.

Notice may be given where the person giving the notice is already the director of a company which proposes to adopt a prohibited name". (4431893)

#### NOTICE TO THE CREDITORS OF AN INSOLVENT COMPANY REGARDING THE RE-USE OF A PROHIBITED NAME - RULE 22.4 OF THE INSOLVENCY (ENGLAND AND WALES) RULES 2016 KENNEDYS BAR AND RESTAURANT LIMITED 07652210

Trading Name: Kennedys Bar And Restaurant Limited

Registered office: 1 Little Stonegate York YO1 8AX

Registered office: Trading address:1 Little Stonegate York YO1 8AX On 2 August 2023 the above-named company went into insolvent liquidation.

I, Karen Waugh of 1 Little Stonegate York YO1 8AX was a director of the above-named company during the 12 months ending with the day before it went into liquidation.

I give notice that it is my intention to act in all or any of the ways specified in section 216(3) of the Insolvency Act 1986 in connection with, or for the purposes of, the carrying on of the whole or substantially the whole of the business of the insolvent company under the following name:

Kennedys Bar and Restaurant

Rule 22.5 - Statement as to the effect of the notice under rule 22.4(2):

"Section 216(3) of the Insolvency Act 1986 lists the activities that a director of a company that has gone into insolvent liquidation may not undertake unless the court gives permission or there is an exception in the Insolvency Rules made under the Insolvency Act 1986. (This includes the exceptions in Part 22 of the Insolvency (England and Wales) Rules 2016). These activities are-

(a) acting as a director of another company that is known by a name which is either the same as a name used by the company in insolvent liquidation in the 12 months before it entered liquidation or is so similar as to suggest an association with that company;

(b) directly or indirectly being concerned or taking part in the promotion, formation or management of any such company; or

(c) directly or indirectly being concerned in the carrying on of a business otherwise than through a company under a name of the kind mentioned in (a) above. This notice is given under rule 22.4 of the Insolvency (England and Wales) Rules 2016 where the business of a company which is in, or may go into, insolvent liquidation is, or is to be, carried on otherwise than by the company in liquidation with the involvement of a director of that company and under the same or a similar name to that of that company.

The purpose of the giving of this notice is to permit the director to act in these circumstances where the company enters (or has entered) insolvent liquidation without the director committing a criminal offence and in the case of the carrying on of the business through another company, being personally liable for that company's debts.

Notice may be given where the person giving the notice is already the director of a company which proposes to adopt a prohibited name".

(4431897)

#### NOTICE TO THE CREDITORS OF AN INSOLVENT COMPANY OF THE RE-USE OF A PROHIBITED NAME RULE 22.4 OF THE INSOLVENCY (ENGLAND AND WALES) RULES 2016 LANGLEYS RESTAURANT LIMITED

08455339

(In Liquidation)

Registered office: 1 Park Road, Hampton Wick, Kingston Upon Thames, England, KT1 4AS

On 31 August 2023, the above-named company went into insolvent liquidation.

I, Bevan Hoogewerf, of 79 Berrylands, Surbiton, KT5 8JX was a director of the above-named company during the 12 months ending with the day before it went into liquidation.

I give notice that it is my intention to act in one or more of the ways specified in section 216(3) of the Insolvency Act 1986 in connection with, or for the purposes of, the carrying on of the whole or substantially the whole of the business of the insolvent company under the following name:

Langley's Restaurant

## Rule 22.5 - Statement as to the effect of the notice under rule 22.4(2)

Section 216(3) of the Insolvency Act 1986 lists the activities that a director of a company that has gone into insolvent liquidation may not undertake unless the court gives permission or there is an exception in the Insolvency Rules made under the Insolvency Act 1986. (This includes the exceptions in Part 22 of the Insolvency (England and Wales) Rules 2016). These activities are -

(a) acting as a director of another company that is known by a name which is either the same as a name used by the company in insolvent liquidation in the 12 months before it entered liquidation or is so similar as to suggest an association with that company;

(b) directly or indirectly being concerned or taking part in the promotion, formation or management of any such company; or

(c) directly or indirectly being concerned in the carrying on of a business otherwise than through a company under a name of the kind mentioned in (a) above. This notice is given in pursuance of Rule 22.4 of the Insolvency (England and Wales) Rules 2016 where the business of a company which is in, or may go into, insolvent liquidation is, or is to be, carried on otherwise than by the company in liquidation with the involvement of a director of that company and under the same or a similar name to that of that company.

The purpose of the giving of this notice is to permit the director to act in these circumstances where the company enters (or has entered) insolvent liquidation without the director committing a criminal offence and in the case of the carrying on of the business through another company, being personally liable for that company's debts.

Notice may be given where the person giving the notice is already the director of a company which proposes to adopt a prohibited name. (4431898)

#### NOTICE TO THE CREDITORS OF AN INSOLVENT COMPANY REGARDING THE RE-USE OF A PROHIBITED NAME - RULE 22.4 OF THE INSOLVENCY (ENGLAND AND WALES) RULES 2016 MEDITERRANEAN RESTAURANT & BAR LIMITED 12060561

Registered office: 6 High Street, Harrow, England, HA1 3LW Trading Name: Eighty-Six Restaurant Bar

On 04 September 2023 the above-named company will enter into insolvent liquidation.

I, Massih Mirzai of 86 High Street, Harrow, HA1 3LW am a director of the above-named company during the 12 months ending with the day before it enters into liquidation.

give notice that it is my intention to act in all or any of the ways specified in section 216(3) of the Insolvency Act 1986 in connection with, or for the purposes of, the carrying on of the whole or substantially the whole of the business of the insolvent company under the following name:

Bar 86 Ltd trading as Eighty-Six Restaurant Bar

## Rule 22.5 - Statement as to the effect of the notice under rule 22.4(2):

"Section 216(3) of the Insolvency Act 1986 lists the activities that a director of a company that has gone into insolvent liquidation may not undertake unless the court gives permission or there is an exception in the Insolvency Rules made under the Insolvency Act 1986. (This includes the exceptions in Part 22 of the Insolvency (England and Wales) Rules 2016). These activities are-

(a) acting as a director of another company that is known by a name which is either the same as a name used by the company in insolvent liquidation in the 12 months before it entered liquidation or is so similar as to suggest an association with that company;

(b) directly or indirectly being concerned or taking part in the promotion, formation or management of any such company; or

(c) directly or indirectly being concerned in the carrying on of a business otherwise than through a company under a name of the kind mentioned in (a) above. This notice is given under rule 22.4 of the Insolvency (England and Wales) Rules 2016 where the business of a company which is in, or may go into, insolvent liquidation is, or is to be, carried on otherwise than by the company in liquidation with the involvement of a director of that company and under the same or a similar name to that of that company.

The purpose of the giving of this notice is to permit the director to act in these circumstances where the company enters (or has entered) insolvent liquidation without the director committing a criminal offence and in the case of the carrying on of the business through another company, being personally liable for that company's debts.

Notice may be given where the person giving the notice is already the director of a company which proposes to adopt a prohibited name".

(4431896)

#### NOTICE TO THE CREDITORS OF AN INSOLVENT COMPANY OF THE RE-USE OF A PROHIBITED NAME RULE 22.4 OF THE INSOLVENCY (ENGLAND AND WALES) RULES

#### 2016 MUSHROOM MANAGEMENT LTD

#### 10693088

Registered office: 11 Roman Way Business Centre, Berry Hill, Droitwich Spa, Worcestershire, WR9 9AJ

On 24th July 2023 the above-named company went into insolvent liquidation. I, Edward Surman of Third Floor, Library Building, Sun Street, Tewkesbury, GL20 5NX, was a director of the above-named company during the 12 months ending with the day before it went into liquidation.

I give notice that it is my intention to act in one or more of the ways specified in section 216(3) of the Insolvency Act 1986 in connection with, or for the purposes of, the carrying on of the whole or substantially the whole of the business of the insolvent company under the following name: Mushroombiz

I would not otherwise be permitted to undertake those activities without the leave of the court or the application of an exception created by Rules made under the Insolvency Act 1986.

Breach of the prohibition created by section 216 of The Insolvency Act 1986 is a criminal offence.

## Rule 22.5 - Statement as to the effect of the notice under rule 22.4(2):

Section 216(3) of the Insolvency Act 1986 lists the activities that a director of a company that has gone into insolvent liquidation may not undertake unless the court gives permission or there is an exception in the Insolvency Rules made under the Insolvency Act 1986. (This includes the exceptions in Part 22 of the Insolvency (England and Wales) Rules 2016.) These activities are -

(a) acting as a director of another company that is known by a name which is either the same as a name used by the company in insolvent liquidation in the 12 months before it entered liquidation or is so similar as to suggest an association with that company;

(b) directly or indirectly being concerned or taking part in the promotion, formation or management of any such company; or

(c) directly or indirectly being concerned in the carrying on of a business otherwise than through a company under a name of the kind mentioned in (a) above.

This notice is given under rule 22.4 of the Insolvency (England and Wales) Rules 2016 where the business of a company which is in, or may go into, insolvent liquidation is, or is to be, carried on otherwise than by the company in liquidation with the involvement of a director of that company and under the same or a similar name to that of that company.

The purpose of giving this notice is to permit the director to act in these circumstances where the company enters (or has entered) insolvent liquidation without the director committing a criminal offence and in the case of the carrying on of the business through another company, being personally liable for that company's debts.

Notice may be given where the person giving the notice is already the director of a company which proposes to adopt a prohibited name.

(4431012)

#### PICK ME UP LTD

12055484

Trading Name: Lykke

Registered office: 16 Fitzalan Square, Sheffield, S1 2AZ

Principal trading address: Unit 8, 11 New Era Square, Sheffield, S2 4RB

Notice to the creditors of an insolvent company of the re-use of a prohibited name - Rule 22.4 of the Insolvency (England and Wales) Rules 2016

On 29 August 2023 the company entered administration.

I, Alex Moore of KMU Group, 14 Fitzalan Square, Sheffield, S2 4RB was a director of the above-named company on the day it entered administration.

I give notice that I am acting and intend to continue to act in one or more of the ways to which section 216(3) of the Insolvency Act 1986 would apply if the above-named company were to go into insolvent liquidation.

Rule 22.5 - Statement as to the effect of the notice under rule 22.4(2):

Section 216(3) of the Insolvency Act 1986 lists the activities that a director of a company that has gone into insolvent liquidation may not undertake unless the court gives permission or there is an exception in the Insolvency Rules made under the Insolvency Act 1986. (This includes the exceptions in Part 22 of the Insolvency (England and Wales) Rules 2016). These activities are- (a) acting as a director of another company that is known by a name which is either the same as a name used by the company in insolvent liquidation in the 12 months before it entered liquidation or is so similar as to suggest an association with that company; (b) directly or indirectly being concerned or taking part in the promotion, formation or management of any such company; or (c) directly or indirectly being concerned in the carrying on of a business otherwise than through a company under aname of the kind mentioned in (a) above.

This notice is given in pursuance of Rule 22.4 of the Insolvency (England and Wales) Rules 2016 where the business of a company which is in, or may go into, insolvent liquidation is, or is to be, carried on otherwise than by the company in liquidation with the involvement of a director of that company and under the same or a similar name to that of that company.

The purpose of the giving of this notice is to permit the director to act in these circumstances where the company enters (or has entered) insolvent liquidation without the director committing a criminal offence and in the case of the carrying on of the business through another company, being personally liable for that company's debts.

Notice may be given where the person giving the notice is already the director of a company which proposes to adopt a prohibited name. Ag FJ52947 (4430101)

#### **Administration**

#### **APPOINTMENT OF ADMINISTRATORS**

In the High Court of Justice, Business and Property Courts of England and Wales

Court Number: CR-2023-004115 BUCKS CONCRETE LIMITED

(Company Number 11923085)

Nature of Business: Manufacture of Concrete Products

Registered office: Ability House, 121 Brooker Road, Waltham Abbey, Essex, EN9 1JH

Principal trading address: Finmere Quarry, Banbury Road, Finmere, Buckinghamshire, MK18  $4\mathrm{AJ}$ 

Administrator: *Edwin D S Kirker* (IP number 8227) of Kirker & Co, Centre 645, 2 Old Brompton Road, London, SW7 3DQ .

Date of Appointment: 27 August 2023

For further details contact Edwin Kirker on 020 7580 6030 or at edwin@kirker.co.uk (4431948)

In the High Court of Justice Business and Property Courts of England and Wales Insolvency and Companies Court (ChD)

#### No CR-2023-004529. M.PRICE LIMITED

(Company Number 00604114)

Nature of Business: Glazing

Registered office: 3rd Floor, Westfield House, 60 Charter Row, Sheffield, S1 3FZ

Joint Administrator: *Kris Anthony Wigfield* (IP number 22672) of Begbies Traynor (SY) LLP, 3rd Floor, Westfield House 60 Charter Row Sheffield S1 3FZ.

Joint Administrator: *John Hedger* (IP number 9601) of Seneca Insolvency Practitioners Ltd, Speedwell Mill Old Coach Road Tansley Matlock Derbyshire DE4 5FY.

Date of Appointment: 25 August 2023

For further details contact Avis Wong on 0114 2755033 or at sheffield.north@btguk.com (4431007)

In the High Court of Justice, Business and Property Courts of England Office Holder/s: Stuart Rathmell, of STUART RATHMELL and Wales, Insolvency and Companies List (ChD) INSOLVENCY and, Egyptian Mill, Egyptian Street, Bolton BL1 2HS, No CR-2023- 004676 of 2023 Telephone: 01204867615, Email address: ROMANSLEIGH DEVELOPMENTS LIMITED stuart.rathmell.insolvency@outlook.com Office Holder Number/s: 10050 (Company Number 09906141) Registered office: C/O L T Management Services Ltd, 31 New Road, Date of appointment: 08 August 2023 Haverscroft Industrial Estate, Attleborough, NR17 1YE By whom Appointed: Members and Creditors Principal trading address: Odam Hill, Romansleigh, South Molton, Tuesday 08 August 2023 (4431771) EX36 4NB Nature of Business: Recreational vehicle parks, trailer parks and camping grounds. Name of Company: L.SMART ENGINEERING LTD Company Number: 10839624 Date of Appointment: 24 August 2023. Joint Administrator's Name and Address: Sarah Collins (IP No. 23872) Company Type: Registered Company of Alvarez & Marsal Europe LLP, Suite 3 Regency House, 91 Western Nature of the business: Engineering Road, Brighton, BN1 2NW. Type of Liquidation: Creditors' Voluntary Telephone: +44 (0) 20 7715 5200. Registered office: 51 Pretoria Road, Tonyrefail, Porth CF39 8PW Joint Administrator's Name and Address: Robert Croxen (IP No. 9700) Principal trading address: 51 Pretoria Road, Tonyrefail, Porth CF39 of Alvarez & Marsal Europe LLP, Suite 3 Regency House, 91 Western 8PW Holder/s: Stuart Rathmell, of STUART RATHMELL Road, Brighton, BN1 2NW. Telephone: +44 (0) 20 7715 5200. Office For further information contact Daniel Cudlip at the offices of Alvarez INSOLVENCY and, Egyptian Mill, Egyptian Street, Bolton BL1 2HS, & Marsal Europe LLP on +44 (0) 161 504 1700, or Telephone: 01204867615, Email address: INS\_ROMRDL@alvarezandmarsal.com stuart.rathmell.insolvency@outlook.com (4431864) 25 August 2023 Office Holder Number/s: 10050 Date of appointment: 17 August 2023 By whom Appointed: Members and Creditors In the High Court of Justice Thursday 17 August 2023 (4431773)Court Number: CR-2023-004436 SUREGREEN LIMITED (Company Number 08563588) Name of Company: ROONEY ACADEMY LTD Nature of Business: Retail sale via mail order houses or via internet Company Number: 12256964 Registered office: Viridis House, Toppesfield Road, Finchingfield, Company Type: Registered Company Braintree, Essex, England, CM7 4NZ Nature of the business: Education Principal trading address: Viridis House, Toppesfield Road, Type of Liquidation: Creditors' Voluntary Finchingfield, Braintree, Essex, England, CM7 4NZ Registered office: 50 Redwood Drive, Chorley PR7 3BW Date of Appointment: 24 August 2023 Principal trading address: 50 Redwood Drive, Chorley PR7 3BW Office Holder Details: Lee De'ath (IP No. 9316) of Begbies Traynor Office Holder/s: Stuart Rathmell and Stuart Rath, of STUART (Central) LLP, Town Wall House, Balkerne Hill, Colchester, Essex, CO3 RATHMELL INSOLVENCY and, Egyptian Mill, Egyptian Street, Bolton 3AD and Kirstie Jane Provan (IP No. 009681) of Begbies Traynor BL1 2HS. Telephone: 01204867615, Email address: (London) LLP, 31st Floor, 40 Bank Street, London, E14 5NR and Tom stuart.rathmell.insolvency@outlook.com Gardiner (IP No. 17990) of Begbies Traynor (Central) LLP, Town Wall Office Holder Number/s: 10050 House, Balkerne Hill, Colchester, Essex, CO3 3AD Date of appointment: 25 August 2023 Further details contact: Charlie Robinson of Begbies Traynor (Central) By whom Appointed: Members and Creditors Friday 25 August 2023 (4431784) LLP, Tel: 01206 217900 or by email: charlie.robinson@btguk.com (4430106) Ag FJ52954 Name of Company: SENTIRO (UK) LIMITED Company Number: 06329599 **Creditors' voluntary liquidation** Company Type: Registered Company Nature of the business: Other telecommunications activities APPOINTMENT OF LIQUIDATORS Type of Liquidation: Creditors' Voluntary Registered office: 9 Birnam House, 20 Cambridge Park, Twickenham, Name of Company: DIAMOND KEBAB BRISTOL LTD Middlesex TW1 2JD England Company Number: 12008331 Principal trading address: 9 Birnam House, 20 Cambridge Park, Company Type: Registered Company Twickenham, Middlesex TW1 2JD England Office Holder/s: Christopher David Horner, of Robson Scott Nature of the business: Unlicenced restaurants and cafes Type of Liquidation: Creditors' Voluntary Associates Ltd T/A BusinessRescueExpert and, 47-49 Duke Street, Registered office: 28 Park Street, Bristol BS1 5JA Darlington, Co. Durham DL3 7SD, Telephone: 01325 365950, Email Principal trading address: 28 Park Street, Bristol BS1 5JA address: admin@businessrescueexpert.co.uk Office Holder/s: M. Usman Nazir, of A N G, Castlemead, Lower Castle Office Holder Number/s: 16150 Street, Bristol BS1 3AG, Telephone: 01173252505, Email address: Date of appointment: 30 August 2023 info@ang.uk.com By whom Appointed: Members and Creditors Office Holder Number/s: 11290 Wednesday 30 August 2023 (4430935) Date of appointment: 29 August 2023 By whom Appointed: Members (4431710) Name of Company: HUNGRY BUDDHA LIMITED Company Number: 10359313 Name of Company: WYRE OILS LIMITED Company Type: Registered Company Company Number: 07469229 Nature of the business: Catering and retail food outlet Company Type: Registered Company Trading as: Hungry Buddha Nature of the business: Fuel supplier Type of Liquidation: Creditors' Voluntary Type of Liquidation: Creditors' Voluntary Registered office: 79a High Road, Willesden, London NW10 2SU Registered office: 17 Victoria Road East, Thornton-cleveleys FY5 5HT Principal trading address: 5b Manor Way, Old Woking, Surrey GU22 Principal trading address: 17 Victoria Road East, Thornton-cleveleys 9JX Previous Trading Address: 6 Friary Street, Guildford GU1 4EH FY5 5HT Office Holder/s: Kian Seng Tan, of 79a High Road, Willesden, London NW10 2SU, Telephone: 020 7586 1280, Email address: mail@kstan.co.uk (KST/IY/L1947)

Office Holder Number/s: 8032	Company Number: 09479429
Date of appointment: 29 August 2023	Company Type: Registered Company Nature of the business: TRANSPORT
By whom Appointed: Members and Creditors (4431642)	Type of Liquidation: Creditors' Voluntary
	Registered office: 37 Clorain Road, Liverpool L33 6XY
Name of Company: PDR ELECTRICS LTD	Principal trading address: 37 Clorain Road, Liverpool L33 6XY
Company Number: 08809960	Office Holder/s: Stuart Rathmell, of STUART RATHMELL
Company Type: Registered Company	INSOLVENCY and, Egyptian Mill, Egyptian Street, Bolton BL1 2HS,
Nature of the business: Electrical engineering Type of Liquidation: Creditors' Voluntary	Telephone: 01204867615, Email address: stuart.rathmell.insolvency@outlook.com
Registered office: 17 Victoria Road East, Thornton-cleveleys FY5 5HT	Office Holder Number/s: 10050
Principal trading address: 17 Victoria Road East, Thornton-cleveleys	Date of appointment: 31 August 2023
FY5 5HT	By whom Appointed: Members and Creditors
Office Holder/s: Stuart Rathmell, of STUART RATHMELL	Thursday 31 August 2023 (4432060)
INSOLVENCY and, Egyptian Mill, Egyptian Street, Bolton BL1 2HS,	
Telephone: 01204867615, Email address: stuart.rathmell.insolvency@outlook.com	Name of Company, BARTAK LIMITED
Office Holder Number/s: 10050	Name of Company: PARTAK LIMITED Company Number: 11880678
Date of appointment: 18 August 2023	Company Type: Registered Company
By whom Appointed: Members and Creditors	Nature of the business: Other transportation support activities
Friday 18 August 2023 (4431808)	Type of Liquidation: Creditors' Voluntary
	Registered office: 21 Harbord Close, London SE5 8AG
Name of Company: BOSIE LEE'S CAEE LIMITED	Principal trading address: 21 Harbord Close, London SE5 8AG
Name of Company: ROSIE LEE'S CAFE LIMITED Company Number: 12411794	Office Holder/s: Christopher David Horner, of Robson Scott Associates T/a Businessrescueexpert, 49 Duke Street, Darlington, Co.
Company Type: Registered Company	Durham DL3 7SD, Telephone: 01325365950, Email address:
Nature of the business: Unlicensed restaurants and cafes	admin@businessrescueexpert.co.uk (Partak Limited - Appointment)
Type of Liquidation: Creditors' Voluntary	Office Holder Number/s: 16150
Registered office: 21 Highfield Road, Dartford DA1 2JS	Date of appointment: 31 August 2023
Principal trading address: 3 Station Road, Strood, Rochester ME2 4AX	By whom Appointed: Members and Creditors Thursday 31 August 2023 (4427105)
Office Holder/s: Isobel Susan Brett, of Bretts Business Recovery	(4427103)
Limited and, 21 Highfield Road, Dartford, Kent DA1 2JS, Telephone:	
01474 532862, Email address: enquiries@brettsbr.co.uk (Z3190)	PURSUANT TO SECTION 109 OF THE INSOLVENCY ACT 1986
Office Holder Number/s: 9643	Name of Company: LINKS RESOURCING LIMITED
Date of appointment: 30 August 2023	Company Number: 07934804
By whom Appointed: Members and Creditors Thursday 31 August 2023 (4431355)	Registered office: Suite 3 Dukes House 4-6 High Street Windsor SL4 1LD
	Principal trading address: Riverside Centre 42 High Street Kingston
Name of Company, ELECTRA ELIDORE LTD	KT1 1HL
Name of Company: ELECTRA EUROPE LTD Company Number: 08733875	Nature of Business: Other Activities of Employment Placement Agencies
Company Type: Registered Company	Type of Liquidation: Creditors Voluntary
Nature of the business: Electrical Contracting	Liquidator's name and address: Lee James Cotton, Robert Day and
Type of Liquidation: Creditors' Voluntary	Company Limited, The Old Library The Walk Winslow Buckingham
Registered office: 14 Brockles Mead, Harlow, Essex CM19 4PS	MK18 3AJ, Tel: 01296 711644 E-mail: mail@robertday.biz and <i>Richard</i>
Principal trading address: 14 Brockles Mead, Harlow, Essex CM19 4PS	Behagg, Robert Day and Company Limited, The Old Library The Walk Winslow Buckingham MK18 3AJ 01296 711644 mail@robertday.biz
Office Holder/s: Leigh Holmes and Susan Purnell, of Purnells,	Office Holder Numbers: 9610 and 12790.
Goldfields House, 18a Gold Tops, Newport, South Wales NP20 4PH	Date of Appointment: 24 August 2023
UK, Telephone: 01633 214712, Email address: leigh@purnells.co.uk	By whom Appointed: Members and Creditors (4431881)
(E1106)	
Office Holder Number/s: 9390 and 9386	
Date of appointment: 30 August 2023 By whom Appointed: Members and Creditors	Name of Company: <b>3D CITY SIGNS LIMITED</b> Company Number: 07037635
Thursday 31 August 2023 (4432055)	Nature of Business: Shopfitters
(TIOE000)	Registered office: 142 Thornton Road, Bradford, BD1 2DX
	Type of Liquidation: Creditors
Name of Company: M.R.S. ENGINEERING SERVICES LIMITED	Date of Appointment: 25 August 2023
Company Number: 11596202	Liquidator's name and address: Nicola Baker (IP No. 15852) of
Company Type: Registered Company	Rushtons Insolvency Limited, 6 Festival Building, Ashley Lane,
Nature of the business: Engineering Type of Liquidation: Creditors' Voluntary	Saltaire, BD17 7DQ By whom Appointed: Members and Creditors
Registered office: 1 Stephenson Court, Middlesborough TS6 5UT	Ag FJ52940 (4430239)
Principal trading address: 1 Stephenson Court, Middlesborough TS6	(
5UT	
Office Holder/s: Stuart Rathmell, of STUART RATHMELL	Name of Company: A LAWRANCE BUILDING SERVICES LIMITED
INSOLVENCY and, Egyptian Mill, Egyptian Street, Bolton BL1 2HS,	Company Number: 11544564
Telephone: 01204867615, Email address:	Nature of Business: Development of building projects; Other building
stuart.rathmell.insolvency@outlook.com Office Holder Number/s: 10050	completion and finishing Registered office: 4 Fairacres, Ruislip, Middlesex, HA4 8AN
Date of appointment: 22 August 2023	Type of Liquidation: Creditors
By whom Appointed: Members and Creditors	Date of Appointment: 23 August 2023
Tuesday 22 August 2023 (4432058)	Liquidator's name and address: Farheen Qureshi (IP No. 11392) of
	Parker Getty Limited, Devonshire House, 582 Honeypot Lane,
Name of Company's ACK LOCIETICS LTD	Stanmore, Middx, HA7 1JS
Name of Company: ACK LOGISTICS LTD	By whom Appointed: Members and Creditors

#### Ag FJ52889

(4430268)

Ag FJ52889 (4430268)	
	Name of Company: AMALGAMATED EURO PRODUCTS UK LTD
	Trading Name: AEP UK Ltd
Name of Company: ABBEY QUILTING LIMITED	Company Number: 07977391
Company Number: 09462764	Registered office: 120 Bark Street, Bolton, BL1 2AX
Nature of Business: Other manufacturing not elsewhere classified	Principal trading address: Unit 3 Rufus Business Centre, Ravensbury
Registered office: 1066 London Road, Leigh-on-Sea, Essex, SS9 3NA	Terrace, London, SW18 4RL
Type of Liquidation: Creditors	Nature of Business: Other manufacturing not elsewhere classified
Date of Appointment: 24 August 2023	Type of Liquidation: Creditors Voluntary Liquidation
Liquidator's name and address: Dominik Thiel-Czerwinke (IP No.	Date of Appointment: 21 August 2023
009636) and Louise Donna Baxter (IP No. 009123) both of Begbies	By whom Appointed: Members and Creditors
Traynor (Central) LLP, 1066 London Road, Leigh-on-Sea, Essex, SS9	Liquidator's name and address: Rikki Burton (IP No. 14430) of
3NA	Anderson Brookes Insolvency Practitioners Limited, 6th Floor, 120
By whom Appointed: Members and Creditors	Bark Street, Bolton, BL1 2AX.
Ag FJ52918 (4430225)	For further information contact Steven Marshall at the offices of
	Anderson Brookes Insolvency Practitioners Limited on 0800 1804
O	933, or steven@andersonbrookes.co.uk.
Company Number: 09903105	30 August 2023 (4430992)
Name of Company: AI BUS ENGINEERING SOLUTIONS LTD	
Nature of Business: Repair and maintenance of other transport equipment not elsewhere classified	
Registered office: 45 Caravel Close, Grays, RM16 6QB	Name of Company: ARAMOKO BROADBAND LIMITED
Principal trading address: 45 Caravel Close, Grays, RM16 6QB	Company Number: 05137469 Nature of Business: Other information technology service activities
Type of Liquidation: Creditors Voluntary Liquidation	Registered office: 5 Barnfield Crescent, Exeter, EX1 1QT
Liquidator's name and address: Nimish Patel (IP number 8679) of	Type of Liquidation: Creditors
Hudson Weir Limited, 58 Leman Street, London, E1 8EU.	Date of Appointment: 23 August 2023
Liquidator's name and address: Joint Liquidator: Hasib Howlader (IP	Liquidator's name and address: David Gerard Kirk (IP No. 8830) and
number 20350) of Hudson Weir Limited, 58 Leman Street, London, E1	Daniel Robert Jeeves (IP No. 26032) both of Kirks, 5 Barnfield
8EU.	Crescent, Exeter, EX1 1QT
Date of Appointment: 23 August 2023	By whom Appointed: Members and Creditors
By whom Appointed: Creditors and Members	Ag FJ52883 (4430247)
For further details contact Tiffany Truter on 020 7099 6086 or at	<b>.</b>
tiffany@hudsonweir.co.uk (4431614)	
	Name of Company: AUTO REPAINT LTD
	Company Number: 08720034
Company Number: 12051373	Nature of Business: Car body repair
Name of Company: AIC LONDON LIMITED	Registered office: LevelQ, Sheraton House, Surtees Way, Surtees
Trading Name: Kebaby	Business Park, Stockton on Tees, TS18 3HR
Nature of Business: Unlicenced Restaurants and Cates	Type of Liquidation: Creditors
Registered office: 19 London Road, Morden, Surrey, SM4 5HT	Date of Appointment: 24 August 2023
Principal trading address: 19 London Road, Morden, Surrey, SM4 5HT	Liquidator's name and address: <i>Ian James Royle</i> (IP No. 18934) and
Type of Liquidation: Creditors Voluntary Liquidation	David Adam Broadbent (IP No. 9458) both of Begbies Traynor
Liquidator's name and address: <i>Robert Cundy</i> (IP number 9495) of	(Central) LLP, LevelQ, Sheraton House, Surtees Way, Surtees
Edge Recovery Limited, 5/7 Ravensbourne Road, Bromley, Kent, BR1 1HN.	Business Park, Stockton on Tees, TS18 3HR
Date of Appointment: 29 August 2023	By whom Appointed: Creditors Ag FJ52888 (4430256)
By whom Appointed: Creditors	Ag 1 032000 (4400200)
For further details contact Zoe Herrington on 020 8315 7430 or at	
zoe.herrington@edgerecovery.com (4431374)	Name of Company: AUTOFIX SERVICE AND REPAIRS LTD
	Company Number: 10310470
	Nature of Business: Maintenance and repair of motor vehicles
	Registered office: 31 Dashwood Avenue, High Wycombe, HP12 3DZ
Name of Company: ALCASO LOGISTIC LIMITED	and it is in the process of being changed to The Old Town Hall, 71
Company Number: 10653751	Christchurch Road, Ringwood, BH24 1DH
Registered office: 120 Bark Street, Bolton, Greater Manchester, BL1	Type of Liquidation: Creditors
2AX	Date of Appointment: 24 August 2023
Principal trading address: Schoolfield Corner Church Lane,	Liquidator's name and address: David Meany (IP No. 9453) and Frank
Dogmersfield, Hook, RG27 8SY	Wessely (IP No. 7788) both of Quantuma Advisory Limited, The Old
Nature of Business: Freight transport by road	Town Hall, 71 Christchurch Road, Ringwood, BH24 1DH
Type of Liquidation: Creditors Voluntary Liquidation	By whom Appointed: The Company
Date of Appointment: 25 August 2023	Ag FJ52948 (4430236)
By whom Appointed: Members and Creditors	
Liquidator's name and address: Rikki Burton (IP No. 14430) of	Nome of Company' BATTERSEA CRANICH LTD
Anderson Brookes Insolvency Practitioners Limited, 6th Floor, 120 Bark Street Bolton BL1 24X	Name of Company: BATTERSEA SPANISH LTD.
Bark Street, Bolton, BL1 2AX. For further information contact Laura Wright at the offices of	Company Number: 08846270 Nature of Business: Cultural Education
Anderson Brookes Insolvency Practitioners Limited on 0800 1804	Registered office: Fortis Insolvency Limited, 683-693 Wilmslow Road,
933, or laura@andersonbrookes.co.uk.	Manchester, M20 6RE (previously Battersea Spanish, 263-265
30 August 2023 (4430989)	Battersea Park Rd, London, SW11 4NE)
(1100000)	Type of Liquidation: Creditors
	Date of Appointment: 25 August 2023
	Liquidator's name and address: Daniel Taylor (IP No. 21050) of Fortis
	Insolvency Limited, 683-693 Wilmslow Road, Manchester, M20 6RE
	By whom Appointed: Members and Creditors
	Ag FJ52875 (4430201)

	COMPANIES
Name of Company: BIRTHDAY BLASTER LTD	Company Number: 05065103
Company Number: 07717741	Name of Company: BODEANS LIMITED
Nature of Business: Other amusement and recreation activities not	Trading Name: Fulham Bodeans
elsewhere classified.	Nature of Business: Restaurant
Registered office: Begbies Traynor 2 Harcourt Way, Meridian Business Park, Leicester, LE19 1WP	Registered office: Gable House, 239 Regents Park Road, London, N3 3LF
Type of Liquidation: Creditors	Principal trading address: 6 Broadway Chambers, Hilary Close,
Date of Appointment: 24 August 2023	FulhamRoad, SW6 1EA
Liquidator's name and address: Thomas Harris (IP No. 23370) and	Type of Liquidation: Creditors Voluntary Liquidation
Martin Richard Buttriss (IP No. 9291) both of Begbies Traynor (Central)	Liquidator's name and address: Bijal Shah (IP number 8717) of Edge
LLP, 2 Harcourt Way, Meridian Business Park, Leicester, LE19 1WP	Recovery Limited, 27 Church Street, Rickmansworth, Hertfordshire,
By whom Appointed: Members and Creditors	WD3 1DE.
Ag FJ52975 (4430237)	Date of Appointment: 29 August 2023 By whom Appointed: Creditors
	For further details contact Jack Webb on 020 8315 7430 or at
Name of Company: BJM FENCING LIMITED	Jack.webb@edgerecovery.com (4431341)
Company Number: 09136868	
Nature of Business: Other construction installation	
Registered office: 1066 London Road, Leigh-on-Sea, Essex, SS9 3NA	Company Number: 10963958
Type of Liquidation: Creditors Date of Appointment: 24 August 2023	Name of Company: <b>BROOKLANDS PRESTIGE CARS LIMITED</b> Nature of Business: Sale of used cars
Liquidator's name and address: <i>Wayne MacPherson</i> (IP No. 9445) and	Registered office: Arrans, Office 3, Swan Park Business Centre,
Jamie Taylor (IP No. 002748) both of Begbies Traynor (Central) LLP,	Kettlebrook Road, Tamworth B77 1AG
1066 London Road, Leigh-on-Sea, Essex, SS9 3NA	Principal trading address: 5 Cotton Drive, Dalehouse Lane,
By whom Appointed: Members and Creditors	Kenilworth, Warwickshire, CV8 2EB
Ag FJ52993 (4430232)	Type of Liquidation: Creditors Voluntary Liquidation
	Liquidator's name and address: Liquidator: <i>Robert Paul Gibbons</i> (IP number 9079) of Arrans Limited, Office 3, Swan Park Business Centre,
Name of Company: BL TRUCKING LTD	Kettlebrook Road, Tamworth, B77 1AG.
Company Number: 09861833	Date of Appointment: 22 August 2023
Nature of Business: Freight Transport	By whom Appointed: Members and Creditors
Registered office: LevelQ, Sheraton House, Surtees Way, Surtees	For further details contact: +44 (0)1827 60020 (4430906)
Business Park, Stockton on Tees, TS18 3HR	
Type of Liquidation: Creditors Date of Appointment: 24 August 2023	Name of Company: CHASE RENOVATIONS LIMITED
Liquidator's name and address: <i>Ian James Royle</i> (IP No. 18934) and	Company Number: 12442568
David Adam Broadbent (IP No. 9458) both of Begbies Traynor	Nature of Business: Drylining
(Central) LLP, LevelQ, Sheraton House, Surtees Way, Surtees	Registered office: LevelQ, Sheraton House, Surtees Way, Surtees
Business Park, Stockton on Tees, TS18 3HR	Business Park, Stockton on Tees, TS18 3HR
By whom Appointed: Members and Creditors	Type of Liquidation: Creditors
Ag FJ52881 (4430180)	Date of Appointment: 24 August 2023 Liquidator's name and address: <i>Ian James Royle</i> (IP No. 18934) and
	David Adam Broadbent (IP No. 9458) both of Begbies Traynor
Name of Company: BLYTHWOOD CARS LTD	(Central) LLP, LevelQ, Sheraton House, Surtees Way, Surtees
Company Number: 11195425	Business Park, Stockton on Tees, TS18 3HR
Nature of Business: Sale of used cars and light motor vehicles	By whom Appointed: Creditors and Members
Registered office: 1066 London Road, Leigh-on-Sea, Essex, SS9 3NA Type of Liquidation: Creditors	Ag FJ52906 (4430269)
Date of Appointment: 24 August 2023	
Liquidator's name and address: <i>Louise Donna Baxter</i> (IP No. 009123)	Name of Company: CHURCH LANE FISH BAR LIMITED
and Wayne MacPherson (IP No. 9445) both of Begbies Traynor	Company Number: 10286806
(Central) LLP, 1066 London Road, Leigh-on-Sea, Essex, SS9 3NA	Registered office: 23 Church Lane, London, E11 1HG
By whom Appointed: Members and Creditors	Principal trading address: 23 Church Lane, London, E11 1HG
Ag FJ52924 (4430231)	Nature of Business: Take-away food shops and mobile food stands Type of Liquidation: Creditors' Voluntary
	Liquidator's name and address: <i>Mansoor Mubarik</i> , of Capital Books
	(UK) Limited, 66 Earl Street, Maidstone ME14 1PS, Telephone: 01622
Name of Company: BOARDWALK (RENDLESHAM) LTD	754 927, Email address: mmubarik@capital-books.co.uk, Fax: 01622
Trading Name: The Board Walk Cafe	321 000
Company Number: 11033940 Registered office: 120 Bark Street, Bolton, BL1 2AX	Office Holder Number: 009667. Date of Appointment: 18 August 2023
Principal trading address: Walnut Tree, Rendlesham, Suffolk, IP12	By whom Appointed: Members and Creditors (4431883)
2GG	, , , , , , , , , , , , , , , , , , , ,
Nature of Business: Licensed clubs	
Type of Liquidation: Creditors Voluntary Liquidation	Name of Company: CUSTOM JOINERY & BUILDING LTD
Date of Appointment: 25 August 2023	Company Number: 09626093
By whom Appointed: Members and Creditors	Nature of Business: Other building completion and finishing Previous Name of Company: IBE Joinery and Building Limited
Liquidator's name and address: <i>Rikki Burton</i> (IP No. 14430) of Anderson Brookes Insolvency Practitioners Limited, 6th Floor, 120	Registered office: 30 Birkenhead Road, Hoylake, Wirral, CH47 3BW
Bark Street, Bolton, BL1 2AX.	Type of Liquidation: Creditors
For further information contact Laura Wright at the offices of	Date of Appointment: 24 August 2023
Anderson Brookes Insolvency Practitioners Limited on 0800 1804	Liquidator's name and address: Jason Dean Greenhalgh (IP No.
933, or laura@andersonbrookes.co.uk. 30 August 2023 (4431002)	009271) and <i>Paul Stanley</i> (IP No. 008123) both of Begbies Traynor
30 August 2023 (4431002)	(Central) LLP, No 1 Old Hall Street, Liverpool, L3 9HF

By whom Appointed: Members and Creditors Ag FJ52916

(4430263)

Name of Company: EUROPOST EXPRESS (UK) LTD. Name of Company: DEALS UNITED WORLDWIDE LTD Company Number: 09694048 Company Number: 12866177 Nature of Business: International Cargo and Courier Services Registered office: C12 Marquis Court, Marquisway, Team Valley, Registered office: 6th Floor, 120 Bark St, Bolton, Greater Manchester, Gateshead, NE11 0RU BL1 2AX Principal trading address: Suite 165, Unit 3a, 34-35 Hatton Garden, Type of Liquidation: Creditors EC1N 8DX Date of Appointment: 25 August 2023 Liquidator's name and address: Eric Walls (IP No. 9113) and Wayne Nature of Business: Wholesale of computers, computer peripheral Harrison (IP No. 9703) both of KSA Group Ltd, C12 Marquis Court, equipment and software Marguisway, Team Valley, Gateshead, NE11 0RU Type of Liquidation: Creditors Voluntary Liquidation Date of Appointment: 24 August 2023 By whom Appointed: Members and Creditors (4430183) By whom Appointed: Members and Creditors Ag FJ52876 Liquidator's name and address: Rikki Burton (IP No. 14430) of Anderson Brookes Insolvency Practitioners Limited, 6th Floor, 120 Company Number: 06414686 Bark Street, Bolton, BL1 2AX. Name of Company: EVINCE TECHNOLOGY LIMITED For further information contact Laura Wright at the offices of Nature of Business: Other research and experimental development on Anderson Brookes Insolvency Practitioners Limited on 0800 1804 natural sciences and engineering 933, or laura@andersonbrookes.co.uk. 30 August 2023 (4430998) Registered office: Netpark Plexus Thomas Wright Way, Sedgefield, Stockton-On-Tees, England, TS21 3FD Principal trading address: Netpark Plexus Thomas Wright Way, Name of Company: DK1 LTD Sedgefield, Stockton-On-Tees, England, TS21 3FD Type of Liquidation: Creditors Voluntary Liquidation Company Number: 12235595 Nature of Business: Other business support service activities not Liquidator's name and address: Joint Liquidator: Michael Chamberlain (IP number 8735) of Quantuma Advisory (Yorkshire) Limited, elsewhere classified Resolution House, 12 Mill Hill, Leeds, LS1 5DQ. Registered office: Nickron House, Bury Lane, Rickmansworth, Herts, Liquidator's name and address: Joint Liquidator: Gareth Peckett (IP **WD3 1DN** number 9647) of Quantuma Advisory (Yorkshire) Limited, Resolution Type of Liquidation: Creditors Date of Appointment: 24 August 2023 House, 12 Mill Hill, Leeds, LS1 5DQ. Date of Appointment: 11 August 2023 Liquidator's name and address: Farheen Qureshi (IP No. 11392) of Parker Getty Limited, Devonshire House, 582 Honeypot Lane, By whom Appointed: Members and Creditors Stanmore, Middx, HA7 1JS For further details contact Caitlin Auplish on 0113 242 0808 or at By whom Appointed: Members and Creditors Caitlin.Auplish@quantuma.com (4431682) (4430261) Ag FJ52892 Name of Company: EVOLUTIV CONSTRUCTION LIMITED Name of Company: DOUBLE ANCHOR LIMITED Company Number: 10816943 Company Number: 11835382 Nature of Business: Construction of domestic buildings Registered office: 154 Marsh Lane, Stanmore, HA7 4HT Nature of Business: Information technology consultancy and Type of Liquidation: Creditors information technology service activities Registered office: PO Box 4385 - 11835382, Cardiff, CF14 8LH Date of Appointment: 24 August 2023 Liquidator's name and address: Farheen Qureshi (IP No. 11392) of Type of Liquidation: Creditors Parker Getty Limited, Devonshire House, 582 Honeypot Lane, Date of Appointment: 16 August 2023 Liquidator's name and address: Emma Mifsud (IP No. 21070) of Opus Stanmore, Middx, HA7 1JS Restructuring LLP, Third Floor, One Park Row, Leeds, LS1 5HN and By whom Appointed: Members and Creditors Colin David Wilson (IP No. 9478) of Opus Restructuring LLP, 1 Radian Ag FJ52893 (4430271) Court, Knowlhill, Milton Keynes, MK5 8PJ By whom Appointed: Members and Creditors Name of Company: FARRER'S CONSTRUCTION LTD Ag FJ52877 (4430192) Company Number: 10929566 Nature of Business: Construction of domestic buildings Registered office: (to be changed to) Sussex Innovation, One Croydon, 12-16 Addiscombe Road, Croydon, CR0 0XT Name of Company: EASICUT PRECISION GRINDING COMPANY Type of Liquidation: Creditors LTD Company Number: 07536702 Date of Appointment: 22 August 2023 Liquidator's name and address: Matthew Reeds (IP No. 28040) and Registered office: 120 Bark Street, Bolton, BL1 2AX Paul Bailey (IP No. 9428) both of Bailey Ahmad Limited, Sussex Principal trading address: 22 Leestone Road, Wythenshawe, Innovation, One Croydon, 12-16 Addiscombe Road, Croydon, CRO Manchester, M22 4RN 0XT Nature of Business: Other engineering activities Type of Liquidation: Creditors Voluntary Liquidation By whom Appointed: Members and Creditors Ag FJ52998 (4430251) Date of Appointment: 25 August 2023 By whom Appointed: Members and Creditors Liquidator's name and address: Rikki Burton (IP No. 14430) of Anderson Brookes Insolvency Practitioners Limited, 6th Floor, 120 Company Number: 09544132 Name of Company: FASHION COUTURE LIMITED Bark Street, Bolton, BL1 2AX. Nature of Business: Retail sale via stalls and markets of textiles, For further information contact Steven Marshall at the offices of clothing and footwear Anderson Brookes Insolvency Practitioners Limited on 0800 1804 Registered office: 400 Victoria Road East, Leicester, LE5 0LG 933, or steven@andersonbrookes.co.uk. (4430993) Principal trading address: 400 Victoria Road East, Leicester, LE5 0LG 30 August 2023 Type of Liquidation: Creditors Voluntary Liquidation Liquidator's name and address: Neil Charles Money (IP number 8900) of CBA Business Solutions, 126 New Walk, Leicester, LE1 7JA. Date of Appointment: 29 August 2023 By whom Appointed: The Members and Creditors For further details contact Steven Glanvill on 0116 262 6804 (4431383)

(4430986)

Company Number: 10609051 Name of Company: FAT LEMON PRODUCTIONS LIMITED Nature of Business: Vide Oraclations Activities Registered office: 6th Floor Charlotte Building, 17 Gresse Street, London, W1T 1QL Principal trading address: Cobbetts, Easton Royal, Pewsey, SN9 5LS Type of Liquidation: Creditors Voluntary Liquidation Date of Appointment: 25 August 2023 By whom Appointed: The Members and Creditors Por further details contact Victoria Ely on 0116 262 6804 (443102) Nature of Business: Floor And wall covering Liquidator's name and address: Alexander Kinninmonth (IP No. 9119) Type of Liquidation: Creditors Registered office: 31 Evans Avenue, Watford, W025 0EX in the Process of being changed to FRP Advisory Trading limited, Mountbatten House, Grosvenor Square, Southampton, SO15 2RP Type of Liquidation: Creditors Date of Appointemt: 25 August 2023 Liquidator's name and address: Alexander Kinninmonth (IP No. 9119) Type of Liquidation: Creditors Date of Appointemt: 27 August 2023 Liquidator's name and address: Alexander Kinninmonth (IP No. 9119) Type of Liquidation: Creditors Add <i>Amse Prio</i> (PN No. 22850) both of FRP Advisory Trading limited, Mountbatten House, Grosvenor Square, Southampton, SO15 2RP Type of Liquidation: Creditors Add <i>Amse Prio</i> (PN No. 22850) both of FRP Advisory Trading limited, Mountbatten House, Grosvenor Square, Southampton, SO15 2RP Type of Liquidation: Creditors Ag FJS2989 (443024) Mountbatten House, Grosvenor Square, Southampton, SO15 2RP Type of Liquidation: Creditors Ag FJS2989 (443024) Name of Company: <b>EVORD MANAGEMENT LTD</b> Company Number: 1051806 Nature of Business: Activities of head offices; Management; Other Bate of Appointment: 77 August 2023 Liquidator's name and address: Activities of head offices; Management; Other Bate of Appointment: 72 August 2023 Liquidator's name and address: Steven John Parker (IP No. 5989) and Registered office: Chancery House, 30 St Johns Road, Woking Registered office: Chancery House, 30 St Johns Road, Woking R
Registered office: 6th Floor Charlotte Building, 17 Gresse Street, London, W1T 1QL Principal trading address: Cobbetts, Easton Royal, Pewsey, SN9 5LS Type of Liquidation: Creditors Voluntary Liquidation of CBA Business: Neil Charlest Money (P number 8900) of CBA Business: Neiholas Cusack (IP No. 17490) of Parker Andrews Limited, 5th Floor, The Union Building, 51-59 Res Nature of Business: Freight transport/ IT Services Nature of Business: Neight anapport, TI Services Nature of Business: Neight anapport, TI Services Nature of Business: Advance (V MD25 0EX in the process of being changed to FFP Advisory Trading Limited, Mountbatten House, Grosvenor Square, Southampton, S015 2RP By whom Appointed: Creditors and Members Ag FJS2969 (443025) Nature of Company: FJORD MANAGEMENT LTD Company Number: 1051800 Nature of Business: Activities of head offices; Management Ag FJS2969 (443024) Nature of Business: Activities of head offices; Management Ag FJS2969 (243024) Nature of Business: Activities of head offices; Management Name of Company: <b>GSI ENERGY LIMITED</b> Company Number: 1051806 Nature of Business: Activities of head offices; Management Name of Company: <b>GSI ENERGY LIMITED</b> Company Number: 0517AG Type of Liquidation: Creditors Date of Appointed: Members and Creditors Mature of Business: Activities of head offices; Management Name of Company: <b>GSI ENERGY LIMITED</b> Company Number: 0516416 Nature of Business: Activities of head offices; Management Su
Principal trading address: Cobbetts, Easton Royal, Pewsey, SN9 5LS Type of Liquidation: Creditors Voluntary Liquidation Liquidator's name and address: <i>Neil Charles Money</i> (IP number 8900) Tate of Appointment: 25 August 2023 By whom Appointed: The Members and Creditors For further details contact Victoria Ely on 0116 262 6804 (443102) Name of Company, <b>FENDA CONSULTING LTD</b> Company Number: 09667039 Nature of Business: Freight transport/ IT Services Registered office: 33 Evans Avenue, Watford, WD25 0EX in the process of being changed to FRP Advisory Trading Limited, Mountbatten House, Grosvenor Square, Southampton, S015 2RP Type of Liquidation: Creditors Date of Appointment: 25 August 2023 (443025) Date of Appointment: 25 August 2023 (443025) And <i>Lanse Prior</i> (IP No. 29250) both of FRP Advisory Trading Limited, Mountbatten House, Grosvenor Square, Southampton, S015 2RP By whom Appointed: Creditors and Members Ag FJS2969 (443025) Nature of Business: Advities of head offices; Management; Other Sy whom Appointed: Creditors and Members Ag FJS2969 (443025) Nature of Business: Advities of head offices; Management; Other Sy whom Appointed: Creditors and Members Ag FJS2969 Nature of Business: Activities of head offices; Management; Other Sy whom Appointed: Creditors and Members Ag FJS2969 Nature of Business: Activities of head offices; Management; Other Sy whom Appointed: Creditors and Members Ag FJS2969 Nature of Business: Activities of head offices; Management; Other Surrey, GU21 75A Type of Liquidation: Creditors Date of Appointment: 24 August 2023 Liquidator's name and address: Steven John Parker (IP No. 8989) and Nature of Business: Activities of head offices; Management; Other Surrey, GU21 75A Type of Liquidation: Creditors Date of Appointment: 24 August 2023 Liquidator's name and address: Nicola Elaine Layland (IP No. 01765) Date of Appointment: 24 August 2023 Liquidator's name and address: Nicola Elaine Layland (IP No. 01765) Date of Appointment: 24 August 2023 Liquidator's name and address: Nicola Elaine Layla
Liquidator's name and address: Neil Charles Money (IP number 8900) of CBA Business Solutions, 126 New Walk, Leicester, LE1 7JA. Date of Appointment: 25 August 2023 Type of Liquidation: Creditors Date of Appointment: 24 August 2023 Date of Appointment: 25 August 2023 Date of Appointment: 26 August 2023 Date of Appoint
By whom Appointed: The Members and Creditors For further details contact Victoria Ely on 0116 262 6804 (4431023) Liquidator's name and address: <i>Nicholas Cusack</i> (IP No. 1749) of Parker Andrews Limited, 5th Floor, The Union Building, 51-59 Res Lane, Norwich, NR1 1BY By whom Appointed: Members and Creditors Ag FJ52959 (443025) Nature of Business: Freight transport/ IT Services Registered office: 33 Evans Avenue, Watford, WD25 0EX in the process of being changed to FRP Advisory Trading Limited, Mountbatten House, Grosvenor Square, Southampton, S015 2RP Type of Liquidation: Creditors Date of Appointment: 25 August 2023 Liquidator's name and address: <i>Alexander Kinninmonth</i> (IP No. 9019) By whom Appointed: Creditors and Members Ag FJ52969 (443025) Nature of Company: <b>FJORD MANAGEMENT LTD</b> Company Number: 0518080 Nature of Business: Activities of head offices; Management, onsultancy activities other than financial management; Company Number: 0518080 Nature of Business: Activities of head offices; Management, or Business: Activities of head offices; Management, or Business: and creditors Date of Appointment: 27 July 2023 Liquidator's name and address: <i>Steven John Parker</i> (IP No. 8989) and Liquidator's name and address: <i>Steven John Parker</i> (IP No. 8989) and Date of Appointment: 27 July 2023 Liquidator's name and address: <i>Nicola</i> Elainet Alyaines Park, Whiteley, Fareham, Hampshire, PO15 7AG Type of Liquidation: Creditors Date of Appointment: 27 July 2023 Liquidator's name and address: <i>Nicola</i> Elainet Alyainet (IP No. 01765): and <i>Can Derek Faulds</i> (IP No. 008767) both of Leonard Curits, 158 Parkway, Solent Business Park, Whiteley, Fareham, Hampshire, PO1 7AG Ag FJS3018 (443024) By whom Appointed: Members and Creditors
Parker Andrews Limited, 5th Floor, The Union Building, 51-59 Ros Lane, Norwich, NR1 1BY By whom Appointed: Members and Creditors Ag FJ52959 (443025) Nature of Business: Freight transport/ IT Services Registered office: 32 Vans Avenue, Watford, WD25 0EX in the process of being changed to FRP Advisory Trading Limited, Mountbatten House, Grosvenor Square, Southampton, SO15 2RP Date of Appointment: 25 August 2023 Liquidator's name and address: <i>Alexander Kinninmonth</i> (IP No. 9019) By whom Appointed: Creditors and Members Ag FJ52969 (443025) Nature of Business: Activities of head offices; Management; Other Dates of Appointment: 20518806 Nature of Business: Activities of head offices; Management; Other Surrey, GU21 7SA Type of Liquidation: Creditors Date of Appointment: 27 July 2023 Liquidator's name and address: Satewand ber Kines, Advising Registered office: Chancery House, 30 St Johns Road, Woking, Type of Liquidation: Creditors Date of Appointment: 27 July 2023 Liquidator's name and address: Nature of Business: Rotivities of head offices; Management; Other Dusiness support service activities of head offices; Management; Other Dusiness support service activities of head offices; Management; Other Dusiness cativities of head offices; Management; Other Date of Appointment; 27 July 2023 Liquidator's name and address: <i>Nicola Elaine Layland</i> (IP No. 01765) Date of Appointment; 27 July 2023 Liquidator's name and address: <i>Nicola Elaine Layland</i> (IP No. 01765) Date of Appointment; 27 July 2023 Liquidator's name and address: <i>Nicola Elaine Layland</i> (IP No. 01765) Date of Appointment; 24 August 2023 Liquidator's name and address: <i>Nicola Elaine Layland</i> (IP No. 01765) Date of Appointment; 24 August 2023 Liquidator's name and address: <i>Nicola Elaine Layland</i> (IP No. 01765) Date of Appointment; 24 August 2023 Liquidator's name and address: <i>Nicola Elaine Layland</i>
Company Number: 09667039Ag FJ52959(443025)Nature of Business: Freight transport/ IT ServicesRegistered office: 33 Evans Avenue, Watford, WD25 0EX in the process of being changed to FRP Advisory Trading Limited, Mountbatten House, Grosvenor Square, Southampton, SO15 2RP Type of Liquidation: CreditorsName of Company: <b>GREEN DOOR CLINIC LIMITED</b> Company Number: 11051601Date of Appointment: 25 August 2023Registered office: 30 Weymouth Street, London W1G 7BS Type of Liquidator's name and address: Alexander Kinninmonth (IP No. 9019) and James Prior (IP No. 29250) both of FRP Advisory Trading Limited, Mountbatten House, Grosvenor Square, Southampton, SO15 2RP By whom Appointed: Creditors and Members Ag FJ52969Name of Company: Creditors Date of Appointment: 17 August 2023 Liquidator's name and address: Anlexas Arakapiotis (IP No. 20910) of Kallis & Company, Mountview Court, 1148 High Road, Whetstorn London, N20 0FA By whom Appointed: Creditors and Members Ag FJ52969Name of Company: <b>GSI ENERGY LIMITED</b> Company Number: 05414416 Nature of Business: Activities of head offices; Management consultancy activities other than financial management; Other Business support service activities not elsewhere classified Registered office: Chancery House, 30 St Johns Road, Woking, Type of Liquidation: Creditors Date of Appointment: 27 July 2023 Liquidator's name and address: Silveen John Parker (IP No. 9008) both of Opus Restructuring LLP, 322 High Holborn, London, WC1V 7PB By whom Appointed: Members and Creditors Ag FJ33018Name of Company: Cerditors Type of Liquidation: Creditors Date of Appointment: 27 August 2023 Liquidator's name and address: Nicola Elaine Layland (IP No. 01765) and Carl Derek Faulds (IP No. 008767) both of Leonard Curtis, 158 Parkway, Solent Business Park, Whiteley, Far
Nature of Business: Freight transport/ IT ServicesRegistered office: 33 Evans Avenue, Watford, WD25 0EX in the process of being changed to FRP Advisory Trading Limited, Mountbatten House, Grosvenor Square, Southampton, S015 2RPName of Company: GREEN DOOR CLINIC LIMITED Company Number: 11051601 Nature of Business: Psychiatric Therapy Clinic Registered office: 30 Weymouth Street, London W1G 7BS Type of Liquidation: Creditors Date of Appointment: 27 July 2023 Liquidator's name and address: Arakapiotis (IP No. 20910) of Kallis & Company, Mountview Court, 1148 High Road, Whetstom London, N20 0RA By whom Appointed: Creditors and Members Ag FJ52969Name of Company: GSI ENERGY LIMITED Company Number: 05414416 Nature of Business: Activities of head offices; Management, Company Number: 10518806Name of Company: GSI ENERGY LIMITED Company Number: 05414416 Nature of Business: Activities on telsewhere classified Registered office: 1580 Parkway, Solent Business Park, Whitele Fareham, Hampshire P015 7AG Type of Liquidatior's name and address: <i>Nicola Elaine Layland</i> (IP No. 01765) and <i>Carl Derek Faulds</i> (IP No. 008767) both of Leonard Curtis, 158 Parkway, Solent Business Park, Whiteley, Fareham, Hampshire, P01 FAG By whom Appointed: Members and Creditors Ag FJ53018Name of Company: GSI Energy LIMITED Company Number: 05414416 Nature of Business: Activities not elsewhere classified Registered office: 1580 Parkway, Solent Business Park, Whiteley Fareham, Hampshire, P015 7AG Type of Liquidatior's name and address: <i>Nicola Elaine Layland</i> (IP No. 01765) and <i>Carl Derek Faulds</i> (IP No. 008767) both of Leonard Curtis, 158 Parkway, Solent Business Park, Whiteley, Fareham, Hampshire, P01 TAG
process of being changed to FRP Advisory Trading Limited, Mountbatten House, Grosvenor Square, Southampton, SO15 2RP Type of Liquidation: Creditors Date of Appointment: 25 August 2023 Liquidator's name and address: Alexander Kinnimonth (IP No. 9019) and James Prior (IP No. 29250) both of FRP Advisory Trading Limited, Mountbatten House, Grosvenor Square, Southampton, SO15 2RP By whom Appointed: Creditors and Members Ag FJ52969 Nature of Business: Activities of head offices; Management; Company Number: 10518806 Nature of Business: Activities of head offices; Management; Company Number: 10518806 Nature of Business: Activities of head offices; Management; Company Number: 10518806 Nature of Business: Activities of head offices; Management; Consultancy activities of head offices; Management; Consultancy activities of head offices; Management; Consultancy activities of alexander Classified Registered office: Chancery House, 30 St Johns Road, Woking Type of Liquidation: Creditors Date of Appointment: 27 July 2023 Liquidator's name and address: Steven John Parker (IP No. 9008) both of Opus Restructring LLP, 322 High Holborn, London, WCIV 7PB By whom Appointed: Members and Creditors Ag FJ53018 (4430248)
Mountbatten House, Grosvenor Square, Southampton, SO15 2RPCompany Number: 11051601Type of Liquidation: CreditorsNature of Business: Psychiatric Therapy ClinicDate of Appointment: 25 August 2023Type of Liquidation: Creditorsand James Prior (IP No. 29250) both of FRP Advisory Trading Limited,Type of Liquidator's name and address: Alexander Kinninmonth (IP No. 9019)and James Prior (IP No. 29250) both of FRP Advisory Trading Limited,Type of Liquidator's name and address: Andreas Arakapiotis (IP No. 20910) ofBy whom Appointed: Creditors and Members(4430255)Ag FJ52969(4430255)Name of Company: FJORD MANAGEMENT LTDCompany Number: 10518806Nature of Business: Activities of head offices; Management consultancy activities onte rebancies and elevence classifiedName of Company: GSI ENERGY LIMITED Company Number: 05414416Registered office: Chancery House, 30 St Johns Road, Worrey, GU21 7SANown Spointment: 27 July 2023Liquidator's name and address: Steven John Parker (IP No. 9008) both of Opus Restructuring LLP, 322 High Holborn, London, WC1V 7PB By whom Appointed: Members and Creditors Ag FJ53018Nature of Business Park, Whiteley, Fareham, Hampshire, PO1 7AGBy whom Appointed: Members and Creditors Ag FJ53018(4430248)
Type of Liquidation: CreditorsNature of Business: Psychiatric Therapy ClinicDate of Appointment: 25 August 2023Eigistered office: 30 Weymouth Street, London W1G 7BSLiquidator's name and address: Alexander Kinninmonth (IP No. 9019)Date of Appointment: 17 August 2023Mountbatten House, Grosvenor Square, Southampton, SO15 2RPDate of Appointment: 17 August 2023By whom Appointed: Creditors and Members(4430255)Ag FJ52969(4430255)Name of Company: FJORD MANAGEMENT LTDCompany Number: 10518806Company Number: 10518806Nature of Business: Activities of head offices; Management; OtherNature of Business: Activities on telsewhere classifiedNature of Business: Gas EngineeringRegistered office: Chancery House, 30 St Johns Road, Woking,Nature of Business: Gas EngineeringSurrey, GU21 7SAType of Liquidation: CreditorsType of Liquidation: CreditorsDate of Appointment: 27 July 2023Liquidator's name and address: Steven John Parker (IP No. 8989) andTimothy John Edward Dolder (IP No. 9008) both of OpusRestructuring LLP, 322 High Holborn, London, WC1V 7PBParkway, Solent Business Park, Whiteley, Fareham, Hampshire, PO1Parkway, Solent Business Park, Whiteley, Fareham, Ham
Liquidator's name and address: Alexander Kinnimonth (IP No. 9019) and James Prior (IP No. 29250) both of FRP Advisory Trading Limited, Mountbatten House, Grosvenor Square, Southampton, SO15 2RP By whom Appointed: Creditors and Members Ag FJ52969 (4430255) Name of Company: <b>FJORD MANAGEMENT LTD</b> Company Number: 10518806 Nature of Business: Activities of head offices; Management consultancy activities other than financial management; Other business support service activities not elsewhere classified Registered office: Chancery House, 30 St Johns Road, Woking, Surrey, GU21 7SA Surrey, GU21 7SA Date of Appointment: 27 July 2023 Liquidator's name and address: Steven John Parker (IP No. 8989) and Timothy John Edward Dolder (IP No. 9008) both of Oups Restructuring LLP, 322 High Holborn, London, WC1V 7PB By whom Appointed: Members and Creditors Ag FJ53018 (4430248)
Mountbatten House, Grosvenor Square, Southampton, SO15 2RPLiquidator's name and address: Andreas Arakapiotis (IP No. 20910) of Kallis & Company, Mountview Court, 1148 High Road, Whetston London, N20 0RA By whom Appointed: Members and Creditors Ag FJ52942Name of Company: FJORD MANAGEMENT LTD Company Number: 10518806 Nature of Business: Activities of head offices; Management consultancy activities other than financial management; Other business support service activities not elsewhere classified Registered office: Chancery House, 30 St Johns Road, Woking, Surrey, GU21 7SAName of Company: GSI ENERGY LIMITED Company Number: 0541416 Nature of Business: Gas Engineering Registered office: 1580 Parkway, Solent Business Park, Whitele Fareham, Hampshire P015 7AG Type of Liquidation: Creditors Date of Appointment: 27 July 2023 Liquidator's name and address: <i>Nicola Elaine Layland</i> (IP No. 017652 and <i>Carl Derek Faulds</i> (IP No. 008767) both of Leonard Curtis, 158 Parkway, Solent Business Park, Whiteley, Fareham, Hampshire, P01 Spato Appointed: Members and Creditors Parkway, Solent Business Park, Whiteley, Fareham, Hampshire, P01 Parkway, Sol
By whom Appointed: Creditors and Members Ag FJ52969Kallis & Company, Mountview Court, 1148 High Road, Whetston London, N20 0RA By whom Appointed: Members and Creditors Ag FJ52942Name of Company: FJORD MANAGEMENT LTD Company Number: 10518806 Nature of Business: Activities of head offices; Management consultancy activities other than financial management; Other business support service activities not elsewhere classified Registered office: Chancery House, 30 St Johns Road, Woking, Surrey, GU21 7SA Type of Liquidation: Creditors Date of Appointment: 27 July 2023 Liquidator's name and address: Steven John Parker (IP No. 9008) both of Opus Restructuring LLP, 322 High Holborn, London, WC1V 7PB By whom Appointed: Members and Creditors Ag FJ53018Kallis & Company, Mountview Court, 1148 High Road, Whetston London, N20 0RA By whom Appointed: Members and CreditorsBy whom Appointed: Members and Creditors Ag FJ53018Kallis & Company, Mountview Court, 1148 High Road, Whetston London, N20 0RA By whom Appointed: Members and Creditors
Ag FJ52942(4430243)Name of Company: FJORD MANAGEMENT LTD(4430243)Company Number: 10518806Nature of Business: Activities of head offices; ManagementName of Company: GSI ENERGY LIMITEDConsultancy activities other than financial management; OtherName of Company: Umber: 05414416business support service activities not elsewhere classifiedNature of Business: Gas EngineeringRegistered office: Chancery House, 30 St Johns Road, Woking,Registered office: 1580 Parkway, Solent Business Park, WhiteleSurrey, GU21 7SAType of Liquidation: CreditorsDate of Appointment: 27 July 2023Date of Appointment: 24 August 2023Liquidator's name and address: Steven John Parker (IP No. 8989) andLiquidator's name and address: Nicola Elaine Layland (IP No. 017652)Timothy John Edward Dolder (IP No. 9008) both of OpusParkway, Solent Business Park, Whiteley, Fareham, Hampshire, PO1By whom Appointed: Members and CreditorsTAGAg FJ53018(4430248)
Name of Company: FJORD MANAGEMENT LTDCompany Number: 10518806Nature of Business: Activities of head offices; Management consultancy activities other than financial management; Other business support service activities not elsewhere classifiedName of Company: GSI ENERGY LIMITED Company Number: 05414416Registered office: Chancery House, 30 St Johns Road, Woking, Surrey, GU21 7SANature of Business: Gas Engineering Registered office: 1580 Parkway, Solent Business Park, Whitele Fareham, Hampshire PO15 7AGType of Liquidation: Creditors Date of Appointment: 27 July 2023Type of Liquidation: Creditors Date of Appointment: 27 July 2023Liquidator's name and address: Steven John Parker (IP No. 8989) and Timothy John Edward Dolder (IP No. 9008) both of Opus By whom Appointed: Members and CreditorsType of Liquidator's name and address: Nicola Elaine Layland (IP No. 017652) and Carl Derek Faulds (IP No. 008767) both of Leonard Curtis, 158 Parkway, Solent Business Park, Whiteley, Fareham, Hampshire, PO1 7AGAg FJ53018(4430248)
Nature of Business: Activities of head offices; Management consultancy activities other than financial management; Other business support service activities not elsewhere classified Registered office: Chancery House, 30 St Johns Road, Woking, Surrey, GU21 7SA Type of Liquidation: Creditors Date of Appointment: 27 July 2023 Liquidator's name and address: Steven John Parker (IP No. 8989) and Timothy John Edward Dolder (IP No. 9008) both of Opus Restructuring LLP, 322 High Holborn, London, WC1V 7PB By whom Appointed: Members and CreditorsName of Company: GSI ENERGY LIMITED Company Number: 05414416 Nature of Business: Gas Engineering Registered office: 1580 Parkway, Solent Business Park, Whitele Fareham, Hampshire PO15 7AG Type of Liquidation: Creditors Date of Appointment: 24 August 2023 Liquidator's name and address: Nicola Elaine Layland (IP No. 01765) and Carl Derek Faulds (IP No. 008767) both of Leonard Curtis, 158 Parkway, Solent Business Park, Whiteley, Fareham, Hampshire, PO1 7AG By whom Appointed: Members and CreditorsName of Company: GSI ENERGY LIMITED Company Number: 05414416 Nature of Business: Gas Engineering Registered office: 1580 Parkway, Solent Business Park, Whitele Fareham, Hampshire PO15 7AG Type of Liquidation: Creditors Date of Appointment: 24 August 2023 Liquidator's name and address: Nicola Elaine Layland (IP No. 01765) and Carl Derek Faulds (IP No. 008767) both of Leonard Curtis, 158 Parkway, Solent Business Park, Whiteley, Fareham, Hampshire, PO1 7AG By whom Appointed: Members and Creditors
consultancy activities other than financial management; Other business support service activities not elsewhere classified Registered office: Chancery House, 30 St Johns Road, Woking, Surrey, GU21 7SA Type of Liquidation: Creditors Date of Appointment: 27 July 2023 Liquidator's name and address: Steven John Parker (IP No. 8989) and Timothy John Edward Dolder (IP No. 9008) both of Opus Restructuring LLP, 322 High Holborn, London, WC1V 7PB By whom Appointed: Members and CreditorsCompany Number: 05414416 Nature of Business: Gas Engineering Registered office: 1580 Parkway, Solent Business Park, Whitele Fareham, Hampshire PO15 7AG Type of Liquidation: Creditors Date of Appointment: 24 August 2023 Liquidator's name and address: Steven John Parker (IP No. 8989) and Timothy John Edward Dolder (IP No. 9008) both of Opus Restructuring LLP, 322 High Holborn, London, WC1V 7PB By whom Appointed: Members and CreditorsParkway, Solent Business Park, Whiteley, Fareham, Hampshire, PO1 7AG By whom Appointed: Members and Creditors
Registered office: Chancery House, 30 St Johns Road, Woking, Surrey, GU21 7SARegistered office: 1580 Parkway, Solent Business Park, Whitele Fareham, Hampshire PO15 7AGType of Liquidation: CreditorsType of Liquidation: CreditorsDate of Appointment: 27 July 2023Date of Appointment: 24 August 2023Liquidator's name and address: Steven John Parker (IP No. 8989) and Timothy John Edward Dolder (IP No. 9008) both of Opus Restructuring LLP, 322 High Holborn, London, WC1V 7PBLiquidator's name and address: Nicola Elaine Layland (IP No. 01765) and Carl Derek Faulds (IP No. 008767) both of Leonard Curtis, 1580 Parkway, Solent Business Park, Whiteley, Fareham, Hampshire, PO1 7AGAg FJ53018(4430248)
Type of Liquidation: CreditorsType of Liquidation: CreditorsDate of Appointment: 27 July 2023Date of Appointment: 24 August 2023Liquidator's name and address: Steven John Parker (IP No. 8989) andLiquidator's name and address: Nicola Elaine Layland (IP No. 01765)Timothy John Edward Dolder (IP No. 9008) both of OpusLiquidator's name and address: Nicola Elaine Layland (IP No. 01765)Restructuring LLP, 322 High Holborn, London, WC1V 7PBParkway, Solent Business Park, Whiteley, Fareham, Hampshire, PO1By whom Appointed: Members and Creditors7AGAg FJ53018(4430248)
Liquidator's name and address: Steven John Parker (IP No. 8989) and Timothy John Edward Dolder (IP No. 9008) both of Opus Restructuring LLP, 322 High Holborn, London, WC1V 7PB By whom Appointed: Members and CreditorsLiquidator's name and address: Nicola Elaine Layland (IP No. 01765) and Carl Derek Faulds (IP No. 008767) both of Leonard Curtis, 158 Parkway, Solent Business Park, Whiteley, Fareham, Hampshire, PO1 7AGAg FJ53018(4430248)
Restructuring LLP, 322 High Holborn, London, WC1V 7PBParkway, Solent Business Park, Whiteley, Fareham, Hampshire, PO1By whom Appointed: Members and Creditors7AGAg FJ53018(4430248)By whom Appointed: Members and Creditors
Ag FJ53018 (4430248) By whom Appointed: Members and Creditors
Ag FJ52869 (443018)
Company Number: 07728703
Name of Company: FLEET AGENTS LIMITED Nature of Business: Sale of used cars and light motor vehicles Name of Company: GUMBOO HOTELS LIMITED
Registered office: to be changed to TruSolv Ltd, Grove House, Trading Name: The Flying Horse
Meridians Cross, Ocean Village, Southampton, Hampshire, SO14 3TJ, having previously been 1 & 2 Studley Court Mews, Studley Court, Guildford Road, Chobham, Woking, Surrey, GU24 8EB. Company Number: 07586157 Registered office: 211 Manchester New Road, Middleton, Manchester M24 1JT
Principal trading address: Folly Farm, Folly Lane, Bramley, Tadley, Principal trading address: 37 Packer St, Rochdale, OL16 1NJ
Hampshire, RG26 5BD       Nature of Business: Hotel and Public House         Type of Liquidation: Creditors Voluntary Liquidation       Type of Liquidation: Creditors' Voluntary
Liquidator's name and address: <i>Matthew Hoy</i> (IP number 24550) of Date of Appointment: 24 August 2023 TruSolv Ltd, Grove House, Meridians Cross, Ocean Village, By whom Appointed: Members and Creditors
TruSolv Ltd, Grove House, Meridians Cross, Ocean Village, By whom Appointed: Members and Creditors Southampton, Hampshire, SO14 3TJ. Joint Liquidator's Name and Address: Andrew Mark
Liquidator's name and address: Shane Biddlecombe (IP number 9425) Bland(IP No. 9472) of DMC Becovery Limited 41
of TruSolv Ltd, Grove House, Meridians Cross, Ocean Village, Southampton, Hampshire, SO14 3TJ. Greek Street, Stockport, Cheshire, SK3 8AXEmail:
Date of Appointment: 23 August 2023 creditors@dmcrecovery.co.uk. Telephone: 0161 474
By whom Appointed: Members and Creditors 0920. Joint Liquidator's Name and Address: Janet 0920. Joint Liquidator's Name and Address: Janet Mayo(IP No. 9595) of DMC Recovery Limited, 41
Greek Street, Stockport, Cheshire, SK3 8AXEmail:
Company Number: 03801292 janet.mayo@dmcrecovery.co.uk. Telephone: 0161
Name of Company: <b>FUTURE ENGRAVING LIMITED</b> Nature of Business: Emboss and foil die production
Registered office: 2nd Floor, 14 Castle Street, Liverpool, L2 0NE Recovery Limited on 0161 474 0920, 000

Registered office: 2nd Floor, 14 Castle Street, Liverpool, L2 0NE Principal trading address: 98 Appin Road, Argyle Industrial Estate, Birkenhead, Wirral, CH41 9HH

Type of Liquidation: Creditors Voluntary Liquidation

Liquidator's name and address: John P Fisher (IP number 9420) of Parkin S Booth Ltd, 2nd Floor, 14 Castle Street, Liverpool, L2 0NE. Date of Appointment: 29 August 2023 By whom Appointed: Creditors

LONDON GAZETTE | CONTAINING ALL NOTICES PUBLISHED ONLINE ON 31 AUGUST 2023 | 17495

sandeep.borse@dmcrecovery.co.uk.

25 August 2023

Company Number: 09650124	Name of Company: INDUSTRIAL FORMS LTD
Name of Company: HADERA LIMITED	Company Number: 10384440
Previous Name of Company: Outu & K Transport Limited and Outu &	Nature of Business: Manufacture and Design
K Limited	Registered office: 29-31 Moorland Road, Stoke-On-Trent, ST6 1DS
Nature of Business: Wholesale of wine, beer, spirits, other alcoholic	Type of Liquidation: Creditors
beverages, wholesale of coffee, tea, cocoa and spices. Other	Date of Appointment: 25 August 2023
transportation support activities Registered office: Unit 2 Spinnaker Court, 1C Becketts Place,	Liquidator's name and address: <i>Simon Weir</i> (IP No. 9099) of DSi Business Recovery, 2 Lakeside, Calder Island Way, Wakefield, WF2
Hampton Wick, Kingston upon Thames, KT1 4EQ	7AW
Type of Liquidation: Creditors Voluntary Liquidation	By whom Appointed: Members and Creditors
Liquidator's name and address: Douglas John Pinteau (IP number	Ag FJ52882 (4430250)
19590) of WSM Marks Bloom LLP, Unit 2 Spinnaker Court, 1C	-
Becketts Place, Hampton Wick, Kingston upon Thames, KT1 4EQ.	
Liquidator's name and address: Richard Andrew Segal (IP number	Name of Company: JACKSON & O'LEARY CATERING LIMITED
2685) of WSM Marks Bloom LLP, Unit 2 Spinnaker Court, 1C	Company Number: 07090528
Becketts Place, Hampton Wick, Kingston upon Thames, KT1 4EQ.	Trading Name: Finbarr's Restaurant Nature of Business: Licensed Restaurant
Date of Appointment: 17 August 2023 By whom Appointed: Members and Creditors	Registered office: Aykley Heads House, Aykley Heads, Durham, DH1
For further details contact Ella Foggensteiner on 020 8939 8240	5TS in process of being changed to F17 Evolve Business Centre,
(4431839)	Cygnet Way, Houghton le Spring, Tyne & Wear, DH4 5QY
	Type of Liquidation: Creditors
Name of Company: HB PIZZA LTD	Date of Appointment: 24 August 2023
Company Number: 13563264	Liquidator's name and address: Steven Philip Ross (IP No. 9503) and
Trading Name: Papa Johns	Allan David Kelly (IP No. 9156) both of FRP Advisory Trading Limited,
Nature of Business: Take-away food shops and mobile food stands	F17 Evolve Business Centre, Cygnet Way, Houghton le Spring, Tyne
Registered office: 4 Whitchurch Parade, Whitchurch Lane, Edgware, HA8 6LR in process of being changed to c/o FRP Advisory Trading	and Wear, DH4 5QY By whom Appointed: Members and Creditors
Limited, 4 Beaconsfield Road, St Albans, AL1 3RD	Ag FJ52938 (4430246)
Type of Liquidation: Creditors	(
Date of Appointment: 24 August 2023	
Liquidator's name and address: Simon Carvill-Biggs (IP No. 15930)	Company Number: 09556359
and Miles Needham (IP No. 14372) both of FRP Advisory Trading	Name of Company: JAMES JONES@BF LTD
Limited, 4 Beaconsfield Road, St Albans, Hertfordshire, AL1 3RD	Trading Name: James Jones Beautiful Furnishings
By whom Appointed: Members and Creditors	Nature of Business: Wholesale of furniture, carpets and lighting
Ag FJ52962 (4430258)	equipment Registered office: 63 Newport Road, Middlesborough, TS1 1LB To be
	changed to: c/o Clark Business Recovery Limited, 8 Fusion Court,
Name of Company: HL & HL ANTIQUITIES LIMITED	Aberford Road, Garforth, Leeds, LS25 2GH
Company Number: 07270802	Principal trading address: 63 Newport Road, Middlesborough, TS1
Nature of Business: Other retail sale not in stores, stalls or markets	1LB & The Cleveland Centre, 85 Grange Road, Middlesborough, TS1
Registered office: 43 Essex Street, London, WC2R 3JF	2LS 80 High Street, Redcar, RS10 3DL
Type of Liquidation: Creditors	Type of Liquidation: Creditors Voluntary Liquidation
Date of Appointment: 24 August 2023 Liquidator's name and address: <i>Steven John Parker</i> (IP No. 8989) and	Liquidator's name and address: <i>Phil Clark</i> (IP number 23530) of Clark Business Recovery Limited, 8 Fusion Court, Aberford Road, Garforth,
<i>Trevor John Binyon</i> (IP No. 9285) both of Opus Restructuring LLP, 322	Leeds, LS25 2GH.
High Holborn, London, WC1V 7PB	Liquidator's name and address: Dave Clark (IP number 9565) of Clark
By whom Appointed: Creditors	Business Recovery Limited, 8 Fusion Court, Aberford Road, Garforth,
Ag FJ52941 (4430242)	Leeds, LS25 2GH.
	Date of Appointment: 29 August 2023
	By whom Appointed: Members and Creditors
Name of Company: HT ENTERPRISE SHEFF LIMITED	For further details contact Daniel Overend on 0113 243 8617 or at daniel@clarkbr.co.uk (4431402)
Company Number: 12380692 Nature of Business: Unlicensed restaurants and cafes	Gamereolarion.co.uk (4431402)
Registered office: 598 Bradford Road, Batley, F17 8LP	
Type of Liquidation: Creditors	
Date of Appointment: 29 August 2023	Name of Company: JAO CONSULTING LTD
Liquidator's name and address: David Frederick Wilson (IP No.	Company Number: 10079516
006074) of DFW Associates, 29 Park Square West, Leeds, LS1 2PQ	Registered office: 120 Bark Street, Bolton, Greater Manchester, BL1
By whom Appointed: Members and Creditors under the deemed	2AX Driveria al Angelia and Ang
consent procedure Ag FJ53017 (4430259)	Principal trading address: 1 Clover End, Buckingham, Buckinghamshire, MK18 1FT
Ag 1 8380 17 (4400200)	Nature of Business: Business and domestic software development
	Type of Liquidation: Creditors Voluntary Liquidation
Name of Company: ICONIC LEISURE LIMITED	Date of Appointment: 24 August 2023
Company Number: 12660703	By whom Appointed: Members and Creditors
Trading Name: The Aviary, Whalley	Liquidator's name and address: Rikki Burton (IP No. 14430) of
Nature of Business: Bar / Restaurant	Anderson Brookes Insolvency Practitioners Limited, 6th Floor, 120
Registered office: The Aviary, Queen Street, Whalley, Clitheroe,	Bark Street, Bolton, BL1 2AX.
Lancashire, BB7 9TA Type of Liquidation: Creditors	For further information contact Steven Marshall at the offices of Anderson Brookes Insolvency Practitioners Limited on 0800 1804
Date of Appointment: 23 August 2023	933, or steven@andersonbrookes.co.uk.
Liquidator's name and address: Zane Collins (IP No. 25952) of	30 August 2023 (4430996)
Insolvency One Limited, 1 Aire Street, Suite 2.03, Leeds, West	

(4430240)

Yorkshire, LS1 4PR

Ag FJ52992

By whom Appointed: Members and Creditors

(4429292)

(4430238)

(4431816)

(4431631)

#### Name of Company: JESSICA WILSON LIMITED Company Number: 11762162 Name of Company: LFW ENTERPRISE LTD Nature of Business: Manufacture of jewellery and related articles Trading Name: The Old Farm Inn and The Plough Registered office: 13 Wharf Lane, Cow Roast, Tring, HP23 5FQ and it Company Number: 12298816 is in the process of being changed to The Old Town Hall, 71 Registered office: 100, St James Road, Northampton, NN5 5LF Christchurch Road, Ringwood, BH24 1DH Principal trading address: 16 Church Road, Totternhoe, Dunstable, Type of Liquidation: Creditors LU6 1RE Date of Appointment: 24 August 2023 Nature of Business: Pub/restaurant Liquidator's name and address: David Meany (IP No. 9453) and Frank Type of Liquidation: Creditors' Voluntary Wessely (IP No. 7788) both of Quantuma Advisory Limited, The Old Date of Appointment: 21 August 2023 Town Hall, 71 Christchurch Road, Ringwood, BH24 1DH By whom Appointed: Members and Creditors By whom Appointed: The Company Joint Liquidator's Name and Address: Thomas Ag FJ52914 (4430228) Edward Guthrie(IP No. 15012) of BRI Business Recovery and Insolvency, 7 Paynes Park, Hitchin, Hertfordshire, SG5 1EHTelephone: 01462 429718. Name of Company: JOYNER ESTATE LIMITED Joint Liquidator's Name and Address: Lauren Louise Company Number: 09445858 Auburn(IP No. 20894) of BRI Business Recovery and Registered office: 120 Bark Street, Bolton, BL 1 2AX Insolvency, 7 Paynes Park, Hitchin, Hertfordshire, Principal trading address: 189 Lynchford Road, Farnborough, SG5 1EHTelephone: 01462 429718. Hampshire, GU14 6HD Nature of Business: Advertising agencies For further information contact Emma Johnson at the offices of BRI Type of Liquidation: Creditors Voluntary Liquidation Business Recovery and Insolvency on 01604 754352, or Date of Appointment: 25 August 2023 ejohnson@briuk.co.uk. By whom Appointed: Members and Creditors 22 August 2023 Liquidator's name and address: Rikki Burton (IP No. 14430) of Anderson Brookes Insolvency Practitioners Limited, 6th Floor, 120 Bark Street, Bolton, BL1 2AX. Name of Company: LISSUS LTD For further information contact Laura Wright at the offices of Company Number: 12203658 Anderson Brookes Insolvency Practitioners Limited on 0800 1804 Trading Name: Zebrano 933, or laura@andersonbrookes.co.uk. Nature of Business: Bar / restaurant 30 August 2023 (4431000) Registered office: c/o Begbies Traynor, 2/3 Pavilion Buildings, Brighton, East Sussex, BN1 1EE formerly 2nd Floor, Northumberland House, 303-306 High Holborn, London, WC1V 7JZ Company Number: 08781877 Type of Liquidation: Creditors Name of Company: KC COMPOSITES LIMITED Date of Appointment: 23 August 2023 Nature of Business: Manufacture of carbon fibre motorcycle parts Liquidator's name and address: Jonathan James Beard (IP No. 9552) Registered office: G204 Weston House, Allen House Business Centre, and Gavin Savage (IP No. 9950) both of Begbies Traynor (Central) The Mattings, Station Road, Sawbridgeworth, CM21 9FP LLP, 2/3 Pavilion Buildings, Brighton, East Sussex BN1 1EE Principal trading address: Unit 4, Water Hill Farm, Hatfield Road, By whom Appointed: Members and Creditors Hertford, SG13 8LE Ag FJ52996 Type of Liquidation: Creditors Voluntary Liquidation Liquidator's name and address: Alan J Clark (IP number 8760) of Carter Clark, Recovery House, 15-17 Roebuck Road, Hainault Name of Company: LIVETHEDREAM UK LTD Business Park, Ilford, Essex, IG6 3TU. Company Number: 06009881 Date of Appointment: 25 August 2023 Trading Name: Proof Bar & Kitchen By whom Appointed: The members and creditors Nature of Business: Public houses and bars For further details contact Lisa Portway on 020 8559 5087 or at Registered office: C/O Begbies Traynor, 340 Deansgate, Manchester, Jenny.poleykett@carterclark.co.uk (4431359) M3 4LY Type of Liquidation: Creditors Date of Appointment: 17 August 2023 Name of Company: LAHORE QILA LTD Liquidator's name and address: Paul Stanley (IP No. 008123) and Company Number: 10903290 Dean Watson (IP No. 009661) both of Begbies Traynor (Central) LLP, Nature of Business: Food takeaway 340 Deansgate, Manchester, M3 4LY Registered office: 229 Whalley New Road, Blackburn, BB1 9TL By whom Appointed: Members and Creditors Type of Liquidation: Creditors Ag FJ53060 Date of Appointment: 29 August 2023 Liquidator's name and address: James David Robinson (IP No. 16092) of Finn Associates, Tong Hall, Tong, West Yorkshire BD4 0RR Company Number: 11875442 By whom Appointed: Members and Creditors Name of Company: MOTO DOCTOR LTD Ag FJ53025 (4430245) Nature of Business: Sale, maintenance and repair of motorcycles and related parts and accessories. Registered office: 11 Courtleigh Gardens, NW11 9JX Name of Company: LEIGH SAFETY MANAGEMENT LIMITED Principal trading address: 11 Courtleigh Gardens, NW11 9JX Company Number: 07524224 Type of Liquidation: Creditors Voluntary Liquidation Nature of Business: Other service activities not elsewhere classified Liquidator's name and address: Nimish Patel (IP number 8679) of Registered office: Charter House, 103-105 Leigh Road, Leigh on Sea, Hudson Weir Limited, 58 Leman Street, London, E1 8EU. Essex SS9 1JL (in the process of being changed to Jupiter House, Liquidator's name and address: Hasib Howlader (IP number 20350) of Warley Hill Business Park, The Drive, Brentwood, Essex, CM13 3BE) Hudson Weir Limited, 58 Leman Street, London, E1 8EU. Type of Liquidation: Creditors Date of Appointment: 23 August 2023 Date of Appointment: 24 August 2023 By whom Appointed: Creditors and Members For further details contact Tiffany Truter on 020 7099 6086 or at

Liquidator's name and address: Paul Atkinson (IP No. 9314) and Glyn Mummery (IP No. 8996) both of FRP Advisory Trading Limited, Jupiter House, Warley Hill Business Park, The Drive, Brentwood, Essex, CM13 3BE

By whom Appointed: Creditors Ag FJ52907

(4430272)

tiffany@hudsonweir.co.uk

Name of Company: NAYLOR GEORGE RECRUITMENT LIMITED	Company Number: 11680889
Company Number: 08835452 Nature of Business: Other activities of employment placement	Name of Company: <b>PIZZA JUNCTION 11 LTD</b> Nature of Business: Take-away food shops and mobile food stands
agencies Registered office: Wilkin Chapman LLP, Cartergate House, 26 Chantry Lane,Grimsby,North East Lincolnshire, DN31 2LJ	Registered office: to be changed to Grove House, Meridians Cross, Ocean Village, Southampton SO14 3TJ, having previously been 43 Sherwood Road, Luton, Bedfordshire, LU4 8LQ
Type of Liquidation: Creditors	Principal trading address: Ground Floor, 152 Dunstable Road, Luton,
Date of Appointment: 24 August 2023	LU1 1EW
Liquidator's name and address: Matthew Dix (IP No. 21492) of Wilkin	Type of Liquidation: Creditors Voluntary Liquidation
Chapman LLP, Cartergate House, 26 Chantry Lane, Grimsby, DN31 2LJ	Liquidator's name and address: <i>Matthew Hoy</i> (IP number 24550) of TruSolv Ltd, Grove House, Meridians Cross, Ocean Village,
By whom Appointed: Members and Creditors	Southampton, SO14 3TJ.
Ag FJ53021 (4430243)	Liquidator's name and address: Shane Biddlecombe (IP number 9425)
	of TruSolv Ltd, Grove House, Meridians Cross, Ocean Village,
SECTION 109(1) INSOLVENCY ACT 1986 AND RULE 6.23	Southampton, SO14 3TJ. Date of Appointment: 23 August 2023
INSOLVENCY RULES 2016 (AS AMENDED)	By whom Appointed: Members and Creditors
Name of Company: NEWPORT FABRICATIONS LTD	For further details contact Kay Lloyd on 0808 196 8676 or at
Company Number: 11939699 Registered office: 42 Lytton Road, Barnet EN5 5BY	kay.lloyd@trusolv.co.uk (4431667)
Principal trading address: 1 Vermont Place, Tongwell, Milton Keynes	
MK15 8JA	Name of Company: POLAR MANUFACTURING LTD
Nature of Business: Supply and installation of steel work	Company Number: 07821431
Type of Liquidation: Creditors' Voluntary Liquidation	Nature of Business: Other manufacturing not elsewhere classified
Liquidator's name and address: <i>Paul Weber</i> FCA FCCA MIPA FABRP and <i>Martin Linton</i> FCA FABRP MIPA of Leigh Adams Limited, Maple	Registered office: Unit 35a Bowthorpe Industrial Estate, Barnard Road, Norwich, NR5 9JB
House, High Street, Potters Bar, Herts, EN6 5BS	Type of Liquidation: Creditors
Office Holder Numbers: 9400 and 5998.	Date of Appointment: 25 August 2023
Date of Appointment: 21 August 2023	Liquidator's name and address: Jeremy Karr (IP No. 9540) and Simon
By whom Appointed: Members and creditors Contact information Zuzana Drengubiakova by email at	<i>Killick</i> (IP No. 23390) both of Begbies Traynor (London) LLP, 31st Floor, 40 Bank Street, London, E14 5NR
mail@leighadams.co.uk (4431882)	By whom Appointed: Members and Creditors
	Ag FJ52964 (4430260)
Name of Company: NISAVA LTD	
Company Number: 12128026	Name of Company: PREMIER FURNITURE WAREHOUSE LTD
Trading Name: Kod Pirketi Nature of Business: Licensed restaurants	Company Number: 09005895 Nature of Business: Retail of furniture, lighting, and similar (not
Registered office: 47 High Street, Acton, London, W3 6ND	musical instruments or scores) in specialised store
Type of Liquidation: Creditors	Registered office: 18-20 High Street, Shaftesbury, SP7 8JG and it is in
Date of Appointment: 24 August 2023	the process of being changed to The Old Town Hall, 71 Christchurch
Liquidator's name and address: <i>Farheen Qureshi</i> (IP No. 11392) of Parker Getty Limited, Devonshire House, 582 Honeypot Lane,	Road, Ringwood, BH24 1DH Type of Liquidation: Creditors
Stanmore, Middx, HA7 1JS	Date of Appointment: 25 August 2023
By whom Appointed: Members and Creditors	Liquidator's name and address: Simon Campbell (IP No. 10150) and
Ag FJ52886 (4430253)	David Meany (IP No. 9453) both of Quantuma Advisory Limited, The
	Old Town Hall, 71 Christchurch Road, Ringwood, BH24 1DH By whom Appointed: The Company
Name of Company: PALLAS TRANSPORT SERVICES LIMITED	Ag FJ52900 (4430227)
Company Number: 06971129	
Nature of Business: Freight transport by road	Commony Number 10001170
Registered office: Unit 1, Bewick Main, Birtley, Chester-Le-Street, DH2 1BQ	Company Number: 12091173 Name of Company: <b>PROJECTS UK (EAST ANGLIA) LIMITED</b>
Type of Liquidation: Creditors	Previous Name of Company: C & W Projects Limited
Date of Appointment: 24 August 2023	Nature of Business: Roofing Company
Liquidator's name and address: Gillian Margaret Sayburn (IP No.	Registered office: The Pippins, Waterhouse Lane, Ardleigh, Essex,
10830) and <i>Andrew Little</i> (IP No. 9668) both of Begbies Traynor (Central) LLP, Ground Floor, Portland House, 54 New Bridge Street	CO7 7TE Principal trading address: Langham Barns Business Centre, Langham
West, Newcastle upon Tyne, NE1 8AP	Lane, Langham Colchester CO4 5ZS
By whom Appointed: Members and Creditors	Type of Liquidation: Creditors Voluntary Liquidation
Ag FJ52999 (4430257)	Liquidator's name and address: Liquidator: Sonia Baigent (IP number
	9494) of Assist Business Consulting Ltd, 5 Old Forge Road, Ashby Magna, Lutterworth, Leics, LE17 5NL.
Name of Company: PEDRICK'S ZERO WASTE LTD	Date of Appointment: 18 August 2023
Company Number: 11681568	By whom Appointed: Members
Trading Name: Pedrick's Zero Waste Shop Nature of Business: Other retail sale of food in specialised stores	For further details contact Sonia Baigent on 01455 560042 or at sonia@assistbc.com (4432022)
Registered office: 41 Croydon Road, Caterham, CR3 6PB	(++02022)
Type of Liquidation: Creditors	
Date of Appointment: 25 August 2023	
Liquidator's name and address: Christopher Herron (IP No. 8755) and Nicola Jayne Fisher (IP No. 9090) both of Herron Fisher, Satago	
Cottage, 360a Brighton Road, Croydon, CR2 6AL	
By whom Appointed: Members and Creditors	
Ag FJ53008 (4430254)	

	COMPANIES
Name of Company: <b>QUIRK PROJECT SUPPORT LTD</b> Company Number: 10602731 Nature of Business: Scenery & Props Storage Registered office: 63 Birtles Road, Macclesfield, SK10 3JG Type of Liquidation: Creditors Date of Appointment: 25 August 2023 Liquidator's name and address: <i>Darren Brookes</i> (IP No. 9297) of Milner Boardman & Partners, The Old Bank, 187a Ashley Road, Hale, Cheshire, WA15 9SQ By whom Appointed: Members and Creditors Ag FJ52934 (4430233)	Name of Company: <b>SHOWCASE DESIGN LIMITED</b> Company Number: 13422532 Nature of Business: Manufacture of other furniture Registered office: 1 Park Road, Hampton Wick, Kingston Upon Thames, KT1 4AS and it is in the process of being changed to 1st Floor, 21 Station Road, Watford, WD17 1AP Type of Liquidation: Creditors Date of Appointment: 24 August 2023 Liquidator's name and address: <i>Nicholas Simmonds</i> (IP No. 9570) and <i>Chris Newell</i> (IP No. 13690) both of Quantuma Advisory Limited, 1st Floor, 21 Station Road, Watford, Herts, WD17 1AP By whom Appointed: The Company Ag FJ52923 (4430213)
Name of Company: REAL FINANCIAL ADVICE LIMITED	
Company Number: 12123343 Nature of Business: Mortgage & Protection Advisors Registered office: Suite 20a, Off Liverpool Road, Empire Business Park, Burnley, Lancashire, BB12 6HH Type of Liquidation: Creditors Date of Appointment: 25 August 2023 Liquidator's name and address: <i>Jonathan Mark Taylor</i> (IP No. 10570) of T H Financial Recovery, Suites 101 & 102, Empire Way Business Park, Liverpool Road, Burnley, BB12 6HH By whom Appointed: Members and Creditors Ag FJ52870 (4430195)	Name of Company: <b>SINGLETON'S DAIRY LIMITED</b> Company Number: 00393664 Nature of Business: Butter and cheese production Registered office: c/o Kroll Advisory Ltd, The Chancery, 58 Spring Gardens, Manchester, M2 1EW Type of Liquidation: Creditors Date of Appointment: 23 August 2023 Liquidator's name and address: <i>Andrew Knowles</i> (IP No. 24850) and <i>Steven Muncaster</i> (IP No. 9446) both of Kroll Advisory Ltd, The Chancery, 58 Spring Gardens, Manchester, M2 1EW By whom Appointed: Made pursuant to Schedule B1 Paragraph 83 of
Name of Company: ROBERT OLIVIA CARS LTD.	the Insolvency Act 1986 Ag FJ52974 (4430230)
Company Number: 10754141 Nature of Business: Buying and selling of used cars Registered office: 1 Matchingfield, Kelvedon Hatch, Brentwood, CM15 0XB Type of Liquidation: Creditors Date of Appointment: 24 August 2023 Liquidator's name and address: <i>Richard Hunt</i> (IP No. 21772) of SFP Restructuring Limited, 9 Ensign House, Admirals Way, Marsh Wall, London, E14 9XQ By whom Appointed: Members and Creditors Ag FJ52994 (4430234)	Name of Company: <b>STONEGROVE FACILITIES MANAGEMENT</b> <b>LIMITED</b> Company Number: 06831659 Registered office: 4 Bloors Lane, Rainham, Gillingham, ME8 7EG Principal trading address: Floors 1 and 2, Hamilton House, 84-86 High Street, Rainham, ME8 7JH Nature of Business: Facilities Managment Type of Liquidation: Creditors' Voluntary Date of Appointment: 21 August 2023 By whom Appointed: Members & Creditors
Name of Company: <b>SANDIWAY SUPPLIES LIMITED</b> Company Number: 13117337 Registered office: Blakeden Cottage, Blakeden Ln, Chester Ln, Winsford, CW7 2QL Principal trading address: Blakeden Cottage, Blakeden Ln, Chester Ln, Winsford, CW7 2QL Nature of Business: Animal Food Retailer Type of Liquidation: Creditors' Voluntary Date of Appointment: 23 August 2023 By whom Appointed: Members and Creditors Liquidator's name and address: <i>Gareth Howarth</i> (IP No. 18816) of Path Business Recovery Limited, Flint Glass Works, 64 Jersey Street, Manchester, M4 6JW. Telephone: 0161 413 0999. For further information contact George Blake at the offices of Path Business Recovery Limited on 0161 413 0999, or george.blake@pathbr.co.uk.	Joint Liquidator's Name and Address: Amanda Janice Ireland(IP No. 9274) of Kent Insolvency Limited, 2 The Links, Herne bay, Kent, CT6 7GQTelephone: 01795433655. Joint Liquidator's Name and Address: Alison Jane Collier(IP No. 12252) of Kent Insolvency Limited, 2 The Links, Herne bay, Kent, CT6 7GQTelephone: 01795433655. For further information contact Alison Jane Collier at the offices of Kent Insolvency Limited on 01795 433655, or alison.collier@compassinsolvency.co.uk. 23 August 2023 (4429206) Name of Company: <b>STONES AQUATICS LTD</b> Company Number: 11897786
goorge.balace.pumbliol.dit.         23 August 2023       (4429212)         Name of Company: SANGEETA HAIR SOLUTIONS LTD         Company Number: 07520921         Nature of Business: Hair and Beauty         Registered office: Flat 24 Dorchester House, 228 Great Western Road,         London, W11 1BE         Type of Liquidation: Creditors         Date of Appointment: 23 August 2023         Liquidator's name and address: <i>Richard Hunt</i> (IP No. 21772) of SFP         Restructuring Limited, 9 Ensign House, Admirals Way, Marsh Wall,         London, E14 9XQ         By whom Appointed: Members and Creditors	Trading Name: Stones Aquatics & Natureopolis Registered office: C/O Fortis Insolvency Limited, 683-693 Wilmslow Road, Manchester, M20 6RE (previously 50 Maple Street, Sheerness, ME12 1XJ) Type of Liquidation: Creditors Date of Appointment: 29 August 2023 Liquidator's name and address: <i>Daniel Taylor</i> (IP No. 21050) of Fortis Insolvency Limited, 683-693 Wilmslow Road, Manchester, M20 6RE By whom Appointed: Creditors and Members Ag FJ52920 (4430216)

(4430241)

Ag FJ52976

#### Name of Company: T&L PLAY LIMITED

Company Number: 12265016

Nature of Business: Retail sale via mail order houses or via internet Registered office: The registered office of the Company will be changed to Sussex Innovation, One Croydon, 12-16 Addiscombe Road, Croydon, CR0 0XT

Type of Liquidation: Creditors

Date of Appointment: 25 August 2023

Liquidator's name and address: Matthew Reeds (IP No. 28040) and Paul Bailey (IP No. 9428) both of Bailey Ahmad Business Recovery, Sussex Innovation, One Croydon, 12-16 Addiscombe Road, Croydon, CR0 0XT

By whom Appointed: Members and Creditors

Ag FJ52927

(4430244)

Company Number: 11454330

Name of Company: THE GINGER DOG PUB COMPANY LTD

Trading Name: The Rose and Crown & The Red Lion

Nature of Business: Public houses and bars

Registered office: to be changed to Grove House, Meridians Cross, Ocean Village, Southampton SO14 3TJ, having previously been Rose & Crown Woodside Road, Winkfield, Windsor, SL4 2DP

Principal trading address: Rose & Crown Woodside Road, Winkfield, Windsor, SL4 2DP & 41 The Green, Woodburn Green, GP10 0EU

Type of Liquidation: Creditors Voluntary Liquidation

Liquidator's name and address: Shane Biddlecombe (IP number 9425) of TruSolv Ltd, Grove House, Meridians Cross, Ocean Village, Southampton, SO14 3TJ.

Liquidator's name and address: Matthew Hoy (IP number 24550) of TruSolv Ltd, Grove House, Meridians Cross, Ocean Village, Southampton, SO14 3TJ.

Date of Appointment: 23 August 2023

By whom Appointed: Members and Creditors

For further details contact Kay Lloyd on 0808 196 8676 or at (4431705) kay.lloyd@trusolv.co.uk

#### Company Number: 11669244

#### Name of Company: THE SPA & BEAUTY CLINIC LTD

Nature of Business: Other services activities not elsewhere classified Registered office: The Chapel, Bridge Street, Driffield, YO25 6DA Principal trading address: 1 & 3, Red Lion St, Goole DN14 6BX

Type of Liquidation: Creditors Voluntary Liquidation

Liquidator's name and address: Joint Liquidator: John William Butler (IP number 9591) of Redman Nichols Butler, The Chapel, Bridge Street, Driffield, East Yorkshire, YO25 6DA.

Liquidator's name and address: Joint Liquidator: Andrew James Nichols (IP number 8367) of Redman Nichols Butler, The Chapel, Bridge Street, Driffield, East Yorkshire, YO25 6DA.

Date of Appointment: 24 August 2023

By whom Appointed: Members and Creditors For further details contact: 01377 257788

(4431635)

Company Number: 10054075

# Name of Company: TICKBOX TRAINING LIMITED

Nature of Business: Training and Education

Registered office: Dunheved 10 Park Place Cornwall PL27 7EA

Principal trading address: Dunheved 10 Park Place Cornwall PL27 7EA

Type of Liquidation: Creditors Voluntary Liquidation

Liquidator's name and address: Joint Liquidator: Scott Kippax (IP number 26030) of Begbies Traynor (Central) LLP, Balliol House Southernhay Gardens Exeter EX1 1NP.

Liquidator's name and address: Joint Liquidator: Julie Anne Palmer (IP number 8835) of Begbies Traynor (Central) LLP, Units 1-3 Hilltop Business Park Devizes Road Salisbury Wiltshire SP3 4UF.

Date of Appointment: 24 August 2023

By whom Appointed: Members and Creditors

For further details contact Nikita Iggulden on 01392 260800 or at Nikita.lggulden@btguk.com (4431912)

#### Company Number: 07197292

Name of Company: TRENDY TRAINERS LIMITED

Nature of Business: Wholesale

Registered office: 82a High Street Hertfordshire EN6 5AB

Principal trading address: 82a High Street Hertfordshire EN6 5AB

Type of Liquidation: Creditors Voluntary Liquidation

Liquidator's name and address: Joint Liquidator: Scott Kippax (IP number 26030 ) of Begbies Traynor (Central) LLP, Balliol House Southernhay Gardens Exeter EX1 1NP

Liquidator's name and address: Joint Liquidator: Julie Anne Palmer (IP number 8835) of Begbies Traynor (Central) LLP, Units 1-3 Hilltop Business Park Devizes Road Salisbury Wiltshire SP3 4UF . Date of Appointment: 23 August 2023

By whom Appointed: Members and Creditors

For further details contact Nikita Iggulden on 01392 260800 or at Nikita.lggulden@btguk.com (4431901)

# Name of Company: TRISTAN THOMSON PHOTOGRAPHY LTD

Company Number: 11068924

Nature of Business: Photography

Registered office: Wits End, Fulstone Hall Lane, Holmfirth, HD9 7DL

Type of Liquidation: Creditors

Date of Appointment: 29 August 2023

Liquidator's name and address: Nicola Baker (IP No. 15852) of Rushtons Insolvency Limited, 6 Festival Building, Ashley Lane, Saltaire, BD17 7DQ

By whom Appointed: Members and Creditors Ag FJ52913

(4430270)

# Name of Company: TURCO'S HOSPITALITY LIMITED

Company Number: 10830373

Trading Name: Al Dente Pasta Bar and Kitchen Nature of Business: Licensed restaurant

Registered office: Yare House, 62-64 Thorpe Road, Norwich, NR1 1RY

Type of Liquidation: Creditors

Date of Appointment: 24 August 2023

Liquidator's name and address: Nicholas Cusack (IP No. 17490) of Parker Andrews Limited, 5th Floor, The Union Building, 51-59 Rose Lane, Norwich, NR1 1BY

By whom Appointed: Members and Creditors Ag FJ52885

(4430177)

#### Company Number: 10049902

Name of Company: UNFAIR ADVANTAGE ENTERPRISES LIMITED Nature of Business: Management consultancy activities other than financial management

Registered office: 8 Domingo Street, London, EC1Y 0TA

Principal trading address: 8 Domingo Street, London, EC1Y 0TA

Type of Liquidation: Creditors Voluntary Liquidation Liquidator's name and address: Nimish Patel (IP number 8679) of Hudson Weir Limited, 58 Leman Street, London, E1 8EU.

Liquidator's name and address: Hasib Howlader (IP number 20350) of Hudson Weir Limited, 58 Leman Street, London, E1 8EU.

Date of Appointment: 22 August 2023

By whom Appointed: Creditors and Members

For further details contact Jinjin Qi on 020 7099 6086 or at jinjin@hudsonweir.co.uk (4431625)

#### Name of Company: UNFITTED LTD

Company Number: 07499222

Nature of Business: Kitchen retail sales

Registered office: The Old Rectory, Main Street, Glenfield, Leicester, LE3 8DG (Formerly) Hope House, High Street, Moreton-In-Marsh, GL56 0LH

Type of Liquidation: Creditors

Date of Appointment: 19 July 2023

Liquidator's name and address: Gavin Bates (IP No. 8983) of Smart Business Recovery Limited, The Old Rectory, Main Street, Glenfield, Leicester, LE3 8DG

By whom Appointed: Members and Creditors Ag FJ52905

(4430273)

#### Name of Company: WILL'S JOINERY LIMITED

Company Number: 11080743

Nature of Business: Joinery Installation

Registered office: 7 Davenport Road, Stockport, Cheshire SK7 4EZ Type of Liquidation: Creditors

Date of Appointment: 25 August 2023

Liquidator's name and address: *Daniel Ormerod* (IP No. 26930) and *Martin Maloney* (IP No. 9628) both of Leonard Curtis, Leonard Curtis House, Elms Square, Bury New Road, Whitefield, Greater Manchester, M45 7TA

By whom Appointed: Members and Creditors Ag FJ52971

(4430262)

### **MEETINGS OF CREDITORS**

#### 80 OAK LDN LIMITED

(Company Number 11966152)

Registered office: 5 Churchill Place, 10th Floor, London, E14 5HU Principal trading address: N/A

This notice is given pursuant to Rule 15.13 of the Insolvency (England and Wales) Rules 2016.

Notice is hereby given that a Virtual Meeting of the creditors of the above named Company is being convened and will be held on 1 August 2023 (Decision Date) at 2pm. Michael Rispoli, the Director of the Company is the convener of the meeting ("the Convener").

The primary purpose of the Virtual Meeting is for creditors to consider a decision on the nomination of liquidators, and if the creditors think fit, to appoint a liquidation committee. In addition, in the event that a committee is not formed, creditors will also be asked to consider a decision in relation to the payment of the costs of assistance with preparation of the Company's statement of affairs and seeking the creditors' decision on the nomination of liquidators, as an expense of the liquidation. Creditors participating in the Virtual Meeting or attending by proxy, (unless a committee is formed), will also be asked to consider approving the basis of the Joint Liquidators drawing their disbursements, details of which have been provided in a report to creditors.

In order to be entitled to vote at the meeting (either in person or by proxy), creditors must have submitted a proof in respect of their claim (unless a proof has already been submitted) to the Convener via the contact details below, by no later than 4pm on the business day immediately before the day of the Decision Date.

Any creditor entitled to attend and vote at the meeting is entitled to do so either in person or by proxy. If you cannot attend and wish to be represented at the meeting, a completed proxy form must be delivered to the Convener via the contact details below, prior to the meeting.

Log-in details for the virtual meeting will be supplied to creditors on request.

Any person who requires further information may contact Shanice Kearns by telephone on 020 8343 5900 or alternatively by e-mail at ET-Team@btguk.com.

*Michael Rispoli,* Director/Convener 24 August 2023

Ag FJ52972

(4430207)

### ATISA LTD

(Company Number 11414134)

Registered office: 88 Hall Road, London, E6 2ND

This notice is given pursuant to Rule 15.13 of the Insolvency (England and Wales) Rules 2016.

Notice is hereby given that a virtual meeting of the creditors of the above-named Company is being convened and will be held 5 September 2023 at 11:00am. Mr Aurel Olteanu, the director of the Company, is the convener of the meeting ("the Convener").

The primary purpose of the virtual meeting is for creditors to consider a decision on the nomination of liquidators, and if the creditors think fit, to appoint a liquidation committee.

In addition, in the event that a committee is not formed, creditors will also be asked to consider a decision in relation to the payment of the costs of assistance with preparation of the Company's statement of affairs and seeking the creditors' decision on the nomination of liquidators, as an expense of the liquidation. Creditors participating in the virtual meeting or attending by proxy, (unless a committee is formed), will also be asked to consider approving the basis of the joint liquidators' remuneration and whether they also approve the joint liquidators drawing their disbursements, details of which have been provided in a report to creditors.

In order to be entitled to vote at the meeting (either in person or by proxy), creditors must have submitted a proof in respect of their claim (unless a proof has already been submitted) to the Convener via the contact details below, by no later than 4pm on 4 September 2023.

Any creditor entitled to attend and vote at the meeting is entitled to do so either in person or by proxy. If you cannot attend and wish to be represented at the meeting, a completed proxy form must be delivered to the Convener via the contact details below, prior to the meeting.

Log-in details for the virtual meeting will be supplied to creditors on request.

For further details contact Emily Higgins on 0208 370 7250 or at Emily.Higgins@btguk.com

Dated: 24 August 2023 Mr Aurel Olteanu, Convener

(4431954)

### CISO CONSULTANCY LIMITED

(Company Number 09964693)

Registered office: Optionis House 840 Ibis Court, Centre Park, Warrington, WA1 1RL

Principal trading address: 6 Sandhurst Drive, Ilford, Essex, IG3 9DB

Notice is hereby given that a virtual meeting of the creditors of the above-named Company will be held on 12 September 2023 at 2.30 pm for the purpose provided for in Section 100 of the Insolvency Act 1986.

Creditors entitled to attend and vote at the meeting may do so either in person or by proxy. A creditor can attend the meeting in person and vote, and is entitled to vote if they have delivered proof of their debt by no later than 4.00 pm on the business day before the meeting. If a creditor cannot attend in person, or do not wish to attend but still wish to vote at the meeting, they can either nominate a person to attend on their behalf, or they may nominate the Chair of the meeting, who will be a director of the Company, to vote on their behalf. Creditors must deliver their proxy, together with proof of their debt, by no later than the commencement of the meeting. Creditors must deliver all proofs of their debt and proxies to Company Liquidations Limited, Saxon House, Saxon Way, Cheltenham, GL52 6QX.

Unless they surrender their security, secured creditors must give particulars of their security, the date when it was given and the estimated value at which it is assessed if they wish to vote at the meeting.

At the meeting, creditors may receive information about, or be called upon to approve, the costs of preparing the statement of affairs and convening the meeting of creditors, and may be requested to consider a resolution specifying the terms on which the Liquidator is to be remunerated.

Simon Barriball and Helen Whitehouse of Company Liquidations Limited, Saxon House, Saxon Way, Cheltenham, GL52 6QX are qualified to act as Insolvency Practitioners in relation to the Company and will, during the period before the day of the meeting, furnish creditors free of charge with such information concerning the Company's affairs as they may reasonably require. Creditors can contact them on 01242 576555 or by email at info@companyliquidations.co.uk.

Nadeem Bukhari, Director

25 August 2023

Ag FJ52872

(4430167)

#### **CLARKSON ENGINEERS LIMITED**

(Company Number 03250730)

Registered office: 54 Sandringham Road, Retford, DN22 7QW

Principal trading address: 54 Sandringham Road, Retford, DN22 7QW Notice is hereby given, pursuant to Rule 15.13 of the INSOLVENCY (ENGLAND AND WALES) RULES 2016, that the Director of the abovenamed Company (the 'convener)') is seeking a decision from creditors on the nomination of Joint Liquidators by way of a virtual meeting. A resolution to wind up the Company is to be considered on 8 September 2023. Decisions regarding the Joint Liquidators' remuneration and the formation of a liquidation committee may also be sought at the meeting. The meeting will be held as a virtual meeting by Microsoft Teams Conference call on 8 September 2023 at 11.15 am. Details of how to access the virtual meeting are included in the notice delivered to creditors. If any creditor has not received this notice or requires further information please contact the nominated Joint Liquidators using the details below.

Fiona Grant and Emma Dowd of Wilson Field Limited are persons qualified to act as insolvency practitioners in relation to the company who, during the period before the meeting date, will furnish creditors free of charge with such information concerning the Company's affairs as they may reasonably require.

A creditor may appoint a person as a proxy-holder to act as their representative and to speak, vote, abstain or propose resolutions at the meeting. A proxy for a specific meeting must be delivered to the chair before the meeting. Proxies may be delivered to Wilson Field, The Manor House, 260 Ecclesall Rd South, Sheffield, S11 9PS. In order to be counted a creditor's vote must be accompanied by a proof in respect of the creditor's claim (unless it has already been given). A vote will be disregarded if a creditor's proof in respect of their claim is not received by 4pm on 7 September 2023 (unless the chair of the meeting is content to accept the proof later). Proofs may be delivered to Wilson Field Ltd, The Manor House, 260 Ecclesall Rd South, Sheffield, S11 9PS.

Names and address of nominated Liquidators: Fiona Grant (IP No. 9444) and Emma Dowd (IP No. 17650) both of Wilson Field Limited, The Manor House, 260 Ecclesall Road South, Sheffield, S11 9PS.

For further details contact: The nominated Liquidators, Tel: 0114 2356780, Email: b.hindle@wilsonfield.co.uk. Alternative contact: Benedict Hindle.

John Vincent Clarkson, Director

29 August 2023 Ag FJ52937

(4430223)

# DASH PUBLISHING LIMITED

(Company Number 03684852) Registered office: C/O Condy Mathias, 6 Houndiscombe Road, Plymouth, Devon, PL4 6HH

Principal trading address: Unit 1, Drakes Court, Langage Industrial Estate, Plymouth, PL7 5JY

Notice is hereby given, pursuant to Rule 15.13 of the INSOLVENCY (ENGLAND AND WALES) RULES 2016, that the Director of the abovenamed Company (the 'convener)') is seeking a decision from creditors on the nomination of Joint Liquidators by way of a virtual meeting. A resolution to wind up the Company is to be considered on 7 September 2023. Decisions regarding the Joint Liquidators' remuneration and the formation of a liquidation committee may also be sought at the meeting. The meeting will be held as a virtual meeting by Microsoft Teams Conference call on 7 September 2023 at 11.15 am. Details of how to access the virtual meeting are included in the notice delivered to creditors. If any creditor has not received this notice or requires further information please contact the nominated Joint Liquidators using the details below.

Gemma Louise Roberts and Kelly Burton of Wilson Field Limited are persons qualified to act as insolvency practitioners in relation to the company who, during the period before the meeting date, will furnish creditors free of charge with such information concerning the Company's affairs as they may reasonably require.

A creditor may appoint a person as a proxy-holder to act as their representative and to speak, vote, abstain or propose resolutions at the meeting. A proxy for a specific meeting must be delivered to the chair before the meeting. Proxies may be delivered to Wilson Field, The Manor House, 260 Ecclesall Rd South, Sheffield, S11 9PS. In order to be counted a creditor's vote must be accompanied by a proof in respect of the creditor's claim (unless it has already been given). A vote will be disregarded if a creditor's proof in respect of their claim is not received by 4pm on the business day prior to the day of the meeting (unless the chair of the meeting is content to accept the proof later). Proofs may be delivered to Wilson Field Ltd, The Manor House, 260 Ecclesall Rd South, Sheffield, S11 9PS.

Names and address of nominated Liquidators: Gemma Louise Roberts (IP No. 9701) and Kelly Burton (IP No. 11750) both of Wilson Field Limited, The Manor House, 260 Ecclesall Road South, Sheffield, S11 9PS.

For further details contact: The nominated Liquidators, Tel: 0114 2356780, Email: s.furbank@wilsonfield.co.uk. Alternative contact: Stephen Furbank.

*David Hildrew,* Director 29 August 2023 Ag FJ52968

Ag FJ52968

#### DISCIDIUM LTD

(Company Number 10042570)

Registered office: 4646 Brindles Field, Tonbridge, Kent, TN9 2YS Principal trading address: Suite 2 70 Churchill Square Business

Centre Churchill Square, Kings Hill, West Malling, ME19 4YU

Notice is hereby given, pursuant to Rule 15.13 of the Insolvency (England and Wales) Rules 2016, that the Director of the abovenamed Company (the 'convener') is seeking a decision from creditors on the nomination of a Liquidator by way of a physical meeting. A resolution to wind up the Company is to be considered on 5 September 2023. Decisions regarding the nominated Liquidator's pre appointment costs for convening the physical meeting and the formation of a liquidation committee may also be sought at the meeting.

The meeting will be held at Stanmore House, 64-68 Blackburn Street, Radcliffe, Manchester M26 2JS on 5 September 2023 at 12.00 pm. As a result of the requirement to hold this physical meeting the original deemed consent procedure is superseded. The Directors have discretion to permit remote attendance (meaning attending and being able to participate in the meeting without being in the place where it is being held) if such a request to do so is received in advance of the meeting.

Stephen Lancaster of Lucas Ross Limited, Stanmore House, 64-68 Blackburn Street, Radcliffe, Manchester, M26 2JS is a person qualified to act as an insolvency practitioner in relation to the company who, during the period before the meeting date, will furnish creditors free of charge with such information concerning the Company's affairs as they may reasonably require.

A creditor may appoint a person as a proxy-holder to act as their representative and to speak, vote, abstain or propose resolutions at the meeting. A proxy for a specific meeting must be delivered to the chair before the meeting. Proxies may be delivered to Lucas Ross Limited. Stanmore House, 64-68 Blackburn Street, Radcliffe, Manchester, M26 2JS or by email to help@lucasross.co.uk

In order to be counted a creditor's vote must be accompanied by a proof in respect of the creditor's claim (unless it has already been given). A vote will be disregarded if a creditor's proof in respect of their claim is not received by 4.00 pm on 4 September 2023 (unless the chair of the meeting is content to accept the proof later). Proofs may be delivered to Lucas Ross Limited, Stanmore House, 64-68 Blackburn Street, Radcliffe, Manchester, M26 2JS or by email to help@lucasross.co.uk.

Name and address of nominated Liquidator: Stephen Lancaster (IP No. 13910) of Lucas Ross Limited, Stanmore House, 64-68 Blackburn Street, Radcliffe, Manchester, M26 2JS

Further details contact: Stephen Lancaster, Tel: 0161 509 5095. Alternative contact: Kelly Twist

*Kevin Tallett,* Director 30 August 2023 Ag FJ53010

(4430189)

#### **DOT & CROSS DESIGNS LTD**

(Company Number 11826733)

Registered office: Sterling House, 71 Francis Road, Edgbaston, B16 8SP

Principal trading address: Unit 4, Reliance Trading Estate, Chestom Road, Bilston, West Midlands, WV14 0RD

This notice is given pursuant to Rule 15.13 of the Insolvency (England and Wales) Rules 2016.

Notice is hereby given that a virtual meeting of the creditors of the above named Company is being convened and will be held 7 September 2023 at 10:30am. Simran Sumbal, the director of the Company is

the convener of the meeting ("the Convener").

The primary purpose of the virtual meeting is for creditors to consider a decision on the nomination of liquidators, and if the creditors think fit, to appoint a liquidation committee.

(4430214)

(4431014)

In addition, in the event that a committee is not formed, creditors will also be asked to consider a decision in relation to the payment of the costs of assistance with preparation of the Company's statement of affairs and seeking the creditors' decision on the nomination of liquidators, as an expense of the liquidation.

Creditors participating in the virtual meeting or attending by proxy, (unless a committee is formed), will also be asked to consider approving the basis of the joint liquidators remuneration and whether they also approve the joint liquidators drawing their disbursements, details of which have been provided in a report to creditors.

In order to be entitled to vote at the meeting (either in person or by proxy), creditors must have submitted a proof in respect of their claim (unless a proof has already been submitted) to the Convener via the contact details below, by no later than 4pm on 6 September 2023.

Any creditor entitled to attend and vote at the meeting is entitled to do so either in person or by proxy. If you cannot attend and wish to be represented at the meeting, a completed proxy form must be delivered to the

Convener via the contact details below, prior to the meeting.

Log-in details for the virtual meeting will be supplied to creditors on request.

For further details contact Tommy White on 0208 370 7250 or at Tommy.White@btguk.com

Dated: 25 August 2023

Simran Sumbal, Convener

(4432011)

### **G S MAYFAIR LTD**

(Company Number 09460323)

Trading Name: Olive Kitchen

Registered office: 264 High Street, Beckenham, Kent, BR3 1DZ

Principal trading address: 67 High Street, Epping, CM16 4BA

NOTICE IS HEREBY GIVEN pursuant to Section 100 of the Insolvency Act 1986 and R6.14 and R15.8 of the Insolvency Rules (England & Wales) 2016 ("the Rules") that a Virtual Meeting of the Creditors of the above named Company is being proposed in accordance with resolutions passed by the Board of Directors. The virtual meeting will be held as follows:

Date: 29 August 2023

Time: 12:15

To access the virtual meeting: Dial-in number UK: 0843 373 0843, Dial-in number UK mobiles abroad: 0843 373 0999 & Enter PIN code: 60964673

(Calls cost 5.8p/min plus VAT and network charge from a UK landline) A meeting of shareholders has been called and will be held prior to the virtual meeting of creditors to consider passing a resolution for voluntary winding up of the Company. A list of the names and addresses of the Company's creditors will be available for inspection free of charge at the offices of Edge Recovery Limited, 27 Church Street, Rickmansworth, Hertfordshire, WD3 1DE between 10.00am and 4.00pm on the two business days preceding the date of the creditors meeting.

Any creditor entitled to attend and vote at this virtual meeting may do so either in person or by proxy. Creditors wishing to vote at the virtual meeting must (unless they are individual creditors attending in person) lodge their proxy with the convener before they may be used at the meeting. Unless there are exceptional circumstances, a creditor will not be entitled to vote unless their written statement of claim, (proof), which clearly sets out the name and address of the creditor and the amount claimed, has been lodged and admitted for voting purposes. Proofs must be lodged by 4pm the business day before the meeting.

Unless they surrender their security, secured creditors must give particulars of their security, the date when it was given and the estimated value at which it is assessed if they wish to vote at the meeting.

The resolutions to be taken at the creditors' meeting may include the appointment by creditors of a liquidator, a resolution specifying the terms on which the Liquidator is to be remunerated, and the meeting may receive information about, or be called upon to approve, the costs of preparing the statement of affairs and convening the meeting. Creditors should note that if no non-associate votes be received in relation to fee approval then votes of those asses

Nominated Liquidator: *Robert Cundy* (IP number 9495) of Edge Recovery Limited, 5/7 Ravensbourne Road, Bromley, Kent, BR1 1HN. For further details contact Emma Holmes on 020 8315 7430 or at emma.holmes@edgerecovery.com Dated, 16 August 2023 Kyriakos Pourikou, Director and Convener

#### HATCH LONDON LTD

(Company Number 11874754)

Registered office: 15a Hall Gate, Doncaster, DN1 3NA

Principal trading address: Lower Ground Flat, 5 Chelsea Embankment, London, SW3 4LF

Notice is hereby given, pursuant to Rule 15.13 of the INSOLVENCY (ENGLAND AND WALES) RULES 2016 that the Directors of the above-named Company (the 'conveners') are seeking a decision from creditors on the nomination of a Liquidator by way of a virtual meeting. A resolution to wind up the Company is to be considered on 8 September 2023.

The meeting will be held as a virtual meeting by video conference on 8 September 2023 at 1.15 pm. Details of how to access the virtual meeting are included in the notice delivered to creditors. If any creditor has not received this notice or requires further information please contact the nominated Liquidator using the details below.

Christopher Brooksbank of CB Business Recovery is a person qualified to act as an insolvency practitioner in relation to the company who, during the period before the meeting date, will furnish creditors free of charge with such information concerning the Company's affairs as they may reasonably require.

A creditor may appoint a person as a proxy-holder to act as their representative and to speak, vote, abstain or propose resolutions at the meeting. A proxy for a specific meeting must be delivered to the chair before the meeting. Proxies may be delivered to CB Business Recovery, Hillside, PO Box 205, Liversedge, West Yorkshire, WF17 6WL.

In order to be counted a creditor's vote must be accompanied by a proof in respect of the creditor's claim (unless it has already been given). A vote will be disregarded if a creditor's proof in respect of their claim is not received by 4pm on 7 September 2023 (unless the chair of the meeting is content to accept the proof later). Proofs may be delivered to CB Business Recovery, Hillside, PO Box 205, Liversedge, West Yorkshire, WF17 6WL.

The Directors of the Company, before the meeting date and before the end of the period of seven days beginning with the day after the day on which the company passed a resolution for winding up, are required by Section 99 of the Insolvency Act 1986: (i) to make out a statement in the prescribed form as to the affairs of the company, and (ii) send the statement to the Company's creditors.

Name and address of nominated Liquidator: *Christopher Brooksbank* (IP No. 9658) of CB Business Recovery, Hillside, PO Box 205, Liversedge, West Yorkshire, WF17 6WL

Contact details for nominated Liquidator: Email: chris@cb-br.co.uk *Ryan Murray,* Director

25 August 2023 Ag FJ52936

(4430217)

### HERITAGE WATCHES LIMITED

(Company Number 09715126)

Registered office: PO Box 2690, c/o MRP Accountants, Buckhurst Hill, Essex IG8 1TY

Principal trading address: (former) 3rd Floor, 86-90 Paul Street, Hoxton, London EC2A 4NE

Notice is hereby given under Section 100 of the INSOLVENCY ACT 1986 and Rule 6.14 and 15.8 of the INSOLVENCY (ENGLAND & WALES) RULES 2016 that a virtual meeting of the creditors of the above named Company has been convened by Anthony Mark Torrice, the director of the Company in accordance with resolutions passed by the Board of Directors.

The virtual meeting will be held on 13 September 2023 at 10.15 am. To access the virtual meeting, which will be held via GoTo Meetings conferencing platform, contact DKF Insolvency Limited on behalf of the Convener - details below. This virtual meeting may be recorded (audio) in order to establish and maintain records of the existence of relevant facts or decisions that are taken at the meeting. By attending this meeting, you consent to being recorded. Where any recording of the meeting also entails the processing of personal data, such personal data shall be treated in accordance with the Data Protection Act 2018.

A meeting of shareholders has been called and will be held prior to the virtual meeting of creditors to consider passing a resolution for voluntary winding up of the Company.

Any creditor entitled to attend and vote at this virtual meeting is entitled to do so either in person or by proxy. Creditors wishing to vote at the virtual meeting must (unless they are individual creditors attending in person) lodge their proxy with the convener before they may be used at the meeting.

Unless there are exceptional circumstances, a creditor will not be entitled to vote unless his written statement of claim ('proof'), which clearly sets out the name and address of the creditor and the amount claimed, has been lodged and admitted for voting purposes. Proofs must be lodged by 4pm the business day before the meeting.

Unless they surrender their security, secured creditors must give particulars of their security, the date when it was given and the estimated value at which it is assessed if they wish to vote at the meeting.

The resolutions to be taken at the creditors' meeting may include the appointment by creditors of a Liquidator, a resolution specifying the terms on which the Liquidator is to be remunerated, and the meeting may receive information about, or be called upon to approve, the costs of preparing the statement of affairs and convening the procedure to seek a decision from creditors on the nomination of a liquidator.

Darren Wilson (IP No. 9518) is qualified to act as Insolvency Practitioner in relation to the above Company and a list of names and addresses of the Company's creditors will be available for inspection at the offices of DKF Insolvency Limited, 6 Royal Mews, Southend on Sea, Essex SS1 1DB on the two business days preceding the meeting.

In case of queries, please contact this office on 01702 333105or emaildarren@dkfinsolvency.com or katie@dkfinsolvency.com

Anthony Mark Torrice, Director

23 August 2023

Ag FJ52887

(4430184)

### **INSTALLARE LTD**

(Company Number 13396746)

Trading Name: Installare

Registered office: 15 Hearle Way, Hatfield, AL10 9EW

Principal trading address: 16 Northumberland Avenue, Margate, CT9 3BP

NOTICE IS HEREBY GIVEN pursuant to Rule 15.13 of the Insolvency (England & Wales) Rules 2016 and under Section 100 of the Insolvency Act 1986, that the Director(s) of the above named Company, is/are seeking a decision from creditors, on the nomination of Joint Liquidators, by way of a virtual meeting. A resolution to wind up the Company is to be considered on 08 September 2023.

The virtual meeting will be held by telephone on 08 September 2023, at 10.15am. Details of how to access the virtual meeting are included in the notice delivered to creditors. If any creditor has not received this notice, or requires further information, please contact Bridgewood Financial Solutions Limited using the details below.

Creditors entitled to attend and vote at the meeting, may do so either in person or by proxy. A creditor can attend the virtual meeting in person and vote and is entitled to vote, if they have delivered proof of their debt, by no later than 4.00 p.m on the business day before the meeting. If a creditor cannot attend in person, or does not wish to attend but still wishes to vote at the meeting, they can either nominate a person to attend on their behalf or they may nominate the Chair of the meeting, who will be a Director of the Company, to vote on their behalf.

Creditors must deliver their proxy by no later than the commencement of the meeting. Creditors must deliver all proofs of their debt and proxies to Bridgewood Financial Solutions Limited, Cumberland House, 35 Park Row, Nottingham NG1 6EE, or can be emailed to creditors@bridgewood.co.uk Creditors failing to lodge a proof of their debt or proxy as indicated, will lead to their vote(s) being disregarded. Unless they surrender their security, secured creditors must give particulars of their security, the date when it was given and the estimated value at which it is assessed, if they wish to vote at the meetina.

At the virtual meeting, creditors may receive information about, or be called upon to approve, the costs of preparing the statement of affairs and convening the virtual meeting of creditors and may be requested to consider a resolution specifying the terms on which the Joint Liquidators are to be remunerated.

Thomas Grummitt, IP Number 25976 and Andrew Smith, IP Number 19210, both of Bridgewood Financial Solutions Limited are qualified to act as the Insolvency Practitioners in relation to the Company and will, during the period before the day of the meeting, furnish creditors free of charge with such information concerning the Company's affairs as they may reasonably require. Creditors can contact Charlotte Stuckes on Tel: 0115 871 2934 or email charlotte.stuckes@bridgewood.co.uk. Paul Darby, Director

(4431835)

#### JASON JONES MOTORS LIMITED

(Company Number 11383937)

Registered office: Suite 4 Aus-Bore House, 19-25 Manchester Road, Wilmslow, SK9 1BQ

Principal trading address: Portway House, Main Street, Nailsea, Bristol, BS48 1AD

NOTICE IS HEREBY GIVEN that the director of the Company is convening a virtual meeting of creditors to be held on 7 September 2023 at 10.15, for the purpose of deciding on the nomination of a liquidator.

Creditors can access the virtual meeting as follows: Contact NTF Corporate Solutions Limited on 01625 540744 on the morning of the meeting

In order to be entitled to vote creditors must deliver proxies and proofs to Suite 4 Aus-Bore House.19-25 Manchester Road. Wilmslow SK9 1BQ by 16:00 on the business day before the meeting.

The convener of the meeting is Jason Paul Jones

NOTE: the meeting may be suspended or adjourned by the chair of the meeting (and must be adjourned if it is so resolved at the meetina).

Nominated Liquidator: James Kaye (IP number 9373) of NTF Corporate Solutions Limited, Suite 4 Aus-Bore House, 19-25 Manchester Road, Wilmslow, Cheshire, SK9 1BQ.

Nominated Liquidator: Nicholas Morgan (IP number 9078) of NTF Corporate Solutions Limited, Suite 4 Aus-Bore House, 19-25 Manchester Road, Wilmslow, Cheshire, SK9 1BQ.

For further details contact Mark Salmon on 01625 540744 or at info@ntffs.co.uk (4431582)

#### LTCG LTD

(Company Number 11082153)

Registered office: 7 Orbison Court, Crownhill, Milton Keynes, MK8 0HF

Principal trading address: 7 Orbison Court, Crownhill, Milton Keynes, MK8 0HF

Notice is hereby given under Section 100 of the INSOLVENCY ACT 1986 and Rules 6.14 and 15.8 of the Insolvency (England & Wales) Rules 2016 that a virtual meeting of the creditors of the above named Company is being convened by Leonard Tyson, the director of the Company in accordance with resolutions passed by the Board of Directors. The virtual meeting will be held on 14 September 2023 at 3.00 pm.

To access the virtual meeting, which will be held via a telephone conference, contact CFS Restructuring LLP, on behalf of the convener details below.

A meeting of shareholders has been called and will be held prior to the virtual meeting of creditors to consider passing a resolution for voluntary winding up of the Company.

Any creditor entitled to attend and vote at this virtual meeting is entitled to do so either in person or by proxy. Creditors wishing to vote at the virtual meeting must (unless they are individual creditors attending in person) lodge their proxy with the convener before they may be used at the meeting.

Unless there are exceptional circumstances, a creditor will not be entitled to vote unless his written statement of claim, ('proof'), which clearly sets out the name and address of the creditor and the amount claimed, has been lodged and admitted for voting purposes. Proofs must be lodged by 4.00 pm the business day before the meeting. Unless they surrender their security, secured creditors must give particulars of their security, the date when it was given and the estimated value at which it is assessed if they wish to vote at the meetina.

The resolutions to be taken at the creditors meeting may include the appointment by creditors of a Liquidator, a resolution specifying the terms on which the Liquidator is to be remunerated, and the meeting may receive information about, or be called up to approve, the costs of preparing the statement of affairs and convening the procedure to seek a decision from creditors on the nomination of a Liquidator.

Joseph G M Sadler and Andrew J Cordon (IP Nos. 9048 and 9687) are qualified to act as Insolvency Practitioners in relation to the above Company and a list of names and addresses of the Company's creditors will be available for inspection at CFS Restructuring LLP, 22 Regent Street, Nottingham, NG1 5BQ on the two business days preceding the meeting.

In case of queries, please contact Mark Cooper on 0115 838 7330 or email mark@cfs-llp.com.

Leonard Tyson, Director

24 August 2023

Ag FJ52874

(4430182)

### MAISIES N'ICE CREAM LTD

(Company Number 06918321)

Registered office: 5 London Road, Rainham, Gillingham ME8 7RG

Principal trading address: Hempstead Valley Shopping Centre, Gillingham, Kent ME5 8QU

Notice is hereby given pursuant to Rule 15.13 of the Insolvency (England and Wales) Rules 2016 that a virtual meeting of the creditors of the Company will be held at 10.45 am on 6 September 2023 (the Decision Date). A resolution to wind up the company is to be considered on 6 September 2023. Decisions to be sought are: The creditors nomination of a Liquidator. It is proposed that David Elliott, of Valentine & Co., Galley House, Moon Lane, Barnet, EN5 5YL, be appointed. David Elliott is qualified to act as an insolvency practitioner in relation to the above; Whether a liquidation committee should be established if sufficient creditors are willing to be members of a committee and if so, who the creditors' wish to nominate for membership of the committee; Specifying the terms on which the liquidator is to be remunerated, and the meeting may receive information about, or be called upon to approve, the costs of preparing the statement of affairs and preparing the financial information report for creditors. The dial-in telephone number for the virtual meeting is +44 (0)330 015 4192. Details of how to access the virtual meeting are included in the notice delivered to creditors. If any creditor has not received this notice or requires further information please contact the nominated Liquidator using the details below.

A creditor may appoint a person as a proxy-holder to act as their representative and to speak, vote, abstain or propose resolutions at the meeting. A proxy for a specific meeting must be delivered to the chair before the meeting. In order to be counted a creditor's vote must be accompanied by a proof in respect of the creditor's claim (unless it has already been given). A vote will be disregarded if a creditor's proof in respect of their claim is not received by 4.00 pm on the business day before the meeting date. Proofs and proxies may be delivered to Valentine & Co, Galley House, Moon Lane, Barnet, EN5 5YL.

The Director(s) of the Company, before the meeting date and before the end of the period of seven days beginning with the day after the day on which the Company passed a resolution for winding up, are required by Section 99 of the Insolvency Act 1986: (i) to make out a statement in the prescribed form as to the affairs of the Company, and (ii) send the statement to the Company's creditors.

A list of the names and addresses of the Company's creditors may be inspected free of charge, at the offices of Valentine & Co, Galley House, Moon Lane, Barnet, EN5 5YL between 10.00 am and 4.00 pm on the two business days preceding the date of the meeting.

Proposed Liquidator and Address: David Elliott (IP No. 8595) of Valentine & Co, Galley House, Moon Lane, Barnet, EN5 5YL.

For further details contact: Chuks Chinaka, Tel: 020 8343 3710.

*Tracy Heyburn,* Director 25 August 2023

Ag FJ53019

(4430204)

### **MOTORGROUP 1 LIMITED**

(Company Number 10833373) Trading Name: MTG Motors Registered office: 1 Bicester Road, Kidlington, OX5 2LA Principal trading address: N/A Notice is hereby given that a virtual meeting of the creditors of the above-named Company is being convened by Daniela Asparuhova, to be held on 6 September 2023 at 11:30 am for the purpose provided for in section 100 of the Insolvency Act 1986 and to consider a resolution specifying the terms on which the appointed Liquidator is to be remunerated.

Creditors entitled to attend and vote at the virtual meeting may do so personally or by proxy. A creditor can attend the virtual meeting in person and vote, and is entitled to vote if they have delivered proof of their debt by no later than 4 p.m. on the business day before the meeting. If a creditor cannot attend in person, or does not wish to attend but still wishes to vote at the meeting, they can either nominate a person to attend on their behalf, or they may nominate the Chair of the meeting, who will be a director of the Company, to vote on their behalf. Creditors must deliver their proxy by no later than the commencement of the meeting. Creditors must deliver all proofs of their debt and proxies to Kallis & Company, Mountview Court, 1148 High Road, Whetstone, London, N20 ORA. Creditors failing to lodge a proof of their debt or proxy as indicated will lead to their vote(s) being disregarded.

Unless they surrender their security, secured creditors must give particulars of their security, the date when it was given and the estimated value at which it is assessed if they wish to vote at the meeting.

At the meeting, creditors may receive information about, or be called upon to approve, the costs of preparing the statement of affairs and convening the meeting of creditors.

A list of names and addresses of the Company s creditors will be available for inspection free of charge at Kallis & Company, Mountview Court, 1148 High Road, Whetstone, London, N20 0RA between 10 a.m. and 4 p.m. on the two business days prior to the decision date.

Creditors can contact Kallis & Company on 020 8446 6699 or by email at Yusuf@kallis.co.uk

Daniela Asparuhova, Director

23 August 2023 Ag FJ52939

(4430198)

#### NESS STORES LIMITED

(Company Number 08842673)

Registered office: 5 Beauvais Square, Shortstown, Bedford, MK42 OGG

Principal trading address: 5 Beauvais Square, Shortstown, Bedford, MK42 OGG

NOTICE IS HEREBY GIVEN pursuant to Section 100 of the Insolvency Act 1986 and R6.14 and R15.8 of the Insolvency Rules (England & Wales) 2016 ("the Rulesn) that a Virtual Meeting of the Creditors of the above named Company is being proposed In accordance with resolutions passed by the Board of Directors. The virtual meeting will be held as follows:

Date: 12 September 2023

Time: 11:15

To access the virtual meeting: Dial-in number UK: 0843 373 0843, Dial-in number UK mobiles abroad: +44843 373 0999 & Enter PIN code: 60964673

(Calls cost S.8plmin plus VAT and network charge from a UK laodlinc) A meeting of shareholders has been called and will be held prior to the virtual meeting of creditors to consider passing a resolution for voluntary winding up of the Company. A list of the names and addresses of the Company's creditors will be available for inspection free of charge at the offices of Edge Recovery Limited, 27 Church Street, Rickmansworth, Hertfordshire, WD3 1DE between 10.00am and 4.00pm on the two business days preceding the date of the creditors meeting.

Any creditor entitled to attend and vote at this virtual meeting may do so either in person or by proxy. Creditors wishing to vote at the virtual meeting must (unless they are individual creditors attending in person) lodge their proxy with the convener before they may be used at the meeting. Unless there are exceptional circumstances, a creditor will not be entitled to vote unless their written statement of claim, (proof), which clear1y sets out the name and address of the creditor and the amount claimed, has been lodged and admitted for voting purposes. Proofs must be lodged by 4pm the business day before the meeting. Unless they surrender their security, secured creditors must give particulars of their security, the date when it was given and the estimated value at which it is assessed if they wish to vote at the meeting.

The resolutions to be taken at the creditors' meeting may include the appointment by creditors of a liquidator, a resolution specifying the terms on which the Liquidator Is to be remunerated, and the meeting may receive Information about, or be called upon to approve, the costs of preparing the statement of affairs and convening the meeting. Creditors should note that if no non-associate votes be received in relation to fee approval then votes of those assessed by the insolvency practitioner to be an associate will be taken into account.

Nominated Liquidator: *Bijal Shah* (IP number 8717) of Edge Recovery, 5/7 Ravensboume Road, Bromley, Kent. BR1 1HN.

For further details contact Emma Holmes on emma.holmes@edgerecovery.com or at 020 8315 7430

Dated: 29 August 2023

Robert Dalgamo, Director and Convener

(4431321)

# ODYSSEY BREW CO LTD

(Company Number 09974491) Registered office: Artisans` House, 7 Queensbridge, Northampton

NN4 7BF

Principal trading address: The Brewery, National Trust Brockhampton Estate, Whitbourne Estate, WR6 5SH

Notice is hereby given by Mitchell Evans, director of the company and convener of this decision, that a virtual meeting of the creditors of the above-named Company will be held on 07 September 2023 at 4.15 p.m. for the purpose provided for in section 100 of the Insolvency Act 1986.

Creditors entitled to attend and vote at the meeting may do so either in person or by proxy. A creditor can attend the meeting in person and vote, and is entitled to vote if they have delivered proof of their debt by no later than 4 p.m. on the business day before the meeting. If a creditor cannot attend in person, or do not wish to attend but still wish to vote at the meeting, they can either nominate a person to attend on their behalf, or they may nominate the Chair of the meeting, who will be a director of the Company, to vote on their behalf. Creditors must deliver their proxy, together with proof of their debt, by no later than the commencement of the meeting. Creditors must deliver all proofs of their debt and proxies to PBC Business Recovery & Insolvency Ltd. 9-10 Scirocco Close. Moulton Park, Northampton, NN3 6AP or vote@pbcbusinessrecovery.co.uk.

Unless they surrender their security, secured creditors must give particulars of their security, the date when it was given and the estimated value at which it is assessed if they wish to vote at the meeting.

At the meeting, creditors may receive information about, or be called upon to approve, the costs of preparing the statement of affairs and convening the meeting of creditors, and may be requested to consider a resolution specifying the terms on which the Liquidator Is to be remunerated.

Gary S Pettit of PBC Business Recovery & Insolvency Ltd (IP number 9066), is qualified to act as insolvency practitioner in relation to the company and will, during the period before the decision date, furnish creditors free of charge with such information concerning the company's affairs as they may reasonably require. Creditors can contact PBC Business Recovery & Insolvency Ltd by email at natashapink@pbcbusinessrecovery.co.uk

Mitchell Evans, Director

(4431329)

#### OOOHH LTD.

(Company Number 09313937)

Registered office: 54 Whitehall Lane, Buckhurst Hill, IG9 5JG

Principal trading address: 54 Whitehall Lane, Buckhurst Hill, IG9 5JG Notice is hereby given pursuant to Rule 15.13 OF THE INSOLVENCY (ENGLAND AND WALES) RULES 2016 that a virtual meeting of the creditors of the above-named Company will be held at 10.00 am on 5 September 2023 (the Decision Date). A resolution to wind up the Company is to be considered on 5 September 2023. Decisions to be sought are: The Creditors nomination of a Liquidator. It is proposed that Avner Radomsky and Michael Goldstein of RG Insolvency, Devonshire House, Manor Way, Borehamwood, Hertfordshire, WD6 1QQ be appointed. Avner Radomsky and Michael Goldstein are qualified to act as insolvency practitioners in relation to the above.

Whether a liquidation committee should be established if sufficient creditors are willing to be members of a committee and if so, who the creditors' wish to nominate for membership of the committee. If a Liquidation Committee is not established: That the unpaid pre-appointment costs in respect of the preparation of the Statement of Affairs and for assistance given in seeking a decision from creditors on the nomination of a liquidator in the sum of £1,800, exclusive of VAT and disbursements, of RG Insolvency be approved for payment.

The dial in telephone number for the virtual meeting is +44 (0)330 998 1253. Details of how to access the virtual meeting are included in the notice delivered to creditors. If any creditor has not received this notice or requires further information please contact the nominated Liquidators using the details below.

A creditor may appoint a person as a proxy-holder to act as their representative and to speak, vote, abstain or propose resolutions at the meeting. A proxy for a specific meeting must be delivered to the chair before the meeting. In order to be counted a creditor's vote must be accompanied by a proof in respect of the creditor's claim (unless it has already been given). A vote will be disregarded if a creditor's proof in respect of their claim is not received by 4 pm on the business day before the meeting date. Proofs and proxies may be delivered to RG Insolvency, Devonshire House, Manor Way, Borehamwood, Hertfordshire, WD6 1QQ.

The Directors of the Company, before the meeting date and before the end of the period of seven days beginning with the day after the day on which the Company passed a resolution for winding up, are required by Section 99 of the Insolvency Act 1986: (i) to make out a statement in the prescribed form as to the affairs of the Company, and (ii) send the statement to the Company's creditors.

A list of the names and addresses of the Company's creditors may be inspected free of charge, at the offices of RG Insolvency, Devonshire House, Manor Way, Borehamwood, Hertfordshire, WD6 1QQ between 10.00 am and 4.00 pm on the two business days preceding the date of the Meeting. Please call to arrange a time.

Name and address of proposed Liquidators: *Avner Radomsky* (IP No. 12290) and *Michael Goldstein* (IP No. 12532) both of RG Insolvency, Devonshire House, Manor Way, Borehamwood, Hertfordshire, WD6 1QQ

Further details contact: The proposed Liquidators, Tel: 020 3603 7871. Alternative contact: Michala Curran.

Zoe Smith, Director/Convener

22 August 2023 Ag FJ52897

(4430219)

#### RED CLOUD INFRASTRUCTURE SERVICES LIMITED

(Company Number 09749323)

Registered office: 10 Harlyn Avenue, Bramhall, Stockport, Cheshire SK7 2HN

#### Principal trading address: N/A

Notice is hereby given, pursuant to RULE 15.13 of the Insolvency (England and Wales) Rules 2016 , that the Director of the abovenamed Company (the 'convener') is seeking a decision from creditors on the nomination of a Liquidator by way of a virtual meeting. A resolution to wind up the Company is to be considered on 5 September 2023.

The meeting will be held as a virtual meeting by telephone conference on 5 September 2023 at 2.15 pm. Details of how to access the virtual meeting are included in the notice delivered to creditors. If any creditor has not received this notice or requires further information please contact the nominated liquidator using the details below.

Natalie Hughes of Simply Corporate Limited, The Town Hall, Burnley Road, Padiham, BB12 8BS will provide reasonable information to creditors during business hours on the two business days prior to the meeting date upon receipt of requests for such information.

A creditor may appoint a person as a proxy-holder to act as their representative and to speak, vote, abstain or propose resolutions at the meeting. A proxy for a specific meeting must be delivered to the chair before the meeting. Proxies may be delivered to Simply Corporate Limited, The Town Hall, Burnley Road, Padiham, BB12 8BS.

In order to be counted a creditor's vote must be accompanied by a proof in respect of the creditor's claim (unless it has already been given). A vote will be disregarded if a creditor's proof in respect of their claim is not received by 4pm on the business day before the meeting date (unless the chair of the meeting is content to accept the proof later). Proofs may be delivered to Simply Corporate Limited, The Town Hall, Burnley Road, Padiham, BB12 8BS or emailed to: info@simplycorporate.co.uk

Name and address of nominated Liquidator: Natalie Hughes (IP No. 14336) of Simply Corporate Limited, The Town Hall, Burnley Road, Padiham, BB12 8BS

Further details contact: Natalie Hughes, Tel: 01282 222420, Email: info@simplvcorporate.co.uk. Alternative contact: Catherine Unsworth. Stephen Wallace, Director

29 August 2023

Ag FJ52884

(4430187)

#### **S L J MEDIA LIMITED**

(Company Number 05329473)

Trading Name: S L J Media Limited

Registered office: 13c Clifford Court, Cooper Way, Carlisle, CA3 0JG

Principal trading address: 13c Clifford Court, Cooper Way, Carlisle, CA3 0JG

NOTICE IS HEREBY GIVEN pursuant to Rule 15.13 of the Insolvency (England & Wales) Rules 2016 and under Section 100 of the Insolvency Act 1986, that the Director(s) of the above named Company, is/are seeking a decision from creditors, on the nomination of Joint Liquidators, by way of a virtual meeting. A resolution to wind up the Company is to be considered on 7 September 2023.

The virtual meeting will be held by telephone on 7 September 2023, at 10.15am. Details of how to access the virtual meeting are included in the notice delivered to creditors. If any creditor has not received this notice, or requires further information, please contact Bridgewood Financial Solutions Limited using the details below.

Creditors entitled to attend and vote at the meeting, may do so either in person or by proxy. A creditor can attend the virtual meeting in person and vote and is entitled to vote, if they have delivered proof of their debt, by no later than 4.00 p.m on the business day before the meeting. If a creditor cannot attend in person, or does not wish to attend but still wishes to vote at the meeting, they can either nominate a person to attend on their behalf or they may nominate the Chair of the meeting, who will be a Director of the Company, to vote on their behalf.

Creditors must deliver their proxy by no later than the commencement of the meeting. Creditors must deliver all proofs of their debt and proxies to Bridgewood Financial Solutions Limited, Cumberland House, 35 Park Row, Nottingham NG1 6EE, or can be emailed to creditors@bridgewood.co.uk Creditors failing to lodge a proof of their debt or proxy as indicated, will lead to their vote(s) being disregarded. Unless they surrender their security, secured creditors must give particulars of their security, the date when it was given and the estimated value at which it is assessed, if they wish to vote at the meeting.

At the virtual meeting, creditors may receive information about, or be called upon to approve, the costs of preparing the statement of affairs and convening the virtual meeting of creditors and may be requested to consider a resolution specifying the terms on which the Joint Liquidators are to be remunerated.

Thomas Grummitt, IP Number 25976 and Andrew Smith, IP Number 19210, both of Bridgewood Financial Solutions Limited are qualified to act as the Insolvency Practitioners in relation to the Company and will, during the period before the day of the meeting, furnish creditors free of charge with such information concerning the Company's affairs as they may reasonably require. Creditors can contact Daniel Harrison on Tel: 0115 871 2940 or email daniel.harrison@bridgewood.co.uk. Stuart Leslie Jefferies, Director (4431394)

### SNAP HR LIMITED

(Company Number 06797548)

("THE COMPANY")

Registered office: SUITE 23, 5TH FLOOR, 63 66 HATTON GARDEN, LONDON, EC1N 8LE; CURRENT REGISTERED OFFICE OF THE COMPANY: SUITE 23, 5TH FLOOR, 63 66 HATTON GARDEN, LONDON, EC1N 8LE FORMER REGISTERED OFFICE OF THE COMPANY: 1018 VESTRY STREET, LONDON, N1 7RE

Principal trading address: 19A QUEENS PARADE, HANGER LANE, LONDON, W5 3HU

Notice is hereby given, pursuant to Rule 15.13 of the Insolvency (England and Wales) Rules 2016, that the Director of the above named Company (the 'convener') is seeking a decision from creditors on the nomination of a Liquidator by way of a virtual meeting. A resolution to wind up the Company is to be considered on 05/10/2023.

Decisions regarding the costs of preparing the statement of affairs, convening the meeting of creditors, Liquidator remuneration and the formation of a liquidation committee may also be sought at the meeting. The meeting will be held as a virtual meeting on 05/10/2023 at 11am. Details of how to access the virtual meeting are included in the notice delivered to creditors. If any creditor has not received this notice or requires further information, please contact the nominated Liquidator using the contact particulars referred to below.

Elliot Green of Oliver Elliot, Moda Business Centre, Stirling Way, Borehamwood, WD6 2BW is a person qualified to act as insolvency practitioners in relation to the Company who, during the period before the meeting date, will furnish creditors free of charge with such information concerning the Company's affairs as they may reasonably require.

A creditor may appoint a person as a proxyholder to act as their representative and to speak, vote, abstain or propose resolutions at the meeting. A proxy for a specific meeting must be delivered to the chair before the meeting. Proxies may be delivered to Moda Business Centre, Stirling Way, Borehamwood, WD6 2BW. In order to be counted a creditor's vote must be accompanied by a proof in respect of the creditor's claim (unless it has already been given). A vote will be disregarded if a creditor's proof in respect of their claim is not received by 23.59 pm on the business day before the meeting (unless the chair of the meeting is content to accept the proof later).

Proofs may be delivered to Moda Business Centre, Stirling Way, Borehamwood, WD6 2BW.

Names and address of nominated Liquidator: Elliot Green (IP No. 9260) of Oliver Elliot, Moda Business Centre, Stirling Way, Borehamwood, WD6 2BW

CONTACT PARTICULARS CORRESPONDENCE ADDRESS: OLIVER ELLIOT. MODA BUSINESS CENTRE, STIRLING WAY. BORFHAMWOOD. WD6 2BW FMAII ADDRESS: CONTACT@OLIVERELLIOT.CO.UK TELEPHONE NUMBER: 020 3925 3613 PERSON TO CONTACT IN FIRST INSTANCE: ELLIOT GREEN Stuart Gowan (Convener) DATE: 22 August 2023

(4431880)

#### STARDUST ENTERTAINMENT (UK) LIMITED

(Company Number 10818115)

Trading Name: Black Diamond Gentlemen's Club

Registered office: 36 Alie Street, London, E1 8DA

This notice is given pursuant to Rule 15.13 of the Insolvency (England and Wales) Rules 2016

Notice is hereby given that a virtual meeting of the creditors of the above-named Company is being convened and will be held on 7 September 2023 at 12:00pm. Jianping He, the sole director of the Company is the convener of the meeting ("the Convener").

The primary purpose of the virtual meeting is for creditors to consider a decision on the nomination of Liquidators, and if the creditors think fit, to appoint a liquidation committee.

In addition, in the event that a committee is not formed, creditors will also be asked to consider a decision in relation to the payment of the costs of assistance with preparation of the Company's statement of affairs and seeking the creditors' decision on the nomination of Liquidators, as an expense of the liquidation.

Creditors participating in the virtual meeting or attending by proxy, (unless a committee is formed), will also be asked to consider approving the basis of the Joint Liquidators remuneration and whether they also approve the Joint Liquidators drawing their disbursements, details of which have been provided in a report to creditors.

In order to be entitled to vote at the meeting (either in person or by proxy), creditors must have submitted a proof in respect of their claim (unless a proof has already been submitted) to the Convener via the contact

details below, by no later than 4pm on 6 September 2023.

Any creditor entitled to attend and vote at the meeting is entitled to do so either in person or by proxy. If you cannot attend and wish to be represented at the meeting, a completed proxy form must be delivered to the Convener via the contact details below, prior to the meetina.

Log-in details for the virtual meeting will be supplied to creditors on request.

For further details contact Danielle Craven on 020 8370 7250 or at danielle.craven@btguk.com

Dated: 25 August 2023 Jianping He, Convener

(4431998)

#### THE BOOTLEGGER'S CLUB LTD

(Company Number 09639193)

Registered office: 25-26 Lime Street, London, EC3M 7HR

Principal trading address: 25-26 Lime Street, London, EC3M 7HR NOTICE IS HEREBY GIVEN pursuant to Section 100 of the Insolvency

Act 1986 and R6.14 and R15.8 of the Insolvency Rules (England & Wales) 2016 ("the Rules") that a Virtual Meeting of the Creditors of the above named Company is being proposed in accordance with resolutions passed by the Board of Directors. The virtual meeting will be held as follows:

Date: 11 September 2023

Time: 11:30

To access the virtual meeting: Dial-in number UK: 0843 373 0843, Dial-in number UK mobiles abroad: 0843 373 0999 & Enter PIN code: 60964673

(Calls cost 5.8p/min plus VAT and network charge from a UK landline) A meeting of shareholders has been called and will be held prior to the virtual meeting of creditors to consider passing a resolution for voluntary winding up of the Company. A list of the names and addresses of the Company's creditors will be available for inspection free of charge at the offices of Edge Recovery Limited, 27 Church Street, Rickmansworth, Hertfordshire, WD3 1DE between 10.00am and 4.00pm on the two business days preceding the date of the creditors meeting.

Any creditor entitled to attend and vote at this virtual meeting may do so either in person or by proxy. Creditors wishing to vote at the virtual meeting must (unless they are individual creditors attending in person) lodge their proxy with the convener before they may be used at the meeting. Unless there are exceptional circumstances, a creditor will not be entitled to vote unless their written statement of claim, (proof), which clearly sets out the name and address of the creditor and the amount claimed, has been lodged and admitted for voting purposes. Proofs must be lodged by 4pm the business day before the meeting.

Unless they surrender their security, secured creditors must give particulars of their security, the date when it was given and the estimated value at which it is assessed if they wish to vote at the meeting.

The resolutions to be taken at the creditors' meeting may include the appointment by creditors of a liquidator, a resolution specifying the terms on which the Liquidator is to be remunerated, and the meeting may receive information about, or be called upon to approve, the costs of preparing the statement of affairs and convening the meeting. Creditors should note that if no non-associate votes be received in relation to fee approval then votes of those assessed by the insolvency practitioner to be an associate will be taken into account.

Nominated Liquidator: Bijal Shah (IP number 8717) of Edge Recovery, 27 Church Street, Rickmansworth, Hertfordshire, WD3 1DE.

For further details contact Jack Webb on 020 8315 7430 or at jack.webb@edgerecovery.com

Unni Krishnan Sathar, Director and Convener (4431586)

#### VIECURA MEDICAL UK LIMITED

(Company Number 07711195)

Registered office: Sunnybank, Braithwaite Road, Keighley, BD22 6PA Principal trading address: Sunnybank, Braithwaite Road, Keighley, **BD22 6PA** 

Nature of Business: Manufacture of other textiles not elsewhere classified.

Type of Liquidation: Creditors Voluntary Liquidation.

Date of meeting: 7 September 2023.

NOTICE IS HEREBY GIVEN pursuant to Rule 15.13 of the Insolvency (England and Wales) Rules 2016, that a virtual meeting of the creditors of the above named Company will be held on the date and time specified in this notice for the purposes mentioned in Section 100 of the Insolvency Act 1986. The Insolvency Practitioner named below is qualified to act in this matter. A list of names and addresses of the Company's creditors may be inspected, free of charge, at the address given below, between 10.00am and 4.00pm on the two business days preceding the date of the meeting.

Resolutions to be taken at the meeting may include a resolution specifying the terms on which the liquidator is to be remunerated, and the meeting may receive information about, or be called upon to approve the costs of preparing the statement of affairs and convening the meeting.

Creditors entitled to attend and vote at the virtual meeting may participate either in person or by proxy. A creditor can attend the virtual meeting and vote, and is entitled to vote if they have delivered proof of their debt by no later than 4 pm on the business day before the meeting. If a creditor cannot attend, or does not wish to attend, but still wishes to vote at the virtual meeting, they can ether nominate a person to attend on their behalf, or they may nominate the Chairman of the virtual meeting, who will be a director of the Company, to vote on their behalf. Such creditors must still submit the proof of their claim by no later than 4 pm on the business day before the meeting but proxies will be accepted by the Chairman up to the commencement of the virtual meeting. For the purposes of voting, any secured creditors are required (unless they surrender their security) to lodge a statement with the Insolvency Practitioners prior to the Meeting, giving particulars of their security, the date when it was given and its assessed value. Creditors must deliver proof of their claim and their proxy using the details provided below.

By Order of the Board

Mr Raymond Fielding, Director

Insolvency Practitioner's Name and Address: Rikki Burton (IP No. 14430) of Anderson Brookes Insolvency Practitioners Limited, 6th Floor, 120 Bark Street, Bolton, BL1 2AX.

For further information contact Danny McDowall at the offices of Anderson Brookes Insolvency Practitioners Limited on 0800 1804 933, or danny@andersonbrookes.co.uk. 30 August 2023

(4430990)

### WHITBY APPLETON LTD

(Company Number 07118464)

Registered office: Aston Court, Kingsmead Business Park, Frederick Place, High Wycombe, Buckinghamshire, HP11 1LA

This notice is given pursuant to Rule 15.13 of the Insolvency (England and Wales) Rules 2016.

Notice is hereby given that a virtual meeting of the creditors of the above named Company is being convened and will be held 5 September 2023 at 11:00am. Mr Richard Whitby, the director of the Company is the convener of the meeting ("the Convener").

The primary purpose of the virtual meeting is for creditors to consider a decision on the nomination of liquidators, and if the creditors think fit, to appoint a liquidation committee.

In addition, in the event that a committee is not formed, creditors will also be asked to consider a decision in relation to the payment of the costs of assistance with preparation of the Company's statement of affairs and seeking the creditors' decision on the nomination of liquidators, as an expense of the liquidation.

Creditors participating in the virtual meeting or attending by proxy, (unless a committee is formed), will also be asked to consider approving the basis of the joint liquidators remuneration and whether they also approve the joint liquidators drawing their disbursements, details of which have been provided in a report to creditors.

In order to be entitled to vote at the meeting (either in person or by proxy), creditors must have submitted a proof in respect of their claim (unless a proof has already been submitted) to the Convener via the contact details below, by no later than 4pm on 4 September 2023.

Any creditor entitled to attend and vote at the meeting is entitled to do so either in person or by proxy. If you cannot attend and wish to be represented at the meeting, a completed proxy form must be delivered to the Convener via the contact details below, prior to the meeting.

Log-in details for the virtual meeting will be supplied to creditors on request.

For further details contact Viktorija Sabasova on 0208 370 7250 or at Viktorija.Sabasova@btguk.com

Dated: 25 August 2023

Mr Richard Whitby, Convener

(4431958)

# WHITE MOSS SERVICES LTD

(Company Number 12432069)

Registered office: Tudor House, 250 Menlove Avenue, Liverpool, L18 3JF

This notice is given pursuant to Rule 15.13 of the Insolvency (England and Wales) Rules 2016.

Notice is hereby given that a virtual meeting of the creditors of the above-named Company is being convened and will be held 7 September 2023 at 11am. Shaun Smith, the director of the Company is the convener of the meeting ("the Convener").

The primary purpose of the virtual meeting is for creditors to consider a decision on the nomination of liquidators, and if the creditors think fit, to appoint a liquidation committee.

In addition, in the event that a committee is not formed, creditors will also be asked to consider a decision in relation to the payment of the costs of assistance with preparation of the Company's statement of affairs and seeking the creditors' decision on the nomination of liquidators, as an expense of the liquidation.

Creditors participating in the virtual meeting or attending by proxy, (unless a committee is formed), will also be asked to consider approving the basis of the joint liquidator's remuneration and whether they also approve the joint liquidators drawing their disbursements, details of which have been provided in a report to creditors.

In order to be entitled to vote at the meeting (either in person or by proxy), creditors must have submitted a proof in respect of their claim (unless a proof has already been submitted) to the Convener via the contact details below, by no later than 4pm on 6 September 2023.

Any creditor entitled to attend and vote at the meeting is entitled to do so either in person or by proxy. If you cannot attend and wish to be represented at the meeting, a completed proxy form must be delivered to the Convener via the contact details below, prior to the meeting.

Log-in details for the virtual meeting will be supplied to creditors on request.

For further details contact Viktorija Sabasova on 0208 370 7250 or at Viktorija.Sabasova@btguk.com

Dated: 24 August 2023

Shaun Smith, Convener

(4431965)

# YOUROUTLETCHOICEGB LTD

(Company Number 14191581) Registered office: 20 Belle Vue Avenue, Doncaster, DN4 5DX

Principal trading address: 20 Belle Vue Avenue, Doncaster, DN4 5DX

Notice is hereby given under Section 100 of the INSOLVENCY ACT 1986 and Rule 6.14 and 15.8 of the INSOLVENCY (ENGLAND AND WALES) RULES 2016, that a virtual meeting of the creditors of the above named Company is being proposed by the director of the Company in accordance with resolutions passed by the Board of Directors. The virtual meeting will be held via Zoom, on 12 September 2023 at 11.15 am.

A meeting of shareholders has been called and will be held prior to the virtual meeting of creditors, to consider passing a resolution for voluntary winding up of the Company. Any creditor entitled to attend and vote at this virtual meeting is entitled to do so either in person or by proxy. Creditors wishing to vote at the virtual meeting must (unless they are individual creditors attending in person) lodge their proxy with the convener before they may be used at the meeting. The convener can be contacted care of Simon Weir, DSi Business Recovery, 2 Lakeside, Calder Island Way, Wakefield, WF2 7AW.

Unless there are exceptional circumstances, a creditor will not be entitled to vote unless his written statement of claim, ('proof'), which clearly sets out the name and address of the creditor and the amount claimed, has been lodged and admitted for voting purposes. Proofs must be lodged by 4.00 pm on the business day before the meeting.

Unless they surrender their security, secured creditors must give particulars of their security, the date when it was given and the estimated value at which it is assessed if they wish to vote at the meeting.

The resolutions to be taken at the creditors' meeting may include the appointment by creditors of a Liquidator, a resolution specifying the terms on which the Liquidator is to be remunerated, and the meeting may receive information about, or be called up to approve, the costs of preparing the statement of affairs and convening the meeting.

Simon Weir is qualified to act as an Insolvency Practitioner in relation to the above Company and a list of names and addresses of the Company's creditors will be available for inspection at the offices of DSi Business Recovery, 2 Lakeside, Calder Island Way, Wakefield, WF2 7AW on the two business days preceding the meeting.

In case of queries, please contact Simon Weir on 01924 790880 or email info@dsinsol.com.

James Henry Law, Director

29 August 2023 Ag FJ52944

(4430210)

### **NOTICES TO CREDITORS**

#### ARAMOKO BROADBAND LIMITED

(Company Number 05137469)

Registered office: 5 Barnfield Crescent, Exeter, Devon, EX1 1QT

Principal trading address: N/A Notice is given that under RULE 6.23 of the Insolvency (England and

Wales) Rules 2016 ("the Rules") that the company was placed into Creditors' Voluntary Liquidation (insolvent liquidation) and *David Gerard Kirk* (IP No. 8830) and *Daniel Robert Jeeves* (IP No. 26032) both of Kirks, 5 Barnfield Crescent, Exeter, EX1 1QT were appointed Joint Liquidators by the creditors on 23 August 2023.

Notice is further given that the Creditors are required to prove their debts on or before 29 September 2023 by sending full details of their claims to the Liquidators at Kirks, 5 Barnfield Crescent, Exeter, EX1 1QT. Creditors must also, if so requested by the Liquidators, provide such further details and documentary evidence to support their claims as may appear to the Liquidators to be necessary.

Please note that no further public notice will be made and therefore the Liquidators shall be entitled to make any distribution without regard to any claims not proved in the manner required by statute.

In case of queries, please contact Daniel Jeeves on 01392 474303 or email Nathan@kirks.co.uk.

David Gerard Kirk, Joint Liquidator 25 August 2023 Ag FJ52883

(4430266)

#### EUROPOST EXPRESS (UK) LTD.

(Company Number 09694048)

Registered office: 4 Corringham Road, Wembley, HA9 9QA

Principal trading address: 4 Corringham Road, Wembley, HA9 9QA We, *Eric Walls* (IP No. 9113) and *Wayne Harrison* (IP No. 9703) both of KSA Group Ltd, C12 Marquis Court, Marquisway, Team Valley, Gateshead, NE11 0RU give notice that on 25 August 2023 we were appointed Joint Liquidators of Europost Express (UK) Ltd by resolutions of members and creditors.

Notice is hereby given that the creditors of the above named company, which is being voluntarily wound up, are required, on or before 25 November 2023 to send in their full forenames and surnames, their addresses and descriptions, full particulars of their debts or claims, and the names and addresses of their Solicitors (if any), to the undersigned E Walls and W Harrison of KSA Group Ltd, C12 Marquis Court, Marquis Way, Team Valley, Gateshead, NE11

0RU, the Joint Liquidators of the said Company and, if so required by notice in writing from the said Joint Liquidators, are, personally or by their Solicitors, to come in and prove their debts or claims at such time and place as shall be specified in such notice, or in default thereof they will be excluded from the benefit of any distribution.

further details contact: The Liquidators, E-mail: For insolvency@ksagroup.co.uk, Tel: 0191 482 3343. Alternative contact: Craig Harmon

E Walls, Joint Liquidator 29 August 2023 Ag FJ52876

(4430265)

### WPF THERAPY LTD

(Company Number 01214251)

Registered office: 25 Farringdon Street, London, EC4A 4AB Principal trading address: 23 Magdalen Street, London, SE1 2EN

Notice is hereby given that the creditors of the above-named company, which is being voluntarily wound up, whose claims exceed £1,000 and who have not already proved their debt are requested, on or before 30 September 2023, to send in their names and addresses and to submit their proof of debt to the undersigned at RSM UK Restructuring Advisory LLP, 25 Farringdon Street, London, EC4A 4AB and, if so requested by the Joint Liquidators, to provide such further details or produce such documentary or other evidence as may appear to be necessary.

A creditor with a debt which does not exceed £1,000 (according to the accounting records or the statement of affairs of the above-named company) is not required to prove its debt.

A creditor who has not proved his debt before the declaration of any dividend is not entitled to disturb, by reason that he has not participated in it, the distribution of that dividend or any other dividend before his debt was proved.

Joint Liquidator: Matthew Haw (IP number 9627) of RSM UK Restructuring Advisory LLP, 25 Farringdon Street, London, EC4A 4AB.

Joint Liquidator: Matthew Wild (IP number 9300) of RSM UK Restructuring Advisory LLP, Third Floor, One London Square, Cross Lanes, Guildford, GU1 1UN.

Date of Appointment: 04 August 2023

For further details contact Samantha Hawkins on 0203 201 8000 Nature of company's business: Other human health activities

(4431943)

(4431769)

#### **RESOLUTION FOR WINDING-UP**

#### WYRE OILS LIMITED

(Company Number: 07469229)

trading as WYRE OILS LIMITED

Registered Office: 17 Victoria Road East, Thornton-cleveleys FY5 5HT Principal Trading Address: 17 Victoria Road East, Thornton-cleveleys FY5 5HT

Nature of Business: Fuel supplier

At a Extraordinary Meeting of the Members of the above-named Company, duly convened, and held remotely on Tuesday 08 August 2023, the following Resolution/s was/were duly passed:

1. (Special Resolution) THAT it has been proved to the satisfaction of this meeting that the company cannot, by reason of its liabilities, continue its business and that it is advisable to wind up the same, and accordingly that the Company be wound up voluntarily

2. (Ordinary Resolution) Stuart John Rathmell of Stuart Rathmell Insolvency Egyptian Mill Egyptian Street Bolton BL1 2HS be and is hereby appointed

For further details, please contact: Stuart Rathmell, (10050), STUART RATHMELL INSOLVENCY, Egyptian Mill, Egyptian Street, Bolton BL1 2HS. Telephone: 01204867615, Email address: stuart.rathmell.insolvency@outlook.com. Stuart Macmillan, Chairman

Tuesday 08 August 2023

ROONEY ACADEMY LTD

(Company Number: 12256964) trading as ROONEY ACADEMY LTD Registered Office: 50 Redwood Drive, Chorley PR7 3BW Principal Trading Address: 50 Redwood Drive, Chorley PR7 3BW Nature of Business: Education

At a Extraordinary Meeting of the Members of the above-named Company, duly convened, and held remotely on Friday 25 August 2023, the following Resolution/s was/were duly passed:

1. (Special Resolution) THAT it has been proved to the satisfaction of this meeting that the company cannot, by reason of its liabilities, continue its business and that it is advisable to wind up the same, and accordingly that the Company be wound up voluntarily

2. (Ordinary Resolution) Stuart John Rathmell of Stuart Rathmell Insolvency Egyptian Mill Egyptian Street Bolton BL1 2HS be and is hereby appointed

For further details, please contact: Stuart Rathmell, (10050), STUART RATHMELL INSOLVENCY, Egyptian Mill, Egyptian Street, Bolton BL1 01204867615. Email 2HS. Telephone: address: stuart.rathmell.insolvency@outlook.com. Anthony Lipson, Chairman (4431776)

Friday 25 August 2023

#### PARTAK LIMITED

(Company Number: 11880678)

trading as Partak Limited

Registered Office: 21 Harbord Close, London SE5 8AG

Principal Trading Address: 21 Harbord Close, London SE5 8AG

Nature of Business: Other transportation support activities

At a General Meeting of the Members of the above-named Company, duly convened, and held remotely on Thursday 31 August 2023, the following Resolution/s was/were duly passed:

1. (Special Resolution) That the Company be wound up voluntarily 2. (Ordinary Resolution) That Christopher David Horner, of Robson Scott Associates Ltd t/a BusinessRescueExpert, 47/49 Duke Street, Darlington, DL3 7SD, be appointed as Liquidator of the Company For further details, please contact: Christopher David Horner, (16150), Robson Scott Associates T/a Businessrescueexpert, 49 Duke Street, Darlington, Co. Durham DL3 7SD, Telephone: 01325365950, Email address: admin@businessrescueexpert.co.uk, (Partak Limited Resolutions).

Oliver Esteves Villa, Chairman Thursday 31 August 2023

(4427099)

#### SENTIRO (UK) LIMITED

(Company Number: 06329599) trading as SENTIRO (UK) LIMITED

Registered Office: 9, Birnam House, Twickenham TW1 2JD

Principal Trading Address: 9, Birnam House, Twickenham TW1 2JD

Nature of Business: Other telecommunications activities At a General Meeting of the Members of the above-named Company, duly convened, and held remotely on Wednesday 30 August 2023, the following Resolution/s was/were duly passed:

1. (Special Resolution) That the Company be wound up voluntarily 2. (Ordinary Resolution) That Christopher David Horner, of Robson Scott Associates Ltd t/a BusinessRescueExpert, 47/49 Duke Street, Darlington, DL3 7SD, be appointed as Liquidator of the Company For further details, please contact: Christopher David Horner, (16150), Robson Scott Associates Ltd t/a BusinessRescueExpert, 49 Duke Street, Darlington, Co. Durham DL3 7SD, Telephone: 01325 365 950, Email address: admin@businessrescueexpert.co.uk. Martin Bergin, Chairman

Thursday 31 August 2023

(4430950)

#### DIAMOND KEBAB BRISTOL LTD

(Company Number: 12008331)

trading as Diamond Kebab Bristol Ltd

Registered Office: 28 Park Street, Bristol BS1 5JA

Principal Trading Address: 28 Park Street, Bristol BS1 5JA

Nature of Business: Unlicenced restaurants and cafes

At a General Meeting of the Members of the above-named Company, duly convened, and held at A N G, Castlemead, Lower Castle Street, Bristol BS1 3AG on Tuesday 29 August 2023, the following Resolution/s was/were duly passed:

1. (Special Resolution) That the Company be wound up voluntarily.

2. (Ordinary Resolution) That M. Usman Nazir, be and is hereby appointed Liquidator for the purposes of such winding-up.

For further details, please contact: M. Usman Nazir, (11290), A N G, Castlemead, Lower Castle Street, Bristol BS1 3AG, Telephone: 01173252505, Email address: info@ang.uk.com.

Matin Khan, Chairman

5UT

(Company Number: 11596202)

M.R.S. ENGINEERING SERVICES LIMITED

trading as M.R.S. ENGINEERING SERVICES LIMITED

Nature of Business: STUART RATHMELL INSOLVENCY

2023, the following Resolution/s was/were duly passed:

accordingly that the Company be wound up voluntarily

Registered Office: 1 Stephenson Court, Middlesborough TS6 5UT

Principal Trading Address: 1 Stephenson Court, Middlesborough TS6

At a Extraordinary Meeting of the Members of the above-named

Company, duly convened, and held remotely on Tuesday 22 August

1. (Special Resolution) THAT it has been proved to the satisfaction of

this meeting that the company cannot, by reason of its liabilities, continue its business and that it is advisable to wind up the same, and

2. (Ordinary Resolution) Stuart John Rathmell of Stuart Rathmell

Insolvency Egyptian Mill Egyptian Street Bolton BL1 2HS be and is

For further details, please contact: Stuart Rathmell, (10050), STUART

RATHMELL INSOLVENCY, Egyptian Mill, Egyptian Street, Bolton BL1

01204867615,

(4431803) PDR ELECTRICS LTD

(Company Number: 08809960)

trading as PDR ELECTRICS LTD

Registered Office: 17 Victoria Road East, Thornton-cleveleys FY5 5HT Principal Trading Address: 17 Victoria Road East, Thornton-cleveleys FY5 5HT

Nature of Business: Electrical engineering

At a Extraordinary Meeting of the Members of the above-named Company, duly convened, and held remotely on Friday 18 August 2023, the following Resolution/s was/were duly passed:

1. (Special Resolution) THAT it has been proved to the satisfaction of this meeting that the company cannot, by reason of its liabilities, continue its business and that it is advisable to wind up the same, and accordingly that the Company be wound up voluntarily

2. (Ordinary Resolution) Stuart John Rathmell of Stuart Rathmell Insolvency Egyptian Mill Egyptian Street Bolton BL1 2HS be and is hereby appointed

For further details, please contact: Stuart Rathmell, (10050), STUART RATHMELL INSOLVENCY, Egyptian Mill, Egyptian Street, Bolton BL1 2HS, or STUART RATHMELL INSOLVENCY, Telephone: 01204867615, Email address:

stuart.rathmell.insolvency@outlook.com. Phillip Robinson. Chairman

Friday 18 August 2023

(4431811)

Matthew Stephenson, Chairman Tuesday 22 August 2023

Telephone:

stuart.rathmell.insolvency@outlook.com.

(4432057)

(4432062)

address:

Email

#### ACK LOGISTICS LTD

hereby appointed

2HS.

(Company Number: 09479429)

trading as ACK LOGISTICS LTD

Registered Office: 37 Clorain Road, Liverpool L33 6XY

Principal Trading Address: 37 Clorain Road, Liverpool L33 6XY Nature of Business: TRANSPORT

At a Extraordinary Meeting of the Members of the above-named Company, duly convened, and held remotely on Thursday 31 August 2023, the following Resolution/s was/were duly passed:

1. (Special Resolution) THAT it has been proved to the satisfaction of this meeting that the company cannot, by reason of its liabilities, continue its business and that it is advisable to wind up the same, and accordingly that the Company be wound up voluntarily

2. (Ordinary Resolution) Stuart John Rathmell of Stuart Rathmell Insolvency Egyptian Mill Egyptian Street Bolton BL1 2HS be and is hereby appointed

For further details, please contact: Stuart Rathmell, (10050), STUART RATHMELL INSOLVENCY, Egyptian Mill, Egyptian Street, Bolton BL1 2HS, Telephone: 01204867615, Email address: stuart.rathmell.insolvency@outlook.com.

GERRARD TYRELL, Chairman

Thursday 31 August 2023

L.SMART ENGINEERING LTD

(Company Number: 10839624)

trading as L.SMART ENGINEERING LTD

Registered Office: 51 Pretoria Road, Tonyrefail, Porth CF39 8PW Principal Trading Address: 51 Pretoria Road, Tonyrefail, Porth CF39

Principal Irading Address: 51 Pretoria Road, Ionyrefail, Porth 8 8PW

Nature of Business: STUART RATHMELL INSOLVENCY

At a Extraordinary Meeting of the Members of the above-named Company, duly convened, and held remotely on Thursday 17 August 2023, the following Resolution/s was/were duly passed:

1. (Special Resolution) THAT it has been proved to the satisfaction of this meeting that the company cannot, by reason of its liabilities, continue its business and that it is advisable to wind up the same, and accordingly that the Company be wound up voluntarily

2. (Ordinary Resolution) Stuart John Rathmell of Stuart Rathmell Insolvency Egyptian Mill Egyptian Street Bolton BL1 2HS be and is hereby appointed

For further details, please contact: Stuart Rathmell, (10050), STUART RATHMELL INSOLVENCY, Egyptian Mill, Egyptian Street, Bolton BL1 2HS, Telephone: 01204867615, Email address: stuart.rathmell.insolvency@outlook.com.

Leon Smart, Chairman Thursday 17 August 2023

(4431774)

A&M SAFETY SERVICES LTD (Company Number: 08167023) trading as A&M SAFETY SERVICES LTD

Registered Office: 11 Ramblers Way, Hemlington, Middlesbrough TS8 9GD

Principal Trading Address: 11 Ramblers Way, Hemlington, Middlesbrough TS8 9GD

Nature of Business: Roped access contractor

At a Extraordinary Meeting of the Members of the above-named Company, duly convened, and held remotely on Tuesday 22 August 2023, the following Resolution/s was/were duly passed:

1. (Special Resolution) THAT it has been proved to the satisfaction of this meeting that the company cannot, by reason of its liabilities, continue its business and that it is advisable to wind up the same, and accordingly that the Company be wound up voluntarily

2. (Ordinary Resolution) Stuart John Rathmell of Stuart Rathmell Insolvency Egyptian Mill Egyptian Street Bolton BL1 2HS be and is hereby appointed

For further details, please contact: Stuart Rathmell, (10050), STUART RATHMELL INSOLVENCY, Egyptian Mill, Egyptian Street, Bolton BL1 2HS, Telephone: 01204867615, Email address: stuart.rathmell.insolvency@outlook.com.

Andrew McKenna, Chairman Tuesday 22 August 2023

(4431814)

#### A&M SAFETY SERVICES LTD

(Company Number: 08167023)

trading as A&M SAFETY SERVICES LTD

Registered Office: 11 Ramblers Way, Hemlington, Middlesbrough TS8 9GD

Principal Trading Address: 11 Ramblers Way, Hemlington, Middlesbrough TS8 9GD

Nature of Business: Roped access contractor

At a Extraordinary Meeting of the Members of the above-named Company, duly convened, and held remotely on Tuesday 22 August 2023, the following Resolution/s was/were duly passed:

1. (Special Resolution) THAT it has been proved to the satisfaction of this meeting that the company cannot, by reason of its liabilities, continue its business and that it is advisable to wind up the same, and accordingly that the Company be wound up voluntarily

2. (Ordinary Resolution) Stuart John Rathmell of Stuart Rathmell Insolvency Egyptian Mill Egyptian Street Bolton BL1 2HS be and is hereby appointed

For further details, please contact: Stuart Rathmell, (10050), STUART RATHMELL INSOLVENCY, Egyptian Mill, Egyptian Street, Bolton BL1 2HS, Telephone: 01204867615, Email address: stuart.rathmell.insolvency@outlook.com.

Andrew McKenna, Chairman Tuesday 22 August 2023

(4431819)

### ROSIE LEE'S CAFE LIMITED

(Company Number: 12411794)

trading as Rosie Lee's Cafe Limited Registered Office: 21 Highfield Road, Dartford DA1 2JS

Principal Trading Address: 3 Station Road, Strood, Rochester ME2 4AX

Nature of Business: Unlicensed restaurants and cafes

At a General Meeting of the Members of the above-named Company, duly convened, and held at 3 Station Road, Strood, Rochester ME2 4AX on Wednesday 30 August 2023, the following Resolution/s was/ were duly passed:

1. (Ordinary Resolution) That Isobel Susan Brett of Bretts Business Recovery Limited, 21 Highfield Road, Dartford, Kent, DA1 2JS, be appointed liquidator of the company for the purposes of the windingup

2. (Special Resolution) That the company be wound up voluntarily

For further details, please contact: Isobel Susan Brett, (9643), Bretts Business Recovery Limited, 21 Highfield Road, Dartford, Kent DA1 2JS, Telephone: 01474 532862, Email address: enquiries@brettsbr.co.uk, (Z3190).

L Austin, Chairman

Thursday 31 August 2023

(4431350)

#### ELECTRA EUROPE LTD

(Company Number: 08733875)

trading as Electra Europe Ltd

Registered Office: 14 Brockles Mead, Harlow, Essex CM19 4PS Principal Trading, Address: 14 Brockles, Mead, Harlow, Essey, CN

Principal Trading Address: 14 Brockles Mead, Harlow, Essex CM19 4PS

Nature of Business: Electrical Contracting

At a General Meeting of the Members of the above-named Company, duly convened, and held at 14 Brockles Mead, Harlow, Essex CM19 4PS on Wednesday 30 August 2023, the following Resolution/s was/ were duly passed:

1. (Special Resolution) That it has been proved to the satisfaction of this Meeting that the Company cannot, by reason of its liabilities, continue its business, and that it is advisable to wind-up the same, and accordingly that the Company be wound up voluntarily.

2. (Ordinary Resolution) That Leigh Holmes (IP Number 9390) & Susan Purnell (IP Number 9386) of Purnells, Goldfields House, 18A Gold Tops, Newport, South Wales, NP20 4PH be and are hereby nominated Joint Liquidators for the purpose of the winding-up.

For further details, please contact: Leigh Holmes, (9390), and Susan Purnell, (9386), Purnells, Goldfields House, 18a Gold Tops, Newport, South Wales NP20 4PH UK, Telephone: 01633 214712, Email address: leigh@purnells.co.uk, (E1106).

Rosario Rocchi, Director Thursday 31 August 2023

(4432056)

# 3D CITY SIGNS LIMITED

(Company Number 07037635)

Registered office: 142 Thornton Road, Bradford, BD1 2DX

Principal trading address: 142 Thornton Road, Bradford, BD1 2DX

Notice is hereby given that the following resolutions were passed on 25 August 2023 as a Special Resolution and an Ordinary Resolution respectively:

"That the Company be wound up voluntarily and that *Nicola Baker* (IP No. 15852) of Rushtons Insolvency Limited, 6 Festival Building, Ashley Lane, Saltaire, BD17 7DQ be appointed Liquidator for the purposes of the voluntary winding up of the Company."

Further details contact: Nicola Baker, Tel: 01274 598585. Alternative contact: Email: sarobinson@rushtonsifs.co.uk

Tabassum Faisal, Director

29 August 2023

Ag FJ52940

(4430196)

#### A LAWRANCE BUILDING SERVICES LIMITED

(Company Number 11544564)

Registered office: 4 Fairacres, Ruislip, Middlesex, HA4 8AN Principal trading address: 4 Fairacres, Ruislip, Middlesex, HA4 8AN Notice is hereby given that the following resolutions were passed on 23 August 2023 as a special resolution and an ordinary resolution respectively: "That the Company cannot, by reason of its liabilities, continue its business, and it is advisable to wind up the same, and accordingly that the Company be wound up voluntarily and that *Farheen Qureshi* (IP No. 11392) of Parker Getty Limited, Devonshire House, 582 Honeypot Lane, Stanmore, Middx, HA7 1JS be appointed Liquidator for the purposes of such voluntary winding up."

Further details contact: The Liquidator, Tel: 020 3475 3900 or email meetings@parkergetty.co.uk Alternative contact: E Popat. *Alfie Lawrance,* Director

23 August 2023 Ag FJ52889

(4430169)

### ABBEY QUILTING LIMITED

(Company Number 09462764)

Registered office: Selinas Lane, Dagenham, Essex, RM8 1ES

Principal trading address: (Formerly) Selinas Lane, Dagenham, Essex, RM8 1ES

At a general meeting of the members of the above named Company, duly convened and held at 1066 London Road, Leigh-on-Sea, Essex SS9 3NA on 24 August 2023 the following resolutions were duly passed as a special resolution and an ordinary resolution respectively: "That the Company be wound up voluntarily and that *Dominik Thiel-Czerwinke* (IP No. 009636) and *Louise Donna Baxter* (IP No. 009123) both of Begbies Traynor (Central) LLP, 1066 London Road, Leigh-on-Sea, Essex, SS9 3NA be and hereby are appointed Joint Liquidators of the Company for the purpose of the voluntary winding-up, and any act required or authorised under any enactment to be done by the Joint Liquidators may be done by all or any one or more of the persons holding the office of liquidator from time to time."

Any person who requires further information may contact the Joint Liquidator by telephone on 01702 467255. Alternatively enquiries can be made to Rosie Thurwood by e-mail at Southendteamd@btguk.com or by telephone on 01702 467255.

*Carmnu Gennaro Zammit,* Chair 24 August 2023

Ag FJ52918

(4430208)

### AI BUS ENGINEERING SOLUTIONS LTD

(Company Number 09903105)

Registered office: 45 Caravel Close, Grays, RM16 6QB

Principal trading address: 45 Caravel Close, Grays, RM16 6QB Notice is hereby given that the following resolutions were passed on as a special resolution and an ordinary resolution respectively:

"That the company be wound up voluntarily" and

"That Nimish Patel and Hasib Howlader both of Hudson Weir Limited, 58 Leman Street, London, E1 8EU be appointed as Joint Liquidators of the Company for the purpose of the voluntary winding up and that the Joint Liquidators may act jointly and severally in relation to the appointment."

Joint Liquidator: *Nimish Patel* (IP number 8679) of Hudson Weir Limited, 58 Leman Street, London, E1 8EU.

Joint Liquidator: *Hasib Howlader* (IP number 20350) of Hudson Weir Limited, 58 Leman Street, London, E1 8EU.

Date of Appointment: 23 August 2023

For further details contact Tiffany Truter on 020 7099 6086 or at tiffany@hudsonweir.co.uk

Mr Audrius Indriulis, Director (4431613)

#### AIC LONDON LIMITED

(Company Number 12051373)

Trading Name: Kebaby

Registered office: 19 London Road, Morden, Surrey, SM4 5HT

Principal trading address: 19 London Road, Morden, Surrey, SM4 5HT (Pursuant to Sections 282 & 283 of the Companies Act 2006 and Sections 84(1) and 100 of the Insolvency Act 1986)

MEMBERS MEETING

At a General Meeting of the above named company duly convened and held at 5/7 Ravensboume Road, Bromley, Kent, BR1 1HN, on 29 August 2023, the following resolutions were duly passed:

Special Resolution:

"That the company be wound up voluntarily" Ordinary Resolution:

(4430161)

"That Robert Cundy of Edge Recovery Limited, 5/7 Ravensbourne Road, Bromley, Kent, BR1 1HN, be appointed liquidator of the company for the purposes of the winding-up."

Liquidator: Robert Cundy (IP number 9495) of Edge Recovery Limited, 5/7 Ravensbourne Road, Bromley, Kent, BR1 1HN.

Date of Appointment: 29 August 2023

For further details contact Zoe Herrington on 020 8315 7430 or at zoe.herrington@edgerecovery.com

Cengiz Bakbak, Chair of the Meeting of Members (4431373)

### ALCASO LOGISTIC LIMITED

(Company Number 10653751)

Registered office: 120 Bark Street, Bolton, Greater Manchester, BL1 2AX

Principal trading address: Schoolfield Corner Church Lane, Dogmersfield, Hook, RG27 8SY

Nature of Business: Freight transport by road.

Type of Liquidation: Creditors Voluntary Liquidation.

Place of meeting: 120 Bark Street, Bolton, Greater Manchester, BL1

2AX.

Date of meeting: 25 August 2023.

Notice is given that at a General Meeting of the Company, duly convened and held at the place and on the date given above, a special resolution was passed that the company be wound up voluntarily; and an ordinary resolution was passed appointing the Liquidator for the purposes of the winding-up.

Date of Appointment: 25 August 2023

Liquidator's Name and Address: Rikki Burton (IP No. 14430) of Anderson Brookes Insolvency Practitioners Limited, 6th Floor, 120 Bark Street, Bolton, BL1 2AX.

For further information contact Laura Wright at the offices of Anderson Brookes Insolvency Practitioners Limited on 0800 1804 933. or laura@andersonbrookes.co.uk. 29 August 2023

(4430988)

#### AMALGAMATED EURO PRODUCTS UK LTD

Trading Name: AEP UK Ltd

(Company Number 07977391)

Registered office: 120 Bark Street, Bolton, BL1 2AX

Principal trading address: Unit 3 Rufus Business Centre, Ravensbury Terrace, London, SW18 4RL

Nature of Business: Other manufacturing not elsewhere classified.

Type of Liquidation: Creditors Voluntary Liquidation.

Place of meeting: 120 Bark Street, Bolton, BI1 2AX.

Date of meeting: 21 August 2023.

Notice is given that at a General Meeting of the Company, duly convened and held at the place and on the date given above, a special resolution was passed that the company be wound up voluntarily; and an ordinary resolution was passed appointing the Liquidator for the purposes of the winding-up.

Date of Appointment: 21 August 2023

Liquidator's Name and Address: Rikki Burton (IP No. 14430) of Anderson Brookes Insolvency Practitioners Limited, 6th Floor, 120 Bark Street, Bolton, BL1 2AX.

For further information contact Steven Marshall at the offices of Anderson Brookes Insolvency Practitioners Limited on 0800 1804 933, or steven@andersonbrookes.co.uk.

30 August 2023

(4430991)

#### ARAMOKO BROADBAND LIMITED

(Company Number 05137469)

Registered office: Sigma House Oak View Close, Edginswell Park, Torquay, TQ2 7FF

Principal trading address: N/A

At a general meeting of the above-named company, duly convened and held at 5 Barnfield Crescent, Exeter, EX1 1QT on 23 August 2023 at 11.00 am, the following resolutions were passed as a Special Resolution and an Ordinary Resolution:

"That the Company be wound up voluntarily and that David Gerard Kirk (IP No. 8830) and Daniel Robert Jeeves (IP No. 26032) both of Kirks, 5 Barnfield Crescent, Exeter, EX1 1QT be and are hereby appointed Joint Liquidators for the purpose of such winding up."

In case of queries, please contact Daniel Jeeves on 01392 474303 or email Nathan@kirks.co.uk.

P Bright, Chair 24 August 2023 Ag FJ52883

### AUTO REPAINT LTD

(Company Number 08720034)

Registered office: 57 High Street, Rowley Regis, B65 0EH

Principal trading address: (Formerly) 57 High Street, Rowley Regis, B65 0EH

At a General Meeting of the Members of the above-named Company, duly convened and held at LevelQ, Sheraton House, Surtees Way, Surtees Business Park, Stockton-on-Tees, TS18 3HR on 24 August 2023 the following Resolutions were duly passed, as a Special Resolution and as an Ordinary Resolution respectively:

"That the Company be wound up voluntarily and that lan James Royle (IP No. 18934) and David Adam Broadbent (IP No. 9458) both of Begbies Traynor (Central) LLP, LevelQ, Sheraton House, Surtees Way, Surtees Business Park, Stockton on Tees, TS18 3HR be and hereby are appointed Joint Liquidators of the Company for the purpose of the voluntary winding-up, and any act required or authorised under any enactment to be done by the Joint Liquidators may be done by all or any one or more of the persons holding the office of Liquidator from time to time."

Any person who requires further information may contact the Joint Liquidators by telephone on 01642 796 640. Alternatively enquiries can be made to Jessica Barton by email at Jessica.Barton@btguk.com or by telephone on 01642 796 640. Thomas Clarke. Chair

24 August 2023

Ag FJ52888

### AUTOFIX SERVICE AND REPAIRS LTD

(Company Number 10310470)

Registered office: 31 Dashwood Avenue, High Wycombe, HP12 3DZ and it is in the process of being changed to The Old Town Hall, 71 Christchurch Road, Ringwood, BH24 1DH

Principal trading address: 31 Dashwood Avenue, High Wycombe, HP12 3DZ

Notice is hereby given that the following resolutions were passed on 24 August 2023 as a special resolution and an ordinary resolution respectively:

"That the Company be wound up voluntarily and that David Meany (IP No. 9453) and Frank Wessely (IP No. 7788) both of Quantuma Advisory Limited, The Old Town Hall, 71 Christchurch Road, Ringwood, BH24 1DH be and are hereby appointed Joint Liquidators for the purpose of such winding up. Any act required or authorised under any enactment to be done by a liquidator may be done by one of them."

Further details contact: Zoe Gannaway, Tel: 01202 970 430 or Email: Zoe.Gannaway@quantuma.com

Gail Mooney, Director 24 August 2023 Ag FJ52948

(4430191)

(4430162)

# BATTERSEA SPANISH LTD.

(Company Number 08846270)

Registered office: Fortis Insolvency Limited, 683-693 Wilmslow Road, Manchester, M20 6RE (previously Battersea Spanish, 263-265 Battersea Park Rd, London, SW11 4NE)

Principal trading address: 263-265 Battersea Park Rd, London, SW11 4NE

Notice is hereby given that the following resolutions were passed on 25 August 2023 as a special resolution and an ordinary resolution respectively:

"That the Company be wound up voluntarily and that Daniel Taylor (IP No. 21050) of Fortis Insolvency Limited, 683-693 Wilmslow Road, Manchester, M20 6RE be appointed as Liquidator for the purposes of such voluntary winding up."

For further details contact: The Liquidator, Tel: 0161 694 9955. Alternative contact: Andrew Fox.

Sara Carvajal Batres, Director

25 August 2023

Ag FJ52875

(4430147)

### BIRTHDAY BLASTER LTD

(Company Number 07717741)

Registered office: 1 The Coach House, Main Street South, Aberford, Leeds, LS25 3DA

Principal trading address: 1 The Coach House, Main Street South, Aberford, Leeds, LS25 3DA

At a General Meeting of the members of the above named Company, duly convened and held at Begbies Traynor, 2 Harcourt Way, Meridian Business Park, Leicester LE19 1WP on 24 August 2023 the following resolutions were duly passed, as a Special Resolution and as an Ordinary Resolution respectively:

"That the Company be wound up voluntarily and that *Thomas Harris* (IP No. 23370) and *Martin Richard Buttriss* (IP No. 9291) both of Begbies Traynor (Central) LLP, 2 Harcourt Way, Meridian Business Park, Leicester, LE19 1WP be and hereby are appointed Joint Liquidators of the Company for the purpose of the voluntary winding-up, and any act required or authorised under any enactment to be done by the Joint Liquidators may be done by all or any one or more of the persons holding the office of liquidator from time to time."

Any person who requires further information may contact the Joint Liquidator by telephone on 0116 406 2965. Alternatively enquiries can be made to Jack Thornber by email at Jack.Thornber@btguk.com or by telephone on 0116 402 8559.

Marcus Julian Sheenan, Chair 24 August 2023

Ag FJ52975

(4430168)

BJM FENCING LIMITED (Company Number 09136868)

Registered office: 1386 London Road, Leigh on Sea, Essex SS9 2UJ Principal trading address: (Former) 1386 London Road, Leigh on Sea,

Essex SS9 2UJ At a General Meeting of the Members of the above-named Company,

duly convened and held at 1066 London Road, Leigh-on-Sea, Essex SS9 3NA on 24 August 2023, the following Resolutions were duly passed, as a Special Resolution and as an Ordinary Resolution respectively:

"That the Company be wound up voluntarily and that *Wayne MacPherson* (IP No. 9445) and *Jamie Taylor* (IP No. 002748) both of Begbies Traynor (Central) LLP, 1066 London Road, Leigh-on-Sea, Essex, SS9 3NA be and hereby are appointed Joint Liquidators of the Company for the purpose of the voluntary winding-up, and any act required or authorised under any enactment to be done by the Joint Liquidators may be done by all or any one or more of the persons holding the office of Liquidator from time to time."

Any person who requires further information may contact the Joint Liquidators by telephone on 01702 467255. Alternatively enquiries can be made to Amber Ali by email at Amber.Ali@btguk.com or by telephone on 01702 467255.

*Ben James Mayo,* Chair 24 August 2023 Ag FJ52993

(4430209)

### **BL TRUCKING LTD**

(Company Number 09861833)

Registered office: 42 Leslie Road, Nottingham, NG7 6PD

Principal trading address: (Formerly) 42 Leslie Road, Nottingham, NG7 6PD

At a General Meeting of the Members of the above-named Company, duly convened and held at LevelQ, Sheraton House, Surtees Way, Surtees Business Park, Stockton-on-Tees, TS18 3HR on 24 August 2023 the following Resolutions were duly passed, as a Special Resolution and as an Ordinary Resolution respectively:

"That the Company be wound up voluntarily and that *Ian James Royle* (IP No. 18934) and *David Adam Broadbent* (IP No. 9458) both of Begbies Traynor (Central) LLP, LevelQ, Sheraton House, Surtees Way, Surtees Business Park, Stockton on Tees, TS18 3HR be and hereby are appointed Joint Liquidators of the Company for the purpose of the voluntary winding-up, and any act required or authorised under any enactment to be done by the Joint Liquidators may be done by all or any one or more of the persons holding the office of Liquidator from time to time."

Any person who requires further information may contact the Joint Liquidators by telephone on 01642 796 640. Alternatively enquiries can be made to Alice Dixon by email at Alice.Dixon@btguk.com or by telephone on 01642 796 640. *Bogdan Laskowski*, Chair

24 August 2023 Ag FJ52881

(4430159)

#### BLYTHWOOD CARS LTD

(Company Number 11195425) Registered office: Unit 4 Cannons Yard, Bedlars Green, Bishop's

Stortford, CM22 7UZ Principal trading address: (Formerly) Unit 4 Cannons Yard, Bedlars Green, Bishop's Stortford, CM22 7UZ

At a general meeting of the members of the above named Company, duly convened and held at 1066 London Road, Leigh-on-Sea, Essex SS9 3NA on 24 August 2023 the following resolutions were duly passed as a special resolution and an ordinary resolution respectively: "That the Company be wound up voluntarily and that *Louise Donna Baxter* (IP No. 009123) and *Wayne MacPherson* (IP No. 9445) both of Begbies Traynor (Central) LLP, 1066 London Road, Leigh-on-Sea, Essex, SS9 3NA be and hereby are appointed Joint Liquidators of the Company for the purpose of the voluntary winding-up, and any act required or authorised under any enactment to be done by the Joint Liquidators may be done by all or any one or more of the persons holding the office of liquidator from time to time."

Any person who requires further information may contact the Joint Liquidator by telephone on 01702 467255. Alternatively enquiries can be made to Amber Ali by e-mail at Amber.Ali@btguk.com or by telephone on 01702 467255.

Ross Gurden, Chair 24 August 2023 Ag FJ52924

g FJ52924

#### **BOARDWALK (RENDLESHAM) LTD**

Trading Name: The Board Walk Cafe

(Company Number 11033940)

Registered office: 120 Bark Street, Bolton, BL1 2AX

Principal trading address: Walnut Tree, Rendlesham, Suffolk, IP12 2GG

Nature of Business: Licensed clubs.

Type of Liquidation: Creditors Voluntary Liquidation.

Place of meeting: 120 Bark Street, Bolton, BL1 2AX.

Date of meeting: 25 August 2023.

Notice is given that at a General Meeting of the Company, duly convened and held at the place and on the date given above, a special resolution was passed that the company be wound up voluntarily; and an ordinary resolution was passed appointing the Liquidator for the purposes of the winding-up.

Date of Appointment: 25 August 2023

Liquidator's Name and Address: *Rikki Burton* (IP No. 14430) of Anderson Brookes Insolvency Practitioners Limited, 6th Floor, 120 Bark Street, Bolton, BL1 2AX.

For further information contact Laura Wright at the offices of Anderson Brookes Insolvency Practitioners Limited on 0800 1804 933, or laura@andersonbrookes.co.uk.

30 August 2023

(4431001)

(4430205)

#### BODEANS LIMITED

(Company Number 05065103)

Trading Name: Fulham Bodeans

Registered office: Gable House, 239 Regents Park Road, London, N3  $\rm 3LF$ 

Principal trading address: 6 Broadway Chambers, Hilary Close, FulhamRoad, SW6 1EA

(Pursuant to Sections 282 & 283 of the Companies Act 2006 and Sections 84(1) and 100 of the Insolvency Act 1986)

MEMBERS MEETING

At a General Meeting of the above named company duly convened and held at 27 Church Street, Rickmansworth, Hertfordshire, WD3 1DE, on 29 August 2023, the following resolutions were duly passed: Special Resolution

"That the company be wound up voluntarily".

Ordinary Resolution

COMPANIES

"That Bijal Shah of Edge Recovery Limited, 27 Church Street, Rickmansworth, Hertfordshire, WD3 1DE, be appointed liquidator of the company for the purposes of the winding-up".

Liquidator: *Bijal Shah* (IP number 8717) of Edge Recovery Limited, 27 Church Street, Rickmansworth, Hertfordshire, WD3 1DE.

Date of Appointment: 29 August 2023

For further details contact Jack Webb on 020 8315 7430 or at Jack.webb@edgerecovery.com

Michael Nicholas, Chair of the Meeting of Members

Date: 29 August 2023

### **BROOKLANDS PRESTIGE CARS LIMITED**

(Company Number 10963958)

Registered office: Arrans, Office 3, Swan Park Business Centre, Kettlebrook Road, Tamworth B77 1AG

Principal trading address: 5 Cotton Drive, Dalehouse Lane, Kenilworth, Warwickshire, CV8 2EB

At a general meeting of the above-named company, duly convened, and held at 4.30 pm on 22 August 2023, the following resolutions were passed:

Special resolution

That the company be wound up voluntarily.

Ordinary resolution

That Robert Gibbons, Licensed Insolvency Practitioner, Office Holder No 9079 of Arrans Limited, Office 3, Swan Park Business Centre, Kettlebrook Road, Tamworth, B77 1AG be and is hereby appointed Liquidator for the purpose of such winding up.

Liquidator: *Robert Paul Gibbons* (IP number 9079) of Arrans Limited, Office 3, Swan Park Business Centre, Kettlebrook Road, Tamworth, B77 1AG.

Date of Appointment: 22 August 2023

For further details contact: +44 (0)1827 60020

B A Chimes, Chair

(4430905)

(4431340)

### CHASE RENOVATIONS LIMITED

(Company Number 12442568)

Registered office: Azzurri House Walsall Business Park, Aldridge, Walsall, West Midlands, England, WS9 0RB

Principal trading address: 4 Pye Green Road, Cannock, Staffordshire, WS11 5RZ

At a General Meeting of the Members of the above-named Company, duly convened and held at LevelQ, Sheraton House, Surtees Way, Surtees Business Park, Stockton-on-Tees, TS18 3HR on 24 August 2023 the following Resolutions were duly passed, as a Special Resolution and as an Ordinary Resolution respectively:

"That the Company be wound up voluntarily and that *Ian James Royle* (IP No. 18934) and *David Adam Broadbent* (IP No. 9458) both of Begbies Traynor (Central) LLP, LevelQ, Sheraton House, Surtees Way, Surtees Business Park, Stockton on Tees, TS18 3HR be and hereby are appointed Joint Liquidators of the Company for the purpose of the voluntary winding-up, and any act required or authorised under any enactment to be done by the Joint Liquidators may be done by all or any one or more of the persons holding the office of Liquidator from time to time."

Any person who requires further information may contact the Joint Liquidators by telephone on 01642 796 640. Alternatively enquiries can be made to Becky Thorpe by email at becky.thorpe@btguk.com or by telephone on 01642 796 640.

Tony Andrew Wilby, Chair

24 August 2023

Ag FJ52906

(4430212)

CHURCH LANE FISH BAR LIMITED

(Company Number 10286806)

Registered office: 23 Church Lane, London, E11 1HG Principal trading address: 23 Church Lane, London, E11 1HG

Take-away food shops and mobile food stands

At a General Meeting of the Members of the above-named Company, duly convened, and held at 122 Hither Green Lane, London SE13 6QA on Friday 18 August 2023, the following Resolution/s was/were duly passed:

1. (Ordinary Resolution)

"That Mansoor Mubarik ACA FCCA FABRP of Capital Books (UK) Limited, 66 Earl Street, Maidstone, Kent, ME14 1PS, United Kingdom (office holder no 009667) be and is hereby appointed Liquidator for the purpose of such winding up."

2. (Special Resolution)

"That the company be wound up voluntarily."

For further details, please contact: Mansoor Mubarik, (009667), Capital Books (UK) Limited, 66 Earl Street, Maidstone ME14 1PS, Telephone: 01622 754 927, Email address: mmubarik@capitalbooks.co.uk, Fax: 01622 321 000. *Huseyin Serdal Demirci,* Chairman Friday 18 August 2023 (4431878)

(-----

### **CUSTOM JOINERY & BUILDING LTD**

(Company Number 09626093)

Previous Name of Company: IBE Joinery and Building Limited

Registered office: 30 Birkenhead Road, Hoylake, Wirral, CH47 3BW Principal trading address: (Former) 30 Birkenhead Road, Hoylake, Wirral, CH47 3BW

At a General Meeting of the members of the above named Company, duly convened and held at 30 Birkenhead Road, Hoylake, Wirral, CH47 3BW on 24 August 2023, the following Resolutions were duly passed as a Special Resolution and as an Ordinary Resolution respectively:

"That the Company be wound up voluntarily and that Jason Dean Greenhalgh (IP No. 009271) and Paul Stanley (IP No. 008123) both of Begbies Traynor (Central) LLP, No 1 Old Hall Street, Liverpool, L3 9HF be and hereby are appointed Joint Liquidators of the Company for the purpose of the voluntary winding-up, and any act required or authorised under any enactment to be done by the Joint Liquidators may be done by all or any one or more of the persons holding the office of Liquidator from time to time."

Any person who requires further information may contact the Joint Liquidator by telephone on 0151 227 4010. Alternatively enquiries can be made to David Jones by email at david.jones@btguk.com

*lan Ellis,* Chair 24 August 2023 Ag FJ52916

(4430222)

#### DEALS UNITED WORLDWIDE LTD

(Company Number 12866177)

Registered office: 6th Floor, 120 Bark St, Bolton, Greater Manchester, BL1 2AX

Principal trading address: Suite 165, Unit 3a, 34-35 Hatton Garden, EC1N 8DX

Nature of Business: Wholesale of computers, computer peripheral equipment and software.

Type of Liquidation: Creditors Voluntary Liquidation.

Place of meeting: 120 Bark Street, Bolton, BL1 2AX.

Date of meeting: 24 August 2023.

Notice is given that at a General Meeting of the Company, duly convened and held at the place and on the date given above, a special resolution was passed that the company be wound up voluntarily; and an ordinary resolution was passed appointing the Liquidator for the purposes of the winding-up.

Date of Appointment: 24 August 2023

Liquidator's Name and Address: *Rikki Burton* (IP No. 14430) of Anderson Brookes Insolvency Practitioners Limited, 6th Floor, 120 Bark Street, Bolton, BL1 2AX.

For further information contact Laura Wright at the offices of Anderson Brookes Insolvency Practitioners Limited on 0800 1804 933, or laura@andersonbrookes.co.uk.

30 August 2023

(4430997)

#### DK1 LTD

(Company Number 12235595)

Registered office: Nickron House, Bury Lane, Rickmansworth, Herts, WD3 1DN  $\,$ 

Principal trading address: Nickron House, Bury Lane, Rickmansworth, Herts, WD3 1DN

Notice is hereby given that the following resolutions were passed on 24 August 2023 as a special resolution and an ordinary resolution respectively:

"That the Company cannot, by reason of its liabilities, continue its business, and it is advisable to wind up the same, and accordingly that the Company be wound up voluntarily and that Farheen Qureshi (IP No. 11392) of Parker Getty Limited, Devonshire House, 582 Honeypot Lane, Stanmore, Middx, HA7 1JS be appointed Liquidator for the purposes of such voluntary winding up."

Further details contact: The Liquidator, Tel: 020 3475 3900 or email meetings@parkergetty.co.uk Alternative contact: E Popat.

David Andrew Kennon, Director

24 August 2023

Ag FJ52892

(4430166)

For

further

Fazel Zohrabpour, Chair

Craig Harmon

29 August 2023

Ag FJ52876

### DOUBLE ANCHOR LIMITED

(Company Number 11835382)

Registered office: PO Box 4385 - 11835382, Cardiff, CF14 8LH

Principal trading address: Unit 100, 61 West Ham Lane, London, E15 4PH

At a general meeting of the above-named company, duly convened, and held at 29 Peninsula Court, 2 Montford Place, London, E15 2QA on 16 August 2023, the following resolutions were passed as Special and Ordinary Resolutions:

"That the Company be wound up voluntarily and that Emma Mifsud (IP No. 21070) and Colin David Wilson (IP No. 9478) both of Opus Restructuring LLP, Third Floor, One Park Row, Leeds, LS1 5HN be and are hereby appointed Joint Liquidators for the purpose of such winding up."

Further details contact: Zachary Scahill, Tel: 020 3326 6454 or email cvlteam@opusllp.com

Ashfaq Shaikh, Chair

16 August 2023

Ag FJ52877 (4430142)

#### EASICUT PRECISION GRINDING COMPANY LTD

(Company Number 07536702)

Registered office: 120 Bark Street, Bolton, BL1 2AX

Principal trading address: 22 Leestone Road, Wythenshawe, Manchester, M22 4RN

Nature of Business: Other engineering activities.

Type of Liquidation: Creditors Voluntary Liquidation.

Place of meeting: 120 Bark Street, Bolton, BL1 2AX.

Date of meeting: 25 August 2023.

Notice is given that at a General Meeting of the Company, duly convened and held at the place and on the date given above, a special resolution was passed that the company be wound up voluntarily; and an ordinary resolution was passed appointing the Liquidator for the purposes of the winding-up.

Date of Appointment: 25 August 2023

Liquidator's Name and Address: Rikki Burton (IP No. 14430) of Anderson Brookes Insolvency Practitioners Limited, 6th Floor, 120 Bark Street, Bolton, BL1 2AX.

For further information contact Steven Marshall at the offices of Anderson Brookes Insolvency Practitioners Limited on 0800 1804 933, or steven@andersonbrookes.co.uk.

(4430994)

Florin-Marius Bibirigea, Director

### FARRER'S CONSTRUCTION LTD

(Company Number 10929566)

Registered office: (to be changed to) Sussex Innovation, One Croydon, 12-16 Addiscombe Road, Croydon, CR0 0XT

Principal trading address: Rae House, Dane Street, Bishop's Stortford, CM23 3BT

Notice is hereby given that the following resolutions were passed on 22 August 2023 by Written Resolution of the members of the above named Company as a special resolution and an ordinary resolution:

"That the Company be wound up voluntarily and that Matthew Reeds (IP No. 28040) and Paul Bailey (IP No. 9428) both of Bailey Ahmad Limited, Sussex Innovation, One Croydon, 12-16 Addiscombe Road, Croydon, CR0 0XT be and are hereby appointed Joint Liquidators for the purpose of such winding up."

EUROPOST EXPRESS (UK) LTD. (Company Number 09694048)

30 August 2023

Registered office: C12 Marquis Court, Marquis Way, Team Valley, Gateshead, NE11 0RU

Principal trading address: 4 Corringham Road, Wembley, HA9 9QA

At a General Meeting of the members of the above named Company, duly convened and held at 99 Bishopsgate. London, EC2M 3XD on 25 August 2023, the following resolutions were duly passed as a Special Resolution and as an Ordinary Resolution:

"That it has been proved to the satisfaction of this meeting that the Company cannot, by reason of its liabilities, continue its business, and that it is advisable to wind up the same, and accordingly that the Company be wound up voluntarily and that Eric Walls (IP No. 9113) and Wayne Harrison (IP No. 9703) both of KSA Group Ltd, C12 Marquis Court, Marquisway, Team Valley, Gateshead, NE11 0RU be and are hereby appointed Joint Liquidators for the purposes of such winding up. The Liquidators are authorised to act jointly or severally in all matters pertaining to the liquidation."

(4430146)

E-mail:

EVINCE TECHNOLOGY LIMITED

details

(Company Number 06414686)

Registered office: Netpark Plexus Thomas Wright Way, Sedgefield, Stockton-On-Tees, England, TS21 3FD

contact:

insolvency@ksagroup.co.uk, Tel: 0191 482 3343. Alternative contact:

The

Liquidators,

Principal trading address: Netpark Plexus Thomas Wright Way, Sedgefield, Stockton-On-Tees, England, TS21 3FD

At a General Meeting of the Members of the above-named Company, duly convened, and held on 11 August 2023, the following Resolutions were duly passed, as a Special Resolution and as an Ordinary Resolution:

That the Company be wound up voluntarily.

That Michael Chamberlain and Gareth Peckett of Quantuma Advisory (Yorkshire) Limited, Resolution House, 12 Mill Hill, Leeds, LS1 5DQ be appointed Joint Liquidators of the Company for the purpose of the voluntary winding-up. Any act required or authorised under any enactment to be done by a Liquidator may be done by one of them.

That creditors confirm the nomination of the Liquidators appointed by themembers and accordingly, Michael Chamberlain and Gareth Peckett remain asLiquidators of the Company.

That the Liquidators be authorised to act jointly and severally in the liquidation.

Joint Liquidator: Michael Chamberlain (IP number 8735) of Quantuma Advisory (Yorkshire) Limited, Resolution House, 12 Mill Hill, Leeds, LS1 5DQ.

Joint Liquidator: Gareth Peckett (IP number 9647) of Quantuma Advisory (Yorkshire) Limited, Resolution House, 12 Mill Hill, Leeds, LS1 5DQ.

Date of Appointment: 11 August 2023

For further details contact Caitlin Auplish on 0113 242 0808 or at Caitlin.Auplish@quantuma.com

Phillip Cammerman, Director (4431683)

#### **EVOLUTIV CONSTRUCTION LIMITED**

(Company Number 10816943)

Registered office: 154 Marsh Lane, Stanmore, HA7 4HT

Principal trading address: 154 Marsh Lane, Stanmore, HA7 4HT

Notice is hereby given that the following resolutions were passed on 24 August 2023 as a special resolution and an ordinary resolution respectively:

That the Company cannot, by reason of its liabilities, continue its business, and it is advisable to wind up the same, and accordingly that the Company be wound up voluntarily and that Farheen Qureshi (IP No. 11392) of Parker Getty Limited, Devonshire House, 582 Honeypot Lane, Stanmore, Middx, HA7 1JS be appointed Liquidator for the purposes of such voluntary winding up."

Further details contact: The Liquidator, Tel: 020 3475 3900, email: meetings@parkergetty.co.uk Alternative contact: E Popat.

24 August 2023 Ag FJ52893

(4430224)

(4430178)

It should be noted that the Company was placed into liquidation via deemed consent and that Paul Bailey and Matthew Reeds were appointed as Joint Liquidators on 22 August 2023.

Further details contact: The Liquidators, Email: creditors@babr.co.uk, Tel: 0208 662 6070.

Richard Farrer, Director 30 August 2023 Ag FJ52998

(4430193)

#### FASHION COUTURE LIMITED

(Company Number 09544132)

Registered office: 400 Victoria Road East, Leicester, LE5 0LG Principal trading address: 400 Victoria Road East, Leicester, LE5 0LG At a General Meeting of the Members of the above-named company, duly convened, and held at 126 New Walk, Leicester, LE1 7JA on 29 August 2023, the following resolutions were passed by the Members: 1 as a Special resolution and 2 as an Ordinary resolution. **Resolutions:** 

1. "That the Company be wound up voluntarily"; and

2. "That Neil Charles Money of CBA Business Solutions, 126 New Walk, Leicester, LE1 7JA, be appointed Liquidator of the Company" Liquidator: Neil Charles Money (IP number 8900) of CBA Business Solutions, 126 New Walk, Leicester, LE1 7JA.

Date of Appointment: 29 August 2023

For further details contact Steven Glanvill on 0116 262 6804

Manjit Singh Khara, Chair of Meeting (4431382)

#### FAT LEMON PRODUCTIONS LIMITED

(Company Number 10609051)

Registered office: 6th Floor Charlotte Building, 17 Gresse Street, London, W1T 1QL

Principal trading address: Cobbetts, Easton Royal, Pewsey, SN9 5LS At a General Meeting of the Members of the above-named company, duly convened, and held at 126 New Walk, Leicester, LE1 7JA on 25 August 2023, the following resolutions were passed by the Members: 1 as a Special resolution and 2 as an Ordinary resolution. Resolutions:

1. "That the Company be wound up voluntarily" and

2. "That Neil Charles Money of CBA Business Solutions, 126 New Walk, Leicester, LE1 7JA, be appointed Liquidator of the Company' Liquidator: Neil Charles Money (IP number 8900) of CBA Business

Solutions, 126 New Walk, Leicester, LE1 7JA.

Date of Appointment: 25 August 2023

For further details contact Victoria Ely on 0116 262 6804 Mr Cabell Hopkins, Chair of Meeting (4431022)

#### FENDA CONSULTING LTD

(Company Number 09667039)

Registered office: 33 Evans Avenue, Watford, WD25 0EX in the process of being changed to FRP Advisory Trading Limited, Mountbatten House, Grosvenor Square, Southampton, SO15 2RP Principal trading address: 33 Evans Avenue, Watford, WD25 0EX

At a general meeting of the above named company, duly convened and held at 33 Evans Avenue, Watford, WD25 0EX on 25 August 2023 at 3.00 pm, the following special and ordinary resolution were duly passed:

"That the company be wound up voluntarily and that Alexander Kinninmonth (IP No. 9019) and James Prior (IP No. 29250) both of FRP Advisory Trading Limited, Mountbatten House, Grosvenor Square, Southampton, SO15 2RP be and are hereby appointed Liquidators for the purposes of such winding up and that anything required or authorised to be done by the Liquidators be done by both or either of them."

Further details contact: Nate Taylor, Email: Nate.Taylor@frpadvisory.com Emmanuel Ozouli, Chair 25 August 2023 Ag FJ52969 (4430176)

### FJORD MANAGEMENT LTD

(Company Number 10518806)

Registered office: Chancery House, 30 St Johns Road, Woking, Surrev. GU21 7SA

Principal trading address: Chancery House, 30 St Johns Road, Woking, Surrey, GU21 7SA

Notice is hereby given that the following resolutions were passed on 27 July 2023, as a special resolution and an ordinary resolution respectively:

"That the Company be wound up voluntarily and that Steven John Parker (IP No. 8989) and Timothy John Edward Dolder (IP No. 9008) both of Opus Restructuring LLP, 322 High Holborn, London, WC1V 7PB be and are hereby appointed Joint Liquidators of the Company for the purposes of the winding up."

details The Joint Further contact: Liquidators, Tel: steve.parker@opusllp.com. Alternative contact: Sushil KC.

Heidi Marie Thomas, Director

27 July 2023

Ag FJ53018

#### FLEET AGENTS LIMITED

(Company Number 07728703)

Registered office: to be changed to TruSolv Ltd, Grove House, Meridians Cross, Ocean Village, Southampton, Hampshire, SO14 3TJ, having previously been 1 & 2 Studley Court Mews, Studley Court, Guildford Road, Chobham, Woking, Surrey, GU24 8EB.

Principal trading address: Folly Farm, Folly Lane, Bramley, Tadley, Hampshire, RG26 5BD

Notice is hereby given that the following resolutions were passed on 23 August 2023 as a Special Resolution and Ordinary Resolutions respectively:

a) "That the Company be wound up voluntarily"

b) That Matthew Hoy and Shane Biddlecombe of TruSolv Ltd, Grove House, Meridians Cross, Ocean Village, Southampton, Hampshire, SO14 3TJ be appointed as Joint Liquidators for the purpose of the voluntary winding up."

c) That the Liquidators be authorised to act jointly and severally in the liquidation.

Joint Liquidator: Matthew Hoy (IP number 24550) of TruSolv Ltd, Grove House, Meridians Cross, Ocean Village, Southampton, Hampshire, SO14 3TJ.

Joint Liquidator: Shane Biddlecombe (IP number 9425) of TruSolv Ltd, Grove House, Meridians Cross, Ocean Village, Southampton, Hampshire, SO14 3TJ.

Date of Appointment: 23 August 2023

For further details contact Neil McMahon on 0808 196 8676 Terence Gary Lewis, Director (4431713)

#### FUTURE ENGRAVING LIMITED

(Company Number 03801292)

Registered office: 2nd Floor, 14 Castle Street, Liverpool, L2 0NE

Principal trading address: 98 Appin Road, Argyle Industrial Estate, Birkenhead, Wirral, CH41 9HH

NOTICE IS GIVEN that at a General Meeting of the above-named Company, duly convened and held at Parkin S Booth Ltd., 2nd Floor, 14 Castle Street, Liverpool, L2 ONE on 29 August 2023 at 11.00am the following resolutions were passed as a Special resolution and Ordinary resolution respectively:

1. "That the Company be wound up voluntarily" and

2. "That ?John Fisher (IP No: 9420) of Parkin S Booth Ltd, 2nd Floor, 14 Castle Street, Liverpool, L2 ONE be appointed Liquidator of the Company.'

Liquidator: John P Fisher (IP number 9420) of Parkin S Booth Ltd, 2nd Floor, 14 Castle Street, Liverpool, L2 ONE.

Date of Appointment: 29 August 2023

For further details contact Christine Tickle on 0151 236 4331 or at ct@parkinsbooth.co.uk

Dated this 29th day of August 2023 (4431643)

### **GRAVITY FLOORING (INTL) LTD**

#### (Company Number 14667808)

Registered office: 71-75 Shelton Street, Covent Garden, London WC2H 9JQ

Principal trading address: Unit 3, Staines Business Park, 96-104 Church Street, Staines, TW18 4YA

At a General Meeting of the above named Company duly convened and held at Parker Andrews Limited, 51-59 Rose Lane, Norwich, Norfolk, NR1 1BY on 24 August 2023, the following Resolutions were passed as a Special Resolution and as an Ordinary Resolution:

"That the Company be wound up voluntarily and that *Nicholas Cusack* (IP No. 17490) of Parker Andrews Limited, 5th Floor, The Union Building, 51-59 Rose Lane, Norwich, NR1 1BY be and is hereby appointed Liquidator for the purpose of such winding up."

Further details contact: Mark Middlemas, Email: Mark.Middlemas@parkerandrews.co.uk

Jonathan Dowling, Chair

24 August 2023

Ag FJ52959

(4430188)

### GREEN DOOR CLINIC LIMITED

(Company Number 11051601)

Registered office: 30 Weymouth Street, London W1G 7BS

Principal trading address: 30 Weymouth Street, London W1G 7BS

At a general meeting of the members of the above named Company, duly convened and held at Mountview Court, 1148 High Road, Whetstone, London N20 0RA on 17 August 2023 at 3.30 pm, the following resolutions were passed by the Members, as a special resolution and an ordinary resolution respectively:

"That the Company be wound up voluntarily and that *Andreas Arakapiotis* (IP No. 20910) of Kallis & Company, Mountview Court, 1148 High Road, Whetstone, London, N20 0RA be appointed Liquidator of the Company."

Further details contact: Yusuf Bham, Tel: 020 8446 6699; Email: Yusuf@kallis.co.uk

Jennifer Dew, Director

17 August 2023 Ag FJ52942

(4430181)

#### GSI ENERGY LIMITED

(Company Number 05414416)

Registered office: 1580 Parkway, Solent Business Park, Whiteley, Fareham, Hampshire PO15 7AG

Principal trading address: 14 Northside, Old Sarum, Salisbury, SP4 6BZ

Notice is hereby given that the following resolutions were passed on 24 August 2023 as special resolutions:

"That the Company be wound up voluntarily and that *Nicola Elaine Layland* (IP No. 017652) and *Carl Derek Faulds* (IP No. 008767) both of Leonard Curtis, 1580 Parkway, Solent Business Park, Whiteley, Fareham, Hampshire, PO15 7AG be and they are hereby appointed Joint Liquidators of the Company and that any act required or authorised to be done by the Liquidators, is to be done by both or either of them for the time being holding office."

Further details contact: The Joint Liquidators, Email: Creditors.south@leonardcurtis.co.uk. Alternative contact: Andrew Rumsey.

Adam Timothy Cubitt, Director 24 August 2023 Ag FJ52869

(4430157)

# **GUMBOO HOTELS LIMITED**

Trading Name: The Flying Horse

(Company Number 07586157)

Registered office: 211 Manchester New Road, Middleton, Manchester, M24 1JT

Principal trading address: 37 Packer St, Rochdale, OL16 1NJ

Notice is given that by written resolutions, the members of the company passed a special resolution that the company be wound up voluntarily, and an ordinary resolution appointing the Joint Liquidators for the purposes of the winding-up. The requisite voting majority was received on 24 August 2023

Benjamin Boothman, Director.

24 August 2023

Joint Liquidator's Name and Address: *Andrew Mark Bland* (IP No. 9472) of DMC Recovery Limited, 41 Greek Street, Stockport, Cheshire, SK3 8AX. Email: creditors@dmcrecovery.co.uk. Telephone: 0161 474 0920.

Joint Liquidator's Name and Address: *Janet Mayo* (IP No. 9595) of DMC Recovery Limited, 41 Greek Street, Stockport, Cheshire, SK3 8AX. Email: janet.mayo@dmcrecovery.co.uk. Telephone: 0161 474 0920.

For further information contact Sandeep Borse at the offices of DMC Recovery Limited on 0161 474 0920, or sandeep.borse@dmcrecovery.co.uk.

(4430987)

#### HADERA LIMITED

25 August 2023

(Company Number 09650124)

Previous Name of Company: Outu & K Transport Limited and Outu & K Limited

Registered office: Unit 2 Spinnaker Court, 1C Becketts Place, Hampton Wick, Kingston upon Thames, KT1 4EQ

Notice is hereby given that the following resolutions were passed on 17 August 2023, as a special resolution and an ordinary resolution respectively:

"That the Company be wound up voluntarily"; and

"That Douglas John Pinteau and Richard Andrew Segal of WSM Marks Bloom LLP, Unit 2, Spinnaker Court, 1c Becketts Place, Hampton Wick, Kingston upon Thames, Surrey, KT1 4EQ be appointed Joint Liquidators for the purposes of voluntary such winding up."

Joint Liquidator: *Douglas John Pinteau* (IP number 19590) of WSM Marks Bloom LLP, Unit 2 Spinnaker Court, 1C Becketts Place, Hampton Wick, Kingston upon Thames, KT1 4EQ.

Joint Liquidator: *Richard Andrew Segal* (IP number 2685) of WSM Marks Bloom LLP, Unit 2 Spinnaker Court, 1C Becketts Place, Hampton Wick, Kingston upon Thames, KT1 4EQ.

Date of Appointment: 17 August 2023

For further details contact Ella Foggensteiner on 020 8939 8240 DATED: 17 August 2023

Kaleb Hadera, Director (4431838)

#### HB PIZZA LTD

(Company Number 13563264)

Trading Name: Papa Johns

Registered office: 4 Whitchurch Parade, Whitchurch Lane, Edgware, HA8 6LR in process of being changed to c/o FRP Advisory Trading Limited, 4 Beaconsfield Road, St Albans, AL1 3RD

Principal trading address: 4 Whitchurch Parade, Whitchurch Lane, Edgware, HA8 6LR

At a General Meeting of the above named Company, duly convened, and held at 4 Whitchurch Parade, Whitchurch Lane, Edgware, HA8 6LR on 24 August 2023 at 4.00 pm, the following resolutions were duly passed as a special resolution and as an ordinary resolution:

"That the Company be wound up voluntarily and that *Simon Carvill-Biggs* (IP No. 15930) and *Miles Needham* (IP No. 14372) both of FRP Advisory Trading Limited, 4 Beaconsfield Road, St Albans, Hertfordshire, AL1 3RD be and are hereby appointed Joint Liquidators for the purposes of such winding up."

For further details contact: The Joint Liquidators, Email: cp.stalbans@frpadvisory.com.

Stuart Andrew Hunt, Chair

24 August 2023 Ag FJ52962

(4430179)

### HT ENTERPRISE SHEFF LIMITED

(Company Number 12380692)

Registered office: 598 Bradford Road, Batley, F17 8LP

Principal trading address: 156 Ecclesall Rd, Broomhall, Sheffield, S11 8JD

Notice is hereby given that the following resolutions were passed on 29 August 2023 as a special resolution and ordinary resolution respectively:

17518 | CONTAINING ALL NOTICES PUBLISHED ONLINE ON 31 AUGUST 2023 | LONDON GAZETTE

"That the Company cannot, by reason of its liabilities, continue its business, and that it is advisable to wind up the same, and accordingly that the company be wound up voluntarily and that David Frederick Wilson (IP No. 006074) of DFW Associates, 29 Park Square West, Leeds, LS1 2PQ be and is hereby appointed as Liquidator of the Company for the purposes of the voluntary winding up."

For further details contact: Chris Wilson. Email: Chris.wilson@dfwassociates.co.uk.

Aisha Patel, Director

29 August 2023

Ag FJ53017

(4430197)

#### ICONIC LEISURE LIMITED

(Company Number 12660703)

Trading Name: The Aviary, Whalley

Registered office: The Aviary, Queen Street, Whalley, Clitheroe, Lancashire, BB7 9TA

Principal trading address: The Aviary, Queen Street, Whalley, Clitheroe, Lancashire, BB7 9TA

At a general meeting of the above-named company, duly convened, and held at 1 Tadis House, 455 Whalley New Rd, Blackburn BB1 9SP at 10.00 am on 23 August 2023, the following resolutions were passed as a Special Resolution and as an Ordinary Resolution:

"That the company be wound up voluntarily and that Zane Collins (IP No. 25952) of Insolvency One Limited, 1 Aire Street, Suite 2.03, Leeds, West Yorkshire, LS1 4PR be and is hereby appointed Liquidator for the purpose of such winding up."

Further details contact: Zane Collins, Tel: 0800 999 1329, Email: Zane.collins@insolvencyone.co.uk

Jabbar Khan. Chair

23 August 2023 Ag FJ52992

(4430165)

**INDUSTRIAL FORMS LTD** 

(Company Number 10384440)

Registered office: 29-31 Moorland Road, Stoke-On-Trent, ST6 1DS Principal trading address: 1A Tricketts Lane, Willaston, Nantwich, CW5 6PS

At a General Meeting of the members of the above named Company, duly convened and held at DSi Business Recovery, 2 Lakeside, Calder Island Way, Wakefield, WF2 7AW on 25 August 2023 at 1.00 pm the following resolutions were duly passed as a Special Resolution and as an Ordinary Resolution:

"That it has been proved to the satisfaction of this meeting that the company cannot, by reason of its liabilities, continue its business, and that it is advisable to wind up the same, and accordingly that the company be wound up voluntarily and that Simon Weir (IP No. 9099) of DSi Business Recovery, 2 Lakeside, Calder Island Way, Wakefield, WF2 7AW be and he is hereby appointed Liquidator for the purposes of such winding up."

For further details contact: Simon Weir, Tel: 01924 790880, Email: info@dsinsol.com

Boren Burman, Director 29 August 2023 Ag FJ52882

(4430158)

### **JACKSON & O'LEARY CATERING LIMITED**

(Company Number 07090528)

Trading Name: Finbarr's Restaurant

Registered office: Aykley Heads House, Aykley Heads, Durham, DH1 5TS in process of being changed to F17 Evolve Business Centre, Cygnet Way, Houghton le Spring, Tyne & Wear, DH4 5QY

Principal trading address: Aykley Heads House, Aykley Heads, Durham, DH1 5TS

At a General Meeting of the above-named Company, duly convened and held at FRP Advisory Trading Limited, F17 Evolve Business Centre, Cygnet Way, Houghton le Spring, Tyne & Wear, DH4 5QY on 24 August 2023 at 3.00 pm, the following resolutions were passed as a Special resolution and Ordinary resolution respectively:

"That the Company be wound up voluntarily and that Steven Philip Ross (IP No. 9503) and Allan David Kelly (IP No. 9156) both of FRP Advisory Trading Limited, F17 Evolve Business Centre, Cygnet Way, Houghton le Spring, Tyne and Wear, DH4 5QY be and are hereby appointed Liquidators for the purposes of such winding up."

Further details contact: The Joint Liquidators, Email: Steven.ross@frpadvisory.com Tom Jackson, Chair 24 August 2023 Ag FJ52938 (4430185)

#### JAMES JONES@BF LTD

(Company Number 09556359)

Trading Name: James Jones Beautiful Furnishings

Registered office: 63 Newport Road, Middlesborough, TS1 1LB To be changed to: c/o Clark Business Recovery Limited, 8 Fusion Court, Aberford Road, Garforth, Leeds, LS25 2GH

Principal trading address: 63 Newport Road, Middlesborough, TS1 1LB & The Cleveland Centre, 85 Grange Road, Middlesborough, TS1 2LS 80 High Street, Redcar, RS10 3DL

Notice is hereby given that the following resolutions were\_ passed on, as a special resolution and as ordinary resolutions respectively:

"That the Company be wound up voluntarily" and

"That Phil Clark and Dave Clark be appointed as Joint Liquidators for the purposes of such voluntary winding up"

"That any act required or authorised under any enactment to be done by the Joint Liquidators may be done by all or any one or more of the persons for the time being holding such office."

Joint Liquidator: Phil Clark (IP number 23530) of Clark Business Recovery Limited, 8 Fusion Court, Aberford Road, Garforth, Leeds, LS25 2GH.

Joint Liquidator: Dave Clark (IP number 9565) of Clark Business Recovery Limited, 8 Fusion Court, Aberford Road, Garforth, Leeds, LS25 2GH.

Date of Appointment: 29 August 2023

For further details contact Daniel Overend on 0113 243 8617 or at daniel@clarkbr.co.uk

Frakah Baza Hanif - Convener (4431401)

#### JAO CONSULTING LTD

(Company Number 10079516)

Registered office: 120 Bark Street, Bolton, Greater Manchester, BL1 2AX

Principal trading address: 1 Clover End. Buckingham. Buckinghamshire, MK18 1FT

Nature of Business: Business and domestic software development.

Type of Liquidation: Creditors Voluntary Liquidation.

Place of meeting: 120 Bark Street, Bolton, Greater Manchester, BL1 2AX.

Date of meeting: 24 August 2023.

Notice is given that at a General Meeting of the Company, duly convened and held at the place and on the date given above, a special resolution was passed that the company be wound up voluntarily; and an ordinary resolution was passed appointing the Liquidator for the purposes of the winding-up.

Date of Appointment: 24 August 2023

Liquidator's Name and Address: Rikki Burton (IP No. 14430) of Anderson Brookes Insolvency Practitioners Limited, 6th Floor, 120 Bark Street, Bolton, BL1 2AX.

For further information contact Steven Marshall at the offices of Anderson Brookes Insolvency Practitioners Limited on 0800 1804 933, or steven@andersonbrookes.co.uk. 30 August 2023

(4430995)

#### JESSICA WILSON LIMITED

(Company Number 11762162)

Registered office: 13 Wharf Lane, Cow Roast, Tring, HP23 5FQ and it is in the process of being changed to The Old Town Hall, 71 Christchurch Road, Ringwood, BH24 1DH

Principal trading address: 35 Bower's Parade, High St, Harpenden, AL5 2RU

Notice is hereby given that the following resolutions were passed on 24 August 2023 as a special resolution and an ordinary resolution respectively:

"That the Company be wound up voluntarily and that *David Meany* (IP No. 9453) and *Frank Wessely* (IP No. 7788) both of Quantuma Advisory Limited, The Old Town Hall, 71 Christchurch Road, Ringwood, BH24 1DH be and are hereby appointed Joint Liquidators for the purpose of such winding up. Any act required or authorised under any enactment to be done by a liquidator may be done by one of them."

Further details contact: Tara Edwards, Tel: 01202 059 460 or Email: Tara.Edwards@quantuma.com

Jessica Jayne Wilson, Director 24 August 2023

Ag FJ52914

(4430220)

### JOYNER ESTATE LIMITED

(Company Number 09445858)

Registered office: 120 Bark Street, Bolton, BL 1 2AX

Principal trading address: 189 Lynchford Road, Farnborough, Hampshire, GU14 6HD

Nature of Business: Advertising agencies.

Type of Liquidation: Creditors Voluntary Liquidation.

Place of meeting: 120 Bark Street, Bolton, BL1 2AX.

Date of meeting: 25 August 2023.

Notice is given that at a General Meeting of the Company, duly convened and held at the place and on the date given above, a special resolution was passed that the company be wound up voluntarily; and an ordinary resolution was passed appointing the Liquidator for the purposes of the winding-up.

Date of Appointment: 25 August 2023

Liquidator's Name and Address: *Rikki Burton* (IP No. 14430) of Anderson Brookes Insolvency Practitioners Limited, 6th Floor, 120 Bark Street, Bolton, BL1 2AX.

For further information contact Laura Wright at the offices of Anderson Brookes Insolvency Practitioners Limited on 0800 1804 933, or laura@andersonbrookes.co.uk.

30 August 2023

(4430999)

### KC COMPOSITES LIMITED

(Company Number 08781877)

Registered office: G204 Weston House, Allen House Business Centre, The Mattings, Station Road, Sawbridgeworth, CM21 9FP

Principal trading address: Unit 4, Water Hill Farm, Hatfield Road, Hertford, SG13 8LE

At a general meeting of the above-named company, duly convened, and held at 45 Taylors Avenue, Hoddesdon, EN11 8QE on 25 August 2023 at 11.00am the following resolutions were passed:

Special resolution:

"That the company be wound up voluntarily."

Ordinary resolution:

"That Alan J Clark (office holder no 8760) of Carter Clark, Recovery House, 15-17 Roebuck Road, Hainault Business Park, Ilford, Essex, IG6 3TU be and are hereby appointed Liquidator for the purpose of such winding up."

Liquidator: *Alan J Clark* (IP number 8760) of Carter Clark, Recovery House, 15-17 Roebuck Road, Hainault Business Park, Ilford, Essex, IG6 3TU.

Date of Appointment: 25 August 2023

For further details contact Lisa Portway on 020 8559 5087 or at Jenny.poleykett@carterclark.co.uk

Lee Gudgeon, Chairman

Date: 25 August 2023

(4431358)

### LAHORE QILA LTD

(Company Number 10903290)

Registered office: 229 Whalley New Road, Blackburn, BB1 9TL

Principal trading address: 229 Whalley New Road, Blackburn, BB1 9TL

Notice is hereby given that the following resolutions were passed on 29 August 2023 as a special resolution and an ordinary resolution respectively:

"That the Company be wound up voluntarily and that *James David Robinson* (IP No. 16092) of Finn Associates, Tong Hall, Tong, West Yorkshire BD4 0RR be appointed as Liquidator for the purposes of such voluntary winding up."

Further details contact: The Liquidator, Email: solutions@finnassociates.com. Alternative contact: Helen Kilby. *Nasir Mehmood Choudry*, Director 29 August 2023 Ag FJ53025 (4430174)

### LEIGH SAFETY MANAGEMENT LIMITED

(Company Number 07524224)

Registered office: Charter House, 103-105 Leigh Road, Leigh on Sea, Essex SS9 1JL (in the process of being changed to Jupiter House, Warley Hill Business Park, The Drive, Brentwood, Essex, CM13 3BE) Principal trading address: 61 Leighville Grove, Leigh on Sea, Essex SS9 2HU

At a General Meeting of the above named Company duly convened and held at Jupiter House, Warley Hill Business Park, The Drive, Brentwood, Essex, CM13 3BE on 24 August 2023 at 3.00 pm the following resolutions were duly passed as a Special Resolution and as an Ordinary Resolution:

"That the Company be wound up voluntarily and that *Paul Atkinson* (IP No. 9314) and *Glyn Mummery* (IP No. 8996) both of FRP Advisory Trading Limited, Jupiter House, Warley Hill Business Park, The Drive, Brentwood, Essex, CM13 3BE be and are hereby appointed Liquidators for the purpose of such winding up."

Further details contact: The Joint Liquidators, Email: CP.Brentwood@FRPadvisory.com

Simon Plummer, Chair

24 August 2023 Ag FJ52907

(4430215)

### LINKS RESOURCING LIMITED

(Company Number 07934804)

Registered office: SUITE 3 DUKES HOUSE, 4-6 HIGH STREET, WINDSOR SL4 1LD

Principal trading address: TRADING ADDRESS – RIVERSIDE CENTRE, 42 HIGH STREET, KINGSTON KT1 1HL

#### THE COMPANIES ACT 2006 & THE INSOLVENCY ACT 1986

At a General Meeting of the Company convened and held at The Old Library, The Walk, Winslow, Buckingham MK18 3AJ on 24 August 2023 at 12.00 Noon the following special resolution numbered one and ordinary resolution numbered two were passed:

1 That the Company be wound up voluntarily.

2 That Lee James Cotton and Richard Behagg of Robert Day and Company Limited, The Old Library, The Walk, Winslow, Buckingham MK18 3AJ be appointed joint liquidators of the Company for the purposes of the voluntary winding-up and that any act required or authorised under any enactment to be done by the liquidator may be done by all or any one or more of the persons for the time being holding such office.

Lee James Cotton (IP No. 9610), Richard Behagg (IP No. 12790) Robert Day and Company Limited, The Old Library, The Walk, Winslow, Buckingham MK18 3AJ, mail@robertday.biz, 01296 711644. *TIMOTHY RICHARD CORNWALLIS DORAN* 

(4431877)

### LISSUS LTD

Chair/Convener

(Company Number 12203658)

Trading Name: Zebrano Registered office: c/o Begbies Traynor, 2/3 Pavilion Buildings, Brighton, East Sussex, BN1 1EE formerly 2nd Floor, Northumberland House, 303-306 High Holborn, London, WC1V 7JZ

Principal trading address: (Formerly) 18 Greek Street, London, W1D 4DS

At a general meeting of the members of the above named company duly convened and held at 18 Greek Street, London, W1D 4DS on 23 August 2023 the following resolutions were duly passed as a special and as an ordinary resolution respectively:

"That the company be wound up voluntarily and that *Jonathan James Beard* (IP No. 9552) and *Gavin Savage* (IP No. 9950) both of Begbies Traynor (Central) LLP, 2/3 Pavilion Buildings, Brighton, East Sussex BN1 1EE be and hereby are appointed as Joint Liquidators of the Company for the purposes of the voluntary winding up, and any act required or authorised under any enactment to be done by the Joint Liquidators may be done by all or any one or more of the persons holding the office of the Liquidator from time to time."

Any person who requires further information may contact the Joint Liquidator by telephone on 01273 322960. Alternatively enquiries can be made to James Trebble by e-mail at James.Trebble@btguk.com or by telephone on 01273 322960.

*E Skuraj,* Chair 23 August 2023 Ag FJ52996

(4430190)

#### LIVETHEDREAM UK LTD

(Company Number 06009881)

Trading Name: Proof Bar & Kitchen

Registered office: 79 Tib Street, Manchester, M4 1LS

Principal trading address: (Former) 30A Manchester Road, Chorltoncum-Hardy, Manchester, M21 9PH

At a General Meeting of the Members of the above-named Company, duly convened and held at 79 Tib Street, Manchester, M4 1LS on 17 August 2023 the following resolutions were duly passed, as a Special Resolution and as an Ordinary Resolution respectively:

"That the Company be wound up voluntarily and that *Paul Stanley* (IP No. 008123) and *Dean Watson* (IP No. 009661) both of Begbies Traynor (Central) LLP, 340 Deansgate, Manchester, M3 4LY be and hereby are appointed Joint Liquidators of the Company for the purpose of the voluntary winding-up, and any act required or authorised under any enactment to be done by the Joint Liquidators may be done by all or any one or more of the persons holding the office of Liquidator from time to time."

Any person who requires further information may contact the Joint Liquidators by telephone on 0161 837 1700. Alternatively enquiries can be made to Alex Coffey by email at Alex.Coffey@btguk.com. or by telephone on 0161 837 1700.

Andrew Hick, Chair 17 August 2023 Ag FJ53060

(4431815)

#### MILLENNIUM M.M. LTD

(Company Number 11502314)

Registered office: Flat 75 Peninsula House, O'Leary Street, Warrington, WA2 7SA

Principal trading address: Flat 75 Peninsula House, O'Leary Street, Warrington, WA2 7SA

Notice is given that by written resolutions, the sole member of the company passed a special resolution that the company be wound up voluntarily, and an ordinary resolution appointing the Joint Liquidators for the purposes of the winding-up. The requisite voting majority was received on 22 August 2023

Mr Marcel Mototolea, Director.

22 August 2023

Joint Liquidator's Name and Address: *Christopher Andersen* (IP No. 16070) of AABRS Limited, Langley House, Park Road, London, N2 8EY. Email: cha@aabrs.com. Telephone: 020 8444 3400.

Joint Liquidator's Name and Address: *Kevin McLeod* (IP No. 9438) of AABRS Limited, Langley House, Park Road, London, N2 8EY. Email: KM@AABRS.com. Telephone: 020 8444 3400.

For further information contact Anish Jadav at the offices of AABRS Limited on 020 8444 3400, or aj@AABRS.com.

23 August 2023

(4429209)

### MOTO DOCTOR LTD

(Company Number 11875442)

Registered office: 11 Courtleigh Gardens, NW11 9JX

Principal trading address: 11 Courtleigh Gardens, NW11 9JX

Notice is hereby given that the following resolutions were passed on 23 August 2023 as a special resolution and an ordinary resolution respectively:

"THAT the Company be wound up voluntarily" and

"THAT Nimish Patel and Hasib Howlader of Hudson Weir Limited, 58 Leman Street, London, E1 8EU be appointed as Joint Liquidators of the Company for the purpose of the voluntary winding up and that the Joint Liquidators may act jointly and severally in relation to the appointment."

Joint Liquidator: *Nimish Patel* (IP number 8679) of Hudson Weir Limited, 58 Leman Street, London, E1 8EU.

Joint Liquidator: *Hasib Howlader* (IP number 20350) of Hudson Weir Limited, 58 Leman Street, London, E1 8EU.

Date of Appointment: 23 August 2023

For further details contact Tiffany Truter on 020 7099 6086 or at tiffany@hudsonweir.co.uk Mr Jakub Kazanecki, Director (4431630)

NAYLOR GEORGE RECRUITMENT LIMITED

(Company Number 08835452)

Registered office: 14 London Road, Newark, Nottinghamshire, NG24 1TW

Principal trading address: 14 London Road, Newark, Nottinghamshire, NG24 1TW

At a General Meeting of the above named Company, duly convened and held at 14 London Road, Newark, Nottinghamshire, NG24 1TW on 24 August 2023, the following resolutions were duly passed as a Special Resolution and as an Ordinary Resolution respectively:-

"That the Company be wound up voluntarily and that *Matthew Dix* (IP No. 21492) of Wilkin Chapman LLP, Cartergate House, 26 Chantry Lane, Grimsby, DN31 2LJ be and is hereby appointed Liquidator of the Company for the purpose of such winding up."

Further details contact: Charlotte Houlsby, Tel: 01472 253931, or Email: charlotte.houlsby@wilkinchapman.co.uk.

Sarah Joanne Naylor, Chair

25 August 2023 Ag FJ53021

(4430170)

#### NEWPORT FABRICATIONS LTD

(Company Number 11939699)

("the Company") In Creditors' Voluntary Liquidation

Registered office: 42 Lytton Road, Barnet EN5 5BY

Principal trading address: 1 Vermont Place, Tongwell, Milton Keynes MK15 8JA

NOTICE IS GIVEN that at a General Meeting of the above-named Company, duly convened and held at Leigh Adams Limited, Maple House, High Street, Potters Bar, Hertfordshire EN6 5BS on 21 August 2023 at 9.30 am the following resolutions were passed as a Special resolution and Ordinary resolution respectively:-

1. "That the Company be wound up voluntarily" and

2. "That Paul Weber (IP No: 9400) and Martin Linton (IP No: 5998) of Leigh Adams Limited, Maple House, High Street, Potters Bar, Herts, EN6 5BS be appointed Joint Liquidators of the Company and that they act either jointly or separately."

For further details contact Zuzana Drengubiakova by email at Zuzana@leighadams.co.uk.

ANDREW SCOTT

Director

Dated this 21 day of August 2023

#### **NISAVA LTD**

(Company Number 12128026)

Trading Name: Kod Pirketi

Registered office: 47 High Street, Acton, London, W3 6ND

Principal trading address: 47 High Street, Acton, London, W3 6ND

Notice is hereby given that the following resolutions were passed on 24 August 2023 as a special resolution and an ordinary resolution respectively:

"That the Company cannot, by reason of its liabilities, continue its business, and it is advisable to wind up the same, and accordingly that the Company be wound up voluntarily and that *Farheen Qureshi* (IP No. 11392) of Parker Getty Limited, Devonshire House, 582 Honeypot Lane, Stanmore, Middx, HA7 1JS be appointed as Liquidator for the purposes of such voluntary winding up."

Further details contact: The Liquidator, Tel: 020 3475 3900, Email: meetings@parkergetty.co.uk. Alternative contact: E Popat.

Deyan Stoyanov, Director

29 August 2023

Ag FJ52886

(4430173)

(4431879)

#### PALLAS TRANSPORT SERVICES LIMITED

(Company Number 06971129)

Registered office: Unit 1, Bewick Main, Birtley, Chester-Le-Street, DH2 1BQ

Principal trading address: (Former) Unit 1, Bewick Main, Birtley, Chester-Le-Street, DH2 1BQ

At a General Meeting of the members of the above named Company, duly convened and held at Begbies Traynor (Central) LLP, Ground Floor, Portland House, 54 New Bridge Street West, Newcastle upon Tyne, NE1 8AP on 24 August 2023, the following resolutions were duly passed as a Special Resolution and as an Ordinary Resolution respectively:-

"That the Company be wound up voluntarily and that *Gillian Margaret Sayburn* (IP No. 10830) and *Andrew Little* (IP No. 9668) both of Begbies Traynor (Central) LLP, Ground Floor, Portland House, 54 New Bridge Street West, Newcastle upon Tyne, NE1 8AP be and hereby are appointed Joint Liquidators of the Company for the purpose of the voluntary winding-up, and any act required or authorised under any enactment to be done by the Joint Liquidators may be done by all or any one or more of the persons holding the office of Liquidator from time to time."

Any person who requires further information may contact the Joint Liquidator by telephone on 0191 2699820. Alternatively enquiries can be made to Charlene Carson by email at charlene.carson@btguk.com or by telephone on 0191 2699820.

Lorna Rose McCaffrey, Chair

24 August 2023 Ag FJ52999

(4430206)

#### PEDRICK'S ZERO WASTE LTD

(Company Number 11681568)

Trading Name: Pedrick's Zero Waste Shop

Registered office: 41 Croydon Road, Caterham, CR3 6PB

Principal trading address: 41 Croydon Road, Caterham, CR3 6PB

At a General Meeting of the Members of the above named Company, duly convened, and held at Satago Cottage, 360a Brighton Road, Croydon, CR2 6AL on 25 August 2023, the following Resolutions were passed by the Members as a Special Resolution and as an Ordinary Resolution:

"That the Company be wound up voluntarily, and that *Christopher Herron* (IP No. 8755) and *Nicola Jayne Fisher* (IP No. 9090) both of Herron Fisher, Satago Cottage, 360a Brighton Road, Croydon, CR2 6AL be appointed Joint Liquidators of the Company and that they be authorised to act either jointly or separately."

For further details contact: The Joint Liquidators, Tel: 020 8688 2100, Email: info@herronfisher.co.uk . Alternative contact: Emma Fisher. *Matthew Lee Baker,* Chair 30 August 2023

Ag FJ53008

(4430202)

#### **PIZZA JUNCTION 11 LTD** (Company Number 11680889)

Registered office: to be changed to Grove House, Meridians Cross, Ocean Village, Southampton SO14 3TJ, having previously been 43 Sherwood Road, Luton, Bedfordshire, LU4 8LQ

Principal trading address: Ground Floor, 152 Dunstable Road, Luton, LU1 1EW

Notice is hereby given that the following resolutions were passed on 23 August 2023 as a Special Resolution and Ordinary Resolutions respectively:

a) "That the Company be wound up voluntarily"

b) "That Matthew Hoy and Shane Biddlecombe of TruSolv Ltd, Grove House, Meridians Cross, Ocean Village, Southampton, SO14 3TJ be appointed as Joint Liquidators for the purpose of the voluntary winding up"

c) "That the Liquidators be authorised to act jointly and severally in the liquidation."

Joint Liquidator: *Matthew Hoy* (IP number 24550) of TruSolv Ltd, Grove House, Meridians Cross, Ocean Village, Southampton, SO14 3TJ.

Joint Liquidator: Shane Biddlecombe (IP number 9425) of TruSolv Ltd, Grove House, Meridians Cross, Ocean Village, Southampton, SO14 3TJ.

Date of Appointment: 23 August 2023

For further details contact Kay Lloyd on 0808 196 8676 or at kay.lloyd@trusolv.co.uk

Mohammed Shabbir - Director (4431666)

### POLAR MANUFACTURING LTD

(Company Number 07821431)

Registered office: Unit 35a Bowthorpe Industrial Estate, Barnard Road, Norwich, NR5 9JB

Principal trading address: N/A

The following Written Resolutions were circulated to all eligible members of the Company on 17 August 2023 and the Written Resolutions were duly passed on the effective date 25 August 2023; as a Special Resolution and as an Ordinary Resolution respectively:

"That the Company be wound up voluntarily and that *Jeremy Karr* (IP No. 9540) and *Simon Killick* (IP No. 23390) both of Begbies Traynor (London) LLP, 31st Floor, 40 Bank Street, London, E14 5NR be and hereby are appointed Joint Liquidators of the Company for the purpose of the voluntary winding-up, and any act required or authorised under any enactment to be done by the Joint Liquidators may be done by all or any one or more of the persons holding the office of liquidator from time to time."

Any person who requires further information may contact the Joint Liquidators by telephone on 020 7516 1500. Alternatively enquiries can be made to Elliot Bero by email at Elliot.Bero@btguk.com

*Mike Harris,* Chair 29 August 2023

Ag FJ52964

(4430194)

#### PREMIER FURNITURE WAREHOUSE LTD

(Company Number 09005895)

Registered office: 18-20 High Street, Shaftesbury, SP7 8JG and it is in the process of being changed to The Old Town Hall, 71 Christchurch Road, Ringwood, BH24 1DH

Principal trading address: 27b Fore Street, Trowbridge, Wiltshire, BA14 8EW

Notice is hereby given that the following resolutions were passed on 25 August 2023, as a special resolution and an ordinary resolution respectively:

"That the Company be wound up voluntarily and that *Simon Campbell* (IP No. 10150) and *David Meany* (IP No. 9453) both of Quantuma Advisory Limited, The Old Town Hall, 71 Christchurch Road, Ringwood, BH24 1DH be and are hereby appointed Joint Liquidators for the purpose of such winding up. Any act required or authorised under any enactment to be done by a liquidator may be done by one of them."

Further details contact: Kirsty Hayter, Tel: 01202 970 467 or Email: Kirsty.Hayter@quantuma.com

*Dermot Donaghy,* Director/Convener 25 August 2023

Ag FJ52900

#### PROJECTS UK (EAST ANGLIA) LIMITED

(Company Number 12091173)

Previous Name of Company: C & W Projects Limited

Registered office: The Pippins, Waterhouse Lane, Ardleigh, Essex, CO7 7TE

Principal trading address: Langham Barns Business Centre, Langham Lane, Langham Colchester CO4 5ZS

At a General Meeting of the above-named Company, duly convened and held at The Pippins, Waterhouse Lane, Ardleigh, Essex, CO7 7TE on 18 August 2023 at 10.00am the following resolutions were passed as a Special resolution and Ordinary resolution respectively:-

That the Company be wound up voluntarily and that Sonia Baigent (IP No 9494) of Assist Business Consulting Ltd, be appointed Liquidator of the Company.

Liquidator: *Sonia Baigent* (IP number 9494) of Assist Business Consulting Ltd, 5 Old Forge Road, Ashby Magna, Lutterworth, Leics, LE17 5NL.

Date of Appointment: 18 August 2023

For further details contact Sonia Baigent on 01455 560042 or at sonia@assistbc.com

Troy Blowers, Director (4432021)

(4430164)

### QUIRK PROJECT SUPPORT LTD

(Company Number 10602731)

Registered office: 63 Birtles Road, Macclesfield, SK10 3JG

Principal trading address: 63 Birtles Road, Macclesfield, SK10 3JG; Unit 12-14 North Quays Business Park, Atlantic Street, Altrincham, WA14 5EW

At a General Meeting of the above-named Company, duly convened, and held at The Old Bank, 187a Ashley Road, Hale, Cheshire, WA15 9SQ on 25 August 2023, at 9.00 am, the following resolutions were duly passed as a Special Resolution and as an Ordinary Resolution respectively:

"That the Company be wound up voluntarily, and that *Darren Brookes* (IP No. 9297) of Milner Boardman & Partners, The Old Bank, 187a Ashley Road, Hale, Cheshire, WA15 9SQ be appointed Liquidator of the Company."

For further details contact: Sally Baggs, E-mail: sallyb@milnerboardman.co.uk *Nicolas Wareham*, Director 25 August 2023

Ag FJ52934

(4430200)

### REAL FINANCIAL ADVICE LIMITED

(Company Number 12123343)

Registered office: Suite 20a, Off Liverpool Road, Empire Business Park, Burnley, Lancashire, BB12 6HH

Principal trading address: N/A

At a General Meeting of the above named Company duly convened and held at Suite 101 & 102, Empire Way Business Park, Liverpool Road, Burnley, BB12 6HH on 25 August 2023, the following Resolutions were duly passed as a Special and an Ordinary Resolution, respectively:

"That it has been resolved by Special Resolution that the Company be wound up voluntarily and that *Jonathan Mark Taylor* (IP No. 10570) of T H Financial Recovery, Suites 101 & 102, Empire Way Business Park, Liverpool Road, Burnley, BB12 6HH be appointed liquidator of the company for the purposes of the winding-up."

Further details contact: The Liquidator, Tel: 01282 332 222

Catherine Jane Humphreys, Director

28 August 2023

Ag FJ52870

(4430137)

### ROBERT OLIVIA CARS LTD.

(Company Number 10754141)

Registered office: 1 Matchingfield, Kelvedon Hatch, Brentwood, CM15 0XB

Principal trading address: 1 Matchingfield, Kelvedon Hatch, Brentwood, CM15 0XB

At a General Meeting of the members of the above-named company, duly convened and held on 24 August 2023, the following resolutions were duly passed as a Special Resolution and as an Ordinary Resolution:

"That the Company be wound up voluntarily and that *Richard Hunt* (IP No. 21772) of SFP Restructuring Limited, 9 Ensign House, Admirals Way, Marsh Wall, London, E14 9XQ be, and is hereby, appointed Liquidator of the Company for the purpose of the winding-up."

A decision confirming the appointment of the Liquidator was made by deemed consent on the same day.

Further details contact: Richard Hunt or Max Edmonds, Tel: 020 7538 2222.

Antony Robert Murphy, Chair 25 August 2023 Ag FJ52994

(4430199)

### SANGEETA HAIR SOLUTIONS LTD

(Company Number 07520921)

Registered office: Flat 24 Dorchester House, 228 Great Western Road, London, W11 1BE

Principal trading address: Ground Floor Rear, 70 North End Road, London, W14 9EP

At a General Meeting of the members of the above-named company, duly convened and held on 23 August 2023, the following resolutions were duly passed as a Special Resolution and as an Ordinary Resolution: "That the Company be wound up voluntarily and that *Richard Hunt* (IP No. 21772) of SFP Restructuring Limited, 9 Ensign House, Admirals Way, Marsh Wall, London, E14 9XQ be, and is hereby, appointed Liquidator of the Company for the purpose of the winding-up."

A decision confirming the appointment of the Liquidator was made by deemed consent on the same day.

Further details contact: Richard Hunt or Max Edmonds, Tel: 020 7538 2222.

*Hinnaa Mahboob,* Chair 24 August 2023 Ag FJ52976

(4430175)

### SHOWCASE DESIGN LIMITED

(Company Number 13422532)

Registered office: 1 Park Road, Hampton Wick, Kingston Upon Thames, KT1 4AS and it is in the process of being changed to 1st Floor, 21 Station Road, Watford, WD17 1AP

Principal trading address: 1 Park Road, Hampton Wick, Kingston Upon Thames, KT1 4AS

Notice is hereby given that the following resolutions were passed on 24 August 2023 as a special resolution and an ordinary resolution respectively:

"That the Company be wound up voluntarily and that *Nicholas Simmonds* (IP No. 9570) and *Chris Newell* (IP No. 13690) both of Quantuma Advisory Limited, 1st Floor, 21 Station Road, Watford, Herts, WD17 1AP be and are hereby appointed Joint Liquidators for the purpose of such winding up. Any act required or authorised under any enactment to be done by a liquidator may be done by one of them."

Further details contact: Richard Sutcliffe, Tel: 07469 311101 or Email: richard.sutcliffe@quantuma.com

Christopher Ezzard, Director

24 August 2023 Ag FJ52923

(4430211)

#### STONES AQUATICS LTD

(Company Number 11897786)

Trading Name: Stones Aquatics & Natureopolis

Registered office: 50 Maple Street, Sheerness, ME12 1XJ

Principal trading address: Natureopolis, The Hop Farm, Beltring, Tonbridge, Kent, TN12 6PY, Previously at Stones Garden Centre, Halfway Road, Minster-on-Sea, Kent, ME12 3AA

At a meeting of the members of the above named Company held at 50 Maple Street, Sheerness, ME12 1XJ on 29 August 2023, the following resolutions were passed as a special resolution and an ordinary resolution:

"That the Company be wound up voluntarily and that *Daniel Taylor* (IP No. 21050) of Fortis Insolvency Limited, 683-693 Wilmslow Road, Manchester, M20 6RE be and is hereby appointed Liquidator for the purpose of such winding up. Any act required or authorised under any enactment to be done by a liquidator may be done by him."

For further details contact: Daniel Taylor, Tel: 0161 694 9955. Alternative contact: Pat Clarke

*Darren Woolway,* Chair 29 August 2023 Ag FJ52920

(4430163)

### **T&L PLAY LIMITED**

(Company Number 12265016)

Registered office: The registered office of the Company will be changed to Sussex Innovation, One Croydon, 12-16 Addiscombe Road, Crovdon, CR0 0XT

Principal trading address: 21 East Street, Bromley BR1 1QE

The following resolutions were passed on 25 August 2023 by written resolution of the members of the Company as a special resolution and an ordinary resolution respectively:

"That the Company be wound up voluntarily and that *Matthew Reeds* (IP No. 28040) and *Paul Bailey* (IP No. 9428) both of Bailey Ahmad Business Recovery, Sussex Innovation, One Croydon, 12-16 Addiscombe Road, Croydon, CR0 0XT be appointed Joint Liquidators for the purposes of such voluntary winding up."

Contact details: The Liquidators, Tel: 0208 662 6070, Email: creditors@babr.co.uk Tanel Ali Ismet. Director 29 August 2023 Ag FJ52927 (4430203)

### THE GINGER DOG PUB COMPANY LTD

(Company Number 11454330)

Trading Name: The Rose and Crown & The Red Lion

Registered office: to be changed to Grove House, Meridians Cross, Ocean Village, Southampton SO14 3TJ, having previously been Rose & Crown Woodside Road, Winkfield, Windsor, SL4 2DP

Principal trading address: Rose & Crown Woodside Road, Winkfield, Windsor, SL4 2DP & 41 The Green, Woodburn Green, GP10 0EU

Notice is hereby given that the following resolutions were passed on 23 August 2023 as a Special Resolution and Ordinary Resolutions respectively:

a) "That the Company be wound up voluntarily"

b) "That Shane Biddlecombe and Matthew Hoy of TruSolv Ltd, Grove House, Meridians Cross, Ocean Village, Southampton, SO14 3TJ be appointed as Joint Liquidators for the purpose of the voluntary winding up."

c) "That the Liquidators be authorised to act jointly and severally in the liquidation

Joint Liquidator: Shane Biddlecombe (IP number 9425) of TruSolv Ltd, Grove House, Meridians Cross, Ocean Village, Southampton, SO14 3TJ.

Joint Liquidator: Matthew Hoy (IP number 24550) of TruSolv Ltd, Grove House, Meridians Cross, Ocean Village, Southampton, SO14 3TJ.

Date of Appointment: 23 August 2023

For further details contact Kay Lloyd on 0808 196 8676 or at kay.lloyd@trusolv.co.uk Jack Stoner - Director

(4431704)

#### THE SPA & BEAUTY CLINIC LTD

(Company Number 11669244)

Registered office: The Chapel, Bridge Street, Driffield, YO25 6DA

Principal trading address: 1 & 3, Red Lion St, Goole DN14 6BX

At a general meeting of the above named company duly convened and held at The Chapel, Bridge Street, Driffield, YO25 6DA, on 24 August 2023 the following resolutions were passed: No 1 as an special resolution and No 2 as an ordinary resolution:-

1. That it has been proved to the satisfaction of this meeting that the company cannot, by reason of its liabilities, continue its business and that it is advisable to wind up the same and accordingly that the company be wound up voluntarily.

2. That John William Butler and Andrew James Nichols of Redman Nichols Butler, The Chapel, Bridge Street, Driffield, East Yorkshire, YO25 6DA, be and are hereby appointed Joint Liquidators of the company for the purpose of the voluntary winding up and that the Joint Liquidators may act jointly or severally in all matters relating to the conduct of the liquidation of the Company.

Joint Liquidator: John William Butler (IP number 9591) of Redman Nichols Butler, The Chapel, Bridge Street, Driffield, East Yorkshire, YO25 6DA

Joint Liquidator: Andrew James Nichols (IP number 8367) of Redman Nichols Butler, The Chapel, Bridge Street, Driffield, East Yorkshire, YO25 6DA

Date of Appointment: 24 August 2023

For further details contact: 01377 257788 Anne Goodare -Chair

# TICKBOX TRAINING LIMITED

(Company Number 10054075)

Registered office: Dunheved 10 Park Place Cornwall PL27 7EA Principal trading address: Dunheved 10 Park Place Cornwall PL27

7EA At a General Meeting of the members of the above named company, duly convened and held at Balliol House, Southernhay Gardens, Exeter EX1 1NP on 24 August 2023 the following resolutions ware duly passed; as a Special Resolution and as an Ordinary Resolution respectively:

1. That the company be wound up voluntarily.

2. That Scott Kippax and Julie Anne Palmer both of Begbies Traynor (Central) LLP, Balliol House Southernhay Gardens Exeter EX1 1NP be and hereby are appointed Joint Liquidators of the Company for the purpose of the voluntary winding up, and any act required or authorised under any enactment to be done by the Joint Liquidators may be done by all or any one or more of the persons holding the office of the Liquidator from time to time.

Joint Liquidator: Scott Kippax (IP number 26030) of Begbies Traynor (Central) LLP, Balliol House Southernhay Gardens Exeter EX1 1NP.

Joint Liquidator: Julie Anne Palmer (IP number 8835) of Begbies Traynor (Central) LLP, Units 1-3 Hilltop Business Park Devizes Road Salisbury Wiltshire SP3 4UF.

Date of Appointment: 24 August 2023

For further details contact Nikita Iggulden on 01392 260800 or at Nikita.lggulden@btguk.com Grant Crawford, Chair

(4431911)

#### TRENDY TRAINERS LIMITED

(Company Number 07197292)

Registered office: 82a High Street Hertfordshire EN6 5AB

Principal trading address: 82a High Street Hertfordshire EN6 5AB

At a General Meeting of the members of the above named company, duly convened and held at Balliol House, Southernhay Gardens, Exeter EX1 1NP on 23 August 2023 the following resolutions ware duly passed; as a Special Resolution and as an Ordinary Resolution respectively:

1. That the company be wound up voluntarily.

2. That Scott Kippax and Julie Anne Palmer both of Begbies Traynor (Central) LLP, Balliol House Southernhay Gardens Exeter EX1 1NP be and hereby are appointed Joint Liquidators of the Company for the purpose of the voluntary winding up, and any act required or authorised under any enactment to be done by the Joint Liquidators may be done by all or any one or more of the persons holding the office of the Liquidator from time to time.

Joint Liquidator: Scott Kippax (IP number 26030 ) of Begbies Traynor (Central) LLP, Balliol House Southernhay Gardens Exeter EX1 1NP .

Joint Liquidator: Julie Anne Palmer (IP number 8835) of Begbies Traynor (Central) LLP, Units 1-3 Hilltop Business Park Devizes Road Salisbury Wiltshire SP3 4UF .

Date of Appointment: 23 August 2023

For further details contact Nikita Iggulden on 01392 260800 or at Nikita.lggulden@btguk.com (4431900)

Hassan Saykal, Chair

#### TRISTAN THOMSON PHOTOGRAPHY LTD

(Company Number 11068924)

Registered office: Wits End, Fulstone Hall Lane, Holmfirth, HD9 7DL

Principal trading address: Wits End, Fulstone Hall Lane, Holmfirth, HD9 7DL

Notice is hereby given that the following resolutions were passed on 29 August 2023 as a Special Resolution and an Ordinary Resolution respectively:

"That the Company be wound up voluntarily and that Nicola Baker (IP No. 15852) of Rushtons Insolvency Limited, 6 Festival Building, Ashley Lane, Saltaire, BD17 7DQ be appointed Liquidator for the purposes of the voluntary winding up of the Company."

Further details contact: Nicola Baker, Tel: 01274 598585. Alternative contact: Rhys Wordsworth Email: rwordsworth@rushtonsifs.co.uk Tristan Thompson, Director

29 August 2023

Ag FJ52913

(4431634)

(4430221)

#### TURCO'S HOSPITALITY LIMITED

#### (Company Number 10830373)

Trading Name: Al Dente Pasta Bar and Kitchen

Registered office: Yare House, 62-64 Thorpe Road, Norwich, NR1 1RY

Principal trading address: 25B St Giles Street, Norwich, NR1 2JN At a General Meeting of the above named Company duly convened and held at 9.45 am on 24 August 2023, the following Resolutions were passed as a Special Resolution and as an Ordinary Resolution:

(4430171)

"That the Company be wound up voluntarily and that *Nicholas Cusack* (IP No. 17490) of Parker Andrews Limited, 5th Floor, The Union Building, 51-59 Rose Lane, Norwich, NR1 1BY be and is hereby appointed Liquidator for the purpose of such winding up." Further details contact: Natalie Clarke, Email: natalie.clark@parkerandrews.co.uk *Adriano Turco,* Chair

24 August 2023 Ag FJ52885

(4430172)

UNFAIR ADVANTAGE ENTERPRISES LIMITED

(Company Number 10049902)

Registered office: 8 Domingo Street, London, EC1Y 0TA

Principal trading address: 8 Domingo Street, London, EC1Y 0TA

Notice is hereby given that the following resolutions were passed on 22 August 2023 as a special resolution and an ordinary resolution respectively:

"THAT the Company be wound up voluntarily" and

"THAT Nimish Patel and Hasib Howlader of Hudson Weir Limited, 58 Leman Street, London, E1 8EU be appointed as Joint Liquidators of the Company for the purpose of the voluntary winding up and that the Joint Liquidators may act jointly and severally in relation to the appointment."

Joint Liquidator: *Nimish Patel* (IP number 8679) of Hudson Weir Limited, 58 Leman Street, London, E1 8EU.

Joint Liquidator: *Hasib Howlader* (IP number 20350) of Hudson Weir Limited, 58 Leman Street, London, E1 8EU.

Date of Appointment: 22 August 2023

For further details contact Jinjin Qi on 020 7099 6086 or at jinjin@hudsonweir.co.uk

Mr Ben James McCormack, Director (4431624)

#### UNFITTED LTD

(Company Number 07499222)

Registered office: The Old Rectory, Main Street, Glenfield, Leicester, LE3 8DG (Formerly) Hope House, High Street, Moreton-In-Marsh, GL56 0LH

Principal trading address: Hope House, High Street, Moreton-In-Marsh, GL56 0LH

Notice is given that at a General Meeting of the above named Company, duly convened and held at Hope House, High Street, Moreton-In-Marsh, GL56 0LH on 19 July 2023 at 10.30 am the following resolutions were passed as a Special Resolution and Ordinary Resolution respectively:-

"That the Company be wound up voluntarily and that *Gavin Bates* (IP No. 8983) of Smart Business Recovery Limited, The Old Rectory, Main Street, Glenfield, Leicester, LE3 8DG be appointed Liquidator of the Company."

Further details contact: Sharon Hill, Tel: 0116 2325117, Email: sharon@smartbusinessrecovery.co.uk.

R Rench, Director

19 July 2023

Ag FJ52905

(4430218)

### WILL'S JOINERY LIMITED

(Company Number 11080743) Registered office: 7 Davenport Road, Stockport, Cheshire SK7 4EZ

Principal trading address: n/a

Notice is hereby given that the following resolutions were passed on 25 August 2023 as a Special Resolution and an Ordinary Resolution respectively:

"That the Company be wound up voluntarily and that *Daniel Ormerod* (IP No. 26930) and *Martin Maloney* (IP No. 9628) both of Leonard Curtis, Leonard Curtis House, Elms Square, Bury New Road, Whitefield, Greater Manchester, M45 7TA be and are hereby appointed Joint Liquidators of the Company for the purposes of the winding up of the Company and that any act required or authorised under any enactment to be done by the Joint Liquidators may be done by all or any one or more of the persons for the time being holding such office."

For further details contact: The Joint Liquidator, Tel: 0161 413 0930 Alternative contact: Hasan Patel. .

*Graham Boshell,* Director 25 August 2023

# Liquidation by the Court

### **APPOINTMENT OF LIQUIDATORS**

In the High Court of Justice

Ag FJ52971

Court Number: 002126 of 2023

COLOURCOAT LTD

(Company Number 10405998) Trading Name: Ecosolve

Registered office: 77 Bohemia Road, St. Leonards-On-Sea, TN37 6RJ Principal trading address: Room 327, Century House, 100 Menzies Road, Hastings, TN38 9BB

NOTICE IS HEREBY GIVEN pursuant to Rule 7.59 of the Insolvency (England & Wales) Rules 2016 that Stratford Hamilton and James Sleight were appointed Joint Liquidators of the above named company on 13 July 2023.

Joint Liquidator: *Stratford Hamilton* (IP number 12212) of PKF GM, 15 Westferry Circus, Canary Wharf, London, E14 4HD.

Joint Liquidator: James Sleight (IP number 9648) of PKF GM, 1 Park Row, Leeds, LS1 5HN .

Date of Appointment: 13 July 2023

For further details contact Michelle Dean on 020 7495 1100 or at michelle.dean@pkfgm.co.uk (4431136)

In the Manchester District Registry Court Number: CR-2023-000675

DETRAFFORD (MBJ) LIMITED

(Company Number 10179260)

Previous Name of Company: De Trafford Estates No.1 Castlefield Block C Ltd

Registered office: St Georges Gardens Spinners Way, Castlefield, Manchester, M15  $4\mathrm{UZ}$ 

Principal trading address: N/A

Pursuant to the Insolvency Rules, we, *Paul Atkinson* (IP No. 9314) and *Glyn Mummery* (IP No. 8996) both of FRP Advisory Trading Limited, Jupiter House, Warley Hill Business Park, The Drive, Brentwood, Essex, CM13 3BE hereby give notice that we were appointed Joint Liquidators of the above named Company on 18 August 2023 by Secretary of State.

Creditors who have not yet proved their debt must forward their proof of debt form to us at the address shown above.

Further details contact: The Joint Liquidators, Email: cp.brentwood@frpadvisory.com. Alternative contact: Saskia Chalmers *Paul Atkinson,* Joint Liqudator

18 August 2023

Ag FJ52966

(4430264)

### **MEETINGS OF CREDITORS**

# VIRTUAL MEETING (COMPANY)

THE INSOLVENCY ACT 1986 In the High Court Of Justice No 003231 of 2022

#### NORTHFIELD FABRICATION LTD

(Company Number 12132404)

Registered office: Queen Street Chambers, Queen Street, SHEFFIELD, S1 1WR

A General Meeting Of Creditors is to take place on : 13th September 2023 at 2:00 pm

Venue: Microsoft Team Platform - Should a creditor wish to participate please contact the official receiver for further details via Reception.leeds@insolvency.gov.uk

Meetings summoned by: Leeds Official Receiver

The Purpose of the Meetings: To appoint a liquidator in place of the Official Receiver. A virtual meeting of creditors scheduled for 13th September at 2pm has been gazetted as required under the Insolvency Act 1986.

In order to be entitled to vote at the meeting, creditors must lodge proxies and any previously un-lodged proofs by 12th September at 4pm at the Official Receiver's address stated below.

It is recommended that you send scanned documents via email if possible or allow a few extra days in case of postal delays.. Official Receiver: Lyndsey Burgess Capacity: Liquidatore Official Receiver Date of Appointment: 22nd February 2023 (4431344)

#### **PETITIONS TO WIND-UP**

In the High Court of Justice CR-2023-MAN-000947 In the matter of X400 LIMITED Trading As: X400 Limited.

and in the Matter of the Insolvency Act 1986,

A Petition to wind up the above-named company of X400 Limited (10315163) of C/o Mazars Llp, 6 Dominus Way, Meridian Business Park, Leicester LE19 1RP, whose nature of business is other business support service activities not elsewhere classified, presented on Wednesday 26 July 2023, at 10:00 by MAZARS LLP, of 6 Dominus Way, Meridian Business Park, Leicester LE19 1RP claiming to be a Creditor of the Company, will be heard at the Business and Property Court in Manchester, 1st Floor, 5 New York Street, Manchester M60 9DJ United Kingdom on Tuesday 12 September 2023, at 10:00 (or as soon thereafter as the Petition can be heard).

Any person intending to appear on the hearing of the Petition (whether to support or oppose it) must give notice of intention to do so to the Petitioners or to their Solicitor in accordance with Rule 7.14 by 16:00 hours on Monday 11 September 2023

The Petitioner's Solicitor is Martin Cork, FREETHS LLP, 1st Floor, 5 New York Street, Manchester M1 4JB,, Telephone: 07921 022478, Email: martin.cork@freeths.co.uk (Reference number: MC.2114758.266.) (4431706)

In the High Court of Justice CR-2023-004166 In the matter of CASTILO UK DEVELOPMENT LTD

Trading As: Castilo UK Development Ltd,

and in the Matter of the Insolvency Act 1986,

A Petition to wind up the above-named company of Castilo UK Development Ltd (11568572) of Flat 3, Alder Lodge, London SW6 6NP, previously of Flat 3, Alder Lodge, London SW6 6NP, whose nature of business is Construction of domestic buildings, presented on Wednesday 02 August 2023, at 10:00 by BRITISH GAS TRADING LIMITED, of Millstream, Maidenhead Road, Windsor SL4 5GD claiming to be a Creditor of the Company, will be heard at the The High Court of Justice Business and Property Courts, Insolvency and Companies List, Rolls Building, 7 Fetter Lane, London EC4A 1NL on Wednesday 13 September 2023, at 10:30 (or as soon thereafter as the Petition can be heard).

Any person intending to appear on the hearing of the Petition (whether to support or oppose it) must give notice of intention to do so to the Petitioners or to their Solicitor in accordance with Rule 7.14 by 16:00 hours on Tuesday 12 September 2023

The Petitioner's Solicitor is Womble Bond Dickinson (UK) LLP, WOMBLE BOND DICKINSON (UK) LLP, Ballard House, West Hoe Road, Plymouth PL1 3AE,, Telephone: 01752 677 843, Email: emma.hatherell@wbd-uk.com (Reference number: ELC3/375936.04738.)

Thursday 31 August 2023

(4431903)

In the High Court of Justice CR-2023-003908 In the matter of BH MADE LTD Trading As: BH MADE LTD,

and in the Matter of the Insolvency Act 1986,

A Petition to wind up the above-named company of BH MADE LTD (13248709) of Office 9 Dalton House, 60 Windsor Avenue, London SW19 2RR, whose nature of business is 46470 - Wholesale of furniture, carpets and lighting equipment, presented on Wednesday 26 July 2023, at 15:15 by H.A UPHOLSTERY LTD, of Unit 4k, Glenfield Park Park, Philips Road, Blackburn, Lancs BB1 5PF claiming to be a Creditor of the Company, will be heard at the Business and Property Court in England & Wales, Rolls Building, Fetter Lane, London EC4A 1NL on Wednesday 13 September 2023, at 10:30 (or as soon thereafter as the Petition can be heard).

Any person intending to appear on the hearing of the Petition (whether to support or oppose it) must give notice of intention to do so to the Petitioners or to their Solicitor in accordance with Rule 7.14 by 16:00 hours on Tuesday 12 September 2023

The Petitioner's Solicitor is Coltman Warner Cranston, Unit 3, Coventry Innovation Village, Coventry CV1 2TL,, Telephone: 02476627262, Email: ddavoile@coltmanco.com (Reference number: C0201681.) Wednesday 30 August 2023

(4430433)

In the High Court of Justice No. CR-2023-LDS-000537 of 2023 In the matter of R FARRELL PLANT HIRE LIMITED

Trading As: R Farrell Plant Hire Limited,

and in the Matter of the Insolvency Act 1986,

A Petition to wind up the above-named company of R Farrell Plant Hire Limited (06730659) of 4b Beulah Place, Luddendenfoot, Halifax HX2 6BA, 4b Beulah Place, Luddendenfoot, Halifax HX2 6BA (where service of the petition was effected), whose nature of business is 08110-Quarrying of ornamental and building stone, limestone, gypsum, chalk and slate , presented on Monday 10 July 2023, at 10:00 by MABEY HIRE LIMITED , of Scout Hill, Ravensthorpe, Dewsbury WF13 3EJ claiming to be a Creditor of the Company, will be heard at the Business & Property Courts in Leeds, 1 Oxford Row, Leeds LS1 3BG on Tuesday 12 September 2023, at 10:00 (or as soon thereafter as the Petition can be heard).

Any person intending to appear on the hearing of the Petition (whether to support or oppose it) must give notice of intention to do so to the Petitioners or to their Solicitor in accordance with Rule 7.14 by 16:00 hours on Monday 11 September 2023

The Petitioner's Solicitor is Andrew Langton, LCF LAW LTD , 33 Park Place, Leeds LS1 2RY,, Telephone: 0113 4873936, Email: andrew.langton@lcf.co.uk (Reference number: AL/MAB0001/11967.)

(4430443)

In the High Court of Justice No. 004271 of 2023 In the matter of MEGMARSTER LTD Trading As: Megmarster Ltd,

and in the Matter of the Insolvency Act 1986,

A Petition to wind up the above-named company of Megmarster Ltd (10612499) of Donde 37 - 39, Honor Oak Park, Lewisha, London SE23 1DZ, whose nature of business is 56101 - Licensed restaurants, presented on Thursday 03 August 2023, at 09:30 by LONDON BOROUGH OF LEWISHAM, of 5th Floor, Laurence House, Catford, London SE6 4RU claiming to be a Creditor of the Company, will be heard at the High Court of Justice, Rolls Building, Fetter Lane, London EC4A 1NL on Wednesday 20 September 2023, at 10:30 (or as soon thereafter as the Petition can be heard).

Any person intending to appear on the hearing of the Petition (whether to support or oppose it) must give notice of intention to do so to the Petitioners or to their Solicitor in accordance with Rule 7.14 by 16:00 hours on Tuesday 19 September 2023

The Petitioner's Solicitor is Mr Paul Bowden, WILKIN CHAPMAN LLP, Cartergate House, 26 Chantry Lane, Grimsby DN31 2LJ,, Telephone: 01472265982, Email: paul.bowden@wilkinchapman.co.uk (4430466)

In the High Court of Justice CR-2023-CR-2023-004378

In the matter of STRADBROKE VILLAGE LIMITED

Trading As: STRADBROKE VILLAGE LIMITED,

and in the Matter of the Insolvency Act 1986,

A Petition to wind up the above-named company of STRADBROKE VILLAGE LIMITED (11413720) of 83 Ashurst Drive, Ilford, Essex IG6 1EP ENGLAND, 83 Ashurst Drive, Ilford, Essex IG6 1EP ENGLAND (where service of the petition was effected), whose nature of business is 68100 - Buying and selling of own real estate, presented on Tuesday 08 August 2023, at 18:00 by DDBG PROPERTIES LIMITED, of Second Floor, 3 Liverpool Gardens, Worthing BN11 1TF claiming to be a Creditor of the Company, will be heard at the HIGH COURT OF JUSTICE, BUSINESS AND PROPERTY COURTS OF ENGLAND AND WALES, 7 Rolls Building, Fetter Lane, London EC4A 1NL on Wednesday 04 October 2023, at 10:30 (or as soon thereafter as the Petition can be heard).

Any person intending to appear on the hearing of the Petition (whether to support or oppose it) must give notice of intention to do so to the Petitioners or to their Solicitor in accordance with Rule 7.14 by 16:00 hours on Tuesday 03 October 2023

The Petitioner's Solicitor is Brian Kilcoyne and Paul Hayes, LEWIS SILKIN LLP, Arbor, 255 Blackfriars Road, London SE1 9AX, Telephone: 020 7074 8119, Email: paul.hayes@lewissilkin.com (Reference number: PH/122477/1.) Tuesday 22 August 2023

(4426770)

In the High Court of Justice CR-2023-004463

#### In the matter of COMPLEX PAINTS LIMITED

Trading As: Complex Paints Limited,

and in the Matter of the Insolvency Act 1986,

A Petition to wind up the above-named company of Complex Paints Limited (NI638440) of Room 1, 39a Old Coach Road, Hillsborough, Lisburn BT26 6PB Northern Ireland, Room 1, 39a Old Coach Road, Hillsborough, Lisburn BT26 6PB Northern Ireland (where service of the petition was effected), whose nature of business is wholesale of chemical products, presented on Monday 14 August 2023, at 11:39 by GC FACTORING LIMITED, of 2 London Square, Cross Lanes, Guildford, Surrey GU1 1UN claiming to be a Creditor of the Company, will be heard at the High Court of Justice Business and Property Courts of England and Wales, Rolls Building, 7 Fetter Lane, London EC4A 1NL on Wednesday 04 October 2023, at 10:30 (or as soon thereafter as the Petition can be heard).

Any person intending to appear on the hearing of the Petition (whether to support or oppose it) must give notice of intention to do so to the Petitioners or to their Solicitor in accordance with Rule 7.14 by 16:00 hours on Tuesday 03 October 2023

The Petitioner's Solicitor is Ben Ashworth, PDT SOLICITORS LLP, Pdt Solicitors Llp, 36-48 Queen Street, Horsham, West Sussex RH13 5AD U.K,, Telephone: 01403 831255, Email: bashworth@pdt.co.uk (Reference number: BA/GRE0027.0042.) (4432010)

#### In the HIGH COURT OF JUSTICE

BUSINESS PROPERTY COURT IN Newcastle District Registry No CR-2023-NCL-000146 of 2023

#### In the Matter of ACM SCAFFOLDING LIMITED

(Company Number 12025626)

and in the Matter of the INSOLVENCY ACT 1986

A Petition to wind up the above-named company (registered no: 12025626 ) of 302 Chester Road, Sutton Coldfield, B73 5BZ presented on 3 July 2023, by DIRECT HOIST LIMITED of Green Workshop, Cleatham Road Business Park, Cleatham Road, Kirton in Lindsey, Gainsborough LinoInshire, DN21 4JR (the Petitioner) claiming to be a creditor of the company, will be heard at Newcastle District Registry Barras Bridge, Newcastle Upon Tyne, NE1 8QF

Date: 12 September 2023

Time: 10:00am (or as soon thereafter as the petition can be heard)

Any person intending to appear on the hearing of the petition (whether to support or oppose it) must give notice of that intention in accordance with rule 7.14 by 16.00 hours on 11/09/2023.

The petitioner's solicitor are Silverback Commercial Law Services Limited, 8 Regents Court, Far Moor Lane, Redditch, B98 0SD; Ref: TS100460/006

Dated: 29/08/2023

(4431841)

In the BUSINESS AND PROPERTY COURTS IN MANCHESTER INSOLVENCY AND COMPANIES LIST (ChD) No CR-2023-MAN-000930 of 2023

# In the Matter of ALDWICK ELECTRICAL LTD

(Company Number 12715617)

#### and in the Matter of the INSOLVENCY ACT 1986

A Petition to wind up the Company presented on 26 July 2023 by EDMUNDSON ELECTRICAL LIMITED of South East Region Credit Office Hanover Place, 8 Church Road Tunbridge Wells Kent TN1 1JP claiming to be a Creditor of the Company of 26 Lion Road Bognor Regis PO21 3LA will be heard at the Manchester Business & Property Court, The Civil Justice Centre 1 Bridge Street West Manchester M60 9DJ at 10:00 am on Tuesday 12 September 2023 (or as soon thereafter as the Petition can be heard).

Any person intending to appear on the hearing of the Petition (whether to support or oppose it) must give notice of intention to do so to the Petitioner or its Solicitor in accordance with Rule 4.16 by 16.00 hours on Monday 11 September 2023.

The Petitioner's Solicitor is Pannone Corporate LLP, 378-380 Deansgate Manchester M3 4LY. Ref: INSOLVENCY/319510. (4431842)

#### In the BUSINESS AND PROPERTY COURTS IN MANCHESTER INSOLVENCY AND COMPANIES LIST (ChD) No CR-2023-

MAN-000943 of 2023

#### In the Matter of FARRAGO DESIGN LTD (Company Number 08858609)

and in the Matter of the INSOLVENCY ACT 1986

A Petition to wind up the Company presented on 26 July 2023 by BOELS RENTAL LIMITED of Dr.Nolenslaan 140 Postfach 666 6130 AR Sittard HOLLAND claiming to be a Creditor of the Company of 1 Alvin Street Gloucester Gloucestershire GL1 3EJ will be heard at the Manchester Business & Property Court, The Civil Justice Centre, 1 Bridge Street West, Manchester M60 9DJ at 10:00 am on Tuesday 12 September 2023 (or as soon thereafter as the Petition can be heard).

Any person intending to appear on the hearing of the Petition (whether to support or oppose it) must give notice of intention to do so to the Petitioner or its Solicitor in accordance with Rule 4.16 by 16.00 hours on Monday 11 September 2023.

The Petitioner's Solicitor is Pannone Corporate LLP, 378-380 Deansgate Manchester M3 4LY. Ref: INSOLVENCY/321189. (4431843)

#### In the HIGH COURT OF JUSTICE

BUSINESS AND PROPERTY COURTS IN LEEDS INSOLVENCY AND COMPANIES LIST (ChD) No CR-2023-LDS-000642 of 2023

In the Matter of HERALD NEWS (UK) LTD

(Company Number 12142810) and in the Matter of the INSOLVENCY ACT 1986

A PETITION to wind up the above named company whose Registered

Office is at 7 Hill Street, Haverfordwest, Pembrokeshire, United Kingdom, SA61 1QQ presented on the 27 July 2023 by NEWSQUEST MEDIA GROUP LIMITED whose Registered Office is at 1st Floor, Chartist Tower, Upper dock Street, Newport, Wales NP20 1DW claiming to be the creditor of the company will be heard at The High Court of Justice, Business and Property Courts in Leeds, Insolvency and Companies List (ChD), The Courthouse, 1 Oxford Row, Leeds LS1 3BG on the 12th day of September 2023 at 10:00 hours (or as soon hereafter as the Petition can be heard)

Any person intending to appear on the hearing of the Petition (whether to support or oppose it) must give Notice of Intention to do so to the Petitioner or its solicitor in accordance with Rule 7.14 of The Insolvency Rules 2016 by 1600 hours 11 September 2023.

The Petitioner's solicitors are Harwood & Co , 8E Josephs Well Hanover Walk Leeds LS3 1AB. Tel: 0113 245 7027, Fax: 0113 242 1329, Email: carol.hagan@harwood.law

(4431827)

#### In the HIGH COURT OF JUSTICE

Dated this 25th day of August 2023

BUSINESS AND PROPERTY COURTS IN LEEDS INSOLVENCY AND COMPANIES LIST (ChD) No CR-2023-LDS-000661 of 2023 In the Matter of LOAD2GO LIMITED

### (Company Number 12259726)

and in the Matter of the INSOLVENCY ACT 1986

A Petition to wind up the above-named Company (registered no. 12259726) of Unit 7 Epic Park, Halesfield 6, Telford, Shropshire, England, TF7 4BF, presented on 7 August 2023 by ZIEGLER UK LIMITED whose registered office is at North 2, North Sea Crossing, Stanford-Le-Hope, Essex, England, SS17 9ER (the Petitioner), claiming to be a Creditor of the Company, will be heard at the Business and Property Courts in Leeds, Leeds Combined Court Centre The Courthouse, 1 Oxford Row, Leeds, LS1 3BG on Tuesday 12 September 2023 at 10:00am (or as soon thereafter as the Petition can be heard)

Any person intending to appear on the hearing of the Petition (whether to support or oppose it) must give notice of intention to do so to the Petitioner or its Solicitor in accordance with Rule 7.14 by 16:00 hours on Monday 11 September 2023.

The Petitioner's Solicitor is DWF Law LLP, Bridgewater Place, Water Lane, Leeds LS11 5DY. Tel: 07599 557 244. Email: Duraid.Al-Musaied@dwf.law (Ref: DAM/D1146841). Dated: 29 August 2023

(4431844)

In the Liverpool District Registry No CR-2023-LIV-000261 of 2023 In the Matter of PAD + LIMITED (Company Number 13076805)

and in the Matter of the INSOLVENCY ACT 1986

A Petition to wind up the above-named company of PAD + Limited of R/O Craven House, Ground Floor, 40-44 Uxbridge Road, Ealing, London, W5 2BS (R/N 13076805) presented on 9 AUGUST 2023 by SITE EQUIPMENT LIMITED of St Andrews Road, Avonmouth, Bristol, BS1 9HS, claiming to be a creditor of the company, will be heard at Liverpool District Registry, 35 Vernon Street, Liverpool, L2 2BX

Date: TUESDAY 3 OCTOBER 2023

Time: 10:30 AM hours (or as soon thereafter as the petition can be heard)

Any person intending to appear on the hearing of the petition (whether to support or oppose it) must give notice of intention to do so to the petitioner or his/its solicitor in accordance with Rule 4.16 by 16.00 hours on 2 October 2023.

The Petitioner's Solicitor is Brabners LLP (Kerry Davies), Horton House, Exchange Flags, Liverpool, L2 3YL.

Dated 25 AUGUST 2023 (4431830)

In the High Court of Justice (Business and Property Courts of England and Wales)

Insolvency and Companies List (ChD) No CR-2023-004319 of 2023 In the Matter of ANATOLIA MK100 LTD

(Company Number 11028205)

and in the Matter of the INSOLVENCY ACT 1986

A Petition to wind up the above-named Company, ANATOLIA MK100 LTD (CRN: 11028205 ), of 3 Blundells Road, Bradville, Milton Keynes, MK13 7HA presented on 09 August 2023 by BRITISH GAS TRADING LIMITED T/A CENTRICA BUSINESS SOLUTIONS of Millstream Maidenhead Road, Windsor, Berkshire, SL4 5GD, (the Petitioner), claiming to be a creditor of the Company will be heard at the Royal Courts of Justice, 7 Rolls Buildings, Fetter Lane, London EC4A 1NL, on 20/09/2023, at 10:30AM hours (or as soon thereafter as the Petition can be heard).

Any persons intending to appear on the hearing of the Petition (whether to support or oppose it) must give notice of intention to do so to the Petitioner's solicitors in accordance with Rule 7.14 by 16:00 hours on 19 September 2023.

The Petitioner's solicitors are Flint Bishop LLP, Pinnacle Building, 2 Prospect Place, Pride Park, Derby DE24 8HG, telephone 01332 947752, email sam.brace@flintbishop.co.uk. (Ref M0373254/SXB)

(4431828)

In the High Court of Justice (Business and Property Courts of England and Wales)

Insolvency and Companies List (ChD) No CR-2023-004318 of 2023 In the Matter of EAT N GRILL LTD

(Company Number 13025248)

and in the Matter of the INSOLVENCY ACT 1986

A Petition to wind up the above-named Company, EAT N GRILL LTD (CRN: 13025248), of 235, ECCLES NEW ROAD, SALFORD, M5 4QG presented on 09 August 2023 by BRITISH GAS TRADING LIMITED T/A CENTRICA BUSINESS SOLUTIONS of Millstream Maidenhead Road, Windsor, Berkshire, SL4 5GD, (the Petitioner), claiming to be a creditor of the Company will be heard at the Royal Courts of Justice. 7 Rolls Buildings, Fetter Lane, London EC4A 1NL, on 20/09/2023, at 10:30AM hours (or as soon thereafter as the Petition can be heard).

Any persons intending to appear on the hearing of the Petition (whether to support or oppose it) must give notice of intention to do so to the Petitioner's solicitors in accordance with Rule 7.14 by 16:00 hours on 19 September 2023.

The Petitioner's solicitors are Flint Bishop LLP, Pinnacle Building, 2 Prospect Place, Pride Park, Derby DE24 8HG, telephone 01332 947752. (Ref M0380986/SXB). (4431826) In the HIGH COURT OF JUSTICE, BUSINESS AND PROPERTY COURTS OF ENGLAND AND WALES, INSOLVENCY AND COMPANIES LIST (ChD)

#### Court Number: CR-2023-004304 In the Matter of JCS (SOUTHERN) LIMITED (Company Number 08608221)

and in the Matter of the INSOLVENCY ACT 1986

A Petition to wind up the above-named Company of 7 Bournemouth Road, Chandler's Ford, Eastleigh, SO53 3DA presented on 8 August 2023 by Tesco Stores Limited of Tesco House, Shire Park, Kestrel Way, Welwyn Garden City AL7 1GA, claiming to be a creditor of the company, will be heard at the Rolls Building, 7 Fetter Lane, Strand, London EC4A 1NL, on 20 September 2023, at 10:30 hours (or as soon thereafter that the Petition can be heard).

Any person intending to appear on the hearing of the Petition (whether to support or oppose it) must give notice of intention to do so to the Petitioner or his Solicitor in accordance with Rule 7.14 by 1600 hours on 19 September 2023.

The petitioner's solicitors are Geldards LLP, The Arc, Enterprise Way, Nottingham NG2 1EN (Ref: RLT.88450.455). (4431787)

In the COUNTY COURT AT READING

No 49 of 2023

# In the Matter of SATORI RISK LIMITED

(Company Number 13097279)

and in the Matter of the INSOLVENCY ACT 1986

A Petition to wind up the above-named Company (Registered No. 13097279) of Greenwood House, London Road, Bracknell, RG12 2AA, presented on 21 March 2023 by PARAGON BUSINESS FINANCE PLC of 51 Homer Road, Solihull, West Midlands B91 3QJ ("the Petitioner"), claiming to be a creditor of the Company, will be heard at the County Court at Reading, 160-163 Friar Street, Reading RG1 1HE Date 12 September 2023

Time 12:00PM (or as soon thereafter as the petition can be heard)

Any person intending to appear on the hearing of the petition (whether to support or oppose it) must give notice of intention to do so to the Petitioner or its solicitor in accordance with Rule 4.16 by 16.00 hours on 11 September 2023.

The Petitioner's Solicitor is Moore Barlow LLP, Gateway House, Tollgate Chandler's Ford, Eastleigh SO53 3TG. (Ref: CEP/72712/1.) (4431845) Dated 31 August 2023

#### In the HIGH COURT OF JUSTICE

BUSINESS AND PROPERTY COURTS OF ENGLAND AND WALES INSOLVENCY AND COMPANIES LIST (ChD) No CR-2023-004068 of 2023

In the Matter of TRADE LEGION LIMITED

(Company Number 12209272)

Trading Name: ZEYTIN CAFE

and in the Matter of the INSOLVENCY ACT 1986

A Petition to wind up the above-named company TRADE LEGION LIMITED T/A ZEYTIN CAFÉ (Registered NO.12209272) of 3 Denne Road, Horsham, RH12 1JE presented on 25 July 2023 By STAR CATERING SUPPLIES LIMITED C/O RSW LAW LIMITED, 33 Bedford Row, London, WC1R 4JH (the Petitioner) claiming to be a creditor of the company, will be heard at The High Court of Justice, The Rolls Building, 7 Rolls Buildings, Fetter Lane, London EC4A 1NL

Date: 13 September 2023

Time: 10:30 am hours (or as soon thereafter as the petition can be heard)

Any person intending to appear on the hearing of the petition (whether to support or oppose it) must give notice of that intention in accordance with rule 7.14 by 16.00 hours on 12 September 2023.

The Petitioner's Solicitor is RSW Law Limited, 33 Bedford Row, London, WC1R 4JH; Telephone: 0203 988 0170; email: chris@rswlaw.co.uk . Ref: CW/LIT203.

(4431829)

#### WINDING-UP ORDERS

#### 786 SOLUTIONS LTD

Dated: 30 August 2023

(Company Number 07862435) Registered office: 233 Windsor Street, BIRMINGHAM, B7 4NJ In the High Court Of Justice No 003660 of 2023 Date of Filing Petition: 7 July 2023 Date of Winding-up Order: 23 August 2023 Date of Resolution for Voluntary Winding-up: 23 August 2023 *Y Hill* 3rd Floor, Cannon House, 18 The Priory Queensway, Birmingham, B4 6FD, telephone: 0300 678 0016 Capacity of office holder(s): Liquidator 23 August 2023 (4430398)

### BACK ROW INVESTMENTS LIMITED

(Company Number 09742723) Registered office: Unit 10, Lower Charlton Trading Estate, SHEPTON MALLET, BA4 5QE In the Bristol District Registry No 000011 of 2023 Date of Filing Petition: 15 February 2023 Date of Winding-up Order: 15 August 2023 *M Mace* 3rd Floor, Senate Court, Southernhay Gardens, Exeter, EX1 1UG, telephone: 0300 678 0016 Capacity of office holder(s): Liquidator 15 August 2023 (4430391)

### CHARLIES HOMES AND GARDENS LIMITED

(Company Number 07449363) Registered office: The Cottage Heathman Street, Nether Wallop, STOCKBRIDGE, SO20 8EW In the High Court Of Justice No 003630 of 2023 Date of Filing Petition: 6 July 2023 Date of Winding-up Order: 23 August 2023 V Prime 1st Floor, Spring Place, 105 Commercial Road, Southampton, SO15 1EG, telephone: 0300 678 0016 Capacity of office holder(s): Liquidator 23 August 2023 (4430406)

### CHRISTIAN CHARLES LTD

(Company Number 09923026) Registered office: 32 Blyth Road, BROMLEY, BR1 3TN In the High Court Of Justice No 003637 of 2023 Date of Filing Petition: 6 July 2023 Date of Winding-up Order: 23 August 2023 *L Cook* 11th Floor, Southern House, Wellesley Grove, Croydon, CR0 1XN, telephone: 0300 678 0016 Capacity of office holder(s): Liquidator 23 August 2023 (4430401)

### **CREHAN CONSULTANCY LTD**

(Company Number 07922659) Registered office: 24 Braemar Road, BRENTFORD, TW8 0NR In the High Court Of Justice No 003636 of 2023 Date of Filing Petition: 6 July 2023 Date of Winding-up Order: 23 August 2023 *J Sullivan* 16th Floor, 1 Westfield Avenue, LONDON, E20 1HZ, telephone: 0300 678 0016 Capacity of office holder(s): Liquidator 23 August 2023 (4430410)

### ELITE ELECTRICAL SOLUTIONS (SOUTH COAST) LTD

(Company Number 13267186) Registered office: 11 Calmore Close, BOURNEMOUTH, BH8 0NN In the High Court Of Justice No 003769 of 2023 Date of Filing Petition: 12 July 2023 Date of Winding-up Order: 23 August 2023 *V Prime* 1st Floor, Spring Place, 105 Commercial Road, Southampton, SO15 1EG, telephone: 0300 678 0016 Capacity of office holder(s): Liquidator 23 August 2023 (4430405) ENDEAVOUR STRATEGIC DEVELOPMENTS LLP

(Company Number OC387343) Registered office: 32 Crescent, SALFORD, M5 4PF In the High Court Of Justice No 003638 of 2023 Date of Filing Petition: 6 July 2023 Date of Winding-up Order: 23 August 2023 *C Megram* 2nd Floor, 3 Piccadilly Place, London Road, Manchester, M1 3BN, telephone: 0300 678 0016 Capacity of office holder(s): Liquidator 23 August 2023 (4430417)

### EVOLVE MANAGEMENT UK LIMITED

(Company Number 06246123) Registered office: 10 The Edge, Clowes Street, Salford, M3 5NB In the High Court Of Justice No 003645 of 2023 Date of Filing Petition: 6 July 2023 Date of Winding-up Order: 23 August 2023 *C Megram* 2nd Floor, 3 Piccadilly Place, London Road, Manchester, M1 3BN, telephone: 0300 678 0016 Capacity of office holder(s): Liquidator 24 August 2023 (4430394)

### FOXTROT COSTUMES AND PROPS LIMITED

(Company Number 09253255) Registered office: 3b Brassie Avenue, LONDON, W3 7DE In the High Court Of Justice No 003483 of 2023 Date of Filing Petition: 6 July 2023 Date of Winding-up Order: 23 August 2023 *J Sullivan* 16th Floor, 1 Westfield Avenue, LONDON, E20 1HZ, telephone: 0300 678 0016 Capacity of office holder(s): Liquidator 23 August 2023 (4430408)

### GLOBAL FINANCIAL EXPERTS LIMITED

(Company Number 10792026) Registered office: 90 High Street, NEWMARKET, CB8 8FE In the High Court Of Justice No 003682 of 2023 Date of Filing Petition: 12 July 2023 Date of Winding-up Order: 23 August 2023 *M Commins* Eastbrook, Shaftesbury Road, Cambridge, CB2 8DR, telephone: 0300 678 0016 Capacity of office holder(s): Liquidator 23 August 2023 (4430411)

### GTG SOLUTIONS LIMITED

(Company Number 08584798) Registered office: 49 Horton Road, Datchet, SLOUGH, SL3 9EP In the High Court Of Justice No 003736 of 2023 Date of Filing Petition: 11 July 2023 Date of Winding-up Order: 23 August 2023 *J Sullivan* 16th Floor, 1 Westfield Avenue, LONDON, E20 1HZ, telephone: 0300 678 0016 Capacity of office holder(s): Liquidator 23 August 2023 (4430403)

### INSTANT MAKR LTD

(Company Number 08342137) Registered office: 2a Triangle Road, LONDON, E8 3RP In the High Court Of Justice No 003644 of 2023 Date of Filing Petition: 6 July 2023 Date of Winding-up Order: 23 August 2023 *J Sullivan* 16th Floor, 1 Westfield Avenue, LONDON, E20 1HZ, telephone: 0300 678 0016 Capacity of office holder(s): Liquidator 23 August 2023 (4430404)

### MAC4YOU LTD

(Company Number 11005215) Registered office: 6 Princes Court, Dukes Avenue, NEW MALDEN, KT3 4HZ In the High Court Of Justice No 003743 of 2023 Date of Filing Petition: 11 July 2023 Date of Winding-up Order: 23 August 2023 *L Cook* 11th Floor, Southern House, Wellesley Grove, Croydon, CR0 1XN, telephone: 0300 678 0016 Capacity of office holder(s): Liquidator 23 August 2023 (4430399)

### MICROGEM INTERNATIONAL PLC

(Company Number 09703359) Registered office: The Innovation Centre, Venture Road, Chilworth, SOUTHAMPTON, SO16 7NP In the High Court Of Justice No 002023 of 2023 Date of Filing Petition: 21 April 2023 Date of Winding-up Order: 23 August 2023 *V Prime* 1st Floor, Spring Place, 105 Commercial Road, Southampton, SO15 1EG, telephone: 0300 678 0016 Capacity of office holder(s): Liquidator 23 August 2023 (4430395)

### **MII TRADE LTD**

(Company Number 10288887) Registered office: 68 Stuart Place, LUTON, LU1 5DL In the High Court Of Justice No 003634 of 2023 Date of Filing Petition: 6 July 2023 Date of Winding-up Order: 23 August 2023 *J Sullivan* 16th Floor, 1 Westfield Avenue, LONDON, E20 1HZ, telephone: 0300 678 0016 Capacity of office holder(s): Liquidator 23 August 2023 (4430415)

### MISTRY MOTORS LIMITED

(Company Number 05150332) Registered office: Unit No: 1, Hartford Street, Denton, MANCHESTER, M34 3FR In the High Court Of Justice No 003395 of 2023 Date of Filing Petition: 27 June 2023 Date of Winding-up Order: 23 August 2023 *C Megram* 2nd Floor, 3 Piccadilly Place, London Road, Manchester, M1 3BN, telephone: 0300 678 0016 Capacity of office holder(s): Liquidator 23 August 2023 (4430397)

### MSM UTILITIES LTD

(Company Number 12299223) Registered office: 68 Eastcourt Lane, GILLINGHAM, ME8 6EX In the High Court Of Justice No 001972 of 2023 Date of Filing Petition: 20 April 2023 Date of Winding-up Order: 23 August 2023 *K Jackson* Fifty, Pembroke Court, Pembroke Chatham Maritime, CHATHAM, ME4 4EL, telephone: 03006780016 Capacity of office holder(s): Liquidator 23 August 2023 (4430393)

### PINNACLE PLASTERING CONTRACTORS LTD

(Company Number 08945079) Registered office: c/o Kreston Reeves LLP 2nd Floor, Innovation House, Ramsgate Road, Sandwich KENT, CT13 9FF In the High Court Of Justice No 003703 of 2023 Date of Filing Petition: 10 July 2023 Date of Winding-up Order: 23 August 2023 *K Jackson* Fifty, Pembroke Court, Pembroke Chatham Maritime, CHATHAM, ME4 4EL, telephone: 03006780016 Capacity of office holder(s): Liquidator 23 August 2023 (4430402)

### SEMPER CONTRACTING LTD

(Company Number 13376770) Registered office: 7 Skelton-on-Ure, RIPON, HG4 5AG In the High Court Of Justice No 003712 of 2023 Date of Filing Petition: 10 July 2023 Date of Winding-up Order: 23 August 2023 R Dury Level 1, Apex Court, City Link, NOTTINGHAM, NG2 4LA, telephone: 0300 678 0016. email: Nottingham.OR@insolvency.gsi.gov.uk Capacity of office holder(s): Liquidator 23 August 2023 (4430400)

### SMITHS PROPERTY MAINTENANCE LTD

(Company Number 06091602) Registered office: Suite 16 Beaufort Court, Admirals Way, South Quay, Docklands, LONDON, E14 9XL In the High Court Of Justice No 003727 of 2023 Date of Filing Petition: 11 July 2023 Date of Winding-up Order: 23 August 2023 *J Sullivan* 16th Floor, 1 Westfield Avenue, LONDON, E20 1HZ, telephone: 0300 678 0016 Capacity of office holder(s): Liquidator 23 August 2023 (4430419)

### START TO FINISH BUILDERS LTD

(Company Number 09797827) Registered office: 23 Lancelot Avenue, ROCHESTER, ME2 2YU In the High Court Of Justice No 003640 of 2023 Date of Filing Petition: 6 July 2023 Date of Winding-up Order: 23 August 2023 *K Jackson* Fifty, Pembroke Court, Pembroke Chatham Maritime, CHATHAM, ME4 4EL, telephone: 03006780016 Capacity of office holder(s): Liquidator 23 August 2023 (4430414)

### SUITCASE GROUP LIMITED

(Company Number 10662138) Registered office: 48 Warwick Street, LONDON, W1B 5AW In the High Court Of Justice No 003587 of 2023 Date of Filing Petition: 7 July 2023 Date of Winding-up Order: 23 August 2023 *J Sullivan* 16th Floor, 1 Westfield Avenue, LONDON, E20 1HZ, telephone: 0300 678 0016 Capacity of office holder(s): Liquidator 23 August 2023 (4430413)

### TRANSLATE MY LAW LONDON LTD (Company Number 08745947)

Registered office: PO Box 4385, 08745947, Companies House Default Address, Cardiff, CF14 8LH In the High Court Of Justice No 003695 of 2023 Date of Filing Petition: 10 July 2023 Date of Winding-up Order: 23 August 2023 *J Sullivan* 16th Floor, 1 Westfield Avenue, LONDON, E20 1HZ, telephone: 0300 678 0016 Capacity of office holder(s): Liquidator 23 August 2023 (4430409)

TRUST	SERV	ICES	VTM	LTD
-------	------	------	-----	-----

(Company Number 11310976) Registered office: 151 Brookhill Road, LONDON, SE18 6UZ In the High Court Of Justice No 003694 of 2023 Date of Filing Petition: 10 July 2023 Date of Winding-up Order: 23 August 2023 *J Sullivan* 16th Floor, 1 Westfield Avenue, LONDON, E20 1HZ, telephone: 0300 678 0016 Capacity of office holder(s): Liquidator 23 August 2023 (4430407)

### VEXTRA CAPITAL LIMITED

(Company Number 13317679) Registered office: 4 Old Park Lane, LONDON, W1K 1QW In the High Court Of Justice No 000919 of 2023 Date of Filing Petition: 21 March 2023 Date of Winding-up Order: 9 August 2023 *J Sullivan* 16th Floor, 1 Westfield Avenue, LONDON, E20 1HZ, telephone: 0300 678 0016 Capacity of office holder(s): Liquidator 9 August 2023 (4430392)

### WESTLAND ENVIRONMENTAL LIMITED

(Company Number 06054949) Registered office: Unit 24 Maple Leaf Business Park, Manston, Ramsgate, CT12 5GB In the High Court Of Justice No 002636 of 2023 Date of Filing Petition: 19 May 2023 Date of Winding-up Order: 23 August 2023 *K Jackson* Fifty, Pembroke Court, Pembroke Chatham Maritime, CHATHAM, ME4 4EL, telephone: 03006780016 Capacity of office holder(s): Liquidator 23 August 2023 (4430396)

### Members' voluntary liquidation

### APPOINTMENT OF LIQUIDATORS

Name of Company: **ACESSAR LIMITED** Company Number: 07607214 Nature of Business: Financial services Registered office: 36 Upton Road, Bexleyheath, Kent, DA6 8LH Type of Liquidation: Members Date of Appointment: 17 August 2023 *John Paul Bell* (IP No. 8608) and *Toyah Marie Poole* (IP No. 9740) both of Clarke Bell Limited, 3rd Floor, The Pinnacle, 73 King Street, Manchester, M2 4NG By whom Appointed: Members Ag FJ53012 (4430156)

### Name of Company: **ACNE PRODUCTION LIMITED** Company Number: 04289550 Nature of Business: Motion picture production activities Previous Name of Company: Acne Films Limited; Trade Offer Limited Registered office: 1 New Street Square, London, EC4A 3HQ Type of Liquidation: Members Date of Appointment: 25 August 2023 *James Douglas Ernle Money* (IP No. 8999) and *Steven Edward Butt* (IP No. 9108) both of Rollings Butt LLP, 6 Snow Hill, London, EC1A 2AY By whom Appointed: Members Ag FJ52928 (4430129)

Name of Company: ALISON FOSTER LIMITED Company Number: 10724246 Nature of Business: Project management services Registered office: DNS House, 382 Kenton Road, Harrow, Middlesex, HA3 8DF Type of Liquidation: Members Date of Appointment: 23 August 2023 John Paul Bell (IP No. 8608) and Toyah Marie Poole (IP No. 9740) both of Clarke Bell Limited, 3rd Floor, The Pinnacle, 73 King Street, Manchester, M2 4NG By whom Appointed: Members (4430140) Ag FJ53006 Name of Company: ARRANDCO AUDIT LIMITED Company Number: 04045321 Nature of Business: Accounting and Auditing activities Previous Name of Company: Baker Tilly Audit Limited Name of Company: PERSONAL INVESTMENT MANAGEMENT LIMITED Company Number: 03792617 Nature of Business: Financial Intermediation Name of Company: RSM & CO (UK) LIMITED Company Number: 04273907 Nature of Business: Financial Intermediation Previous Name of Company: Baker Tilly & Co Limited Registered office: (All) 7th Floor, 21 Lombard Street, London EC3V 9AH Type of Liquidation: (All) Members Date of Appointment: 24 August 2023 Henry Nicholas Page (IP No. 12250) and Edward James Ellis (IP No. 25552) both of Mercer & Hole, 21 Lombard Street, London, EC3V 9AH By whom Appointed: Members Ag FJ53020 (4430133)

### PURSUANT TO SECTION 109 OF THE INSOLVENCY ACT 1986 Name of Company: ASIAN SENSATION LTD

Company Number: 08240164 Trading Name: Man Ho Restaurant Registered office: 12 Burford Close Luton, England LU3 4DS United Kingdom Principal trading address: 72 Dunstable Road Luton, Bedfordshire, England LU1 1EH United Kingdom Nature of Business: 56101 - Licensed restaurants Type of Liquidation: Members Liquidator name and address: Alex Martin-Sklan, Martin Sklan & Co 133 Golders Green Road London NW11 8HJ United Kingdom Office Holder Number: 5815. Date of Appointment: 15 August 2023 By whom Appointed: The Company Contact name: Tanjina Mahbub Email address: ms@martinsklan.co.uk Telephone Number: 020 8458 4433 (4431870)

Name of Company: **BOOTSTRAP SOFTWARE LIMITED** Company Number: 08858028 Nature of Business: Software Consultancy Registered office: 80 Ashton Road, Denton, Manchester, M34 3JF Type of Liquidation: Members Date of Appointment: 23 August 2023 *John Paul Bell* (IP No. 8608) and *Toyah Marie Poole* (IP No. 9740) both of Clarke Bell Limited, 3rd Floor, The Pinnacle, 73 King Street, Manchester, M2 4NG By whom Appointed: Members Ag FJ53016 (4430152)

### COMPANIES

### Name of Company: BSI SPEEDWAY (HOLDINGS) LIMITED

Company Number: 05713374 Nature of Business: Activities of sport clubs Registered office: Building 6, Chiswick Park, 566 Chiswick High Road, London, W4 5HR Type of Liquidation: Members Date of Appointment: 23 August 2023 *Steven Edward Butt* (IP No. 9108) and *James Douglas Ernle Money* (IP No. 8999) both of Rollings Butt LLP, 6 Snow Hill, London, EC1A 2AY By whom Appointed: Members Ag FJ52943 (4430132)

Name of Company: **CERIDWEN MAXIMUS LIMITED** Company Number: 08513411 Nature of Business: Information technology consultancy activities Registered office: W8a Knoll Business Centre, 325-327 Old Shoreham Road, Hove, BN3 7GS Type of Liquidation: Members Date of Appointment: 23 August 2023 *Donald McNaught* (IP No. 9359) of Johnston Carmichael LLP, 227 West George Street, Glasgow, G2 2ND By whom Appointed: The Company Ag FJ52873 (4430122)

### PURSUANT TO SECTION 109 OF THE INSOLVENCY ACT 1986 Name of Company: CLEAR PEN SOLUTIONS LTD

Company Number: 07000964

Registered office: Goodwood House Blackbrook Park Avenue Taunton Somerset TA1 2PX

Nature of Business: Other activities auxiliary to insurance and pension funding

Type of Liquidation: Members

Liquidator name and address *Daniel James*, Albert Goodman LLP, Goodwood House, Blackbrook Park Avenue, Taunton, Somerset TA1 2PX

Office Holder Number: 26292.

Date of Appointment: 23 August 2023

By whom Appointed: Members

Alternative Contact: Telephone: Email: Karolina Struska 01823 250796 karolina.struska@albertgoodman.co.uk (4431869)

Company Number: 09413653

### Name of Company: CRAIC MANAGEMENT LIMITED

Nature of Business: Management consultancy activities other than financial management

Registered office: Unit 14/15 Hall Farm Sywell Aerodrome, Sywell, Northamptonshire, NN6 0BN

Principal trading address: Yewtrees, 49 Dallington Road, Northampton, NN5 7BW

Type of Liquidation: Members Voluntary Liquidation

Liquidator: *Gary Pettit* (IP number 9066) of PBC Business Recovery & Insolvency Ltd., 9-10 Scirocco Close, Moulton Park, Northampton, NN3 6AP .

Date of Appointment: 15 August 2023

By wr	iom Appoint	ea: iviembei	'S			
For	further	details	contact	Natasha	Pink	at
natasl	hapink@pbc	businessred	overy.co.uk		(4430	910)

Name of Company: **DMSI LIMITED** Company Number: 10401901 Nature of Business: Management Consultancy Registered office: Flat 108b, Princes Park Manor Royal Drive, London N11 3FP Type of Liquidation: Members Date of Appointment: 23 August 2023 *Donald McNaught* (IP No. 9359) of Johnston Carmichael LLP, 227 West George Street, Glasgow, G2 2ND By whom Appointed: The Company Ag FJ52880 (4430126)

### Company Number: 13575110

Name of Company: EXPERT REVIEWS HOLDINGS LIMITED

Nature of Business: Publishing of consumer and business journals and periodicals

Registered office: 1st & 2nd Floors Wenlock Works, 1a Shepherdess Walk, England, London, N1 7QE

Type of Liquidation: Members Voluntary Liquidation

Joint Liquidator: *Steven Sherry* (IP number 19752) of PricewaterhouseCoopers LLP, 7 More London Riverside, London SE1 2RT.

Joint Liquidator: *Emma Cray* (IP number 17450) of PricewaterhouseCoopers LLP, One Chamberlain Square, Birmingham, B3 3AX.

Date of Appointment: 18 August 2023

By whom Appointed: The Members of the Company

For further details contact Elizabeth Colley at elizabeth.colley@pwc.com

The Liquidators may act as controllers of personal data as defined by UK data protection law depending upon the specific processing activities undertaken. PricewaterhouseCoopers LLP may act as a processor on the instructions of the Liquidators. Personal data will be kept secure and processed only for matters relating to the Liquidators` appointment. Further details are available in the privacy statement on the PwC.co.uk website or by contacting the Liquidators. (4432031)

Company Number: 06183487

Name of Company: **GARETH BUNN CONSULTING LIMITED** Nature of Business: Training Services Registered office: 202 Sheen Lane, East Sheen, London SW14 8LF Principal trading address: 202 Sheen Lane, East Sheen, London SW14 8LF

Type of Liquidation: Members Voluntary Liquidation

Liquidator: *Jeremy Charles Frost* (IP number 9091) of Frost Group Limited, One Elmfield Park Bromley BR1 1LU.

Date of Appointment: 23 August 2023

By whom Appointed: Members For further details contact Kelly Walford on 0345 260 0101 or at

kellyw@frostbr.co.uk (4432002)

### Name of Company: GREAT HORKESLEY WINES LIMITED Company Number: 03755496 Trading Name: The Wine Centre Nature of Business: Retail sale of beverages in specialised stores Registered office: The Causeway, Great Horkesley, Colchester CO6 4BH. Type of Liquidation: Members Date of Appointment: 29 August 2023 Lee De'ath (IP No. 9316) and Tom Gardiner (IP No. 17990) both of Begbies Traynor (Central) LLP, Town Wall House, Balkerne Hill, Colchester, Essex, CO3 3AD By whom Appointed: Members Ag FJ53022 (4430148) Name of Company: HAYMAR ADVISORY LIMITED Company Number: 09569473 Nature of Business: Financial Services Registered office: Units 1-3 Hilltop Business Park, Devizes Road, Salisbury, Wiltshire, SP3 4UF Type of Liquidation: Members Date of Appointment: 25 August 2023 Simon Lowes (IP No. 9194) and Stephen Mark Powell (IP No. 009561) both of Begbies Traynor (Central) LLP, Units 1-3 Hilltop Business Park, Devizes Road, Salisbury, Wiltshire, SP3 4UF

By whom Appointed: Members Ag FJ52899

(4430127)

### Name of Company: JUICE CONSULTING LTD

Company Number: 04530837

Nature of Business: Accounting and auditing activities; Financial management Registered office: 5 Barnfield Crescent, Exeter, EX1 1QT Type of Liquidation: Members Date of Appointment: 22 August 2023 David Gerard Kirk (IP No. 8830) and Daniel Robert Jeeves (IP No. 26032) both of Kirks, 5 Barnfield Crescent, Exeter, EX1 1QT By whom Appointed: Members (4430125) Ag FJ52871

Company Number: 12939748 Name of Company: KACE CONSULTING LTD Nature of Business: Management Consultancy Registered office: 38 High Street, Watton, Thetford, Norfolk IP2 Principal trading address: 55 Salisbury Road, Ealing, Londo 9TT Type of Liquidation: Members Voluntary Liquidation Liquidator: Jeremy Charles Frost (IP number 9091) of Frost Limited, One Elmfield Park Bromley BR1 1LU. Date of Appointment: 21 August 2023 By whom Appointed: Members For further details contact Kelly Walford on 0345 260 010 kellvw@frostbr.co.uk (4 Company Number: 07095044 Name of Company: LDN CUSTOMER SOLUTIONS LIMITED Nature of Business: IT service activities Registered office: Old World Cottage, Ketton Road, Han

Rutland, LE15 8TH Type of Liquidation: Members Voluntary Liquidation Liquidator: Michael Durkan (IP number 9583) of Durkan Cahi

G2 Montpellier House, Montpellier Drive, Cheltenham, GL50 17 Date of Appointment: 17 August 2023

By whom Appointed: Members

#### further details contact: 01242 250811 For mpd@durkancahill.com (44

Company Number: 13130924 Name of Company: LINDEN SP LIMITED Previous Name of Company: CANONBIE CAPITAL L CANONBIE SP LIMITED LIMITED and CANONBIE SP LIMITED Nature of Business: Loan Management & Advisory Services Registered office: 29 Benham House Coleridge Gardens, 55 Road, London SW10 0RD Principal trading address: 29 Benham House Coleridge Garder Kings Road, London SW10 0RD Type of Liquidation: Members Voluntary Liquidation Liquidator: Jeremy Charles Frost (IP number 9091) of Frost Limited, One Elmfield Park Bromley BR1 1LU. Date of Appointment: 21 August 2023 By whom Appointed: Members For further details contact Kelly Walford on 0345 260 010 kellyw@frostbr.co.uk (4 Name of Company: MOSS AND IVY LIMITED Company Number: 07379767 Nature of Business: Others - Not Reported Previous Name of Company: Kate of Kensington Ltd; Bar Direct Limited Registered office: Begbies Traynor, The Officers Mess B Centre, Royston Road, Duxford, Cambridge CB22 4QH Type of Liquidation: Members Date of Appointment: 24 August 2023

Wayne MacPherson (IP No. 9445) and Dominik Thiel-Czerw No. 009636) both of Begbies Traynor (Central) LLP, 1066 Road, Leigh-on-Sea, Essex, SS9 3NA By whom Appointed: Members Ag FJ52952 (4430136)

### Name of Company: MY PROMOTIONS LTD Company Number: 08278818

Nature of Business: Retail sale of newspapers and stationery in

specialised stores Registered office: Currently The Old School House, West Street,

Southwick, Fareham, PO17 6EA but shortly changing to c/o Begbies Traynor (Central) LLP, Units 1-3 Hilltop Business Park, Devizes Road, Salisbury SP3 4UF

Type of Liquidation: Members

Date of Appointment: 29 August 2023

Julie Anne Palmer (IP No. 008835) and Stephen Mark Powell (IP No. 009561) both of Begbies Traynor (Central) LLP, Units 1-3 Hilltop Business Park, Devizes Road, Salisbury, Wiltshire, SP3 4UF By whom Appointed: Members Ag FJ52960

(4430145)

5 6AE	
h W13 PURSUANT TO SECTION 109 OF THE INSOLVENCY ACT 1986 Name of Company: QM CAPITAL PARTNERS LIMITED	
Company Number: 10737660 Group Registered office: One, Fleet Place, London, EC4M 7WS Principal trading address: One, Fleet Place, London, EC4M 7WS Nature of Business: Activities of financial services holding companie Type of Liquidation: Members' Voluntary Liquidation	es
or at Liquidator(s) name(s) and address(es): <i>Daryl Warwick</i> and <i>Mi</i> , 31763) <i>Kienlen</i> , James Watson House, Montgomery Way, Rosehill, Carlis Cumbria, CA1 2UU Office Holder Numbers: 9500 and 9367. Date of Appointment: 24 August 2023 By whom Appointed: The Members	
Further Contact Details Lene Davidson Tel No. 01228 690200 bleton, (443187	71)
Name of Company: QUASSIA (PROPERTIES) LIMITED	
<ul> <li>, Suite Company Number: 01044492</li> <li>Y. Nature of Business: Other letting and operating of own or leased re estate.</li> </ul>	eal
Previous Name of Company: Quassia (Joinery) Limited r at Registered office: Quassia House, Bearwalden Industrial Estat 31619) Wendons Ambo, Saffron Walden, Essex, CB11 4JX Type of Liquidation: Members	te,
Date of Appointment: 21 August 2023 Andrew Dix (IP No. 009327) of AD Business Recovery Limited, Sw House, Ground Floor, 18 Hoffmanns Way, Chelmsford, Essex, CN	
MITED; 1GU By whom Appointed: Members	
Ag FJ52958 (443014 Kings	13)
is, 552 Name of Company: <b>SKAGEN FINANCE LIMITED</b> Company Number: 09299298	
Nature of Business: Financial management Group Registered office: c/o Mazars LLP, 30 Old Bailey, London, EC4M 7AU Type of Liquidation: Members Date of Appointment: 24 August 2023	U
Simon David Chandler (IP No. 008822) and Guy Robert Thom. or at Hollander (IP No. 009233) both of Mazars LLP, 30 Old Bailey, Londo 31964) EC4M 7AU	
By whom Appointed: Member Ag FJ52921 (443012	23)
Name of Company: <b>SONOCO CONSUMER PRODUCTS LTD</b> Owner Company Number: 02409393	
Registered office: C/O Sonoco Cores & Paper Ltd Stainland Boa siness Mills, Holywell Green, Halifax, England, HX4 9PY	ard
Name of Company: <b>COREPAK LTD</b> Company Number: 02304976 Registered office: C/O Sonoco Cores & Paper Ltd Stainland Boa	ard
<ul> <li>Mills, Holywell Green, Halifax, England, HX4 9PY</li> <li>Name of Company: SONOCO CAPSEALS LINERS LTD Company Number: 00969584</li> </ul>	

Registered office: C/O Sonoco Cores & Paper Ltd Stainland Board Mills, Holywell Green, Halifax, England, HX4 9PY

### COMPANIES

Name of Company: SONOCO BOARD MILLS LTD Company Number: 00196071 Registered office: C/O Sonoco Cores & Paper Ltd Stainland Board Mills, Holywell Green, Halifax, England, HX4 9PY Name of Company: CAPSEALS LTD Company Number: 00562588 Registered office: C/O Sonoco Cores & Paper Ltd Stainland Board Mills, Holywell Green, Halifax, England, HX4 9PY Name of Company: CAPSEALS LINERS LTD Company Number: 00637766 Registered office: C/O Sonoco Cores & Paper Ltd Stainland Board Mills, Holywell Green, Halifax, England, HX4 9PY Name of Company: SONOCO PACKAGING TAPES LTD Company Number: 00354137 Registered office: C/O Sonoco Cores & Paper Ltd Stainland Board Mills, Holywell Green, Halifax, England, HX4 9PY Name of Company: SONOCO UK LEASING LTD Company Number: 00423904 Registered office: C/O Sonoco Cores & Paper Ltd Stainland Board Mills, Holywell Green, Halifax, England, HX4 9PY Name of Company: TPT LIMITED Company Number: 00577310 Nature of Business: 99999 - Dormant Companies Registered office: C/O Sonoco Cores & Paper Ltd Stainland Board Mills, Holywell Green, Halifax, England, HX4 9PY Type of Liquidation: Members Voluntary Liquidation Joint Liquidator: Emma Cray (IP 17450) number of PricewaterhouseCoopers LLP, One Chamberlain Square, Birmingham B3 3AX.

Joint Liquidator: *Steven Sherry* (IP number 19752) of PricewaterhouseCoopers LLP, 7 More London Riverside, London SE1 2RT.

Date of Appointment: 22 August 2023

By whom Appointed: The Members of the Companies

For further details contact Dijesh Patel on 07802 660714 or at dijesh.v.patel@pwc.com

The Liquidators may act as controllers of personal data as defined by UK data protection law depending upon the specific processing activities undertaken. PricewaterhouseCoopers LLP may act as a processor on the instructions of the Liquidators. Personal data will be kept secure and processed only for matters relating to the Liquidators` appointment. Further details are available in the privacy statement on the PwC.co.uk website or by contacting the Liquidators. (4430970)

Company Number: 01260244

### Name of Company: SUIT YOURSELF LIMITED

Nature of Business: Other letting and operating of own or leased real estate.

Registered office: c/o Sterling Ford, Centurion Court, 83 Camp Road, St Albans, Herts AL1 5JN

Principal trading address: (Former): The Roma Building, 32/38 Scrutton Street, Bishopsgate, London EC2A 4RQ

Type of Liquidation: Members Voluntary Liquidation

Liquidator: *Phillip Anthony Roberts* (IP number 6055) of Sterling Ford, Centurion Court, 83 Camp Road, St Albans, Herts AL1 5JN.

Date of Appointment: 18 August 2023

By whom Appointed: Members

For further details contact: 01727 811161 (4431361)

### Name of Company: **THE THREE PARTNERSHIP LIMITED** Company Number: 07709146

Nature of Business: Recruitment Consultant

Registered office: 207 Knutsford Road Grappenhall, Warrington, Cheshire, WA4 2QL Type of Liquidation: Members

Type of Elquidation. Weitibers

Date of Appointment: 24 August 2023

John Paul Bell (IP No. 8608) and Toyah Marie Poole (IP No. 9740) both of Clarke Bell Limited, 3rd Floor, The Pinnacle, 73 King Street, Manchester, M2 4NG By whom Appointed: Members

Ag FJ53011 (4430135)

### **NOTICES TO CREDITORS**

### ACESSAR LIMITED

(Company Number 07607214)

Registered office: 36 Upton Road, Bexleyheath, Kent, DA6 8LH

Principal trading address: 36 Upton Road, Bexleyheath, Kent, DA6  $\rm 8LH$ 

Notice is hereby given that Creditors of the Company are required, on or before 25 September 2023, to prove their debts by delivering their proofs (in the format specified in Rule 14.4 of the INSOLVENCY (ENGLAND AND WALES) RULES 2016) to the Joint Liquidators at Clarke Bell Limited, 3rd Floor, The Pinnacle, 73 King Street, Manchester, M2 4NG.

If so required by notice from the Joint Liquidators, creditors must produce any document or other evidence which the Joint Liquidators consider is necessary to substantiate the whole or any part of a claim. Note: The Directors of the Company have made a declaration of solvency and it is expected that all creditors will be paid in full.

Date of Appointment: 17 August 2023.

Office Holder Details: *John Paul Bell* (IP No. 8608) and *Toyah Marie Poole* (IP No. 9740) both of Clarke Bell Limited, 3rd Floor, The Pinnacle, 73 King Street, Manchester, M2 4NG

For further details contact: The Joint Liquidators, Email: mvl@clarkebell.com. Alternative contact: Thomas Bailey

(4430131)

(4430153)

John Paul Bell, Joint Liquidator 30 August 2023

Ag FJ53012

### ACNE PRODUCTION LIMITED

(Company Number 04289550)

Registered office: 1 New Street Square, London, EC4A 3HQ

Principal trading address: 1 New Street Square, London, EC4A 3HQ Notice is hereby given that creditors of the Company are required, on or before 29 September 2023, to prove their debts by delivering their proofs (in the format specified in Rule 14.4 of the INSOLVENCY (ENGLAND AND WALES) RULES 2016) to the Joint Liquidators at Rollings Butt LLP, 6 Snow Hill, London, EC1A 2AY.

If so required by notice from the Joint Liquidators, creditors must produce any document or other evidence which the Joint Liquidators consider is necessary to substantiate the whole or any part of a claim. Note: The Directors of the Company have made a declaration of solvency and it is expected that all creditors will be paid in full. Date of Appointment: 25 August 2023

Office Holder Details: *James Douglas Ernle Money* (IP No. 8999) and *Steven Edward Butt* (IP No. 9108) both of Rollings Butt LLP, 6 Snow Hill, London, EC1A 2AY

For further details contact: The Joint Liquidators, Tel: 020 7002 7960 or Email: mail@rollingsbutt.com. Alternative contact: Ali Yarwood. *James Douglas Ernle Money,* Joint Liquidator

25 August 2023

Ag FJ52928

### ALISON FOSTER LIMITED

(Company Number 10724246) Registered office: DNS House, 382 Kenton Road, Harrow, Middlesex, HA3 8DP

Principal trading address: Flat 20, Wren Court, 12 The Avenue, Surbiton, Surrey, KT5 8JB

Notice is hereby given that Creditors of the Company are required, on or before 22 September 2023, to prove their debts by delivering their proofs (in the format specified in Rule 14.4 of the INSOLVENCY (ENGLAND AND WALES) RULES 2016) to the Joint Liquidators at Clarke Bell Limited, 3rd Floor, The Pinnacle, 73 King Street, Manchester, M2 4NG.

If so required by notice from the Joint Liquidators, creditors must produce any document or other evidence which the Joint Liquidators consider is necessary to substantiate the whole or any part of a claim. Note: The Directors of the Company have made a declaration of solvency and it is expected that all creditors will be paid in full. Date of Appointment: 23 August 2023.

Office Holder Details: *John Paul Bell* (IP No. 8608) and *Toyah Marie Poole* (IP No. 9740) both of Clarke Bell Limited, 3rd Floor, The Pinnacle, 73 King Street, Manchester, M2 4NG

For further details contact: The Joint Liquidators, Email: mvl@clarkebell.com. Alternative contact: Thomas Bailey *John Paul Bell*, Joint Liquidator 30 August 2023

Ag FJ53006

(4430128)

### AMNET LIMITED

(Company Number 03148086)

Registered office: 1 More London Place, London SE1 2AF

Principal trading address: 10 Triton Street, Regents Place, London, NW1 3BF

As Joint Liquidators of the company, we hereby give notice that we intend to make a final distribution to its creditors. The last date for proving is 17 October 2023 and creditors of the company should by that date send their full names and addresses and particulars of their debts or claims to me, Samantha Jane Keen of Ernst & Young LLP, 1 More London Place, London, SE1 2AF.

In accordance with Rule 14.38(1)(c) of the Insolvency (England and Wales) Rules 2016, we may thereafter make the proposed distribution without regard to the claim of any person in respect of a debt not yet proved.

Date of Appointment: 16 September 2022

Office Holder Details: *Samantha Jane Keen* (IP No. 9250) and *Richard Barker* (IP No. 17150) both of Ernst & Young LLP, 1 More London Place, London, SE1 2AF

For further details contact: The Joint Liquidators, Tel: 020 7951 3427. Alternative contact: Katya Vasileva.

Samantha Jane Keen, Joint Liquidator

29 August 2023 Ag FJ53014

(4430144)

(4430134)

### ARRANDCO AUDIT LIMITED

(Company Number 04045321) PERSONAL INVESTMENT MANAGEMENT LIMITED (Company Number 03792617) RSM & CO (UK) LIMITED

(Company Number 04273907)

Registered office: (All) 7th Floor, 21 Lombard Street, London EC3V 9AH

Principal trading address: (All) 6th Floor, 25 Farringdon Street, London EC4A 4AB

We, *Henry Nicholas Page* (IP No. 12250) and *Edward James Ellis* (IP No. 25552) both of Mercer & Hole, 21 Lombard Street, London, EC3V 9AH were appointed Joint Liquidators of the above named Companies on 24 August 2023.

Notice is hereby given that the creditors of the above Companies, which are being voluntarily wound up, are required, on or before 1 October 2023 to send their full names and addresses together with full particulars of their debts or claims to the joint liquidators of the Companies and, if so required by notice from the Joint Liquidators, provide such further details or produce such documentary evidence which they consider is necessary to substantiate the whole or any part of a claim, in default of which they will be excluded from the benefit of any distribution made before such debts are proved.

Note: These are solvent liquidations and all known creditors have been or will be paid in full.

Further information can be obtained from the Joint Liquidators or the case administrator, Harry Smart, on Tel: 020 7236 2601; Email: Harry.Smart@mercerhole.co.uk.

Edward Ellis, Joint Liquidator

29 August 2023 Ag FJ53020

Ay FJ55020

### IN THE MATTER OF THE INSOLVENCY ACT 1986 ASIAN SENSATION LIMITED

(Company Number 08240164)

Trading Name: Man Ho Restaurant

Registered office: 12 Burford Close, Luton, England, LU3 4DS, United Kingdom

Principal trading address: 72 Dunstable Road, Luton, Bedfordshire, England, LU1 1EH, United Kingdom

Nature of business: Members Voluntary Liquidation

Type of liquidation: 56101 - Licensed restaurants

I, Alex Martin-Sklan of Martin Sklan & Co, 133 Golders Green Road, London, NW11 8HJ, United Kingdom give notice that I was appointed liquidator of the above-named company on 15 August 2023 by a resolution of members.

NOTICE IS HEREBY GIVEN that the creditors of the above-named company which is being voluntarily wound up, are required, on or before 21 September 2023 to prove their debts by sending to the undersigned Alex Martin-Sklan of Martin Sklan & Co, 133 Golders Green Road, London, NW11 8HJ, United Kingdom the liquidator of the company, written statements of the amounts they claim to be due to them from the company and, if so requested, to provide such further details or produce such documentary evidence as may appear to the liquidator to be necessary. Please note that this is a solvent liquidation and therefore the liquidator is entitled to make the distribution without regard to the claim of any person in respect of a debt not proved.

Contact person Alex Martin-Sklan Telephone no. 020 8458 4433 email address ms@martinsklan.co.uk

Alex Martin-Sklan

Liquidator

Dated: 22 August 2023

(4431868)

(4430138)

### **BOOTSTRAP SOFTWARE LIMITED**

(Company Number 08858028)

Registered office: 80 Ashton Road, Denton, Manchester, M34 3JF Principal trading address: 80 Ashton Road, Denton, Manchester, M34 3.IF

Notice is hereby given that Creditors of the Company are required, on or before 22 September 2023, to prove their debts by delivering their proofs (in the format specified in Rule 14.4 of the INSOLVENCY (ENGLAND AND WALES) RULES 2016) to the Joint Liquidators at Clarke Bell Limited, 3rd Floor, The Pinnacle, 73 King Street, Manchester, M2 4NG.

If so required by notice from the Joint Liquidators, creditors must produce any document or other evidence which the Joint Liquidators consider is necessary to substantiate the whole or any part of a claim. Note: The Directors of the Company have made a declaration of solvency and it is expected that all creditors will be paid in full.

Date of Appointment: 23 August 2023.

Office Holder Details: *John Paul Bell* (IP No. 8608) and *Toyah Marie Poole* (IP No. 9740) both of Clarke Bell Limited, 3rd Floor, The Pinnacle, 73 King Street, Manchester, M2 4NG

For further details contact: The Joint Liquidators, Email: mvl@clarkebell.com. Alternative contact: Thomas Bailey

John Paul Bell, Joint Liquidator

30 August 2023

Ag FJ53016

### **BSI SPEEDWAY (HOLDINGS) LIMITED**

(Company Number 05713374)

Registered office: Building 6, Chiswick Park, 566 Chiswick High Road, London, W4 5HR

Principal trading address: Building 6, Chiswick Park, 566 Chiswick High Road, London, W4 5HR

Notice is hereby given that creditors of the Company are required, on or before 25 September 2023 to prove their debts by delivering their proofs (in the format specified in Rule 14.4 of the INSOLVENCY (ENGLAND AND WALES) RULES 2016) to the Joint Liquidators at Rollings Butt LLP, 6 Snow Hill, London, EC1A 2AY.

If so required by notice from the Joint Liquidators, creditors must produce any document or other evidence which the Joint Liquidators consider is necessary to substantiate the whole or any part of a claim. Note: The Directors of the Company have made a declaration of solvency and it is expected that all creditors will be paid in full.

Date of Appointment: 23 August 2023

Office Holder Details: *Steven Edward Butt* (IP No. 9108) and *James Douglas Ernle Money* (IP No. 8999) both of Rollings Butt LLP, 6 Snow Hill, London, EC1A 2AY

For further details contact: The Joint Liquidators, Tel: 020 7002 7960 or Email: mail@rollingsbutt.com. Alternative contact: Laurie'elle Rawlings

Steven Edward Butt, Joint Liquidator

25 August 2023 Ag FJ52943

(4430149)

### CLEAR PEN SOLUTIONS LTD

(Company Number 07000964)

Registered office: Goodwood House, Blackbrook Park Avenue, Taunton, Somerset TA1 2PX

Principal trading address: New Penderel House 4th Floor, 283-288 High Holborn, London WC1V 7HP

On 23 August 2023 the above company was placed in members' voluntary liquidation and Daniel James, IP No. 26292 of Albert Goodman LLP,Goodwood House, Blackbrook Park Avenue, Taunton, Somerset TA1 2PX was appointed liquidator. The company is able to pay all its known creditors in full.

Notice is hereby given that creditors who have not already proved are required, on or before 5 October 2023, to prove their debts by sending Daniel James of Albert Goodman LLP, Goodwood House, Blackbrook Park Avenue, Taunton, Somerset TA1 2PX, the liquidator of the company, written statements of the amounts they claim to be due to them from the company and, if so requested, to provide further details or produce documentary or other evidence as may appear to the liquidator to be necessary. A creditor who has not proved their debt before the declaration of any dividend is not entitled to disturb, by reason that they have not participated in it, the distribution of that dividend or any other dividend declared before their debt was proved. Notice is also hereby given pursuant to Rule 14.28 of the Insolvency (England and Wales) Rules 2016, that the liquidator intends to declare a first and final dividend to unsecured creditors of the company within the period of two months from the last date for proving as specified above.

The liquidator intends that, after paying or providing for the claims of all creditors who have proved their debts, the funds remaining in the hands of the liquidator shall be distributed to shareholders absolutely. Further details contact: Karolina Struska Telephone: 01823 250796 Email: karolina.struska@albertgoodman.co.uk

DANIEL JAMES Liquidator 25 August 2023

(4431872)

### **CRAIC MANAGEMENT LIMITED**

(Company Number 09413653)

Registered office: Unit 14/15 Hall Farm Sywell Aerodrome, Sywell, Northamptonshire, NN6 0BN

Principal trading address: Yewtrees, 49 Dallington Road, Northampton, NN5 7BW

NOTICE IS HEREBY GIVEN that the creditors of the above named Company, which was voluntarily wound up on 15 August 2023 are required, on or before 22 September 2023 to send their full names and addresses together with full particulars of their debts or claims to PBC Business Recovery & Insolvency Ltd, 9/10 Scirocco Close, Moulton Park, Northampton, NN3 6AP and, if so requested by me, to provide such further details or produce such documentary or other evidence as may appear to be necessary, or in default thereof they will be excluded from the benefit of any distribution made before such debts are proved.

Note: This is a solvent liquidation and all known creditors have been or wrn be paid in full.

Liquidator: Gary Pettit (IP number 9066) of PBC Business Recovery & Insolvency Ltd,, 9-10 Scirocco Close, Moulton Park, Northampton, NN3 6AP .

Date of Appointment: 15 August 2023

For	further	details	contact	Natasha	Pink	at
natash	napink@pbc	businessrec	overy.co.uk		(4430	911)

### EXPERT REVIEWS HOLDINGS LIMITED

(Company Number 13575110)

Registered office: 1st & 2nd Floors Wenlock Works, 1a Shepherdess Walk, England, London, N1 7QE

NOTICE IS HEREBY GIVEN that the creditors of the above named company, which is being voluntarily wound up, must send their full names and addresses (and those of their Solicitors, if any), together with full particulars of their debts or claims to steven.a.sherry@pwc.com at PricewaterhouseCoopers LLP, 7 More London Riverside, London SE1 2RT by 25 September 2023.

The distribution may be made without regard to the claim of any person in respect of a debt not proved.

Note: It is anticipated that all known Creditors will be paid in full.

Joint Liquidator: *Steven Sherry* (IP number 19752) of PricewaterhouseCoopers LLP, 7 More London Riverside, London SE1 2RT.

Joint Liquidator: *Emma Cray* (IP number 17450) of PricewaterhouseCoopers LLP, One Chamberlain Square, Birmingham, B3 3AX.

Date of Appointment: 18 August 2023

For further details contact Elizabeth Colley at elizabeth.colley@pwc.com

The Liquidators may act as controllers of personal data as defined by UK data protection law depending upon the specific processing activities undertaken. PricewaterhouseCoopers LLP may act as a processor on the instructions of the Liquidators. Personal data will be kept secure and processed only for matters relating to the Liquidators' appointment. Further details are available in the privacy statement on the PwC.co.uk website or by contacting the Liquidators. (4432029)

### GARETH BUNN CONSULTING LIMITED

(Company Number 06183487)

Registered office: 202 Sheen Lane, East Sheen, London SW14 8LF Principal trading address: 202 Sheen Lane, East Sheen, London SW14 8LF

NOTICE TO CREDITORS

The Company was placed into Member's Voluntary Liquidation on 23 August 2023 and is able to pay all its known creditors in full. Pursuant to Rule 4.182A of the Insolvency Rules 1986, NOTICE IS HEREBY GIVEN that the Liquidator intends to make a first and final distribution to remaining creditors of the above-named Company and that the last date for proving debts against the above-named Company, which is being voluntarily wound up, is 28 September 2023. Claims must be sent to the undersigned, Jeremy Charles Frost of Frost Group Limited, One Elmfield Park, Bromley BR1 1LU, the Liquidator of the Company.

After 28 September 2023, the Liquidator may make that distribution without regard to the claim of any person in respect of a debt not already proved.

The Liquidator intends that after paying or providing for a final distribution in respect of the claims of all creditors who have proved their debts by the above date, the assets remaining in the hands of the Liquidator shall be distributed to the shareholder absolutely.

Liquidator: *Jeremy Charles Frost* (IP number 9091) of Frost Group Limited, One Elmfield Park Bromley BR1 1LU.

Date of Appointment: 23 August 2023

For further details contact Kelly Walford on 0345 260 0101 or at kellyw@frostbr.co.uk (4432003)

### **GREAT HORKESLEY WINES LIMITED**

(Company Number 03755496)

Trading Name: The Wine Centre

Registered office: The Causeway, Great Horkesley, Colchester CO6 4BH.

Principal trading address: The Wine Centre, The Causeway, Great Horkesley, Colchester CO6 4BH.

The Company was placed into members' voluntary liquidation on 29 August 2023 when *Lee De'ath* (IP No. 9316) and *Tom Gardiner* (IP No. 17990) both of Begbies Traynor (Central) LLP, Town Wall House, Balkerne Hill, Colchester, Essex, CO3 3AD were appointed as Joint Liquidators of the Company. The Company is able to pay all its known liabilities in full.

Notice is hereby given that the creditors of the Company are required on or before 29 September 2023 to send in their names and addresses, particulars of their debts or claims and the names and addresses of their Solicitors (if any) to the undersigned Lee De'ath of Begbies Traynor (Central) LLP, Town Wall House, Balkerne Hill, Colchester, Essex, CO3 3AD the Joint Liquidator of the Company and, if so required by notice in writing to prove their debts or claims at such time and place as shall be specified in such notice, or in default shall be excluded from the benefit of any distribution made before such debts are proved.

Any person who requires further information may contact the Joint Liquidator by telephone on 01206 217900. Alternatively, enquiries can be made to Jenson Durbridge by email: jenson.durbridge@btguk.com or telephone 01206 984922.

*Lee De'ath,* Joint Liquidator 30 August 2023 Ag FJ53022

### HAYMAR ADVISORY LIMITED

(Company Number 09569473)

Registered office: Units 1-3 Hilltop Business Park, Devizes Road, Salisbury, Wiltshire, SP3 4UF

Principal trading address: Sefton End, Farthing Green Lane, Stoke Poges, Slough, SL2 4JQ

The Company was placed into members' voluntary liquidation on 25 August 2023 and on the same date, *Simon Lowes* (IP No. 9194) and *Stephen Mark Powell* (IP No. 009561) both of Begbies Traynor (Central) LLP, Units 1-3 Hilltop Business Park, Devizes Road, Salisbury, Wiltshire, SP3 4UF were appointed as Joint Liquidators of the Company.

Notice is hereby given that the Creditors of the Company are required on or before 3 October 2023 to send in their names and addresses, particulars of their debts or claims and the names and addresses of their Solicitors (if any) to the undersigned Simon Lowes of Begbies Traynor (Central) LLP, Units 1-3 Hilltop Business Park, Devizes Road, Salisbury, Wiltshire, SP3 4UF the Joint Liquidator of the Company and, if so required by notice in writing to prove their debts or claims at such time and place as shall be specified in such notice, or in default thereof shall be excluded from the benefit of any distribution made before such debts are proved.

This notice is purely formal, the Company is able to pay all its known creditors in full.

Any person who requires further information may contact the Joint Liquidator by telephone on 01722 435190. Alternatively enquiries can be made to Joanne Mkzie by e-mail at Joanne.Mkzie@btguk.com or by telephone on 01722 435190.

Simon Lowes, Joint Liquidator

29 August 2023 Ag FJ52899

(4430150)

### JUICE CONSULTING LTD

(Company Number 04530837)

Registered office: 5 Barnfield Crescent, Exeter, Devon, EX1 1QT Principal trading address: N/A

We, *David Gerard Kirk* (IP No. 8830) and *Daniel Robert Jeeves* (IP No. 26032) both of Kirks, 5 Barnfield Crescent, Exeter, EX1 1QT give notice that we were appointed joint liquidators of the above named company on 22 August 2023 by a resolution of members.

Notice is hereby given that the creditors of the above named company, which is being voluntarily wound up, are required, on or before 22 September 2023 to prove their debts by sending to the undersigned David Gerard Kirk of Kirks, 5 Barnfield Crescent, Exeter, EX1 1QT, the Joint Liquidator of the company, written statements of the amounts they claim to be due to them from the Company, and, if so requested, to provide such further details or produce such documentary evidence as may appear to the Joint Liquidator to be necessary.

Please note that this is a solvent liquidation and therefore the joint liquidator is entitled to make the distribution without regard to the claim of any person in respect of a debt not proved.

Further enquiries contact: Nathan Jeeves, Email: nathan@kirks.co.uk or telephone 01392 474303.

David Gerard Kirk, Joint Liquidator 25 August 2023

Ag FJ52871

(4430141)

### KACE CONSULTING LTD

(Company Number 12939748)

Registered office: 38 High Street, Watton, Thetford, Norfolk IP25 6AE Principal trading address: 55 Salisbury Road, Ealing, London W13 9TT

### NOTICE TO CREDITORS

The Company was placed into Member's Voluntary Liquidation on 21 August 2023 and is able to pay all its known creditors in full. Pursuant to Rule 4.182A of the Insolvency Rules 1986, NOTICE IS HEREBY GIVEN that the Liquidator intends to make a first and final distribution to remaining creditors of the above-named Company and that the last date for proving debts against the above-named Company, which is being voluntarily wound up, is 26 September 2023. Claims must be sent to the undersigned, Jeremy Charles Frost of Frost Group Limited, One Elmfield Park, Bromley BR1 1LU, the Liquidator of the Company. After 26 September 2023, the Liquidator may make that distribution without regard to the claim of any person in respect of a debt not already proved.

The Liquidator intends that after paying or providing for a final distribution in respect of the claims of all creditors who have proved their debts by the above date, the assets remaining in the hands of the Liquidator shall be distributed to the shareholder absolutely.

Liquidator: *Jeremy Charles Frost* (IP number 9091) of Frost Group Limited, One Elmfield Park Bromley BR1 1LU.

Date of Appointment: 21 August 2023

For further details contact Kelly Walford on 0345 260 0101 or at kellyw@frostbr.co.uk (4431764)

### LINDEN SP LIMITED

(Company Number 13130924)

Previous Name of Company: CANONBIE CAPITAL LIMITED; CANONBIE SP LIMITED LIMITED and CANONBIE SP LIMITED

Registered office: 29 Benham House Coleridge Gardens, 552 Kings Road, London SW10 0RD

Principal trading address: 29 Benham House Coleridge Gardens, 552 Kings Road, London SW10 0RD

NOTICE TO CREDITORS

The Company was placed into Members' Voluntary Liquidation on 21 August 2023 and is able to pay all its known creditors in full. Pursuant to Rule 4.182A of the Insolvency Rules 1986, NOTICE IS HEREBY GIVEN that the Liquidator intends to make a first and final distribution to remaining creditors of the above-named Company and that the last date for proving debts against the above-named Company, which is being voluntarily wound up, is 26 September 2023. Claims must be sent to the undersigned, Jeremy Charles Frost of Frost Group Limited, One Elmfield Park, Bromley BR1 1LU, the Liquidator of the Company.

After 26 September 2023, the Liquidator may make that distribution without regard to the claim of any person in respect of a debt not already proved.

The Liquidator intends that after paying or providing for a final distribution in respect of the claims of all creditors who have proved their debts by the above date, the assets remaining in the hands of the Liquidator shall be distributed to the shareholders absolutely.

Liquidator: *Jeremy Charles Frost* (IP number 9091) of Frost Group Limited, One Elmfield Park Bromley BR1 1LU.

Date of Appointment: 21 August 2023

For further details contact Kelly Walford on 0345 260 0101 or at kellyw@frostbr.co.uk (4431963)

### MOSS AND IVY LIMITED

### (Company Number 07379767)

Previous Name of Company: Kate of Kensington Ltd

Registered office: Begbies Traynor, The Officers' Mess Business Centre, Royston Road, Duxford, Cambridge CB22 4QH

Principal trading address: (Former) 1 Sopwith Crescent, Wickford, Essex SS11 8YU

The Company was placed into members' voluntary liquidation on 24 August 2023 and on the same date, *Wayne MacPherson* (IP No. 9445) and *Dominik Thiel-Czerwinke* (IP No. 009636) both of Begbies Traynor (Central) LLP, 1066 London Road, Leigh-on-Sea, Essex, SS9 3NA were appointed as Joint Liquidators of the Company.

Notice is hereby given that the Creditors of the Company are required on or before 12 October 2023 to send in their names and addresses, particulars of their debts or claims and the names and addresses of their solicitors (if any) to the undersigned Dominik Thiel-Czerwinke of Begbies Traynor (Central) LLP, 1066 London Road, Leigh on Sea, Essex SS9 3NA the Joint Liquidator of the Company and, if so required by notice in writing to prove their debts or claims at such time and place as shall be specified in such notice, or in default thereof shall be excluded from the benefit of any distribution made before such debts are proved.

This notice is purely formal, the Company is able to pay all its known creditors in full.

Any person who requires further information may contact the Joint Liquidators by telephone on 01223 495660. Alternatively enquiries can be made to Carol Wilson by e-mail:carol.wilson@btguk.com or by telephone on 01223 495660.

*Dominik Thiel Czerwinke,* Joint Liquidator 29 August 2023

(4430151)

Ag FJ52952

### MY PROMOTIONS LTD

### (Company Number 08278818)

Registered office: Currently The Old School House, West Street, Southwick, Fareham, PO17 6EA but shortly changing to c/o Begbies Traynor (Central) LLP, Units 1-3 Hilltop Business Park, Devizes Road, Salisbury SP3 4UF

Principal trading address: N/A

The Company was placed into members' voluntary liquidation on 29 August 2023 and on the same date, *Julie Anne Palmer* (IP No. 008835) and *Stephen Mark Powell* (IP No. 009561) both of Begbies Traynor (Central) LLP, Units 1-3 Hilltop Business Park, Devizes Road, Salisbury, Wiltshire, SP3 4UF were appointed as Joint Liquidators of the Company.

Notice is hereby given that the Creditors of the Company are required on or before 26 September 2023 to send in their names and addresses, particulars of their debts or claims and the names and addresses of their Solicitors (if any) to the undersigned Julie Palmer of Begbies Traynor (Central) LLP, Units 1-3 Hilltop Business Park, Devizes Road, Salisbury, Wiltshire, SP3 4UF the Joint Liquidator of the Company and, if so required by notice in writing to prove their debts or claims at such time and place as shall be specified in such notice, or in default thereof shall be excluded from the benefit of any distribution made before such debts are proved.

This notice is purely formal, the Company is able to pay all its known creditors in full.

Any person who requires further information may contact the Joint Liquidator by telephone on 01722 435190. Alternatively enquiries can be made to Nick Walsh by e-mail at Nicholas.Walsh@btguk.com or by telephone on 01722 435190.

J A Palmer, Joint Liquidator

29 August 2023

Ag FJ52960 (4430155)

# NOTICE TO CREDITORS (MEMBERS' VOLUNTARY LIQUIDATION) QM CAPITAL PARTNERS LIMITED

(Company Number 10737660)

Registered office: One, Fleet Place, London, EC4M 7WS

Principal trading address: One, Fleet Place, London, EC4M 7WS

NOTICE IS HEREBY GIVEN that Daryl Warwick and Mike Kienlen of Armstrong Watson LLP, James Watson House, Montgomery Way, Rosehill, Carlisle, Cumbria, CA1 2UU were appointed joint liquidators of the above company by the members on 24 August 2023.

NOTICE IS ALSO HEREBY GIVEN that the creditors of the above named company are required on or before 30 September 2023 to send their names and addresses with particulars of their debt to Daryl Warwick of Armstrong Watson LLP, James Watson House, Montgomery Way, Rosehill, Carlisle, Cumbria, CA1 2UU the joint liquidator of the said company. Failure to submit a claim by this date will result in you being excluded from any distributions paid. Please note that this is a solvent liquidation and all known creditors have been or will be paid in full.

Further Details Lene Davidson Tel No. 01228 690200

Name of Office Holder(s) Daryl Warwick and Mike Kienlen

Office Holder Number (s) 9500 and 9367

Address of Office Holder(s) James Watson House, Montgomery Way, Rosehill, Carlisle, Cumbria, CA1 2UU

Capacity Liquidators

Date of Appointment 24 August 2023

QUASSIA (PROPERTIES) LIMITED

(Company Number 01044492)

Previous Name of Company: Quassia (Joinery) Limited

Registered office: Quassia House, Bearwalden Industrial Estate, Wendons Ambo, Saffron Walden, Essex, CB11 4JX

Principal trading address: n/a

Notice is hereby given pursuant to Rule 14.28 of the INSOLVENCY (ENGLAND AND WALES) RULES 2016, that the Liquidator intends to declare a first and final distribution to all creditors of the Company within the period of two months from the last date for proving specified below.

Creditors who have not yet done so must prove their debts by sending their full names and addresses, particulars of their debts or claims, and the names and addresses of their solicitors (if any), to the Liquidator at Swift House, Ground Floor, 18 Hoffmanns Way, Chelmsford, Essex, CM1 1GU by no later than 21 September 2023 (the last date for proving).

The distribution may be made without regard to the claim of any person in respect of a debt not proved.

Note: The Directors of the Company have made a declaration of solvency and it is expected that all creditors will be paid in full.

In accordance with the provisions of Part 14 of the Insolvency (England and Wales) Rules 2016, a creditor whose debt is a 'small debt' not exceeding  $\pounds1,000$  is deemed to have proved for the purposes of determination and payment of this distribution and is not required to prove their debt as described by this notice unless advised otherwise by the Liquidator.

Date of Appointment: 21 August 2023

Office Holder Details: *Andrew Dix* (IP No. 009327) of AD Business Recovery Limited, Swift House, Ground Floor, 18 Hoffmanns Way, Chelmsford, Essex, CM1 1GU

For further details contact: The Liquidator, Tel: 01245 254791 or email info@adbusinessrecovery.co.uk. Alternative contact: Patricia Marsh.

Andrew Dix, Liquidator

29 August 2023 Ag FJ52958

(4430154)

### SKAGEN FINANCE LIMITED

(Company Number 09299298)

Registered office: c/o Mazars LLP, 30 Old Bailey, London, EC4M 7AU Principal trading address: Farncombe House, Farncombe, Broadway, WR12 7LJ

Notice is hereby given that creditors of the Company are required, on or before 26 September 2023, to prove their debts by delivering their proofs (in the format specified in Rule 14.4 of the INSOLVENCY (ENGLAND AND WALES) RULES 2016) to the Joint Liquidators at Mazars LLP, 30 Old Bailey, London, EC4M 7AU.

If so required by notice from the Joint Liquidators, creditors must produce any document or other evidence which the Joint Liquidators consider is necessary to substantiate the whole or any part of a claim. Note: The Directors of the Company have made a declaration of solvency and it is expected that all creditors will be paid in full. Date of Appointment: 24 August 2023

Office Holder Details: *Simon David Chandler* (IP No. 008822) and *Guy Robert Thomas Hollander* (IP No. 009233) both of Mazars LLP, 30 Old Bailey, London, EC4M 7AU

Further details contact: The Joint Liquidators, Tel: 0121 232 9578. Alternative contact: Dan Carr.

Simon David Chandler, Joint Liquidator

29 August 2023

(4430160)

### SONOCO CONSUMER PRODUCTS LTD

(Company Number 02409393)

Ag FJ52921

(4431863)

Registered office: C/O Sonoco Cores & Paper Ltd Stainland Board Mills, Holywell Green, Halifax, England, HX4 9PY

### COREPAK LTD

(Company Number 02304976) Registered office: C/O Sonoco Cores & Paper Ltd Stainland Board Mills, Holywell Green, Halifax, England, HX4 9PY

SONOCO CAPSEALS LINERS LTD

### (Company Number 00969584)

Registered office: C/O Sonoco Cores & Paper Ltd Stainland Board Mills, Holywell Green, Halifax, England, HX4 9PY

### SONOCO BOARD MILLS LTD

(Company Number 00196071)

Registered office: C/O Sonoco Cores & Paper Ltd Stainland Board Mills, Holywell Green, Halifax, England, HX4 9PY

### CAPSEALS LTD

(Company Number 00562588)

Registered office: C/O Sonoco Cores & Paper Ltd Stainland Board Mills, Holywell Green, Halifax, England, HX4 9PY

### CAPSEALS LINERS LTD (Company Number 00637766)

Registered office: C/O Sonoco Cores & Paper Ltd Stainland Board Mills, Holywell Green, Halifax, England, HX4 9PY

(4431362)

### SONOCO PACKAGING TAPES LTD

(Company Number 00354137)

Registered office: C/O Sonoco Cores & Paper Ltd Stainland Board Mills, Holywell Green, Halifax, England, HX4 9PY

SONOCO UK LEASING LTD

(Company Number 00423904)

Registered office: C/O Sonoco Cores & Paper Ltd Stainland Board Mills, Holywell Green, Halifax, England, HX4 9PY

TPT LIMITED

(Company Number 00577310)

Registered office: C/O Sonoco Cores & Paper Ltd Stainland Board Mills, Holywell Green, Halifax, England, HX4 9PY

NOTICE IS HEREBY GIVEN that the creditors of the above named companies, which are being voluntarily wound up, must send their full names and addresses (and those of their Solicitors, if any), together with full particulars of their debts or claims to emma.cray@pwc.com at PricewaterhouseCoopers LLP, One Chamberlain Square, Birmingham B3 3AX by 2 October 2023.

The distribution may be made without regard to the claim of any person in respect of a debt not proved.

Note: It is anticipated that all known Creditors will be paid in full.

Joint Liquidator: *Emma Cray* (IP number 17450) of PricewaterhouseCoopers LLP, One Chamberlain Square, Birmingham B3 3AX.

Joint Liquidator: *Steven Sherry* (IP number 19752) of PricewaterhouseCoopers LLP, 7 More London Riverside, London SE1 2RT.

Date of Appointment: 22 August 2023

For further details contact Dijesh Patel on 07802 660714 or at dijesh.v.patel@pwc.com

The Liquidators may act as controllers of personal data as defined by UK data protection law depending upon the specific processing activities undertaken. PricewaterhouseCoopers LLP may act as a processor on the instructions of the Liquidators. Personal data will be kept secure and processed only for matters relating to the Liquidators` appointment. Further details are available in the privacy statement on the PwC.co.uk website or by contacting the Liquidators. (4430972)

### SUIT YOURSELF LIMITED

(Company Number 01260244)

Registered office: (Former): The Roma Building, 32/38 Scrutton Street, Bishopsgate, London EC2A 4RQ

Principal trading address: (Former): The Roma Building, 32/38 Scrutton Street, Bishopsgate, London EC2A 4RQ

**Nature of Business:** Other letting and operating of own or leased real estate.

### Date by which proofs must be delivered: 22 September 2023

Notice is hereby given, pursuant to Rule 14.28 of the Insolvency (England and Wales) Rules 2016, that the liquidator of SUIT YOURSELF Limited ("the Company") (in members' voluntary liquidation) intends to make final distributions to creditors. Creditors are required to prove their debts on or before the date by which proofs must be delivered by sending full details of their claims to the liquidator. Creditors must also, if so requested by the liquidator, provide such further details and documentary evidence to support their claims as the liquidator deems necessary.

The intended distributions are final distributions and may be made without regard to the claim of any person in respect of a debt not proved by the date by which proofs must be delivered. Any creditor who has not delivered their proof by that date, or who increases the claim in their proof after that date, will not be entitled to disturb the intended final distributions. The liquidator intends that, after paying or providing for final distributions in respect of creditors who have proved their claims, all assets remaining in the liquidator's hands following the final distributions to creditors shall be distributed to the shareholders of the Company absolutely.

The director has made a statutory declaration that the Company is able to pay all its known liabilities in full with interest at the official rate.

### Date of Appointment: 18 August 2023

**Place to which proofs must be delivered**: Liquidator's Name and Address: Phillip Anthony Roberts (IP No. 6055) of Sterling Ford, Centurion Court, 83 Camp Road, St Albans, Herts AL1 5JN. Telephone: 01727 811161.

Liquidator: *Phillip Anthony Roberts* (IP number 6055) of Sterling Ford, Centurion Court, 83 Camp Road, St Albans, Herts AL1 5JN.

Date of Appointment: 18 August 2023 For further details contact: 01727 811161

### THE THREE PARTNERSHIP LIMITED

(Company Number 07709146)

Registered office: 207 Knutsford Road Grappenhall, Warrington, Cheshire, WA4 2QL

Principal trading address: 207 Knutsford Road Grappenhall, Warrington, Cheshire, WA4 2QL

Notice is hereby given that Creditors of the Company are required, on or before 25 September 2023, to prove their debts by delivering their proofs (in the format specified in Rule 14.4 of the INSOLVENCY (ENGLAND AND WALES) RULES 2016) to the Joint Liquidators at Clarke Bell Limited, 3rd Floor, The Pinnacle, 73 King Street, Manchester, M2 4NG.

If so required by notice from the Joint Liquidators, creditors must produce any document or other evidence which the Joint Liquidators consider is necessary to substantiate the whole or any part of a claim. Note: The Directors of the Company have made a declaration of solvency and it is expected that all creditors will be paid in full. Date of Appointment: 24 August 2023.

Office Holder Details: *John Paul Bell* (IP No. 8608) and *Toyah Marie Poole* (IP No. 9740) both of Clarke Bell Limited, 3rd Floor, The Pinnacle, 73 King Street, Manchester, M2 4NG

For further details contact: The Joint Liquidators, Email: mvl@clarkebell.com. Alternative contact: Thomas Bailey

John Paul Bell, Joint Liquidator

30 August 2023

Ag FJ53011

(4430130)

### **RESOLUTION FOR VOLUNTARY WINDING-UP**

### ACESSAR LIMITED

(Company Number 07607214)

Registered office: 36 Upton Road, Bexleyheath, Kent, DA6 8LH

Principal trading address: 36 Upton Road, Bexleyheath, Kent, DA6 8LH

Notice is hereby given that the following resolutions were passed on 17 August 2023, as a Special Resolution and an Ordinary Resolution respectively:

"That the Company be wound up voluntarily and that *John Paul Bell* (IP No. 8608) and *Toyah Marie Poole* (IP No. 9740) both of Clarke Bell Limited, 3rd Floor, The Pinnacle, 73 King Street, Manchester, M2 4NG be appointed Joint Liquidators for the purposes of the voluntary winding up of the Company."

For further details contact: The Joint Liquidators, Email: mvl@clarkebell.com. Alternative contact: Thomas Bailey

Mark Andrew Prockter, Director

17 August 2023 Ag FJ53012

ACNE PRODUCTION LIMITED

(Company Number 04289550)

Registered office: 1 New Street Square, London, EC4A 3HQ

Principal trading address: 1 New Street Square, London, EC4A 3HQ Notice is hereby given that the following resolutions were passed on

25 August 2023, as a special resolution and an ordinary resolution respectively:

"That the Company be wound up voluntarily and that *James Douglas Ernle Money* (IP No. 8999) and *Steven Edward Butt* (IP No. 9108) both of Rollings Butt LLP, 6 Snow Hill, London, EC1A 2AY be appointed as Joint Liquidators of the Company for the purposes of the voluntary winding up."

For further details contact: The Joint Liquidators, Tel: 020 7002 7960 or Email: mail@rollingsbutt.com. Alternative contact: Ali Yarwood.

Royston Paul Seaward, Director 25 August 2023

Ag FJ52928

(4430112)

(4430118)

### ALISON FOSTER LIMITED

(Company Number 10724246)

Registered office: DNS House, 382 Kenton Road, Harrow, Middlesex, HA3 8DP

Principal trading address: Flat 20, Wren Court, 12 The Avenue, Surbiton, Surrey, KT5 8 JB

Notice is hereby given that the following resolutions were passed on 23 August 2023, as a Special Resolution and an Ordinary Resolution respectively:

"That the Company be wound up voluntarily and that *John Paul Bell* (IP No. 8608) and *Toyah Marie Poole* (IP No. 9740) both of Clarke Bell Limited, 3rd Floor, The Pinnacle, 73 King Street, Manchester, M2 4NG be appointed Joint Liquidators for the purposes of the voluntary winding up of the Company."

For further details contact: The Joint Liquidators, Email: mvl@clarkebell.com. Alternative contact: Thomas Bailey

Dr Alison Fiona Foster, Director

23 August 2023

Ag FJ53006

(4430117)

### ARRANDCO AUDIT LIMITED (Company Number 04045321)

#### (Company Number 04045321) PERSONAL INVESTMENT MANAGEMENT LIMITED (Company Number 03792617)

RSM & CO (UK) LIMITED

(Company Number 04273907)

Registered office: (All) 6th Floor, 25 Farringdon Street, London EC4A 4AB

Principal trading address: (All) 6th Floor, 25 Farringdon Street, London EC4A 4AB

On 24 August 2023 the Members of the above named Companies, passed the following written resolutions as a Special and Ordinary Resolution respectively:

"That the Companies be wound up voluntarily and that *Henry Nicholas Page* (IP No. 12250) and *Edward James Ellis* (IP No. 25552) both of Mercer & Hole, 21 Lombard Street, London, EC3V 9AH be and are hereby appointed Joint Liquidators of the Companies, to act either jointly or severally."

Further information can be obtained from the Joint Liquidators or the case administrator, Harry Smart, at Mercer & Hole, 21 Lombard Street, London, EC3V 9AH or by email to Harry.Smart@mercerhole.co.uk. *John Taylor*, Director

30 August 2023

Ag FJ53020

(4430124)

### ASIAN SENSATION LIMITED

(Company Number 08240164) Trading Name: Man Ho Restaurant

Registered office: 12 Burford Close, Luton, England, LU3 4DS, United Kingdom

Principal trading address: 72 Dunstable Road, Luton, Bedfordshire, England, LU1 1EH, United Kingdom

SPECIAL AND ORDINARY RESOLUTIONS

(Pursuant to Sections 282 & 283 of the Companies Act 2006 and Section 84(1) of the Insolvency Act 1986)

At a General Meeting of the members of the above-named company, duly convened and held at 133 Golders Green Road, London, NW11 8HJ on 15 August 2023 the following resolutions were duly passed as special and ordinary resolutions:

### **Special Resolution**

i. "That the Company be wound-up voluntarily".

### **Ordinary Resolution**

i. "That Alex Martin-Sklan of Martin Sklan & Co, 133 Golders Green Road, London, NW11 8HJ, United Kingdom be and is hereby appointed liquidator of the company".

Names of Insolvency Practitioners: Alex Martin-Sklan

Address of Insolvency Practitioners: 133 Golders Green Road, London, NW11 8HJ, United Kingdom

IP Numbers: 5815

Date of Appointment: 15 August 2023

Contact Name: Tanjina Mahbub Email Address: ms@martinsklan.co.uk

Telephone Number: 020 8458 4433.

Yun Ngor Lau

Chair of the meeting Dated: 15 August 2023

### BOOTSTRAP SOFTWARE LIMITED

(Company Number 08858028)

Registered office: 80 Ashton Road, Denton, Manchester, M34 3JF Principal trading address: 80 Ashton Road, Denton, Manchester, M34 3JF

Notice is hereby given that the following resolutions were passed on 23 August 2023, as a Special Resolution and an Ordinary Resolution respectively:

"That the Company be wound up voluntarily and that *John Paul Bell* (IP No. 8608) and *Toyah Marie Poole* (IP No. 9740) both of Clarke Bell Limited, 3rd Floor, The Pinnacle, 73 King Street, Manchester, M2 4NG be appointed Joint Liquidators for the purposes of the voluntary winding up of the Company."

For further details contact: The Joint Liquidators, Email: mvl@clarkebell.com. Alternative contact: Thomas Bailey

Balaji Sivaram Iyer, Director

23 August 2023 Ag FJ53016

### BSI SPEEDWAY (HOLDINGS) LIMITED

(Company Number 05713374)

Registered office: Building 6, Chiswick Park, 566 Chiswick High Road, London, W4 5HR

Principal trading address: Building 6, Chiswick Park, 566 Chiswick High Road, London, W4 5HR

Notice is hereby given that the following resolutions were passed on 23 August 2023, as a special resolution and an ordinary resolution respectively:

"That the Company be wound up voluntarily and that *Steven Edward Butt* (IP No. 9108) and *James Douglas Ernle Money* (IP No. 8999) both of Rollings Butt LLP, 6 Snow Hill, London, EC1A 2AY be appointed as Joint Liquidators of the Company for the purposes of the voluntary winding up."

For further details contact: The Joint Liquidators, Tel: 020 7002 7960 or Email: mail@rollingsbutt.com. Alternative contact: Laurie'elle Rawlings

*Tyler Nevius,* Director 25 August 2023 Ag FJ52943

(4430113)

### CERIDWEN MAXIMUS LIMITED

(Company Number 08513411)

Registered office: W8a Knoll Business Centre, 325-327 Old Shoreham Road, Hove, BN3 7GS

Principal trading address: N/A

The following written resolutions of the sole member of Ceridwen Maximus Limited were passed on 23 August 2023 as Special and Ordinary Resolutions:

"That pursuant to section 84(1)(b) of the Insolvency Act 1986 the Company be wound up voluntarily and that pursuant to sections 84(1) and 91 of the Insolvency Act 1986 *Donald McNaught* (IP No. 9359) of Johnston Carmichael LLP, 227 West George Street, Glasgow, G2 2ND be appointed Liquidator of the Company for the purposes of winding up the Company's affairs and distributing its assets."

Further details contact: Gemma Mulgrew, Email: gemma.mulgrew@jcca.co.uk, Tel: 0141 286 1435.

Anoop Srivastava, Shareholder

28 August 2023 Ag FJ52873

(4430107)

### **CLEAR PEN SOLUTIONS LTD**

(Company Number 07000964)

Registered office: New Penderel House 4th Floor, 283-288 High Holborn, London WC1V 7HP

Principal trading address: New Penderel House 4th Floor, 283-288 High Holborn, London WC1V 7HP

Passed the 23 August 2023

The member of the above company has duly passed the following WRITTEN RESOLUTIONS: a SPECIAL RESOLUTION

(4430120)

"that the company be wound up voluntarily" and an ORDINARY RESOLUTION "that DANIEL JAMES (IP No. 26292) of Albert Goodman LLP, Goodwood House, Blackbrook Park Avenue, Taunton, Somerset TA1 2PX be appointed liquidator of the company for the purposes of the voluntary winding-up.

Further details contact: Telephone: 01823 210170

Email: steve.tucker@albertgoodman.co.uk

DAWN HARRIS

Director

(4431866)

### **CRAIC MANAGEMENT LIMITED**

(Company Number 09413653)

Registered office: Unit 14/15 Hall Farm Sywell Aerodrome, Sywell, Northamptonshire, NN6 0BN

Principal trading address: Yewtrees, 49 Dallington Road, Northampton, NN5 7BW

At a general meeting of the above-named company, duly convened and held at 9/10 Scirocco Close, Moulton Park, Northampton, NN3 6AP on 15 August 2023 at 04:15 pm the following resolutions were passed:

i. A special resolution that tl1e company be wound up voluntarily.

ii. An ordinary resolution that Gary Pettit of PBC Business Recovery and Insolvency, 9/1 0 Scirocco Close, Moulton Park, Northampton NN3 6AP be and is hereby appointed liquidator of the company and that he is authorised to act.

Liquidator: *Gary Pettit* (IP number 9066) of PBC Business Recovery & Insolvency Ltd,, 9-10 Scirocco Close, Moulton Park, Northampton, NN3 6AP .

Date of Appointment: 15 August 2023

For	further	details	contact	Natasha	Pink	at
natash	napink@pbc	businessrec	overy.co.uk			
Keith I	Huewen, Dir	rector			(4430	909)

### DMSI LIMITED

(Company Number 10401901)

Registered office: Flat 108b, Princes Park Manor Royal Drive, London N11 3FP

Principal trading address: N/A

The following written resolutions of the sole member of DMSI Limited were passed on 23 August 2023 as a Special and an Ordinary Resolution:

"That pursuant to section 84(1)(b) of the Insolvency Act 1986 the Company be wound up voluntarily and that pursuant to sections 84(1) and 91 of the Insolvency Act 1986 *Donald McNaught* (IP No. 9359) of Johnston Carmichael LLP, 227 West George Street, Glasgow, G2 2ND be appointed Liquidator of the Company for the purposes of winding up the Company's affairs and distributing its assets."

Further details contact: Angela Hindmarch, Email: angela.hindmarch@jcca.co.uk, tel: 0141 488 3474.

Jeremy Posner, Sole Shareholder

29 August 2023

Ag FJ52880

(4430109)

### EXPERT REVIEWS HOLDINGS LIMITED

(Company Number 13575110)

Registered office: 1st & 2nd Floors Wenlock Works, 1a Shepherdess Walk, England, London, N1 7QE

By written resolution of the sole member of the above-named company passed on 18 August 2023, the following Resolutions were duly passed, as a Special Resolution and as an Ordinary Resolution:

1 THAT the Company be wound up voluntarily.

Ordinary resolution

2 THAT Steven Sherry and Emma Cray of PricewaterhouseCoopers LLP, 7 More London Riverside, London SE1 2RT be and are hereby appointed Joint Liquidators of the Company for the purposes of such winding up, and any act required or authorised under any enactment to be done by the Joint Liquidators is to be done by all or any one or more of the persons for the time being holding office.

Joint Liquidator: *Steven Sherry* (IP number 19752) of PricewaterhouseCoopers LLP, 7 More London Riverside, London SE1 2RT.

Joint Liquidator: *Emma Cray* (IP number 17450) of PricewaterhouseCoopers LLP, One Chamberlain Square, Birmingham, B3 3AX.

Date of Appointment: 18 August 2023

For further details contact Elizabeth Colley at elizabeth.colley@pwc.com

The Liquidators may act as controllers of personal data as defined by UK data protection law depending upon the specific processing activities undertaken. PricewaterhouseCoopers LLP may act as a processor on the instructions of the Liquidators. Personal data will be kept secure and processed only for matters relating to the Liquidators` appointment. Further details are available in the privacy statement on the PwC.co.uk website or by contacting the Liquidators.

(4432030)

### GARETH BUNN CONSULTING LIMITED

(Company Number 06183487)

Registered office: 202 Sheen Lane, East Sheen, London SW14 8LF Principal trading address: 202 Sheen Lane, East Sheen, London SW14 8LF

At a General Meeting of the Company, duly convened and held at 202 Sheen Lane, East Sheen, London SW14 8LF on 23 August 2023 at 2:45pm the following Resolutions were passed:

That the Company be wound up voluntaril.y

That Jeremy Charles Frost (IP number: 9091) of Frost Group Limited, One Elmfield Park, Bromley BR1 1LU, be appointed Liquidator of the Company.

Liquidator: Jeremy Charles Frost (IP number 9091) of Frost Group Limited, One Elmfield Park Bromley BR1 1LU.

Date of Appointment: 23 August 2023

For further details contact Kelly Walford on 0345 260 0101 or at kellyw@frostbr.co.uk

Gareth Dennis Brotherton Bunn, Director (4432001)

### **GREAT HORKESLEY WINES LIMITED**

(Company Number 03755496)

Trading Name: The Wine Centre

Registered office: The Causeway, Great Horkesley, Colchester CO6 4BH.

Principal trading address: The Wine Centre, The Causeway, Great Horkesley, Colchester CO6 4BH.

At a General Meeting of the Members of the above named Company held on 29 August 2023 the following resolutions were duly passed as a Special Resolution and as an Ordinary Resolution respectively:

"That the Company be wound up voluntarily and that *Lee De'ath* (IP No. 9316) and *Tom Gardiner* (IP No. 17990) both of Begbies Traynor (Central) LLP, Town Wall House, Balkerne Hill, Colchester, Essex, CO3 3AD be and are hereby appointed as Joint Liquidators for the purposes of such winding up and that any power conferred on them by law or by this resolution, may be exercised and any act required or authorised under any enactment to be done by them, may be done any them jointly or by each of them alone."

Any person who requires further information may contact the Joint Liquidator by telephone on 01206 217900. Alternatively, enquiries can be made to Jenson Durbridge by email: jenson.durbridge@btguk.com or telephone 01206 984922.

Anthony Borges, Chair 29 August 2023 Ag FJ53022

(4430121)

### HAYMAR ADVISORY LIMITED

(Company Number 09569473)

Registered office: Units 1-3 Hilltop Business Park, Devizes Road, Salisbury, Wiltshire, SP3 4UF

Principal trading address: Sefton End, Farthing Green Lane, Stoke Poges, Slough, SL2 4JQ

At a General Meeting of the members of the above-named Company duly convened and held at Sefton End, Farthing Green Lane, Stoke Poges, Slough, SL2 4JQ on 25 August 2023, at which the required quorum was present, the following resolutions were duly passed: "That the Company be wound up voluntarily and that *Simon Lowes* (IP No. 9194) and *Stephen Mark Powell* (IP No. 009561) both of Begbies Traynor (Central) LLP, Units 1-3 Hilltop Business Park, Devizes Road, Salisbury, Wiltshire, SP3 4UF be and are hereby appointed as Joint Liquidators for the purposes of such winding up and that any power conferred on them by law or by this resolution, may be exercised and any act required or authorised under any enactment to be done by them, may be done by them jointly or by each of them alone."

Any person who requires further information may contact the Joint Liquidator by telephone on 01722 435190. Alternatively enquiries can be made to Joanne Mkzie by e-mail at Joanne.Mkzie@btguk.com or by telephone on 01722 435190.

*Paul Waller,* Chair 25 August 2023 Ag FJ52899

(4430111)

### JUICE CONSULTING LTD

(Company Number 04530837)

Registered office: Garden Flat, 15c Apsley Road, Clifton, Bristol, BS8 2SH

Principal trading address: N/A

At a General Meeting of the members of the above named company, duly convened and held at Garden Flat, 15c Apsley Road, Clifton, Bristol, BS8 2SH on 22 August 2023, at 10.00 am the following resolutions were duly passed as Special and as Ordinary Resolutions: "That the Company be wound up voluntarily and that *David Gerard Kirk* (IP No. 8830) and *Daniel Robert Jeeves* (IP No. 26032) both of Kirks, 5 Barnfield Crescent, Exeter, EX1 1QT be and are hereby appointed joint liquidators of the company."

Further enquiries contact: Nathan Jeeves, Email: nathan@kirks.co.uk or telephone 01392 474303.

Paul Griffee, Chair 22 August 2023 Ag FJ52871

(4430108)

### KACE CONSULTING LTD

(Company Number 12939748)

Registered office: 38 High Street, Watton, Thetford, Norfolk IP25 6AE Principal trading address: 55 Salisbury Road, Ealing, London W13 9TT

At a General Meeting of the Company, duly convened and held at 38 High Street, Watton, Thetford, Norfolk IP25 6AE on 21 August 2023 at 10:45am the following Resolutions were passed:

That the Company be wound up voluntarily.

That Jeremy Charles Frost (IP number: 9091) of Frost Group Limited, One Elmfield Park, Bromley BR1 1LU, be appointed Liquidator of the Company.

Liquidator: *Jeremy Charles Frost* (IP number 9091) of Frost Group Limited, One Elmfield Park Bromley BR1 1LU.

Date of Appointment: 21 August 2023

For further details contact Kelly Walford on 0345 260 0101 or at kellyw@frostbr.co.uk

Karl Gregory William Blight, Director (4431762)

### LDN CUSTOMER SOLUTIONS LIMITED

(Company Number 07095044)

Registered office: Old World Cottage, Ketton Road, Hambleton, Rutland, LE15 8TH

Passed by written resolutions on 17 August 2023 the following resolutions were passed as a Special Resolution and an Ordinary Resolution respectively:

1. That the Company be placed into Members Voluntary Liquidation.

2. That Michael Patrick Durkan of Durkan Cahill, Suite G2 Montpellier House, Montpellier Drive, Cheltenham, GL50 1TY, be and is hereby appointed Liquidator for the purpose of the winding-up of the Company.

Liquidator: *Michael Durkan* (IP number 9583) of Durkan Cahill, Suite G2 Montpellier House, Montpellier Drive, Cheltenham, GL50 1TY.

Date of Appointment: 17 August 2023

For further details contact: 01242 250811 or at mpd@durkancahill.com (4431618)

### LINDEN SP LIMITED

(Company Number 13130924)

Previous Name of Company: CANONBIE CAPITAL LIMITED; CANONBIE SP LIMITED LIMITED and CANONBIE SP LIMITED Registered office: 29 Benham House Coleridge Gardens, 552 Kings

Road, London SW10 0RD Principal trading address: 29 Benham House Coleridge Gardens, 552

Kings Road, London SW10 0RD At a General Meeting of the Company, duly convened and held at 29 Benham House Coleridge Gardens, 552 Kings Road, London SW10 ORD on 21 August 2023 at 10:15am the following Resolutions were passed:

That the Company be wound up voluntarily.

That Jeremy Charles Frost (IP number: 9091) of Frost Group Limited, One Elmfield Park, Bromley BR1 1LU, be appointed Liquidator of the Company.

Liquidator: Jeremy Charles Frost (IP number 9091) of Frost Group Limited, One Elmfield Park Bromley BR1 1LU.

Date of Appointment: 21 August 2023

For further details contact Kelly Walford on 0345 260 0101 or at kellyw@frostbr.co.uk

Alan Napier Scott Aimers, Director (4431962)

### MB PAINTING CONTRACTOR LIMITED

(Company Number 09143622)

Registered office: 45 Parkfields, Roydon, Harlow, CM19 5JA Principal trading address: 45 Parkfields, Roydon, Harlow, CM19 5JA Place of meeting: 45 Parkfields, Roydon, Harlow, Essex, CM19 5JA. Date of meeting: 21 August 2023.

Time of meeting: 11:00 am.

At a general meeting of the members of the above-named Company, duly convened and held at the place, time and on the date specified above, the following resolutions were passed as a special resolution, and an ordinary resolution respectively: that the Company be wound up voluntarily, and the Liquidator specified below be appointed Liquidator of the Company for the purposes of the voluntary winding up.

Date of Appointment: 21 August 2023

Liquidator's Name and Address: *Richard Frank Simms* (IP No. 9252) of F A Simms & Partners Limited, Alma Park, Woodway Lane, Claybrooke Parva, Lutterworth, Leicestershire, LE17 5FB. Telephone: 01455 555 444.

For further information contact Keeley Skinner at the offices of F A Simms & Partners Limited on 01455 555 444, or keeleys@fasimms.com. 22 August 2023 (4429202)

-

### MOSS AND IVY LIMITED

(Company Number 07379767)

Previous Name of Company: Kate of Kensington Ltd

Registered office: 1 Sopwith Crescent, Wickford, Essex SS11 8YU

Principal trading address: (Formerly) 1 Sopwith Crescent, Wickford, Essex SS11  $8\mathrm{YU}$ 

At a General Meeting of the members of the above named Company held on 24 August 2023 the following Resolutions were passed as a Special Resolution and as an Ordinary Resolution respectively:

"That the Company be wound up voluntarily and that *Wayne MacPherson* (IP No. 9445) and *Dominik Thiel-Czerwinke* (IP No. 009636) both of Begbies Traynor (Central) LLP, 1066 London Road, Leigh-on-Sea, Essex, SS9 3NA be and are hereby appointed as joint liquidators for the purposes of such winding up and that any power conferred on them by law or by this resolution, may be exercised and any act required or authorised under any enactment to be done by them, may be done by them jointly or by each of them alone."

Any person who requires further information may contact Carol Wilson by telephone on 01223 495660 or Email: carol.wilson@btguk.com *Kate Helen Thomas,* Chair

24 August 2023

Ag FJ52952

(4430114)

### MY PROMOTIONS LTD

(Company Number 08278818)

Registered office: Currently The Old School House, West Street, Southwick, Fareham, PO17 6EA but shortly changing to c/o Begbies Traynor (Central) LLP, Units 1-3 Hilltop Business Park, Devizes Road, Salisbury SP3 4UF

Principal trading address: N/A

At a general meeting of the members of the above named Company held on 29 August 2023 the following resolutions were passed as a special resolution and as an ordinary resolution respectively:

"That the Company be wound up voluntarily and that *Julie Anne Palmer* (IP No. 008835) and *Stephen Mark Powell* (IP No. 009561) both of Begbies Traynor (Central) LLP, Units 1-3 Hilltop Business Park, Devizes Road, Salisbury, Wiltshire, SP3 4UF be and are hereby appointed as Joint Liquidators for the purposes of such winding up and that any power conferred on them by law or by this resolution, may be exercised and any act required or authorised under any enactment to be done by them, may be done by them jointly or by each of them alone."

Any person who requires further information may contact Ryan Cullinane by telephone on 023 8021 9820 or by email at ryan.cullinane@btguk.com.

Daniel David Francis, Director 29 August 2023 Ag FJ52960

(4430116)

### RESOLUTIONS QM CAPITAL PARTNERS LIMITED

(Company Number 10737660)

Registered office: One, Fleet Place, London, EC4M 7WS

Principal trading address: One, Fleet Place, London, EC4M 7WS

At a general meeting of the members of the above named company, duly convened and held at James Watson House, Montgomery Way, Rosehill, Carlisle, Cumbria, CA1 2UU on 24 August 2023 the following special resolution was duly passed:

'That the Company be wound-up voluntarily and that Daryl Warwick and Mike Kienlen of Armstrong Watson LLP, James Watson House, Montgomery Way, Rosehill, Carlisle, Cumbria, CA1 2UU are hereby appointed joint liquidators for the purpose of the winding up'

Further Details Lene Davidson Tel No. 01228 690200

Name(s) of Office Holder(s) Daryl Warwick and Mike Kienlen

Office Holder Number(s) 9500 and 9367

Address of Office Holder(s) James Watson House, Montgomery Way, Rosehill, Carlisle, Cumbria, CA1 2UU and James Watson House, Montgomery Way, Rosehill, Carlisle, Cumbria, CA1 2UU Capacity Joint Liquidator

Andrew Adams

(4431867)

### QUASSIA (PROPERTIES) LIMITED

(Company Number 01044492)

Registered office: Quassia House, Bearwalden Industrial Estate, Wendons Ambo, Saffron Walden, Essex, CB11 4JX

Principal trading address: n/a

Notice is hereby given that the following resolutions were passed on 21 August 2023, as a special and ordinary resolution respectively:

"That the Company be wound up voluntarily and that *Andrew Dix* (IP No. 009327) of AD Business Recovery Limited, Swift House, Ground Floor, 18 Hoffmanns Way, Chelmsford, Essex, CM1 1GU be appointed Liquidator for the purposes of such voluntary winding up."

For further details contact: The Liquidator, Tel: 01245 254791 or email info@adbusinessrecovery.co.uk. Alternative contact: Patricia Marsh. *Lance Sykes, Director* 

21 August 2023

Ag FJ52958

(4430115)

SKAGEN FINANCE LIMITED

### (Company Number 09299298)

Registered office: c/o Mazars LLP, 30 Old Bailey, London, EC4M 7AU Principal trading address: Farncombe House, Farncombe, Broadway, WR12 7LJ

Notice is hereby given that the following resolutions were passed on 24 August 2023, as a special resolution and an ordinary resolution respectively:

"That the Company be wound up voluntarily and that *Simon David Chandler* (IP No. 008822) and *Guy Robert Thomas Hollander* (IP No. 009233) both of Mazars LLP, 30 Old Bailey, London, EC4M 7AU be appointed as Joint Liquidators for the purposes of such voluntary winding up."

Further details contact: The Joint Liquidators, Tel: 0121 232 9578. Alternative contact: Dan Carr.

Paul Nicholas Dillon, Director 29 August 2023 Ag FJ52921

(4430110)

### SONOCO CONSUMER PRODUCTS LTD

(Company Number 02409393)

Registered office: C/O Sonoco Cores & Paper Ltd Stainland Board Mills, Holywell Green, Halifax, England, HX4 9PY

### COREPAK LTD

(Company Number 02304976)

Registered office: C/O Sonoco Cores & Paper Ltd Stainland Board Mills, Holywell Green, Halifax, England, HX4 9PY

SONOCO CAPSEALS LINERS LTD

(Company Number 00969584)

Registered office: C/O Sonoco Cores & Paper Ltd Stainland Board Mills, Holywell Green, Halifax, England, HX4 9PY

SONOCO BOARD MILLS LTD

(Company Number 00196071)

Registered office: C/O Sonoco Cores & Paper Ltd Stainland Board Mills, Holywell Green, Halifax, England, HX4 9PY

### CAPSEALS LTD

(Company Number 00562588) Registered office: C/O Sonoco Cores & Paper Ltd Stainland Board

Mills, Holywell Green, Halifax, England, HX4 9PY

CAPSEALS LINERS LTD

### (Company Number 00637766)

Registered office: C/O Sonoco Cores & Paper Ltd Stainland Board Mills, Holywell Green, Halifax, England, HX4 9PY

SONOCO PACKAGING TAPES LTD

(Company Number 00354137)

Registered office: C/O Sonoco Cores & Paper Ltd Stainland Board Mills, Holywell Green, Halifax, England, HX4 9PY

SONOCO UK LEASING LTD

(Company Number 00423904)

Registered office: C/O Sonoco Cores & Paper Ltd Stainland Board Mills, Holywell Green, Halifax, England, HX4 9PY

### TPT LIMITED

(Company Number 00577310)

Registered office: C/O Sonoco Cores & Paper Ltd Stainland Board Mills, Holywell Green, Halifax, England, HX4 9PY

By written resolution of the sole Member of the above-named companies passed on 22 August 2023, the following Resolutions were duly passed, as a Special Resolution and as an Ordinary Resolution: 1 THAT the Companies be wound up voluntarily.

Ordinary resolution

2 THAT Emma Cray and Steven Sherry of PricewaterhouseCoopers LLP, One Chamberlain Square, Birmingham B3 3AX be and are hereby appointed Joint Liquidators of the Companies for the purposes of such winding up, and any act required or authorised under any enactment to be done by the Joint Liquidators is to be done by all or any one or more of the persons for the time being holding office.

Joint Liquidator: *Emma Cray* (IP number 17450) of PricewaterhouseCoopers LLP, One Chamberlain Square, Birmingham B3 3AX.

Joint Liquidator: *Steven Sherry* (IP number 19752) of PricewaterhouseCoopers LLP, 7 More London Riverside, London SE1 2RT.

Date of Appointment: 22 August 2023

For further details contact Dijesh Patel on 07802 660714 or at dijesh.v.patel@pwc.com

The Liquidators may act as controllers of personal data as defined by UK data protection law depending upon the specific processing activities undertaken. PricewaterhouseCoopers LLP may act as a processor on the instructions of the Liquidators. Personal data will be kept secure and processed only for matters relating to the Liquidators` appointment. Further details are available in the privacy statement on the PwC.co.uk website or by contacting the Liquidators.

(4430971)

### SUIT YOURSELF LIMITED

### (Company Number 01260244)

Registered office: (Former): The Roma Building, 32/38 Scrutton Street, Bishopsgate, London EC2A 4RQ

Principal trading address: (Former): The Roma Building, 32/38 Scrutton Street, Bishopsgate, London EC2A 4RQ

At a General Meeting of the above named Company, duly convened, and held at 3:00pm on 18 August 2023 at The Roma Building, 32/38 Scrutton Street, Bishopsgate, London EC2A 4RQ the subjoined RESOLUTIONS were duly passed, viz:-

**RESOLUTIONS:** 

1. As a Special Resolution, THAT the Company be wound up voluntarily.

2. As an Ordinary Resolution, THAT Phillip Anthony Roberts of Sterling Ford be and is hereby appointed liquidator for the purposes of such winding up and that he shall be remunerated and his disbursements paid in accordance with Sterling Ford's Terms & Scale of Charges MVL(C) dated 1 March 2023.

3. As a Special Resolution, THAT in accordance with the provisions of the Company's Articles of Association, the liquidator be and is hereby authorised to divide among the members in specie all or any part of the company's assets.

Liquidator: *Phillip Anthony Roberts* (IP number 6055) of Sterling Ford, Centurion Court, 83 Camp Road, St Albans, Herts AL1 5JN.

Date of Appointment: 18 August 2023

For further details contact: 01727 811161

Sarah Bard (4431360)

### THE THREE PARTNERSHIP LIMITED

(Company Number 07709146)

Registered office: 207 Knutsford Road Grappenhall, Warrington, Cheshire, WA4 2QL

Principal trading address: 207 Knutsford Road Grappenhall, Warrington, Cheshire, WA4 2QL

Notice is hereby given that the following resolutions were passed on 24 August 2023, as a Special Resolution and an Ordinary Resolution respectively:

"That the Company be wound up voluntarily and that *John Paul Bell* (IP No. 8608) and *Toyah Marie Poole* (IP No. 9740) both of Clarke Bell Limited, 3rd Floor, The Pinnacle, 73 King Street, Manchester, M2 4NG be appointed Joint Liquidators for the purposes of the voluntary winding up of the Company."

For further details contact: The Joint Liquidators, Email: mvl@clarkebell.com. Alternative contact: Thomas Bailey

Sarah Louise Barwell, Director

24 August 2023 Ag FJ53011

(4430119)

### **Partnerships**

### **DISSOLUTION OF PARTNERSHIP**

### SERAPHIM SPACE LP (the "Partnership")

### **DISSOLUTION OF A PARTNERSHIP**

Notice is hereby given, pursuant to section 10 of the Limited Partnerships Act 1907, that the Partnership, a limited partnership registered in England and Wales with number LP017106 has been dissolved with effect from 31 March 2022 and that the general partner, Seraphim Space (General Partner) LLP, ceased to be the general partner of the Partnership. (4431887)

### TRANSFER OF INTEREST

FOR A TRANSFER OF LIMITED PARTNERSHIP INTEREST Transfer of Interest in BRIDGEPOINT EUROPE IV 'E' L.P. (Registered No. LP012526) Notice is hereby given by the general partner, pursuant to Section 10 of the Limited Partnerships Act 1907, that as of 18 August 2023, PRIMAPEN KMGK has transferred its interest as a limited partner in Bridgepoint Europe IV 'E' L.P. (the "**Partnership**"), a limited partnership registered in England and Wales with registration number LP012526, to STRATEGIC PARTNERS IX INVESTMENTS L.P. which has been admitted as a new limited partner in the Partnership.

(4431888)

# MACQUARIE INFRASTRUCTURE DEBT (UK INFLATION LINKED) FUND 2 L.P.

(Registered No. LP017476)

Notice is hereby given, pursuant to section 10 of the Limited Partnerships Act 1907, that on 25 August 2023, Mitchells & Butlers CIF Limited in its capacity as trustee of the Mitchell & Butlers Common Investment Fund transferred its limited partnership interests in Macquarie Infrastructure Debt (UK Inflation Linked) Fund 2 L.P., a limited partnership carrying on business at North Suite, First Floor, Regency Court, Glategny Esplanade, St Peter Port, Guernsey, GY1 1WW (the "Partnership"), to Falkirk Council (acting as administering authority of Falkirk Council Pension Fund), Fife Council (acting as administering authority of Fife Council Pension Fund) and Lothian Pension Fund (acting through its administering authority, the City of Edinburgh Council), as new limited partners in the Partnership.

Ben Morgan, Director, MIDF UK2 Guernsey GP Limited, as general<br/>partner of Macquarie Infrastructure Debt (UK Inflation Linked) Fund 2<br/>L.P.(4431890)

# PEOPLE

### **Personal insolvency**

### **BANKRUPTCY ORDERS**

### ABRAHAM, THOMAS DENNIS

2 Sackville Close, Swindon, SN3 3EJ Birth details: 18 March 1987 Thomas Dennis Abraham, Currently not working, of 2 Sackville Close, Swindon, Wiltshire, SN3 3EJ In the Office of the Adjudicator No 5151398 of 2023 Date of Filing Petition: 22 August 2023 Bankruptcy order date: 24 August 2023 Time of Bankruptcy Order: 00:00 Whether Debtor's or Creditor's PetitionDebtor's *T Keller* PO Box 18416, Birmingham, B2 2UB, telephone: 0300 6780 016 Capacity of office holder(s): Trustee 24 August 2023 (4430432)

### **BERAHINO, SAIDO**

37 Old Penkridge Road, CANNOCK, Staffordshire, WS11 1HX Saido Berahino of 37 Old Penkridge Road, CANNOCK, Staffordshire, WS11 1HX In the County Court at Walsall No 3 of 2023 Date of Filing Petition: 11 January 2023 Bankruptcy order date: 21 August 2023 Time of Bankruptcy Order: 11:26 Whether Debtor's or Creditor's PetitionCreditor's Name and address of petitioner: CLIFFORD CLEMENS46, 40210, DUSSELDORF Y Hill 3rd Floor, Cannon House, 18 The Priory Queensway, Birmingham, B4 6FD, telephone: 0300 678 0016 Capacity of office holder(s): Trustee 21 August 2023 (4430424)

### CAMPBELL, STEPHANIE JAYNE

10 Ruby Close, Stoke-On-Trent, ST6 8HS Birth details: 29 September 1987 Stephanie Jayne Campbell, Self Employed, of 10 Ruby Close, Stokeon-Trent, Staffordshire, ST6 8HS and carrying on business as Stephanie Campbell from Garden of Eden, 7 Liverpool Road, Stoke, Staffordshire, ST4 1AR In the Office of the Adjudicator No 5151412 of 2023 Date of Filing Petition: 22 August 2023 Bankruptcy order date: 24 August 2023 Time of Bankruptcy Order: 00:00 Whether Debtor's or Creditor's PetitionDebtor's T Keller PO Box 18416, Birmingham, B2 2UB, telephone: 0300 6780 016 Capacity of office holder(s): Trustee 24 August 2023 (4430429)

### DAVIES, LAURA CLARE

58 Town End Street, Godalming, GU7 1BH Birth details: 3 January 1985 Laura Clare Davies, Self Employed, of 58 Town End Street, Godalming, Surrey, GU7 1BH and carrying on business as Laura Davies In the Office of the Adjudicator No 5151380 of 2023 Date of Filing Petition: 21 August 2023 Bankruptcy order date: 23 August 2023 Time of Bankruptcy Order: 00:00 Whether Debtor's or Creditor's PetitionDebtor's L Cook PO Box 18416, Birmingham, B2 2UB, telephone: 0300 678 0016 Capacity of office holder(s): Trustee (4430423) 23 August 2023

### FRAMPTON, MICHAEL JAMES

10 Council Houses, Henley, Dorchester, DT2 7BW Birth details: 24 July 1987 Michael James Frampton, Employed, of 10 Council Houses, Henley, Dorchester, DT2 7BW, formerly of 26 Brisbane Road, Weymouth, Dorset, DT3 6RD In the Office of the Adjudicator No 5151378 of 2023 Date of Filing Petition: 21 August 2023 Bankruptcy order date: 23 August 2023 Time of Bankruptcy Order: 00:00 Whether Debtor's or Creditor's PetitionDebtor's T Keller PO Box 18416, Birmingham, B2 2UB, telephone: 0300 6780 016 Capacity of office holder(s): Trustee (4430421)23 August 2023

### HARRIS, DAVINIA JANE

19 Rudge Close, Hardwicke, Gloucester, GL2 4ET Birth details: 24 February 1978 Davinia Jane Harris, Employed, of 19 Rudge Close, Hardwicke, Gloucester, GL2 4ET formerly of 21 The Causeway, Gloucester, GL2 4LD In the Office of the Adjudicator No 5151386 of 2023 Date of Filing Petition: 22 August 2023 Bankruptcy order date: 24 August 2023 Time of Bankruptcy Order: 00:00 Whether Debtor's or Creditor's PetitionDebtor's T Keller PO Box 18416, Birmingham, B2 2UB, telephone: 0300 6780 016 Capacity of office holder(s): Trustee 24 August 2023 (4430426)

### HAYDEN, OLIVER

77 Aust Crescent, Bulwark, Chepstow, NP16 5NL Birth details: 26 April 1985 Oliver Hayden, Self Employed, of 77 Aust Crescent, Bulwark, Chepstow, Monmouthshire, NP16 5NL and carrying on business as The Five Alls Hocker Hill Street, Chepstow, Monmouthshire, NP16 5ER In the Office of the Adjudicator No 5151394 of 2023 Date of Filing Petition: 22 August 2023 Bankruptcv order date: 24 August 2023 Time of Bankruptcy Order: 00:00 Whether Debtor's or Creditor's PetitionDebtor's S Baxter PO Box 18416, Birmingham, B2 2UB, telephone: 0300 678 0016 Capacity of office holder(s): Trustee 29 August 2023 (4430430)

HODGSON, ADAM MARCUS
74 Minors Crescent, Darlington, DL3 0DP
Birth details: 4 October 1992
Adam Marcus Hodgson, Employed, of 74 Minors Crescent,
Darlington, County Durham, DL3 0DP formerly of 14 Bedford Street,
Darlington, County Durham, DL1 5JX
In the Office of the Adjudicator
No 5151382 of 2023
Date of Filing Petition: 21 August 2023
Bankruptcy order date: 23 August 2023
Time of Bankruptcy Order: 00:00
Whether Debtor's or Creditor's PetitionDebtor's
T Keller PO Box 18416, Birmingham, B2 2UB, telephone: 0300 6780
016
Capacity of office holder(s): Trustee
23 August 2023 (4430422)

### HORRIGAN, NIGEL KEVIN

Shripney Road, Bognor Regis, PO22 9NE Birth details: 22 May 1967 Nigel Kevin Horrigan, Employed, of Shripney Road, Bognor Regis, West Sussex, PO22 9NE formerly of Flat 4, 118 High Street, Selsey, Chichester, West Sussex, PO20 0QG In the Office of the Adjudicator No 5151348 of 2023 Date of Filing Petition: 20 August 2023 Bankruptcy order date: 23 August 2023 Time of Bankruptcy Order: 00:00 Whether Debtor's or Creditor's PetitionDebtor's K Jackson PO Box 18416, Birmingham, B2 2UB, telephone: 03006780016 Capacity of office holder(s): Trustee (4430418) 29 August 2023

### JONES, SAMANTHA JANE

14 Austen Road, Farnborough, GU14 8LE Birth details: 25 September 1979 Samantha Jane Jones, Currently not working, of 14 Austen Road, Farnborough, Hampshire, GU14 8LE In the Office of the Adjudicator No 5151374 of 2023 Date of Filing Petition: 21 August 2023 Bankruptcy order date: 23 August 2023 Time of Bankruptcy Order: 00:00 Whether Debtor's or Creditor's PetitionDebtor's *T Keller* PO Box 18416, Birmingham, B2 2UB, telephone: 0300 6780 016 Capacity of office holder(s): Trustee 23 August 2023 (4430420)

### MCCULLEY, TOM JAMES

Flat 1, The Hawthorns, Westwells Road, Corsham, SN13 9RA Birth details: 20 May 1980 Tom James Mcculley, Employed, of Flat 1, The Hawthorns, Westwells Road, Corsham, Wiltshire, SN13 9RA formerly of 17 Peregrine Court, Calne, Wiltshire, SN11 9RY and formerly of 34 Pine Close, Corsham, Wiltshire, SN13 0LB In the Office of the Adjudicator No 5151384 of 2023 Date of Filing Petition: 22 August 2023 Bankruptcy order date: 24 August 2023 Time of Bankruptcy Order: 00:00 Whether Debtor's or Creditor's PetitionDebtor's T Keller PO Box 18416, Birmingham, B2 2UB, telephone: 0300 6780 016 Capacity of office holder(s): Trustee 24 August 2023 (4430425)

### PARROTT, DANIELLE

2 Andrews Close, Buckhurst Hill, IG9 5BL Birth details: 28 January 1979 DANIELLE PARROTT - Currently self-employed, also known as Danielle Holland and as Danielle Doran, of 2 Andrews Close, Buckhurst Hill, Essex IG9 5BL, formerly of 11a Queens Road, Buckhurst Hill, Essex IG9 5BZ, currently trading as Danielle Parrott, a dogwalker, at 2 Andrews Close, Buckhurst Hill, Essex IG9 5BL, and formerly employed as a company director In the Office of the Adjudicator No 5151156 of 2023 Date of Filing Petition: 14 August 2023 Bankruptcy order date: 16 August 2023 Time of Bankruptcy Order: 00:00 Whether Debtor's or Creditor's PetitionDebtor's T Keller PO Box 18416, Birmingham, B2 2UB, telephone: 0300 6780 016 Capacity of office holder(s): Trustee 16 August 2023 (4430416) SIDDALL, SUSAN ANN 72 Sefton Drive, Lancaster, LA1 2QD Birth details: 19 June 1962 Susan Ann Siddall, Currently not working, of 72 Sefton Drive,

Lancaster, Lancashire, LA1 2QD In the Office of the Adjudicator No 5151390 of 2023 Date of Filing Petition: 22 August 2023 Bankruptcy order date: 24 August 2023 Time of Bankruptcy Order: 00:00 Whether Debtor's or Creditor's PetitionDebtor's *T Keller* PO Box 18416, Birmingham, B2 2UB, telephone: 0300 6780 016 Capacity of office holder(s): Trustee

(4430427)

24 August 2023

TAYLOR, ALEXANDER JAMES 28 Lock Keepers Court, Blackweir Terrace, Cardiff, CF10 3EZ Birth details: 12 August 1989 Alexander James Taylor, Employed, Director, of 28 Lock Keepers Court, Blackweir Terrace, Cardiff, CF10 3EZ formerly of 2 Beatty Court, Schooner Way, Cardiff, CF10 4DU In the Office of the Adjudicator No 5151388 of 2023 Date of Filing Petition: 22 August 2023 Bankruptcy order date: 24 August 2023 Time of Bankruptcy Order: 00:00 Whether Debtor's or Creditor's PetitionDebtor's T Keller PO Box 18416, Birmingham, B2 2UB, telephone: 0300 6780 016 Capacity of office holder(s): Trustee 24 August 2023 (4430428)

### VICKERS, ROBERT JAMES

35 Leslie Avenue, Yeadon, LEEDS, LS19 7XH Birth details: 19 December 1972 ROBERT JAMES VICKERS, self-employed landscape gardener of 35 Leslie Avenue, Yeadon, LEEDS LS19 7XH In the County Court at Leeds No 47 of 2023 Date of Filing Petition: 10 July 2023 Bankruptcv order date: 23 August 2023 Time of Bankruptcy Order: 12:13 Whether Debtor's or Creditor's PetitionCreditor's Name and address of petitioner: PHILIP BOOTHBooth & Co Ltd, Coopers House, Intake Lane, OSSETT, WF5 0RG L Burgess 3rd Floor, 1 City Walk, Leeds, LS11 9DA, telephone: 0300 678 0016 Capacity of office holder(s): Official Receiver 23 August 2023 (4430412)

### DISCHARGE FROM BANKRUPTCY

### MADEJ. ARTUR

319 Chichester Road, Portsmouth, PO2 0AW Birth details: 12 July 1969 Artur Madej, Employed, of 319 Chichester Road, Portsmouth, PO2 0AW In the Office of the Adjudicator No 5093620 of 2019 Bankruptcy order date: 19 December 2019 Date of certificate of discharge: 18 August 2023 D Brogan 2nd Floor, 3 Piccadilly Place, London Road, Manchester, M1 3BN, telephone: 0161 234 8500 Capacity of office holder(s): Official Receiver 19 December 2019 (4430431)

### **MEETING OF CREDITORS**

In the Office of the Adjudicator

### No 5122954 of 2021 SCOT MICHAEL WILLOCK

### In Bankruptcv

Residential Address: 14 Winward Way, Birmingham, B36 0PP. Date of Birth: 21 March 1974. Occupation: Unemployed

This Notice is given under Rule 15.13 of the Insolvency (England & Wales) Rules 2016 ("the Rules"). It is delivered by the Joint Trustee and the convener of the meeting, James Maxwell, of JT Maxwell Limited, Unit 1 Lagan House, 1 Sackville Street, Lisburn, County Antrim, BT27 4AB.

A meeting of creditors is to take place for the purposes of considering the following:

1. The establishment of a creditor's committee

2. The approval of the Joint Trustees' fees and disbursements

The meeting will be held as follows

Time: 11.00 am

Date: 20 September 2023

To access the virtual meeting: Contact the our office on 02892 446 410 at least one day before the meeting for conference call login details

To be entitled to vote, those attending must submit a proxy form, together with a proof of debt if one has not already been submitted, to the Joint Trustees by one of the following methods:

By post to: Watch Bankruptcy Trustees, Unit 3-4 Lagan House, 1 Sackville Street, Lisburn, BT27 4AB, By email to: contactus@wbtuk.com

All proofs of debt must be delivered by: 4.00 pm on 19 September 2023

All proxy forms must be delivered to the convener or chair before they may be used at the meeting fixed for 11.00 am on 20 September 2023 Date of Appointment: 27 August 2021

For further information please contact: Sean Fay on 02892 446410 or by email at contactus@wbtuk.com

Joint Trustees, James Maxwell ( IP No 24310) and Michael Peacock (IP No 24330) both of Watch Bankruptcy Trustees, Unit 3-4 Lagan House, 1 Sackville Street, Lisburn, BT27 4AB (4430925)

### NOTICES OF DIVIDENDS

In the County Court at Sheffield No 200 of 2000

### **ROBERT LESLIE ANDREWS**

Bankrupt's Status - Formerly in bankruptcy

46 Laverdene Avenue, Totley, Sheffield, South Yorkshire, S17 4HF. NOTE: the above-named was discharged from the proceedings and may no longer have a connection with the addresses listed. Birth details: 01/06/1977

Bankrupt's occupation: delivery driver

Notice is hereby given that I, the Official Receiver of the above, intend to declare a First and Final Dividend to Unsecured Creditors within 2 months of 03/10/2023, being the last day for proving, failing which they will be excluded from the dividend. Claims should be sent to the Official Receiver, at the address below.

Laura Gandy - Official Receiver, PO BOX 16665, Birmingham, B2 2JX; Date of Appointment - 05/06/2000; Reference - PPI/BKT00166478 29 August 2023 (4431288)

In the County Court at Durham No 347 of 2009 Bankruptcy Order Date: 15/05/2009 MR GEORGE JOSEF ARMSTRONG

Bankrupt's Status - Formerly in bankruptcy; Deceased Status - No; Discharge Status - Yes (NOTE: the above-named may be discharged from the proceedings and may no longer have a connection with the addresses listed.)

1A Front Street, Sacriston, Durham DH7 6LD

Birth details: 21/12/1942

Notice is hereby given that I, the Official Receiver of the above, intend to declare a First and Final Dividend to Unsecured Creditors within 2 months of 28 /09/2023, being the last day for proving, failing which they will be excluded from the dividend. Claims should be sent to the Official Receiver, at the address below.

Ms L Haselgrove, LTADT Chatham, PO.Box 16665, Birmingham, B2 2JX, LTADT.Chatham@insolvency.gov.uk, Enquiry Line: 0300 6780 015. PPI/BKT00001862 25 August 2023

(4431299)

In the Canterbury Court

No 274 of 1996

### MR STEPHEN ROBERT ACKRILL

Bankrupt's Status - Formerly in bankruptcy

Stephen Robert Ackrill, unemployed of no fixed abode, lately residing at Honeysuckle Cottage, Selson, Eastry, Sandwich, CT13 OEF, and lately carrying on business as a car repairer under the style "Ackrill Automotive" from High Street, Littlebourne, Canterbury and Wingham Coachworks, Goodnestone Road, Wingham, Canterbury NOTE: the above-named was discharged from the proceedings and may no longer have a connection with the addresses listed. NOTICE OF FINAL INTENDED DIVIDEND.

Birth details: 22/06/1957

Notice is hereby given that I, the Official Receiver of the above, intend to declare a First and Final Dividend to Unsecured Creditors within 4 months of 27/09/2023, being the last day for proving, failing which they will be excluded from the dividend. Claims should be sent to the Official Receiver, at the address below.

Mr Stephen Baxter (Official Receiver) The Insolvency Service, LTADT Cardiff, Dividend Team, 3rd Floor, Companies House, Crown Way, Cardiff CF14 3ZA; clare.thompson@insolvency.gov.uk; Tel: 03003 046117

25 August 2023

(4431301)

In the County Court at Bournemouth No 325 of 2006

Bankruptcy Order Date: 11/05/2006

#### MR PHILLIP BRUCE BONE

Bankrupt's Status - Formerly in bankruptcy; Deceased Status - No; Discharge Status - Yes (NOTE: the above-named may be discharged from the proceedings and may no longer have a connection with the addresses listed.)

26 Meredith Close, Christchurch, Dorset BH23 3TX

Birth details: 28/07/1970

Bankrupt's occupation: Salesman

Notice is hereby given that I, the Official Receiver and Trustee, intend to declare a First and Final Dividend of 3p/£ to Unsecured creditors within two months of 03/10/2023, being the last day for proving, failing which they will be excluded from the dividend.

Ms L Haselgrove, LTADT Chatham, PO.Box 16665, Birmingham, B2 2JX, LTADT.Chatham@insolvency.gov.uk, Enquiry Line: 0300 6780 015, PPI/BKT00182166

29 August 2023

(4431292)

### COMPANIES

### In the County Court at St Albans No 323 of 2004

Bankruptcy Order Date: 30/11/2004

### **MR KALYAN BRATA DAS**

Bankrupt's Status - Formerly in bankruptcy: Deceased Status - No: Discharge Status - Yes (NOTE: the above-named may be discharged from the proceedings and may no longer have a connection with the addresses listed.)

Upper Flat, 33A Yarmouth Road, North Watford, Herts WD24 4DQ Birth details: 13/05/1941

Bankrupt's occupation: Travel consultant

Notice is hereby given that I, the Official Receiver of the above, intend to declare a First and Final Dividend to Unsecured Creditors within 2 months of 03/10/2023, being the last day for proving, failing which they will be excluded from the dividend. Claims should be sent to the Official Receiver, at the address below.

Mrs L Haselgrove, RTLU NW, PO.Box 16665, Birmingham, B2 2JX, RTLU.NW@insolvency.gov.uk, Enquiry Line: 0300 6780 015, PPI/ BKT00190344

29 August 2023

(4431296)

In the County Court at Wakefield

### No 60 of 2009

### MR ANDREW DARREN FIELDHOUSE

Bankrupt's Status - Formerly in bankruptcy

16 Inverness Road, Garforth, Leeds, West Yorkshire, LS25 2LS; 21 Field End Road, Halton, Leeds, West Yorkshire, LS15 0QA. NOTE: the above-named was discharged from the proceedings and may no longer have a connection with the addresses listed.

Birth details: 16/03/1965

Bankrupt's occupation: AppleMac Operator

Notice is hereby given that I, the Official Receiver of the above, intend to declare a First and Final Dividend to Unsecured Creditors within 2 months of 29 /09/2023, being the last day for proving, failing which they will be excluded from the dividend. Claims should be sent to the Official Receiver, at the address below.

Laura Gandy - Official Receiver, PO BOX 16665, Birmingham, B2 2JX; Date of Appointment - 22/01/2009; Reference - PPI/ BKT00135957 (4431309) 25 August 2023

In the County Court at Pontypridd

### No 28 of 2008

### **ROBIN CHARLES GLOSSOP**

Bankrupt's Status - Formerly in bankruptcy

11 HEOL TIR BACH, PENYRHEOL, CAERPHILLY, CF83 2JTT. NOTE: the above-named was discharged from the proceedings and may no longer have a connection with the addresses listed.

Birth details: 28/01/1942 Bankrupt's occupation: RETIRED

Notice is hereby given that I, the Official Receiver of the above, intend to declare a First and Final Dividend to Unsecured Creditors within 2 months of 03/10/2023, being the last day for proving, failing which they will be excluded from the dividend. Claims should be sent to the Official Receiver, at the address below.

Laura Gandy - Official Receiver, PO BOX 16665, Birmingham, B2 2JX; Date of Appointment - 18/02/2008; Reference - PPI/BKT00166322 29 August 2023 (4431300)

### In the Gloucester Court No 48 of 1998 **MR NATHAN SCOTT GRAVES**

### Bankrupt's Status – Formerly in bankruptcy

Nathan Scott Graves, Machine Operator of 39 Hillview Road, Hucclecote, Gloucester GL3 3LR NOTE: the above-named was discharged from the proceedings and may no longer have a connection with the addresses listed. NOTICE OF FINAL INTENDED DIVIDEND.

### Birth details: Unknown

Notice is hereby given that I, the Official Receiver of the above, intend to declare a First and Final Dividend to Unsecured Creditors within 4 months of 27 /09/2023, being the last day for proving, failing which they will be excluded from the dividend. Claims should be sent to the Official Receiver, at the address below.

Mr Stephen Baxter (Official Receiver) The Insolvency Service, LTADT Cardiff, Dividend Team, 3rd Floor, Companies House, Crown Way, Cardiff CF14 3ZA; clare.thompson@insolvency.gov.uk; Tel: 03003 046117 24 August 2023

(4431304)

In the County Court at Croydon No 1545 of 2006

### SUSAN ELIZABETH HILEY

Also known as: SUSAN ELIZAEBETH ROWE

Bankrupt's Status - Formerly in bankruptcy 43 Sliverdale Road, Sydenham, London, SE26 4SG,. NOTE: the above-named was discharged from the proceedings and may no longer have a connection with the addresses listed. Birth details: 09/03/1967

Bankrupt's occupation: Sales Advisor

Notice is hereby given that I, the Official Receiver of the above, intend to declare a First and Final Dividend to Unsecured Creditors within 2 months of 03/10/2023, being the last day for proving, failing which they will be excluded from the dividend. Claims should be sent to the Official Receiver, at the address below.

Laura Gandy - Official Receiver, PO BOX 16665, Birmingham, B2 2JX; Date of Appointment - 01/11/2006; Reference - PPI/BKT00135635 29 August 2023 (4431293)

In the High Court Of Justice

### No 1520 of 2007

### JEAN HYLTON

Bankrupt's Status - Formerly in bankruptcy; Deceased Status - No; Discharge Status - Yes

FLAT 26, QUEENSBRIDGE COURT, QUEENSBRIDGEROAD, LONDON F2 8PG

Birth details: 20/06/1962

Notice is hereby given that I, the Official Receiver of the above, intend to declare a First and Final Dividend to Unsecured Creditors within 2 months of 29 /09/2023, being the last day for proving, failing which they will be excluded from the dividend. Claims should be sent to the Official Receiver, at the address below.

Ms L Gandy, PPI Exeter, PO.Box 16665, Birmingham, B2 2JX, PPI.Exeter@insolvency.gov.uk, Enquiry Line: 0300 6780 015 BKT00186411; BO: 07.02.2007

25 August 2023

(4431316)

### In the Wakefield Court No 192 of 2003 **MS CATHERINE HIORNS**

### Bankrupt's Status - Formerly in bankruptcy

Birch Tree Walk, Knottingley, West Yorkshire, WF11 0PQ, formerly residing at14 Eastfield Avenue, Knottingley, West Yorkshire. NOTE: the above-named was discharged from the proceedings and may no longer have a connection with the addresses listed.

Birth details: 28/04/1950

Bankrupt's occupation: Unemployed

Notice is hereby given that I, the Official Receiver of the above, intend to declare a First and Final Dividend to Unsecured Creditors within 2 months of 29 /09/2023, being the last day for proving, failing which they will be excluded from the dividend. Claims should be sent to the Official Receiver, at the address below.

Lindsey Haselgrove, LTADT Manchester, PO Box 16665, BIRMINGHAM, B2 2JX, 0300 678 0015, PPI/ BKT00190956, Order date: 25/11/2003; RTLU.NW@insolvency.gov.uk

25 August 2023

(4431307)

In the County Court at Gloucester No 21 of 2007

### CHARLOTTE LEE

Bankrupt's Status - Formerly in bankruptcy; Deceased Status - No; Discharge Status - Yes

4 Wedgewood Crescent, Cinderford, Gloucestershire GL14 2DA Birth details: 01/04/1973

Notice is hereby given that I, the Official Receiver of the above, intend to declare a First and Final Dividend to Unsecured Creditors within 2 months of 29 /09/2023, being the last day for proving, failing which they will be excluded from the dividend. Claims should be sent to the Official Receiver, at the address below.

Ms L Gandy, PPI Exeter, PO.Box 16665, Birmingham, B2 2JX. PPI.Exeter@insolvency.gov.uk, Enquiry Line: 0300 6780 015; BKT00116755 - BO: 02.07.2007

25 August 2023

(4431308)

In the County Court at County Court at Bournemouth No 90 of 2003

Bankruptcy Order Date: 08/04/2003

### DAVID JAMES LUFF

Bankrupt's Status - Formerly in bankruptcy: Deceased Status - No: Discharge Status - Yes (NOTE: the above-named may be discharged from the proceedings and may no longer have a connection with the addresses listed.)

12 Briar Close, Mudeford, Dorset BH23 3RB; 2 Old Highway Mews, 141 Leigh Road, Wimborne BH21 2AD

Birth details: 23/04/1975

Bankrupt's occupation: Aircraft Engineer

Notice is hereby given that I, the Official Receiver of the above, intend to declare a First and Final Dividend to Unsecured Creditors within 2 months of 03/10/2023, being the last day for proving, failing which they will be excluded from the dividend. Claims should be sent to the Official Receiver, at the address below.

Ms L Haselgrove, LTADT Chatham, PO.Box 16665, Birmingham, B2 2JX, LTADT.Chatham@insolvency.gov.uk, Enquiry Line: 0300 6780 015, PPI/BKT00005040 29 August 2023

(4431297)

In the County Court at Taunton

### No 49 of 2002

### MS ELIZABETH-ANN LYNCH

Also known as: Elizabeth-Ann Dearne

Bankrupt's Status - Formerly in bankruptcy; Deceased Status - No; Discharge Status - Yes

10 Quirke Street, Minehead, Somerset, TA24 5TZ TA24 5TZ; 12 Meadow View, Timberscombe, Minehead, TA24 7TE TA24 7TE; 10 Holloway Street, Minehead, Somerset, TA24 5NR TA24 5NR

Birth details: 02/10/1967

Bankrupt's occupation: Unemployed

Notice is hereby given that I, the Official Receiver of the above, intend to declare a First and Final Dividend to Unsecured Creditors within 2 months of 29 /09/2023, being the last day for proving, failing which they will be excluded from the dividend. Claims should be sent to the Official Receiver, at the address below.

Mrs L Haselgrove, RTLU NW, PO.Box 16665, Birmingham, B2 2JX, RTLU.NW@insolvency.gov.uk, Enquiry Line: 0300 6780 015 (4431302)

25 August 2023

In the County Court at Leeds

### No 211 of 2007

### MR DAVID JOSEPH SCADDAN

Bankrupt's Status - Formerly in bankruptcy

49 Rycroft Towers, Bramley, LEEDS, LS13 4NH 1 Chaucer Avenue, PUDSEY, LS28 9DN. NOTE: the above-named was discharged from the proceedings and may no longer have a connection with the addresses listed.

Birth details: 23/09/1949

### Bankrupt's occupation: a Forklift Driver

Notice is hereby given that I, the Official Receiver of the above, intend to declare a Final Dividend to Unsecured Creditors within 2 months of 26 /09/2023, being the last day for proving, failing which they will be excluded from the dividend. Claims should be sent to the Official Receiver, at the address below.

Lindsey Haselgrove, RTLU NW, PO.Box 16665, Birmingham, B2 2JX, RTLU.NW@insolvency.gov.uk, Enquiry Line: 0300 6780 015, Order Date: 22/02/2007, PPI/BKT00190916 29 August 2023

(4431312)

In the County Court at Leeds No 148 of 2004 Bankruptcy Order Date: 24/03/2004

MISS ANGELA MARY TALLANT

Bankrupt's Status - Formerly in bankruptcy; Deceased Status - No; Discharge Status - Yes (NOTE: the above-named may be discharged from the proceedings and may no longer have a connection with the addresses listed.)

4 Farrow Hill, LEEDS, West Yorkshire, LS12 3TE

Birth details: 05/01/1969

Bankrupt's occupation: Unemployed

Notice is hereby given that I, the Official Receiver of the above, intend to declare a First and Final Dividend to Unsecured Creditors within 2 months of 03/10/2023, being the last day for proving, failing which they will be excluded from the dividend. Claims should be sent to the Official Receiver, at the address below.

Mrs L Haselgrove, RTLU NW, PO.Box 16665, Birmingham, B2 2JX, RTLU.NW@insolvency.gov.uk, Enquiry Line: 0300 6780 015, PPI/ BKT00190322 (4431290)

29 August 2023

In the County Court at Brighton

### No 1535 of 2005 MR ANDREW START TAYLOR

Bankrupt's Status - Formerly in bankruptcy

31 Friar Walk, Worthing, West Sussex, BN13 1BL. NOTE: the abovenamed was discharged from the proceedings and may no longer have a connection with the addresses listed.

Birth details: 13/09/1966

Bankrupt's occupation: Baker

Notice is hereby given that I, the Official Receiver of the above, intend to declare a First and Final Dividend to Unsecured Creditors within 2 months of 03/10/2023, being the last day for proving, failing which they will be excluded from the dividend. Claims should be sent to the Official Receiver, at the address below.

Lindsey Haselgrove, RTLU NW, PO Box 16665, Birmingham, B2 2JX, RTLU.NW@insolvency.gov.uk, Enquiry Line: 0300 6780 015, Order Date: 10/01/2006, Ref: PPI/BKT00138462 29 August 2023

(4431287)

In the County Court at Blackpool

No 194 of 2004 Bankruptcy Order Date: 14/07/2004

MRS COLLEEN ANNE WALSH

Bankrupt's Status - Formerly in bankruptcy; Deceased Status - No; Discharge Status - Yes (NOTE: the above-named may be discharged from the proceedings and may no longer have a connection with the addresses listed.)

1 Freckleton Court, Lytham, Lancashire FY8 5DS

Birth details: 16/08/1953

29 August 2023

Bankrupt's occupation: Unemployed

Notice is hereby given that I, the Official Receiver of the above, intend to declare a First and Final Dividend to Unsecured Creditors within 2 months of 03/10/2023, being the last day for proving, failing which they will be excluded from the dividend. Claims should be sent to the Official Receiver, at the address below.

Ms L Gandy, LTADT Chatham, PO.Box 16665, Birmingham, B2 2JX, LTADT.Chatham@insolvency.gov.uk, Enquiry Line: 0300 6780 015, PPI/BKT00110243

(4431286)

### In the County Court at Oldham No 85 of 2003

### SYED MOKADDUS ALI

Bankrupt's Status - Formerly in bankruptcy

5 Neath Street, Westwood, Oldham OL9 6AS; 119 Farringdon, Freehold, Rochdale OL11 4JU; 30 Alfred Street, Werneth, Oldham OL9 7AD; NOTE: the above-named was discharged from the proceedings and may no longer have a connection with the addresses listed.

Birth details: 12/04/1939

### Bankrupt's occupation: UNEMPLOYED

Notice is hereby given that I, the Official Receiver of the above, intend to declare a First and Final Dividend to Unsecured Creditors within 2 months of 29 /09/2023, being the last day for proving, failing which they will be excluded from the dividend. Claims should be sent to the Official Receiver, at the address below.

Laura Gandy - Official Receiver, PO BOX 16665, Birmingham, B2 2JX; Date of Appointment - 02/04/2003; Reference - PPI/BKT00116277 25 August 2023 (4431289)

In the County Court at Croydon

### No 1424 of 2008

SUHAIL ASGHAR

Bankrupt's Status - Formerly in bankruptcy; Deceased Status - No; Discharge Status - Yes

146 Finchale Road, Abbey Wood, London SE2 9PW; 57 FuchsiaStreet, Abbey Wood, London SE2 0TA

### Birth details: 17/07/1966

Notice is hereby given that I, the Official Receiver of the above, intend to declare a First and Final Dividend to Unsecured Creditors within 2 months of 29 /09/2023, being the last day for proving, failing which they will be excluded from the dividend. Claims should be sent to the Official Receiver, at the address below.

Ms L Gandy, PPI Exeter, PO.Box 16665, Birmingham, B2 2JX, PPI.Exeter@insolvency.gov.uk, Enquiry Line: 0300 6780 015; BKT00186706 - BO: 29.10.2008 25 August 2023

(4431294)

In the County Court at Norwich

### No 19 of 2004

### MRS BRIDGET BODDY

Bankrupt's Status - Formerly in bankruptcy

31 Russell Street NORWICH Norfolk NR2 4QT. NOTE: the abovenamed was discharged from the proceedings and may no longer have a connection with the addresses listed.

Birth details: 30/12/1953

Bankrupt's occupation: NUMBER PLATE OPERATIVE

Notice is hereby given that I, the Official Receiver of the above, intend to declare a Final Dividend to Unsecured Creditors within 2 months of 26/09/2023, being the last day for proving, failing which they will be excluded from the dividend. Claims should be sent to the Official Receiver, at the address below.

Lindsey Haselgrove, RTLU NW, PO.Box 16665, Birmingham, B2 2JX, RTLU.NW@insolvency.gov.uk, Enquiry Line: 0300 6780 015, Order Date: 09/01/2004, PPI/ BKT00190939 29 August 2023

(4431295)

In the County Court at Walsall No. 323 of 2009

### SIMON ANTHONY BAKER

Formerly in bankruptcy Date of bankruptcy order: 20 March 2009

31 Birch Coppice Gardens, Willenhall WV12 4BL

NOTE: the above-named was discharged from the proceedings and may no longer have a connection with the addresses listed.

### 18 June 1982

Signmaker

Notice is hereby given, pursuant to Rule 11.2 of the Insolvency Rules 1986 that the Official Receiver, having been appointed on 20 March 2009 intend(s) to declare a(n) Dividend of £0.02 to all Unsecured creditors of the Bankrupt's estate within 2 months of the last date for proving specified below.

Creditors who have not yet done so must prove their debts or claims and the names and addresses, particulars of their debts by sending their full names and addresses of their solicitors (if any), to the Official Receiver using the details specified below.

The Official Receiver is/are not obliged to deal with proofs lodged after the last date for proving. Creditors who have not proved their debts by 00:00 on 05 October 2023 may be excluded from the benefit of the dividend or any other dividend declared before their debt is proved.

Office holder details: Lindsey Haselgrove, The Insolvency Service, Po Box 16665. Birmingham B2 2JX. 03003048498. michael.bolitho@insolvency.gov.uk, (BKT00191721/PPI). (4431595)

In the High Court of Justice

### No. 1081 of 2005 STELLA ADJUA MARTINE JULIETTE DADZIE

### Formerly in bankruptcy

Date of bankruptcy order: 07 February 2005

29d Lancaster Road, London W11 1QJ

NOTE: the above-named was discharged from the proceedings and may no longer have a connection with the addresses listed.

09 September 1968

Unemployed

Notice is hereby given, pursuant to Rule 11.2 of the Insolvency Rules 1986 that the Official Receiver, having been appointed on 07 February 2005 intend(s) to declare a(n) Dividend of £0.13 to all Unsecured creditors of the Bankrupt's estate within 2 months of the last date for proving specified below.

Creditors who have not yet done so must prove their debts or claims and the names and addresses, particulars of their debts by sending their full names and addresses of their solicitors (if any), to the Official Receiver using the details specified below.

The Official Receiver is/are not obliged to deal with proofs lodged after the last date for proving. Creditors who have not proved their debts by 00:00 on 05 October 2023 may be excluded from the benefit of the dividend or any other dividend declared before their debt is proved.

Office holder details: Lindsey Haselgrove , The Insolvency Service, Po 16665, Birmingham B2 2JX, 03003048498, Box michael.bolitho@insolvency.gov.uk, (BKT00191659/PPI). (4430957)

In the County Court at Nottingham

### No. 337 of 2004 STEPHEN WILLIAM EDLIN

also known as: STEPHEN WILLIAM SPITZOKOWSKI

Formerly in bankruptcy

Date of bankruptcy order: 07 June 2004

45 Beech Avenue, Newark NG24 4JN

NOTE: the above-named was discharged from the proceedings and may no longer have a connection with the addresses listed.

30 June 1954

Lorry driver

Notice is hereby given, pursuant to Rule 11.6 of the Insolvency Rules 1986 that the Official Receiver, having been appointed on 07 June 2004 intend(s) to declare a(n) First and final dividend to all Creditors of the Bankrupt's estate who have proven their debts within 2 months of the last date for proving specified below.

Last day of proving: 23:59 on 05 October 2023

Office holder details: Lindsey Haselgrove, Po Box 16665, Birmingham +443030031572, jasper.sheriff@insolvency.gov.uk, B2 2JX. (BKT00191980). (4431831)

In the County Court at Torquay and Newton Abbot No 58 of 2002

### ANTHONY GEE

Bankrupt's Status – Formerly in bankruptcy

10 Teign Street, Teignmouth, Devon, TQ14 8EA; 5 Hocking Road, Wyken, Coventry, CV2 5ER. NOTE: the above-named was discharged from the proceedings and may no longer have a connection with the addresses listed.

Birth details: 04/03/1978

Bankrupt's occupation: Shelf Filler

Notice is hereby given that I, the Official Receiver of the above, intend to declare a First and Final Dividend to Unsecured Creditors within 2 months of 29 /09/2023, being the last day for proving, failing which they will be excluded from the dividend. Claims should be sent to the Official Receiver, at the address below.

Laura Gandy – Official Receiver, PO BOX 16665, Birmingham, B2 2JX; Date of Appointment – 14/03/2002; Reference – PPI/BKT00117930 25 August 2023 (4431306)

In the County Court at Newcastle upon Tyne No. 1039 of 2007

### JUNE GORDON

Formerly in bankruptcy

Date of bankruptcy order: 19 June 2007

32 Burnhills Gardens, Greenside, Ryton NE40 4RB

Formerly of: 19 Norfolk Avenue, Birtley, Chester Le Street DH3 2DG

NOTE: the above-named was discharged from the proceedings and may no longer have a connection with the addresses listed. 11 June 1950

Unemploved

Unemployed

Notice is hereby given, pursuant to Rule 11.6 of the Insolvency Rules 1986 that the Official Receiver, having been appointed on 19 June 2007 intend(s) to declare a(n) First and final dividend to all Creditors of the Bankrupt's estate who have proven their debts within 2 months of the last date for proving specified below.

Last day of proving: 01:00 on 05 October 2023

Office holder details: Lindsey Haselgrove, Po Box 16665, Birmingham B2 2JX, +443030031841, Aditi.Ottaway@insolvency.gov.uk, (BKT00183608). (4431560)

This notice is in substitution for that which appeared in The Gazette Notice ID 4399305 URL https://www.thegazette.co.uk/notice/4399305 In the Bradford Court

No 334 of 2001

### MR ANDREW DAVID GILLSON

Bankrupt's Status - Formerly in bankruptcy

21 Vale Mill Lane, Haworth, Keithley, West Yorkshire, BD22 0EF trading with another at The Green Works, Knotts Lane, Colne, Lancashire, BBA 8AD as a Demolition Contractor as DAVID GILLSON & SON. NOTE: the above-named was discharged from the proceedings and may no longer have a connection with the addresses listed.

#### Birth details: 11/04/1968

Bankrupt's occupation: DEMOLITION CONTRACTOR

Notice is hereby given that I, the Official Receiver of the above, intend to declare a First and Final Dividend to Unsecured Creditors within 2 months of 14/09/2023, being the last day for proving, failing which they will be excluded from the dividend. Claims should be sent to the Official Receiver, at the address below.

Lindsey Haselgrove, LTADT Manchester, PO Box 16665, BIRMINGHAM, B2 2JX, 0300 678 0015, PPI/BKT00157582, Order date: 31/07/2001; RTLU.NW@insolvency.gov.uk 24 August 2023 (4431291)

In the County Court at Norwich No. 973 of 2009

### ANDREW PAUL GOWERS

Formerly in bankruptcy

Date of bankruptcy order: 01 June 2009

37 Pleasant Place, Beccles NR34 9YD

Formerly of: 103 Banham Road, Beccles NR34 9JP

NOTE: the above-named was discharged from the proceedings and may no longer have a connection with the addresses listed. 21 January 1965

Blow Moulding Technician

Notice is hereby given, pursuant to Rule 11.2 of the Insolvency Rules 1986 that the Official Receiver, having been appointed on 01 June 2009 intend(s) to declare a(n) Dividend of £0.01 to all Unsecured creditors of the Bankrupt's estate within 2 months of the last date for proving specified below.

Creditors who have not yet done so must prove their debts or claims and the names and addresses, particulars of their debts by sending their full names and addresses of their solicitors (if any), to the Official Receiver using the details specified below.

The Official Receiver is/are not obliged to deal with proofs lodged after the last date for proving. Creditors who have not proved their debts by 00:00 on 28 September 2023 may be excluded from the benefit of the dividend or any other dividend declared before their debt is proved.

Office holder details: Lindsey Haselgrove , The Insolvency Service, Po Box 16665, Birmingham B2 2JX, 03003048498, michael.bolitho@insolvency.gov.uk, (BKT00191625/PPI). (4430948)

In the High Court of Justice

### No. 1496 of 2018 JAMES JOHN GROVER

In Bankruptcy

Date of bankruptcy order: 19 November 2018 32 Lawrence Avenue, Sawbridgeworth CM21 9DN

NOTE: the above-named was discharged from the proceedings and may no longer have a connection with the addresses listed.

24 April 1980

Subcontractor

Notice is hereby given, pursuant to Rule 14.35 of the Insolvency (England and Wales) Rules 2016 that the Official Receiver, having been appointed on 19 November 2018 intend(s) to declare a(n) First and final dividend of  $\pounds 4 p/\pounds$  to all Creditors of the Bankrupt's estate within 2 months of the last date for proving specified below.

Creditors who have not yet done so must prove their debts or claims and the names and addresses, particulars of their debts by sending their full names and addresses of their solicitors (if any), to the Official Receiver using the details specified below.

The Official Receiver is/are not obliged to deal with proofs lodged after the last date for proving. Creditors who have not proved their debts by on 01:00 on 04 October 2023 may be excluded from the benefit of the dividend or any other dividend declared before their debt is proved.

Office holder details: Michael Commins, Po Box 16653, Birmingham B2 2NG, 03003046401, Emma.Abbott@Insolvency.gov.uk, (BKT5490083). (4430882)

In the County Court at Harrogate

### No. 324 of 2006

ANDREW ROBERT HARTLEY

Formerly in bankruptcy

Date of bankruptcy order: 21 November 2006 9 Magdalens Close, Ripon HG4 1HH

NOTE: the above-named was discharged from the proceedings and may no longer have a connection with the addresses listed.

12 October 1979

A Computer Technician

Notice is hereby given, pursuant to Rule 11.6 of the Insolvency Rules 1986 that the Official Receiver, having been appointed on 21 November 2006 intend(s) to declare a(n) Final dividend to all Unsecured creditors of the Bankrupt's estate within 2 months of the last date for proving specified below.

Creditors who have not yet done so must prove their debts or claims and the names and addresses, particulars of their debts by sending their full names and addresses of their solicitors (if any), to the Official Receiver using the details specified below.

The Official Receiver is/are not obliged to deal with proofs lodged after the last date for proving. Creditors who have not proved their debts by on 00:00 on 29 September 2023 may be excluded from the benefit of the dividend or any other dividend declared before their debt is proved.

Office holder details: Lindsey Haselgrove, Po Box 16665, Birmingham B2 2JX, 03006780015, RTLU.NW@INSOLVENCY.GOV.UK, (PPI/BKT00190758). (4429193)

### COMPANIES

In the County Court at Preson No 202 of 2003

Bankruptcy Order Date: 17/11/2003 **MR ANTHONY HERON** 

Bankrupt's Status - Formerly in bankruptcy; Deceased Status - No; Discharge Status - Yes (NOTE: the above-named may be discharged from the proceedings and may no longer have a connection with the addresses listed.)

73 Higher Meadow, Leyland, Lancashire PR25 5RJ.

Birth details: 10/10/1965

Bankrupt's occupation: Drainage Manager

Notice is hereby given that I, the Official Receiver of the above, intend to declare a First and Final Dividend to Unsecured Creditors within 2 months of 20 /09/2023, being the last day for proving, failing which they will be excluded from the dividend. Claims should be sent to the Official Receiver, at the address below.

Ms L Gandy, PPI Exeter, PO.Box 16665, Birmingham, B2 2JX, PPI.Exeter@insolvency.gov.uk, Enquiry Line: 0300 6780 015,Date of bankruptcy order:17/11/2003, PPI/BKT00063649 24 August 2023 (4431303)

In the Macclesfield Court

No 107 of 1994

### **MR MARK HILL**

Bankrupt's Status - Formerly in bankruptcy

Mark Hill of 2 Rose Bank, Bollington, Nr Macclesfield, Cheshire, lately residing at 330 Buxton Road, Macclesfield Cheshire, An engineer NOTE: the above-named was discharged from the proceedings and may no longer have a connection with the addresses listed. NOTICE OF FINAL INTENDED DIVIDEND.

Birth details: 11/05/1964

Notice is hereby given that I, the Official Receiver of the above, intend to declare a First and Final Dividend to Unsecured Creditors within 2 months of 27 /09/2023, being the last day for proving, failing which they will be excluded from the dividend. Claims should be sent to the Official Receiver, at the address below.

Mr Stephen Baxter (Official Receiver) The Insolvency Service, LTADT Cardiff, Dividend Team, 3rd Floor, Companies House, Crown Way, Cardiff CF14 3ZA; clare.thompson@insolvency.gov.uk; Tel: 03003 046117

24 August 2023

(4431305)

In the High Court of Justice No 4268 of 2003 Bankruptcy Order Date: 28/05/2003

**GRANT CARL JAMES** 

Bankrupt's Status - Formerly in bankruptcy; Deceased Status - No; Discharge Status - Yes (NOTE: the above-named may be discharged from the proceedings and may no longer have a connection with the addresses listed.)

92 Sterling Gardens, Sangford Street, London SE14 6DZ; 20 Ravensbourne, BradleyStreet, London NW8 8BE

Birth details: 27/04/1967

Bankrupt's occupation: unemployed

Notice is hereby given that I, the Official Receiver of the above, intend to declare a First and Final Dividend to Unsecured Creditors within 2 months of 03/10/2023, being the last day for proving, failing which they will be excluded from the dividend. Claims should be sent to the Official Receiver, at the address below.

Ms L Haselgrove, LTADT Chatham, PO.Box 16665, Birmingham, B2 2JX, LTADT.Chatham@insolvency.gov.uk, Enquiry Line: 0300 6780 015, PPI/BKT00181881 29 August 2023

(4431317)

In the County Court at Newcastle upon Tyne No. 535 of 2010 **CLARE LOUISE JOHNSON** 

also known as: Clare Louise Pybus

Formerly in bankruptcy

Date of bankruptcy order: 18 February 2010

5 Kennford, Gateshead NE9 6AG

Formerly of: 6 The Avenue, Sheriff Hill, Gateshead NE9 5PN NOTE: the above-named was discharged from the proceedings and

may no longer have a connection with the addresses listed.

### 06 February 1977

Unemployed

Notice is hereby given, pursuant to Rule 11.2 of the Insolvency Rules 1986 that the Official Receiver, having been appointed on 18 February 2010 intend(s) to declare a(n) Dividend of £0.02 to all Unsecured creditors of the Bankrupt's estate within 2 months of the last date for proving specified below.

Creditors who have not yet done so must prove their debts or claims and the names and addresses, particulars of their debts by sending their full names and addresses of their solicitors (if any), to the Official Receiver using the details specified below.

The Official Receiver is/are not obliged to deal with proofs lodged after the last date for proving. Creditors who have not proved their debts by 00:00 on 05 October 2023 may be excluded from the benefit of the dividend or any other dividend declared before their debt is proved.

Office holder details: Lindsey Haselgrove , The Insolvency Service, Po Box 16665. Birmingham B2 2JX. 03003048498. michael.bolitho@insolvency.gov.uk, (BKT00191645/PPI). (4430965)

In the County Court at Leeds

### No. 1347 of 2008

JACK MORLEY

Formerly in bankruptcy

Date of bankruptcy order: 21 November 2008

53 Boggart Hill Crescent, Leeds LS14 1LF Formerly of: 30 Boggart Hill Drive, Leeds LS14 1JY

NOTE: the above-named was discharged from the proceedings and may no longer have a connection with the addresses listed. 12 May 1941

retired

Notice is hereby given, pursuant to Rule 11.6 of the Insolvency Rules 1986 that the Official Receiver, having been appointed on 21 November 2008 intend(s) to declare a(n) First and final dividend to all Creditors of the Bankrupt's estate who have proven their debts within 2 months of the last date for proving specified below.

Last day of proving: 23:59 on 28 September 2023

Office holder details: Lindsey Haselgrove, Po Box 16665, Birmingham jasper.sheriff@insolvency.gov.uk, 2JX, +443030031572, B2 (BKT00181775). (4431390)

In the County Court at Luton

### No 187 of 2000 **MR ANDREW MALONEY**

Bankrupt's Status - Formerly in bankruptcy

34 Sherd Close, Marsh Farm, Luton, Bedfordshire, LU3 3LY. NOTE: the above-named was discharged from the proceedings and may no longer have a connection with the addresses listed.

Birth details: 16/06/1960

Bankrupt's occupation: Builder

Notice is hereby given that I, the Official Receiver of the above, intend to declare a First and Final Dividend to Unsecured Creditors within 2 months of 03/10/2023, being the last day for proving, failing which they will be excluded from the dividend. Claims should be sent to the Official Receiver, at the address below.

Lindsey Haselgrove, RTLU NW, PO Box 16665, Birmingham, B2 2JX, RTLU.NW@insolvency.gov.uk, Enquiry Line: 0300 6780 015, Order Date: 25/01/2001, Ref: PPI/BKT00128081 29 August 2023

(4431311)

In the County Court at Gloucester

### No 113 of 2001 JASON RONALD PARKER

Bankrupt's Status - Formerly in bankruptcy

24 Norton Court, Forest Green, Nailsworth, GL6 0TJ. NOTE: the above-named was discharged from the proceedings and may no longer have a connection with the addresses listed.

Birth details: 15/09/1972

Bankrupt's occupation: Unemployed

Notice is hereby given that I, the Official Receiver of the above, intend to declare a First and Final Dividend to Unsecured Creditors within 2 months of 29 /09/2023, being the last day for proving, failing which they will be excluded from the dividend. Claims should be sent to the Official Receiver, at the address below.

Laura Gandy - Official Receiver, PO BOX 16665, Birmingham, B2 2JX; Date of Appointment - 12/09/2001; Reference - PPI/BKT00118904 25 August 2023 (4431285)

In the County Court at Northampton No 587 of 2010

Bankruptcy Order Date: 07/04/2010

MR ALAN JOHN JOSEPH POWER

Bankrupt's Status - Formerly in bankruptcy; Deceased Status - No; Discharge Status - Yes (NOTE: the above-named may be discharged from the proceedings and may no longer have a connection with the addresses listed.)

451 Lyveden Way, Corby, Northamptonshire, NN18 8RU; 18 Lavender Close, Corby, Northamptonshire, NN18 8P; 15 Beardsley Court, Corby, Northamptonshire, NN18 9LJ.

Birth details: 09/01/1976

Bankrupt's occupation: Warehouse operative

Notice is hereby given that I, the Official Receiver of the above, intend to declare a First and Final Dividend to Unsecured Creditors within 2 months of 29 /09/2023, being the last day for proving, failing which they will be excluded from the dividend. Claims should be sent to the Official Receiver, at the address below.

Mrs L Haselgrove, RTLU NW, PO.Box 16665, Birmingham, B2 2JX, RTLU.NW@insolvency.gov.uk, Enquiry Line: 0300 6780 015, PPI/ BKT00190197

25 August 2023

(4431315)

In the County Court at Kidderminster No 76 of 2007

Bankruptcy Order Date: 21/03/2007

### MANDY ANN PRIEST

Also known as: MANDY ANN HADLEY, MANDY ANN REECE

Bankrupt's Status - Formerly in bankruptcy; Deceased Status - No; Discharge Status - Yes (NOTE: the above-named may be discharged from the proceedings and may no longer have a connection with the addresses listed.)

8 Coronation Way, Comberton, Kidderminster, Worcs DY10 3BA

Birth details: 15/01/1963

Bankrupt's occupation: Unemployed

Notice is hereby given that I, the Official Receiver of the above, intend to declare a First and Final Dividend to Unsecured Creditors within 2 months of 03/10/2023, being the last day for proving, failing which they will be excluded from the dividend. Claims should be sent to the Official Receiver, at the address below.

Ms L Haselgrove, LTADT Chatham, PO.Box 16665, Birmingham, B2 2JX, LTADT.Chatham@insolvency.gov.uk, Enquiry Line: 0300 6780 015. PPI/BKT00182709 (4431298)

29 August 2023

In the County Court at Dewsbury No 579 of 2009

### **MR JASON DEREK ROGERS**

Also known as: JASON DAVID ROGERS

Bankrupt's Status - Formerly in bankruptcy

14 Weavers Croft, Rawfolds, Cleckheaton, West Yorkshire, BD19 5NE. NOTE: the above-named was discharged from the proceedings and may no longer have a connection with the addresses listed.

Birth details: 18/11/1975 Bankrupt's occupation: Sales & customer services

Notice is hereby given that I, the Official Receiver of the above, intend to declare a First and Final Dividend to Unsecured Creditors within 2 months of 29 /09/2023, being the last day for proving, failing which they will be excluded from the dividend. Claims should be sent to the Official Receiver, at the address below.

Laura Gandy - Official Receiver, PO BOX 16665, Birmingham, B2 2JX; Date of Appointment - 13/10/2009; Reference - PPI/BKT00141141 (4431310) 25 August 2023

In the High Court of Justice

No 960 of 2003 Bankruptcy Order Date: 03/02/2003

### **FAZLUR RAHMAN**

Bankrupt's Status - Formerly in bankruptcy; Deceased Status - No; Discharge Status - Yes (NOTE: the above-named may be discharged from the proceedings and may no longer have a connection with the addresses listed.)

28 Underhill House, Burgess Street, London E14 7AX

Birth details: 04/08/1971

Bankrupt's occupation: Cook

Notice is hereby given that I, the Official Receiver of the above, intend to declare a First and Final Dividend to Unsecured Creditors within 2 months of 03/10/2023, being the last day for proving, failing which they will be excluded from the dividend. Claims should be sent to the Official Receiver, at the address below.

Ms L Haselgrove, LTADT Chatham, PO.Box 16665, Birmingham, B2 2JX, LTADT.Chatham@insolvency.gov.uk, Enquiry Line: 0300 6780 015. PPI/BKT00181809 29 August 2023

(4431314)

In the County Court at Nottingham No. 1256 of 2007

### ELAINE LOUISE SCOTNEY

Formerly in bankruptcy

Date of bankruptcy order: 04 July 2007

27 Dylan Thomas Road, Nottingham NG5 5UA

Formerly of: 1 Teesdale Road, Nottingham NG5 1DA

NOTE: the above-named was discharged from the proceedings and may no longer have a connection with the addresses listed.

30 June 1976

Receptionist

Notice is hereby given, pursuant to Rule 11.6 of the Insolvency Rules 1986 that the Official Receiver, having been appointed on 04 July 2007 intend(s) to declare a(n) First and final dividend to all Unsecured creditors of the Bankrupt's estate within 2 months of the last date for proving specified below.

Creditors who have not yet done so must prove their debts or claims and the names and addresses, particulars of their debts by sending their full names and addresses of their solicitors (if any), to the Official Receiver using the details specified below.

The Official Receiver is/are not obliged to deal with proofs lodged after the last date for proving. Creditors who have not proved their debts by on 17:00 on 22 September 2023 may be excluded from the benefit of the dividend or any other dividend declared before their debt is proved.

Office holder details: Lindsey Haselgrove, Po Box 16665, Birmingham B2 2JX, 0300 304 8156, RTLU.NW@insolvency.gov.uk, (PPI/ BKT00191028).

Friday 25 August 2023

(4429122)

In the County Court at Eastbourne

### No 134 of 2003

SARAH JANE SQUIRES

Also known as: SARAH CROLL; SARAH DOVIDONIS

Bankrupt's Status - Formerly in bankruptcy

6 Wannock Road, Eastbourne, East Sussex, BN22 7JU; 22 Amherst Road, Bexhillon Sea, TN40 1QJ; The Devonshire Arms, Devonshire Square, Bexhill on Sea, East Sussex, TN40 1AB. NOTE: the abovenamed was discharged from the proceedings and may no longer have a connection with the addresses listed.

Birth details: 09/01/1973

Bankrupt's occupation: music teacher

(t/a The Limelight School from 6 Wannock Road, Eastbourne, BN227JU; 22 Amherst Road, Devonshire Square, Bexhill on Sea, East Sussex)

Notice is hereby given that I, the Official Receiver of the above, intend to declare a First and Final Dividend to Unsecured Creditors within 2 months of 03/10/2023, being the last day for proving, failing which they will be excluded from the dividend. Claims should be sent to the Official Receiver, at the address below.

Laura Gandy - Official Receiver, PO BOX 16665, Birmingham, B2 2JX; Date of Appointment - 12/09/2003; Reference - PPI/BKT00117973 29 August 2023 (4431313)

In the County Court at Sunderland No. 794 of 2009 BRIAN TURNBULL

In Bankruptcy

Date of bankruptcy order: 22 September 2009 124 Marlborough Street South, South Shields NE33 4DB Formerly of: 71 Ormonde Street, Jarrow NE32 3PB NOTE: the above-named was discharged from the proceedings and may no longer have a connection with the addresses listed. 25 May 1981

HGV Driver

Notice is hereby given, pursuant to Rule 11.6 of the Insolvency Rules 1986 that the Official Receiver, having been appointed on 22 September 2009 intend(s) to declare a(n) Final dividend to all Creditors of the Bankrupt's estate who have proven their debts within 2 months of the last date for proving specified below.

Last day of proving: 01:00 on 28 September 2023

Office holder details: Lindsey Haselgrove, Po Box 16665, Birmingham B2 2JX, +443030031841, Aditi.Ottaway@insolvency.gov.uk, (BKT00183648). (4431770)

## Wills & probate

### **DECEASED ESTATES – LONDON EDITION**

Notice is hereby given pursuant to s. 27 of the Trustee Act 1925, that any person having a claim against or an interest in the estate of any of the deceased persons whose names and addresses are set out below is hereby required to send particulars in writing of his claim or interest to the person or persons whose names and addresses are set out below, and to send such particulars before the date specified in relation to that deceased person displayed below, after which date the personal representatives will distribute the estate among the persons entitled thereto having regard only to the claims and interests of which they have had notice and will not, as respects the property so distributed, be liable to any person of whose claim they shall not then have had notice

Name of Deceased (Surname first)	Address, description and date of death of Deceased	Names addresses and descriptions of Persons to whom notices of claims are to be given and names, in parentheses, of Personal Representatives	Date before which notice of claims to be given	
BIRCH, Margaret	Hythe Road, Worthing, West Sussex, BN11 5DA . 30 May 2022	c/o Adams & Remers, Trinity House, School Hill, Lewes, East Sussex, BN7 2NN. (Elizabeth Cole and Stephen Johnson.)	2 November 2023	(4431906)
ENGLAND, Brian	191 Greenside, Staincross, Barnsley, S75 6AZ . 12 April 2023	MKB Solicitors LLP, 1-11 Huddersfield Road, Barnsley, South Yorkshire S70 2LP. Reference : RLP/ E6335. (Steven Dawson and Martin Sutton.)	1 November 2023	(4431905)
LOMAS, Roger Sydney	477 Tessall Lane, Northfield, Birmingham B31 5HA . 30 January 2022	Gordon Jones & Company, Solicitors, 137 New Road, Rubery, Birmingham, B45 9JR; reference DJW.rk.22/092 Lomas.	1 November 2023	(4431907)
TAYLOR, Christopher Ivan	2 Church Close, Alveston, Stratford- upon-Avon, Warwickshire, CV37 7QG . Company Director Butchers (retired). 14 October 2022	Heath & Blenkinsop Solicitors, 42 Brook Street, Warwick, Warwickshire, CV34 4BL. (Anton Taylor and James Taylor.)	1 November 2023	(4431908)
ADAMS, Maurice Clifford	103 Queens Road, Loughton, Essex, IG10 1RR. 14 June 2022	Foskett Marr Gadsby & Head LLP, 181 High Street, Epping, Essex CM16 4BQ (John Paul Adams, Jennifer Ann Laplain and Julie Elizabeth Jones.)	1 November 2023	(4426146)
ALFORD, Jennifer Ann (Jennifer Ann Adams)	3 Brookfield, Clovelly Road, Bideford, Devon, EX39 3DP. 9 March 2023	Brewer Harding & Rowe LLP, 29 Bridgeland Street, Bideford, Devon, EX39 2PT	1 November 2023	(4428909)
ALLAN, Doreen Evelyn Ruth	Apartment 15, Knightwood Mews, Shannon Way, Chandlers Ford, Hampshire, SO53 4TL. 6 May 2023	Warner Goodman LLP, Portland Chambers, 66 West Street, Fareham, PO16 0JR (Ref: AZM.ALL0000547.2)	1 November 2023	(4428702)
ALLEN, Mrs Jeanette Jeanette Vaughan)	Powys Lodge, 1 Etloe Road, Bristol, BS6 7PG16 St. Edyths Road, Bristol, BS9 2ES. 11 November 2022	Clarke Willmott LLP, Rachel Tonkinson , Blackbrook Gate, Blackbrook Park Avenue, Taunton, TA1 2PG.	1 November 2023	(4431973)
ALLOTT, Timothy Stuart	Church Cottage, Brearton, Harrogate, HG3 3BX. 26 May 2023	Greenwoods Solicitors Ltd, 88-90 High Street, Knaresborough, HG5 0EA (Lesley Ann Allott.)	1 November 2023	(4426128)
ATKINS, Derek	11, Ash Grove, Armthorpe, Doncaster, DN3 3EY. 8 June 2023	Dawson & Burgess, 33 Church Street Armthorpe Doncaster, DN3 3AE (Kristen Adele Craig & Antony Henshaw .)	1 November 2023	(4427433)
AZIZ, Michael Nissan Michael Nissan Azizollahoff)	27 Tenterden Gardens, London, NW4 1TG. 25 May 2023	Pearlmans Solicitors Llp, 39 Finchley Lane, London, Barnet, GB, NW4 1BX	1 November 2023	(4427498)
BAKER, Eileen Mary	70 Shelley Drive Broadbridge Heath Horsham West Sussex, RH12 3NT. 8 July 2023	Coole Bevis LLP, 14 Carfax, Horsham, West Sussex, RH12 1DZ (James Pickles.)	1 November 2023	(4428125)
BAKER, Miss Jane Jennifer (Miss Jane Baker)	Flat 22, Quarry Chase, 30 Poole Road, Bournemouth, BH4 9DD. 14 August 2023	Solomons Law, Paul Solomons , Hamilton Court House, 1-3 Alum Chine Road, BOURNEMOUTH, BH4 8DT.	1 November 2023	(4430981)

Name of Deceased (Surname first)	Address, description and date of death of Deceased	Names addresses and descriptions of Persons to whom notices of claims are to be given and names, in parentheses, of Personal Representatives	Date before which notice of claims to be given	
BARBER, Cedric Clive	4 George Street, Ellesmere Port, CH65 2BR. 8 August 2022	TAYLOR BROWN SOLICITORS, 78 Whitby Road, Ellesmere Port, GB, CH65 0AA (Mark Frederick Howard Brown.)	1 November 2023	(4427333)
BARKER, Richard Ernest	152 Westerfield Road, IPSWICH, IP4 3AF. Solicitor. 12 October 2022	David Barker, The London Gazette (35371), PO Box 3584, Norwich, NR7 7WD.	1 November 2023	(4430894)
BARNES, Helen Pauline Veronica	13 Sutherland Avenue, Broadstone Dorset , BH18 9EB . 15 March 2023	HAROLD G WALKER, 30 West Borough, Wimborne, Dorset, GB, BH21 1NF (Nicola Anne Lowe and Simon Jonathan Nethercott.)	1 November 2023	(4428907)
BARSDELL, Irene Lucy	Forrester Court Nursing Home, 5 Cirencester Street, LONDON, W2 5SR. 8 May 2023	Colligenda Ltd, Adam Shute, St. Marys Court, The Broadway, AMERSHAM, HP7 0UT.	1 November 2023	(4432079)
BARUA, Gautam Prosad	71 Quickswood, London, NW1 2NB. 8 July 2022	Hodge Jones & Allen LLP, 180 North Gower Street, London, NW1 2NB	1 November 2023	(4428154)
BELL, Mr Roderick Charles (Roderick Bell)	5 Penventon View, Helston, Cornwall, TR13 8SD. 6 June 2023	Jessica Forrester c/o Hugh James, Two Central Square, Cardiff, CF10 1FS.	1 November 2023	(4432007)
BERRY, Betty Elizabeth	Oxford Beaumont, Bayworth Lane, Boars Hill, Oxford, . 16 July 2023	RWK Goodman, Godstow Court Minns Business Park 5 West, Oxford, GB, OX2 0JB	1 November 2023	(4431767)
BETTS, Jonathan Scott (Jonathon Scott Betts)	St Christophers, Horseshoe Lane, Chadlington, Chipping Norton, OX7 3NB. 21 March 2023	John Welch & Stammers, 24 Church Green, Witney, Oxfordshire, GB, OX28 4AT	1 November 2023	(4428949)
BICKERTON, Frank	Sutton Hall Farm, Sutton Lane Middlewich, CW10 0ES. 24 September 2021	Knights, The Brampton Newcastle- under-Lyme, Staffordshire, ST5 0QW	1 November 2023	(4425570)
BLAIR , Margaret Jane Elizabeth	32 Hawthorn Avenue, Maltby, Rotherham, South Yorkshire, S66 8BT. 20 November 2022	WALKER & CO SOLICITORS, 82 High Street, Maltby, Rotherham S66 8RL (John George Conlon and Patricia Conlon.)	1 November 2023	(4426838)
BLOOM, Mrs Maureen	131 Main Street, Cayton, SCARBOROUGH, YO11 3TE. 8 October 2021	David Whitehead, The London Gazette (35373), PO Box 3584, Norwich, NR7 7WD.	1 November 2023	(4430896)
BODEN, Derek (Derrick Boden)	20 Condover Close, WALSALL, WS2 0NN. 1 February 2021	Dorothy Vera Heeley, C/O Treethorpe Limited, 9 Caxton House, Broad Street, CAMBRIDGE, CB23 6JN.	1 November 2023	(4432070)
BOOKER, Barry	Ferndale, Redwell Lane, Ightham, Sevenoaks, Kent, TN15 9EH. 20 March 2020	Warners Law LLP, Bank House, Bank Street, Tonbridge, Kent, TN9 1BL (Tina Reynolds and Nigel Paul Hardman.)	1 November 2023	(4425408)
BORNOR, Clive Sinclair	5 Langham Grove, Maidstone, Kent, ME16 0LA. 2 July 2023	Honey Legal, 28 Talbot Lane, Leicester, GB, LE1 4LR (Preet Bansal.)	1 November 2023	(4426125)
BOSTOCK, Irene	11 Oakfield Road, Exeter, Devon, EX4 1BA. 19 April 2023	Trowers & Hamlins LLP, The Senate, Southernhay Gardens, Exeter EX1 1UG (Amanda Rainford, Nicholas Hawker and Valerie Hawker.)	1 November 2023	(4428970)
BOURTON, Sylvia Joyce (Sylvia Joyce McKnight)	34 Villiers Road, Bicester, OX26 2BB. 7 April 2023	S E Solicitors, 52-54 The Green, South Bar Street, Banbury, OX16 9AB (Lucy Gordon.)	1 November 2023	(4427346)
BOWEN, Margaret Doreen	25 Cliston Avenue, Exmouth, Devon, EX8 4LJ. 13 October 2022	Trowers & Hamlins LLP, The Senate, Southernhay Gardens, Exeter, EX1 1UG (Amanda Rainford, Frank Cook and Andrew Bowen.)	1 November 2023	(4428968)
BOWER , Thomas John	52 Riland Road, Sutton Coldfield, B75 7AT. 9 November 2021	Enoch Evans LLP, 5 Emmanuel Court, 2 Mill Street, Sutton Coldfield, B72 1TJ (Mary Bower.)	1 November 2023	(4429110)

Name of Deceased (Surname first)	Address, description and date of death of Deceased	Names addresses and descriptions of Persons to whom notices of claims are to be given and names, in parentheses, of Personal Representatives	Date before which notice of claims to be given	
BOWKETT, Derek Arthur	4 Elmdale, Ewyas Harold, Herefordshire, HR2 0HZ. 24 July 2022	Humfrys & Symonds Solicitors, 1 St John Street, Hereford, HR1 2ND (Tracey Cairns.)	1 November 2023	(4427990)
BRAILSFORD, David Henry	9 Fairway Drive, Beeston, Nottingham, NG9 4BN. 24 March 2022	ProTrust Consulting Ltd, 4 Manor Farm Close, Stone, Aylesbury, Buckinghamshire, HP17 8RQ (Ref: BRA005) (Robert Ian Cartmell.)	1 November 2023	(4427430)
BRISTOW, Julie	14 The Wish, Kenardington, ASHFORD, TN26 2NB. 12 August 2023	Jack Bristow, The London Gazette (35388), PO Box 3584, Norwich, NR7 7WD.	1 November 2023	(4431702)
BROCKLEHURST, George Douglas	25 Stanhope Road, Mickleover, Derby, DE3 9HG. 18 May 2023	SIMPSON JONES SOLICITORS, 2 Stuart Street, Derby, GB, DE1 2EQ (Ref: WIL6951) (Simpson Jones Solicitors.)	1 November 2023	(4426694)
BROWN, Ronald William	120 Pitmaston Road, Birmingham, B28 9PN. 7 January 2022	Moseley Chapman & Skemp Limited, 4 Trinity Place, Midland Drive, Sutton Coldfield B72 1TX (Ref: JT.BR0037/001) (Stanley William Ronald Dawes.)	1 November 2023	(4426891)
BROWN, Rosa Mary	St Ives House, Horton Road, Ashley Heath, Ringwood, BH24 2EE. 27 June 2023	Ellis Jones Solicitors LLP, 302 Charminster Road, Bournemouth, Dorset, BH8 9RU	1 November 2023	(4428965)
BURNS, David George	Millwood, Coldalhurst Lane, Astley, Tyldesley, Manchester, M29 7BS. 3 January 2023	Potter Rees Dolan (a Hugh James Business), 12 Commercial Street, Manchester M15 4PZ (Ref: BUR01465/1 Attn: Gregory Poole) (Kyle David Burns.)	1 November 2023	(4427467)
BURTON, Gillian Mary	Alice Grange Care Home, St Isidores, Kesgrave, Ipswich, IP5 2GA. 8 May 2023	Barker Gotelee, 41 Barrack Square, Martlesham Heath, Ipswich, IP5 3RF (Alan Downs and Carol Rosemary Downs.)	1 November 2023	(4428721)
CALVERT, John Stafford	Flat 2, The Brae, 17 Longdown Road, Farnham, GU10 3JU. 10 October 2022	RWK GOODMAN, 5-6 Northumberland Buildings, Queen Square, BATH BA1 2JE (James David Mcneile.)	1 November 2023	(4426840)
CHADWICK, Edith	53 Hesketh Road, Burscough, Ormskirk, Lancashire, L40 7SG. 28 July 2023	Dickinson Parker Hill, 22 Derby Street, Ormskirk, Merseyside, GB, L39 2BZ	1 November 2023	(4428107)
CHANNON, Colin George	24 Beverley Way, London, SW20 0AG. 23 April 2022	Amphlett Lissimore, 80-86 Westow Street, London, SE19 3AF (Akilah Louise McEwen.)	1 November 2023	(4426354)
CHECKLEY, Florrie	Lower Farm House, Lower Farm Lane, Mollington, Banbury, Oxfordshire, OX17 1BJ. 4 July 2022	Aplin Stockton Fairfax Solicitors, 36 West Bar, Banbury, Oxfordshire OX16 9RU (Ref: 011528/004/MP) (Linda Joyce Parker, Sally Patricia Burrows and Keith Andrew Checkley.)	1 November 2023	(4427530)
CHRISTOPHER, Margaret	The Beeches Residential Home, Beech Road, Armthorpe Doncaster, DN3 2DZ. 8 July 2023	Dawson & Burgess Solicitors, Dawson & Burgess Solicitors 3 South Parade, Hall Cross Hill, Doncaster, DN1 2DZ (Kristen Adele Craig & Antony Henshaw.)	1 November 2023	(4426355)
CHUBB, Eric Francis	22 Porch Way, Whetstone, London, N20 0DS. 3 July 2022	Nash & Co Solicitors Llp, Beaumont House, Beaumont Park, Plymouth, GB, PL4 9BD (Kenneth Victor Chubb.)	1 November 2023	(4425931)
CLARKE, Mr Francis Neil	11 Mansel Court, CAMBRIDGE, CB4 2EU. post room worker (retired) . 26 January 2023	Jonathan Charles Clarke, The London Gazette (35389), PO Box 3584, Norwich, NR7 7WD.	2 November 2023	(4431703)
CLUBB, Charles Joseph	10 A Osborne Gardens, Herne Bay, Kent, CT6 6SH. 13 February 2023	Honey Legal, 28 Talbot Lane, Leicester, GB, LE1 4LR (Adam Sym.)	1 November 2023	(4428961)

Name of Deceased (Surname first)	Address, description and date of death of Deceased	Names addresses and descriptions of Persons to whom notices of claims are to be given and names, in parentheses, of Personal Representatives	Date before which notice of claims to be given	
COOK, Pamela Joan	Rush Hill Mews, Clarks Way, Bath, BA2 2TR. 10 January 2023	RWK GOODMAN, 5-6 Northumberland Buildings, Queen Square, Bath BA1 2JE (Julian Cook, Christopher Cook.)	1 November 2023	(4427379)
CRANE, David Mark	11a church Lane, Cayton Scarborough, YO11 3SA. 30 October 2021	PINKNEY GRUNWELLS LAWYERS LLP, 64 Westborough, Scarborough, GB, YO11 1TS (Richard Ian Sunley and Mrs Kate Una Sunley.)	1 November 2023	(4428723)
CRANE, Philip Edward	Manor House Residential Home, Manor House 58-60 Main Street, Middleton Market Harborough LE16 8YU formerly of 50 Windsor Gardens Lower Street, Kettering Northamptonshire, NN16 8DU. 14 June 2023	Lamb & Holmes, Lamb & Holmes, Uppingham House, Headway Business Park 5 Saxon Way West Corby Northants, NN18 9EZ (Richard Neville White.)	1 November 2023	(4426408)
CRONIN, Alan James	8 Wall Road, Ashford, Kent, TN24 8NZ. 6 May 2023	Hallett & Co, 11 Bank Street, Ashford, Kent, TN23 1DA	1 November 2023	(4428115)
DALTON, Winifred Edna	54 Rylands Road, Kennington, Ashford, Kent, TN24 9LN. 26 May 2023	Hallett & Co, 11 Bank Street, Ashford, Kent, TN23 1DA (Nina Dawn Mantle and Faye Anita Dalton.)	1 November 2023	(4426684)
DARWELL-TAYLOR, Peter Anthony	Liberham Lodge, Rectory Lane, Little Bookham, Leatherhead, Surrey KT23 4DY Formerly of 45 Montrouge Crescent, Epsom, Surrey, KT17 3PB. 5 August 2023	Trowers & Hamlins LLP, The Senate, Southernhay Gardens, Exeter, EX1 1UG (Amanda Rainford, Timothy Darwell-Taylor and Simon Darwell- Taylor.)	1 November 2023	(4429015)
DAVIS, Mary Rose	Rema 22 Parkside, Pinhoe Exeter, EX1 3TN. 18 March 2023	Trowers & Hamlins LLP, Trowers & Hamlins LLP, The Senate Southernhay Gardens, Exeter, EX1 1UG (Amanda Rainford and Anthony Davis.)	1 November 2023	(4428700)
DAWSON, Janet	Dulverton House, 9 Granville Square, Scarborough, YO11 2QZ. 22 February 2023	Thorpe & Co Solicitors, 17 Valley Bridge Parade, Scarborough, GB, YO11 2JX (C D Burnett & C A McNeill.)	1 November 2023	(4427137)
DENNIS, Albert Henry	West View, Northlew, Okehampton, Devon, EX20 3BR. 19 May 2023	Trowers & Hamlins LLP, Trowers & Hamlins LLP, The Senate Southernhay Gardens Exeter, EX1 1UG (Amanda Rainford and Timothy Parsons.)	1 November 2023	(4428703)
DONALDSON, Jacquelyn Elizabeth	10 Dawlish Close, LIVERPOOL, L25 9QR. 19 April 2023	Athena Probate LTD, Sophie Blanchard, 1st Floor, 34 The Oaks, Manston Business Park, RAMSGATE, Kent, CT12 5FN.	1 November 2023	(4432063)
DOREY, Alan Jackson	Paintons Farm, Netherton, Abingdon, OX13 5LS. 5 June 2023	Freeths LLP, 5700 Oxford Business Park South Oxford OX4 2RW	1 November 2023	(4428474)
DOYLES , Adrian Michael	56 Main Street, Wadworth Doncaster, DN11 9AW. 31 August 2022	Andrew Isaacs Law Limited, Unit 7 Atlas Office Park, Balby Carr Bank, Doncaster, DN4 5JT (Amanda King and Reuben Glynn.)	1 November 2023	(4426352)
DREW, Lyndon Haydn	Whitehaven, 5 Cledwyn Gardens, Trecynon, Aberdare, CF44 8HP. 28 July 2023	GGP LAW, 130 High Street, Blackwood, GB, NP12 1AF (Emily Clarke, GGP Law Ltd .)	1 November 2023	(4428146)
DURNIN, Mrs Bridget Mary (Brid Mary Durnin)	Pembroke Lodge, 32 Alexandra Road, READING, RG1 5PF. Parish Assistant (retired). 13 November 2022	Niamh Mary Durnin-Duffy, c/o C&R Legal Limited Solicitors, High Street, Lechlade, Gloucestershire, England, GL7 3AE.	2 October 2023	(4430314)
ELKIN, Steven John	115 Duckpool Road, NEWPORT, NP19 8FLParklands Residential Home, 9 Montgomery Road, Malpas, Newport, NP20 6QE. 5 July 2022	Athena Probate LTD, Sophie Blanchard, 1st Floor, 34 The Oaks, Manston Business Park, RAMSGATE, Kent, CT12 5FN.	1 November 2023	(4432059)

Name of Deceased (Surname first)	Address, description and date of death of Deceased	Names addresses and descriptions of Persons to whom notices of claims are to be given and names, in parentheses, of Personal Representatives	Date before which notice of claims to be given	
ELLIOTT, Michael	58 Seymour Drive, Ellesmere Port, CH66 1LX. 10 August 2022	TAYLOR BROWN SOLICITORS, 78 Whitby Road, Ellesmere Port, GB, CH65 0AA (Mark Frederick Howard Brown.)	1 November 2023	(4427515)
ELLIOTT, Michael	58 Seymour Drive, Ellesmere Port, CH66 1LX. 10 August 2022	Taylor Brown Solicitors, 78 Whitby Road, Ellesmere Port, GB, CH65 0AA (Mark Frederick Howard Brown.)	1 November 2023	(4428456)
ELLIS, John James	Helen Court, Riverview, Maldon Road, Witham, Essex, CM8 1FX. 17 August 2023	Birketts LLP, Brierly Place, New London Road, Chelmsford, Essex CM2 0AP	1 November 2023	(4427348)
ELLIS, William	17 Derby Place, Leeds Road, BRADFORD, BD3 8PL. 17 October 2022	Colligenda Ltd, Adam Shute, St. Marys Court, The Broadway, AMERSHAM, HP7 0UT.	1 November 2023	(4432075)
EVANS, Norman	56 Homerton Road, Pallister Park, Middlesbrough, TS3 8LX. Groundsman (Retired). 21 December 2021	Fidelis Legal Services Limited, Fidelis House, 5 Low Moor Road, Lincoln, LN6 3JY. Ref: FEA1432H. (Helen Claydon and Ian Peace)	6 November 2023	(4430235)
FALLON, EILEEN ELIZABETH	ASHWOOD NURSING HOME, 43 SPALDING COMMON, SPALDING, LINCOLNSHIRE, United Kingdom, PE11 3AU. 22 May 2022	Lynne Thornton Solicitor, 50 CREWE ROAD, SHAVINGTON, CREWE, CHESHIRE, CW2 5JB.	2 November 2023	(4430452)
FANCOURT, John Edward	22 Highfield Road, Dereham, NR19 2EY. 26 January 2023	Scott Rowe Solicitors, Chard Street, Axminster, Devon, EX13 5DS (Philip Nigel White.)	1 November 2023	(4428233)
FARNAN, Mr Thomas Anthony	Fir Tree Lane, Burtonwood, WARRINGTON, WA5 4NN23 Acton Road, Burtonwood, Warrington, WA5 4HT. Retired Builder. 3 November 2021	Nyland & Beattie, Old Town Hall, Victoria Square, WIDNES, WA8 7QZ.	1 November 2023	(4430929)
FILLARY, Derek Eldred	Parklands Nursing Home, 21-27 Thundersley Park Road, BENFLEET, SS7 1EGMundy House Care Home, Church Road, Basildon, SS14 2EY. 23 March 2023	Athena Probate LTD, Sophie Blanchard, 1st Floor, 34 The Oaks, Manston Business Park, RAMSGATE, Kent, CT12 5FN.	1 November 2023	(4431956)
FOX, Miss Linda Mary (Linda Fox)	3 North Mead, Henfield, West Sussex, BN5 9BW. 20 January 2022	Joanna Bridges c/o Ellis-Fermor & Negus, 2 Devonshire Avenue, Beeston, NG9 1BS.	1 November 2023	(4432066)
FOYLE, Alexander Edward	Woodstock, Hall Lane, Holcombe, Dawlish, Devon, EX7 0JP. 13 June 2023	Trowers & Hamlins LLP, The Senate Southernhay Gardens, Exeter EX1 1UG (Amanda Rainford and Maureen Houghton.)	1 November 2023	(4428979)
GADHIA, Rameshchandra Narandas	Flat 35, Lutyens Lodge, 523 Uxbridge Road, Pinner, HA5 4JX. 21 October 2022	Cornerstone Wills, Basepoint Business Centre, 377-399 London Road, Camberley, GU15 3HL (Susan Ann O'Loughlin.)	1 November 2023	(4427514)
GADHIA, Hasumati Rameshchandra	1 Perrin Road, Wembley, HA0 2NW. 28 September 2021	Cornerstone Wills, Basepoint Business Centre, 377-399 London Road, Camberley, GU15 3HL (Susan Ann O'Loughlin.)	1 November 2023	(4427664)
GARRY, Joan	76 Penland Road, Haywards Heath West Sussex, RH16 1PH. 19 April 2022	Rix & Kay, The Courtyard, River Way, Uckfield, East Sussex, TN22 1SL (Ref: JCB/G7446/1) (Peter Sprague Garry.)	1 November 2023	(4427500)
GARWOOD, Stephen Philip	2 Mansbrook Villas, Nacton, Ipswich, Suffolk, IP10 0JL. 10 August 2023	Kerseys Solicitors LLP, 32 Lloyds Avenue, Ipswich, Suffolk IP1 3HD (Lee Perry Garwood.)	1 November 2023	(4429171)
GAY, Trevor Frederick	1 Austens Orchard, Smallhythe Road, Tenterden, Kent, TN30 7LQ. 21 June 2023	Whitehead Monckton Limited, 3-4 Market Square, High Street, Tenterden, Kent TN30 6BN (Ref: LEH/ G44146/1/Gay) (Peter Gay, Simon Gay, Jill Blackford & Helen Gay.)	1 November 2023	(4426899)

Name of Deceased (Surname first)	Address, description and date of death of Deceased	Names addresses and descriptions of Persons to whom notices of claims are to be given and names, in parentheses, of Personal Representatives	Date before which notice of claims to be given	
GOODGER, Mrs Barbara May (Mrs Barbara Goodger, Miss Barbara May Stuart)	Flat 51, Shepheards House, Manor Park Road, CHISLEHURST, BR7 5FT. 2 August 2023	Clarkson Wright & Jakes, Abigail Doust , Valiant House, 12 Knoll Rise, ORPINGTON, BR6 0PG.	1 November 2023	(4431004)
GOULD, Jane Louise (Jane Louise Goodyear)	4 Boskenna Road, Four Lanes, Redruth, Cornwall, TR16 6LQ, formerly of Upton Grey Lodge, Upton Grey, Basingstoke, Hampshire, formerly of Copyhold House, Hazelmere, formerly of 1 Humbley Grove Cottages, South Warnborough, Hook, Hampshire, formerly of 3 Manor Farm Cottages, Upton Grey, Basingstoke, Hampshire, formerly of 17 Little Hoddington, Upton Grey, Basingstoke, Hampshire, . 15 December 2022	DB Law Solicitors LLP, Camborne Business Centre, College Street, Camborne, Cornwall, TR14 7DD (Deborah Mary Tanguay & William David Goodyear.)	1 November 2023	(4426858)
GRAHAM, Ena Dorothy (Dorothy Graham)	Birmingham, B31, . 2 July 2023	Davisons Law, 63-65 Beckbury Road, Weoley Castle, Birmingham, B29 5HS	1 November 2023	(4425449)
GREEN, Mr Peter James (Peter Green)	Halsey House, 31 Norwich Road, Cromer, Norfolk, NR27 0BA25 Roughton Road, Cromer, Norfolk, NR27 0HH. 30 May 2023	Hansells Solicitors, Cambridge House, 26 Tombland, Norwich, NR3 1RE.	1 November 2023	(4430933)
GREVILLE, Marjorie Elizabeth ( (Elizabeth Greville))	5 Glenalmond House, Manor Fields, London, SW15 3LP and 204-1177 Yonge Street, Toronto, Ontario, M4T2Y4, Canada. Lawyer (Retired). 22 November 2022	Winckworth Sherwood LLP, Minerva House, 5 Montague Close, London, SE1 9BB. Ref: ALBE/43336/1/TJSS. (Timothy James Somerville Snaith)	8 November 2023	(4430226)
GRIMES, Terence	6 Vernon Court, Stanmore Middlesex, HA7 2BN. 14 February 2023	HECKFORD NORTON, 19 Leys Avenue, Letchworth, Hertfordshire, SG6 3EB (Thomas Grimes & Nicola Grimes.)	1 November 2023	(4427769)
HARPER, Mrs Rose Marie	56, Grainsby Avenue, Holton le Clay, United Kingdom, DN36 5DA. Retired. 16 February 2023	Florins Wills & Probate Limited, David Harper, 132, Eastgate, Louth, United Kingdom, LN11 9AA.	2 November 2023	(4431806)
HEAD, Miss Amy Peggy	Avery Homes Ltd, Milton Court Care Centre, Milton Keynes, Bucks, United Kingdom, MK7 6JD. Retired Chartered Secretary. 23 January 2022	Lynne Head, The London Gazette (35375), PO Box 3584, Norwich, NR7 7WD.	1 November 2023	(4430898)
HICKS, Bryan Edgar	84 Pole Barn Lane, Frinton on Sea, Essex, CO13 9NH Previously of 65 Rainham Way, Frinton on Sea, Essex, CO13 9NR. 28 March 2023	Sparlings Solicitors, 62A Connaught Avenue, Frinton on Sea, Essex, CO13 9QH	1 November 2023	(4428140)
HILL, John Colin	10 The Ridgeway, RUISLIP, HA4 8QS. 9 July 2023	Valued Estates Ltd, 7 Church Barns, Ware Road, Widford, Hertfordshire, United Kingdom, SG12 8RL.	1 November 2023	(4431957)
HITCHINER, Marion	15 Telford Road Tamworth, B79 8EZ. 19 September 2022	TAYLOR FAWCETT, 14 Albert Street, Harrogate, GB, HG1 1JT (Andrew Stephen Fraser.)	1 November 2023	(4428015)
HOSKIN, Mr Gail	8 Fairfield Close, Newport, Newport, NP18 3DR. 28 July 2023	Kate Toner c/o Jacklyn Dawson Solicitors, Equity Chambers, John Frost Square, Newport, NP20 1PW.	31 October 2023	(4430864)
JARRETT, MR CARL FONSECA (CARL JARRETT)	13 Blackshots Lane, Grays, Essex, RM16 2JT. 21 March 2023	Andrew Hawkins at Kidd Rapinet LLP, WALKER HOUSE, GEORGE STREET, Aylesbury Buckinghamshire, HP20 2HU.	1 November 2023	(4432064)
JACKSON, Dennis	216 Derbyshire Hill Road, Parr, St Helens, WA9 2LU. 23 January 2023	Canter Levin and Berg, St Nicholas House, Old Churchyard, Liverpool, L2 8TX (Martin Malone and Claire Jansz.)	1 November 2023	(4426845)

Name of Deceased (Surname first)	Address, description and date of death of Deceased	Names addresses and descriptions of Persons to whom notices of claims are to be given and names, in parentheses, of Personal Representatives	Date before which notice of claims to be given	
JEFFERSON, Charles Arthur	32 Riversdale Drive, Goole, East Yorkshire, DN14 5LJ. 21 April 2023	Symes Bains Broomer, Port Office, East Parade, Goole, East Yorkshire, DN14 5RB (Andrew Samuel Horwich.)	1 November 2023	(4427373)
JERMY, Joan Mabel	135 Baldock Road, Letchworth Garden City, Hertfordshire, SG6 2EQ. 5 July 2023	HECKFORD NORTON, 19 Leys Avenue, Letchworth, GB, SG6 3EB (Karen Jean Georgiou and Doreen Mary Lovell.)	1 November 2023	(4428258)
JONES, Lynda	2 Swallow Road, Packmoor Stoke on Trent, ST7 4GG. 13 January 2023	CHARLTONS SOLICITORS, Saracen House, 84 High Street, Staffordshire, GB, ST8 6AS (Sharon Farrow.)	1 November 2023	(4428925)
JONES, Sian Alison	4 Belgrave Gardens, Dereham, Norfolk, NR19 1PZ. 29 July 2023	Heritage Limited, 13 Hellesdon Park Road, Drayton High Road, Norwich, Norfolk NR6 5DR (Ref: P2444) (Mr Stuart Jan Burrell.)	1 November 2023	(4428095)
KEAR, Paul Christopher	Flat 5, 150 Highgate, Kendal, Cumbria, LA9 5AH. 27 July 2023	JOHN WELCH & STAMMERS, 24 Church Green, Witney, Oxfordshire, GB, OX28 4AT	1 November 2023	(4426844)
KEELER, Neville David	Sunrise, Vicarage Hill, Petham, Canterbury, Kent, CT4 5RE. 26 November 2021	Boys and Maughan, incorporating Gardner Croft, 2 Castle Street, Canterbury, Kent, GB, CT1 2QH (Timothy Townsend.)	1 November 2023	(4427484)
KENT , Anne Veronica	5 Chapel Lane, Nettleham, Lincoln, LN2 2NX. 23 March 2022	PAGE NELSON SOLICITORS LIMITED, Victory House, Henley Way, Doddington Road, Lincoln, GB, LN6 3QR (Amanda Louise Johnson and Steven Forster.)	1 November 2023	(4425541)
KIM, Mr Chang Jin (Mr Chang Kim)	93 Manor Road South, ESHER, KT10 0QB. 2 November 2022	Herrington Carmichael LLP, Charlotte Woods , Brennan House, , Farnborough Aerospace Centre Business Park, Farnborough, GU14 6XR.	1 November 2023	(4430984)
KOWALEWICZ, Magdolna Maria	7 Bank Top Middleton Manchester, M24 1DW. 27 August 2015	Latimer Lee Solicitors LTD, 13-15 Square Street Ramsbottom BL0 9BL (Stephen Reid Latimer .)	1 November 2023	(4428969)
LAKE, Mr Benjamin Tudor Powell (Lake)	25 Rawlinson Avenue, DERBY, DE23 8WA. 12 February 2022	Kathryn Rachel Sideras, The London Gazette (35378), PO Box 3584, Norwich, NR7 7WD.	3 November 2023	(4430921)
LARNER, Derek Thomas	24 Whitelands Road, Cirencester, Gloucestershire, GL7 1AJ. 1 November 2021	Red Kite Law LLP, 10/12 Dollar Street, Cirencester, Gloucestershire, GL7 2AL (Sara Whitehead.)	1 November 2023	(4428149)
LAWRENCE, Vera Veronica	61 Archers Court, Salisbury, SP1 3WE. 25 March 2023	Hartley & Worstenholme, 20 Bank Street, Castleford WF10 1JD (Ref: AWR/LAW00256/00001) (Jonathan Edward Sharp and Christopher Wilton.)	1 November 2023	(4426802)
LINDLEY, Valerie	16 Kyrle Close, Ironbridge Telford Shropshire, TF8 7BQ. 15 March 2023	CLARKES, Hazledine House, Telford Centre, Telford, Shropshire, TF3 4JL	1 November 2023	(4428136)
LINTON, Daphne Elizabeth Ann	10 Whitecliff Road, Poole, Dorset, BH14 8DU. 16 August 2023	HEPPENSTALLS SOLICITORS LTD, South Wing, Lymington Town Hall, Avenue Road, Lymington, Hampshire, SO41 9BF (Suzanne Edwina Allan.)	1 November 2023	(4428311)
LLOYD, John Howard Patrick	3 Vine House, High Street, Staplehurst, Tonbridge, Kent, TN12 0AR. 2 May 2023	Warners Law LLP, Bank House, Bank Street, Tonbridge, Kent, TN9 1BL (Mary Vesper Shaw.)	1 November 2023	(4428918)
LODGE, Richard	129 Cornflower Drive, Chelmsford, CM1 6XZ. 27 April 2022	TAYLOR FAWCETT, 14 Albert Street, Harrogate, GB, HG1 1JT	1 November 2023	(4428137)
MABEY, Brian Harry	Annabel House, 57 Lower Bristol Road, Weston Super Mare, BS23 2PX. 28 January 2023	JMD Law, 26-28 James Street, Cardiff Bay, CF10 5EX (Ref: 191406) (Jeanette James.)	1 November 2023	(4429058)

Name of Deceased (Surname first)	Address, description and date of death of Deceased	Names addresses and descriptions of Persons to whom notices of claims are to be given and names, in parentheses, of Personal Representatives	Date before which notice of claims to be given	
MACPHAIL, Mary Majorie	Cherry Blossom Manor, German Road, Bramley, Tadley, RG26 5GF. 22 April 2022	BATES SOLICITORS, 43 Essex Street, London, WC2R 3JF (Ian Macdonald and Isobel Macdonald.)	1 November 2023	(4429159)
MATALA, Ngoywakabira	18 Hawthorn Avenue, Thornton Heath Surrey, CR7 8BU. 19 February 2023	TILBURY GODDARD, 727B London Road, Thornton Heath, Surrey, CR7 6AU (Ref: SW/Matala) (Ntumba Katshimuaji Matala.)	1 November 2023	(4428620)
MATON, Mrs Brenda Olive	26 Capella Gardens, Dibden, SOUTHAMPTON, Hampshire, SO45 5UE. Retired Housekeeper. 4 April 2023	Alison Michelle Conner, 38 Whitewater Rise, Dibden Purlieu, SOUTHAMPTON, Hampshire, SO45 4BY.	1 November 2023	(4428941)
MATTHEWS, Mr Robert John	The Willows, Purls Bridge Drove, MARCH, PE15 0ND. 29 June 2022	Ward Gethin Archer Solicitors, 3 Regis Place, North Lynn Industrial Estate, KINGS LYNN, PE30 2JN.	1 November 2023	(4430928)
MAUVAN, Derrick William	84 Grandison Rise, Tupsley Hereford, HR1 1PR. 13 April 2023	Lanyon Bowdler, Bath Street Offices, Bath Street, Hereford, GB, HR1 2GY	1 November 2023	(4426194)
MCDONNELL, Eunice	14 Laurel Drive, Willaston, Wirral, CH64 1TW. 13 June 2023	Lloyds Bank Plc, Po Box 5005, Lancing, GB, BN99 8AZ (Lloyds Bank Plc.)	1 November 2023	(4428375)
MCQUISTAN, Andrea Elizabeth	87 Stone Lane, Worthing West Sussex, BN13 2BD. 1 May 2021	STREATHERS, 1 Crouch End Hill, London, GB, N8 8DH	1 November 2023	(4428133)
MCLACHRIE, Miss Fiona Jane	18 Swanfield Road, STOURBRIDGE, West Midlands, DY8 5UR. Incident Command Despatcher. 9 August 2023	Heather Mclachrie, The London Gazette (35362), PO Box 3584, Norwich, NR7 7WD.	3 November 2023	(4430385)
MILLER, Peter	73 Fisher Avenue, Warrington, WA2 9PG. 10 August 2023	Vincent Sykes Solicitors LLP, 4 West Street, Oundle, Peterborough, PE8 4EF (Garry Peter Miller.)	1 November 2023	(4426176)
MITCHELL, Mark	65 Rydal Avenue, Warrington, WA4 6AT. 8 August 2021	Susan Howarth & Company, 41 Chesterway, Northwich, GB, CW9 5JE (Sarah Alderton.)	1 November 2023	(4428418)
MITTAL, Indu (Indu Gupta)	Wimbledon Beaumont Care Home, 35 Arterberry Road, Wimbledeon, SW20 8AG. 4 September 2022	Laurus Law, 30 Dukes Place, London, EC3A 7LP	1 November 2023	(4426776)
MORGAN, John Francis	16 Millway, Wellington, HEREFORD, HR4 8AS. 10 November 2021	Athena Probate LTD, Sophie Blanchard, 1st Floor, 34 The Oaks, Manston Business Park, RAMSGATE, Kent, CT12 5FN.	1 November 2023	(4430468)
MORRIS, Brian Albert	35 Merton Drive Weston Village Weston-super-Mare, BS24 7EQ. 8 February 2023	Wards Solicitors, 195/197 High Street, Worle, Weston-super-Mare BS22 6JS (Jenny Pierce and Thomas William Scoffham.)	1 November 2023	(4428134)
MOSLEY, Joyce	Willow House Residential Home, 396 Halifax Road, Liversedge West Yorkshire WF15 6NG. 18 March 2023	Catteralls Solicitors Limited, 15 King Street, Wakefield WF1 2SL	1 November 2023	(4430099)
MOYSEY, Gladys Eileen	9 Noverton Lane, Prestbury, Cheltenham, GL52 5BA. 30 November 2022	Harrison Clark Rickerbys, Ellenborough House, Wellington Street, Cheltenham, GL50 1YD	1 November 2023	(4426782)
MUTTER, Janet Christine	11 Hanover Court, Beaminster, Dorset, DT8 3LR. 21 July 2023	Kitson & Trotman, The Champions, Beaminster, DT8 3AN (Ref: TS/ M018172) (Tracy Scammell.)	1 November 2023	(4425936)
NABB, Valerie Irene	Larkhall Springs, Swainswick Gardens, Bath, BA1 6TL, formerly of Milestone, Warminster Road, Monkton Combe, Bath, BA2 7HY, . 2 June 2023	Hopkins Law Ltd, 50/50a High Street, Cowbridge, Vale of Glamorgan, CF71 7AH (Louise Stanlie Brownsey and Elizabeth Frances Louise Hall.)	1 November 2023	(4426993)
NEARN, Norman Frederick	70 Meadow View, Carlisle, CA1 3JL. 29 April 2023	ATKINSON RITSON SOLICITORS, 15 Fisher Street, Carlisle, CA3 8RW (Christine Broad & Christopher Richard Packham.)	1 November 2023	(4428919)

Name of Deceased (Surname first)	Address, description and date of death of Deceased	Names addresses and descriptions of Persons to whom notices of claims are to be given and names, in parentheses, of Personal Representatives	Date before which notice of claims to be given	
NEWCOMBE, Frances Mary	17 Maryfield Avenue, Exeter, Devon, EX4 6JN. 24 July 2022	Trowers & Hamlins LLP, The Senate, Southernhay Gardens, Exeter, EX1 1UG (Roger Newcombe c/o Amanda Rainford.)	1 November 2023	(4428951)
NEWMAN, Maureen Jennie Bernadette	21 Woodhatch Spinney, Coulsdon, Surrey, CR5 2SU. 12 April 2023	Gumersalls Solicitors, Mezzanine Floor, Epsom Square, Epsom, KT19 8AG. (Deborah Simone Valentina Isles and Laura Anne Pawley)	9 November 2023	(4430229)
NEWTON, Vera	Conway, Vicarage Road, Ambleside, Cumbria, LA22 9DH. 17 April 2022	Milne Moser Solicitors, 100 Highgate Kendal Cumbria, LA9 4HE (Paul Graham Williams.)	1 November 2023	(4429047)
NICHOL, Jean Margaret	33 Kingsclere Road, Bicester, OX26 2HJ. 14 January 2023	RACHEL STAFFORD LEGAL SERVICES, 22 Whittle Road, Thame, Thame, Oxfordshire, GB, OX9 3PF (Rachel Louise Stafford.)	1 November 2023	(4427509)
NORTON, Sally Elizabeth	Flat 22 Apple Wharf, The Plains, Totnes, Devon, TQ9 5QL. 19 April 2023	BOYCE HATTON, 58 The Terrace, Torquay, GB, TQ1 1DE (Ref: KLS/ NOR93/2) (Alexander Norton.)	1 November 2023	(4429049)
ODDY, Paul Anthony	1- 6 Boxhill Close, Honicknowle, PLYMOUTH, Devon, England, PL5 3QB. 13 March 2023	Paula Yearling, 34 Oak Hill, Hollesley, WOODBRIDGE, IP12 3JY.	1 November 2023	(4430542)
OWEN, Miss Mandy Mary (Mandy Owen, Mandy Mary Allen, MANDY M OWEN, MANDY M ALLEN)	8 Camino Road, Birmingham, West Midlands, B32 3XE. 11 July 2023	Chris Ellison c/o Irwin Mitchell LLP, Riverside East, 2 Millsands, Sheffield, S3 8DT.	1 November 2023	(4432009)
PACEY, Barbara	92 Dewe Road, BRIGHTON, BN2 4BD. 20 August 2022	Lisa Jane Mansell, The London Gazette (35364), PO Box 3584, Norwich, NR7 7WD.	1 October 2023	(4430442)
PARKE, Mr Raymond Walter (Mr Raymond Parke)	2 Tennyson Rise, EAST GRINSTEAD, West Sussex, RH19 1SQ. 5 June 2023	Mayo Wynne Baxter LLP, Fiona Dodd , 3 Bell Lane, LEWES, BN7 1JU.	1 November 2023	(4431003)
PARRISH, Annie Mary	190 The Roundabout, Birmingham, B31 2TX. 26 April 2023	Star Legal Limited, Strand House, 70 The Strand, Bromsgrove Worcestershire, B61 8DQ (Margaret Yvonne Dutton and Nancy Elizabeth Hatenboer.)	1 November 2023	(4427740)
PARSONS, Jean Margaret	10 Hillside Gardens, Brockham, Betchworth, Surey, RH3 7EW. 27 September 2022	DOWNS SOLICITORS, 156 High Street, Dorking, GB, RH4 1BQ (Amber Lea O'Connor.)	1 November 2023	(4429031)
PAYNE, Richard Gordon	Flat 2, Room 2, Purbeck Lodge, 15 Bonnets Lane, WAREHAM, Dorset, BH20 4HA. 7 November 2022	Athena Probate LTD, Sophie Blanchard, 1st Floor, 34 The Oaks, Manston Business Park, RAMSGATE, Kent, CT12 5FN.	1 November 2023	(4431974)
PEARCE, Alan Derek	13 Two Stones Crescent, Kenfig Hill, Bridgend, CF33 6DZ. 23 January 2022	Chattertons, 30 Avenue Road, Grantham Lincolnshire, NG31 6TH (Jennifer Anne Pearce.)	1 November 2023	(4428144)
PEPPER, Marilyn	Everdale Grange, 78-80 Lutterworth Road, LEICESTER, Leicestershire, LE2 8PG. 5 November 2021	Athena Probate LTD, Sophie Blanchard, 1st Floor, 34 The Oaks, Manston Business Park, RAMSGATE, Kent, CT12 5FN.	1 November 2023	(4431984)
PIESLAK, Janet Augusta Elizabeth	Limetree Care Home, 8 Limetree Close, LONDON, SW2 3EN28-29 Oakdale Road, London, SW16 2HJ. 5 April 2023	Michael Stephen Pieslak, C/O Treethorpe Limited, 9 Caxton House, Broad Street, CAMBRIDGE, CB23 6JN. (David Andrew Anderson, )	1 November 2023	(4432069)
PITMAN, Hubert Cecil	Coppins, 11B Lees Heights, Chalbury, Chipping Norton, Oxfordshire, OX7 3EZ. 25 July 2023	Lloyds Bank Plc, PO Box 5005, Lancing, GB, BN99 8AZ (Vanessa Berwick.)	1 November 2023	(4428260)
PLAYFORD, Terence	3 Copley Gardens, Weston-Super- Mare, BS22 6HX. 6 February 2022	Premier Solicitors, Premier House, Lurke Street, Bedford, Bedfordshire, GB, MK40 3HU (Sheila Hammond.)	1 November 2023	(4427404)

Name of Deceased (Surname first)	Address, description and date of death of Deceased	Names addresses and descriptions of Persons to whom notices of claims are to be given and names, in parentheses, of Personal Representatives	Date before which notice of claims to be given	
POWELL, David	OWELL, David 22 Bruce Crescent, Hartlepool, TS24 Tilly Bailey & Irvine LLP, Castle 9HR. 12 July 2022 House, 11 Bridge Road, Stockton on Tees TS18 3AD (Ref: JT/157278.2) (Carol Robinson and Janet Cooper.)		1 November 2023	(4426893)
PROTHEROE, Mrs Jean Mary (Jean Mary Hughes)	56 South View, Kenfig Hill, Bridgend, CF33 6DGPicton Court Care Home, 200 West Road, Nottage, Porthcawl, CF36 3RT. 20 October 2021	Michelmores Trust Corporation Limited, Woodwater House, Pynes Hill, EXETER, EX2 5WR.	1 November 2023	(4431960)
PROTHEROE, Mr Ronald Thomas	56 South View, Kenfig Hill, Bridgend, CF33 6DG. 9 April 2021	Michelmores Trust Corporation Limited, Woodwater House, Pynes Hill, Exeter, EX2 5WR.	1 November 2023	(4431999)
PURDUE-SMITH, Peter John	8 Chester Crescent, Cirencester, Gloucestershire, GL7 1HE. 12 June 2022	Red Kite Law LLP, 10/12 Dollar Street, Cirencester, Gloucestershire, GL7 2AL (Christopher Mills and Sheila Telling .)	1 November 2023	(4428124)
PYKETT, Maureen Ann	28 Money Hill Road, RICKMANSWORTH, WD3 7EQ. 25 July 2023	Valued Estates Ltd, 7 Church Barns, Ware Road, Widford, Hertfordshire, United Kingdom, SG12 8RL.	1 November 2023	(4431972)
QUICKFALL, Margaret Rose	37 Hathersage Avenue, North Hykeham, LN6 8LJ. 12 August 2023	Bridge McFarland LLP, Sibthorp House, 351-355 High Street	1 November 2023	(4432019)
RENN, Mr Michael John	Uppsala, Sweden. 17 March 2023	Artillerigatan 6B, Uppsala, Sweden 755 91; Telephone number: 0046 706955756; Email address: gunilla@renn.se. (Gunilla Renn.)	1 November 2023	(4430098)
ROBERTS, Colin	50 Norrington Road, Northfield, Birmingham, B31 5PH. 29 September 2022	TAYLOR FAWCETT, 14 Albert Street, Harrogate, GB, HG1 1JT	1 November 2023	(4427686)
ROBINSON, Millicent Rose	19 Francis Street Pontardawe Abertawe, SA8 4EQ. 29 May 2021	Harrison Clark Rickerbys Limited, Hodge House, 114-116 St Mary Street Cardiff CF10 1DY	1 November 2023	(4431669)
ROBINSON, Jean Marion	Formerly of 15 Cleet Court, Berwick upon Tweed, Northumberland, TD15 1HH. 15 April 2023	Sanderson McCreath & Edney, 4 Quay Walls, Berwick upon Tweed, TD15 1HD	1 November 2023	(4428156)
ROBINSON, Mrs Jessie (Jessie Page)	10 Ravenswood, Northwich, Cheshire, CW8 1NW. 27 June 2023	Jessica Forrester c/o Hugh James, Two Central Square, Cardiff, CF10 1FS.	1 November 2023	(4432008)
ROUTLEY, Kathleen	21 Briarfield Gardens, Horndean, Waterlooville, Hampshire, PO8 9HX. 23 July 2023	LARCOMES LEGAL LIMITED, Appletree House, 17 St Georges Walk, Waterlooville, Hampshire, PO7 7TU (Lisa Jane Moore.)	1 November 2023	(4427529)
SAYER, Mr Philip	46 Lindley Road, WALTON-ON- THAMES, KT12 3HA. 21 January 2022	Guy Peter Ellison Curry, 1 The Crescent, TAUNTON, TA1 4EA.	1 November 2023	(4431620)
SCARISBRICK, Ronald Joseph	Hethersett Hall Care Home, Norwich Road, Hethersett, Norwich, NR9 3AP. 26 July 2023	Heritage Limited, 13 Hellesdon Park Road, Drayton High Road, Norwich, Norfolk NR6 5DR (Ref: P2448) (Mr Stuart Jan Burrell.)	1 November 2023	(4428475)
SCHOFIELD, David Howard	Yarlside Cottage, Bracewell, Skipton, BD23 3JU. 11 June 2022	Hartlaw LLP, 63 St James Street, Wetherby, LS22 6RS (Ref: BH/ 16400-5/Schofield) (Howard Duncan Pask.)	1 November 2023	(4426998)
SCOTT, Anthony Edward	Carolyne House, Waterson Road Grays, RM16 4LD. 27 February 2023	Co-op Legal Services Limited, Aztec 650, Aztec West, Almondsbury, Bristol, BS32 4SD (Ref: CW / 6341659P / Scott) (The Co-operative Trust Corporation.)	1 November 2023	(4427503)

Name of Deceased (Surname first)	Address, description and date of death of Deceased	Names addresses and descriptions Date before which of Persons to whom notices of notice of claims to be claims are to be given and names, in parentheses, of Personal Representatives		
SHAW, Thomas Francis	Rose Cottage, Brancepeth Court,, Hall Lane, Crook, DL15 0PW1 Sycamore Grove, Willington, Crook, DL15 0BWBRANCEPETH COURT, 1A LINCOLN DRIVE, WILLINGTON, Crook, DL15 0PR. 17 May 2023	John Shaw, C/O Treethorpe Limited, 45 Beaufort Court, Admirals Way, London, E14 9XL.	1 November 2023	(4432078)
SHIELD, Margaret June (Barrow)	16 Nuns Avenue, Milnthorpe, LA7 7JX. 8 May 2023	RATCLIFFE & BIBBY SOLICITORS, The Bank, 44 Market Street, Carnforth, GB, LA5 9JX	1 November 2023	(4427347)
SIMMONDS, Mr Sidney Thomas Ivor (Sidney Simmonds)	57 Mostyn Road, Stourport-on- Severn, Worcestershire, DY13 8PW. 10 July 2023	Zoe Perry c/o Thursfields Legal Limited, 14 Church Street, Kidderminster, Worcestershire, DY10 2AH.	31 October 2023	(4430868)
SIMPSON, Richard (Dick)	13 Hill Crescent, Finstock, Chipping Norton, Oxfordshire, OX7 3BT. 4 July 2023	Helen Simpson, 31 School Road, Finstock, Chipping Norton, Oxfordshire OX7 3DN (Helen Simpson, Kevin Simpson.)	1 November 2023	(4428463)
SIMPSON, Malcolm	54 Castle Street, Skipton, North Yorkshire, BD23 2DJ. 25 May 2023	Walker Foster, 3 High Street, Skipton, North Yorkshire, BD23 1AA (Maxine Susan Heppenstall.)	1 November 2023	(4426169)
SKUES, Albert Edward Nicholas ( (Nicholas Skues))	27 Sunstar Lane, Polegate, East Sussex, BN26 5HS. Physics Teacher (Retired). 5 October 2022	Hart Reade LLP, Solicitors, 104 South Street, Eastbourne, East Sussex, BN21 4LW. Ref: HDS.LH.Skues.129560/0001.	9 November 2023	(4430267)
SMALDON, Michael Dowell	5 Hillcrest Road, Hayle, Cornwall, TR27 4NR. 27 November 2022	CVC Solicitors, Unit N Questmap Business Park, Longrock, Penzance, TR20 8AS (Ref: CH/SMA81/1) (Michael Timothy Smaldon.)	1 November 2023	(4426126)
SMITH, Francis Geoffrey	46 Galsworthy Road Goring By Sea Worthing, BN12 6LW. 21 March 2022	Hodges and Lings Law Limited t/a Lings Solicitors, 110 George V Avenue Worthing, BN11 5RR (Philip Hodges.)	1 November 2023	(4428430)
SMITH, Peter	Brooklands House, Bloxham Road, Milton, Banbury, Oxfordshire . 14 April 2022	SE Solicitors, 52-54 The Green, South Bar Street, Banbury, OX16 9AB (Lucy Gordon.)	1 November 2023	(4428001)
SNEDDEN, Robert (Jock Snedden)	17 Alton Road, Leicester, Leicestershire, LE2 8QB. 30 November 2022	Josiah Hincks Solicitors, 33 Leicester Road, Blaby, LE8 4GR (Ronald Andrew Eagle and James Andrew Stuart Snedden .)	1 November 2023	(4428142)
SOOLEY, Mr Terence Neil	16 Burney Road, KINGS LYNN, PE30 5LD. Technical Draughtsman (retired). 16 August 2023	Ward Gethin Archer Solicitors, 3 Regis Place, North Lynn Industrial Estate, KINGS LYNN, PE30 2JN.	1 November 2023	(4430932)
SPINK, Miss Frances Lesley	18 Dorset Mount, LEEDS, LS8 3QX. Housewife. 3 August 2023	Christine Whyte, 21 Coronation Road, Sharlston Common, WAKEFIELD, WF4 1AH.	1 November 2023	(4430870)
STANLEY, Beryl Jean	4 Windmill Park, Windmill Lane, Balsall Common, Coventry, CV7 7GZ. 5 February 2023	Probate Solicitors Limited, 1705 High Street, Knowle, Solihull, B93 0LN (Probate Solicitors Limited.)	1 November 2023	(4425579)
STEVENSON, Paul Miller	8 Alandale Road, Teignmouth, TQ14 8NW. 24 February 2023	Legal And Contract Services Ltd, 1A Millenniu Building, Unit 5 Dawlish Businss Park, Dawlish, EX7 0NH (Michael Graham Bartlett And Wendy Anne Bartlett.)	1 November 2023	(4427441)
STONE, Stephen Ray	The Pavillion, Buckland in the Moor, Ashburton, Newton Abbot, Devon, TQ13 7HN. 6 August 2022	Ashfords LLP, Ashford House, Grenadier Road, Exeter EX1 3LH (Ref: VME/399411-1) (Dawn Sandra Stone and Timothy Ralph Heal.)	1 November 2023	(4427519)
SWANBOROUGH, Robert Charles	Appletree Court Care Home, 158 Burnt Oak Broadway, Edgware, Middx, HA8 0AX. 7 August 2023	Boyes Sutton & Perry, 20 Wood street, Barnet, Herts, EN5 4BJ (Ref: FT/Swanborough) (Frances Helen Thompson and Mamode Yousouf Ali Dookhun.)	1 November 2023	(4426725)

Name of Deceased (Surname first)	Address, description and date of death of Deceased	Names addresses and descriptions of Persons to whom notices of claims are to be given and names, in parentheses, of Personal Representatives	Date before which notice of claims to be given	
THOMAS, Ronald Charles	11 Netley Gardens, MORDEN, SM4 6JW. 16 June 2023	Valued Estates Ltd, 7 Church Barns, Ware Road, Widford, Hertfordshire, United Kingdom, SG12 8RL.	1 November 2023	(4431961)
THOMPSON, Derek	80 Hollow Road, Anstey Leicestershire, LE7 7FS. 2 January 2023	SHAKESPEARE MARTINEAU LLP, One Colton Square, Leicester, GB, LE1 1QH	1 November 2023	(4429112)
THORNTON, Donald Edgar	Westacre Nursing Home, Sleepers Hill, Winchester, SO22 4NE. 22 March 2023	Gardner Leader LLP, Whitehart House, Market PI, Newbury, RG14 5BA (Rosemary Thornton.)	1 November 2023	(4428263)
THROUP, Heather	142 Keighley Road, Skipton, North Yorkshire, BD23 2QT. 30 January 2023	Walker Foster, 3 High Street, Skipton, North Yorkshire, BD23 1AA (Maxine Susan Heppenstall.)	1 November 2023	(4427499)
TIBBLE, Mrs Irene Ann	64 Eden Close, SLOUGH, SL3 8TZ. 12 August 2023	Ward Gethin Archer Solicitors, 3 Regis Place, North Lynn Industrial Estate, KINGS LYNN, PE30 2JN.	1 November 2023	(4431352)
TOMKINS, Alan	15 Ham Island, Old Windsor, WINDSOR, SL4 2JY. 17 June 2022	L,A,S Tomkins, The London Gazette (35387), PO Box 3584, Norwich, NR7 7WD.	1 November 2023	(4431641)
TOWNSEND, Kathleen Marie	84 Moss Lane, Burscough, Lancashire, L40 4AX. 16 January 2020	Dickinson Parker Hill, 22 Derby Street, Ormskirk, Merseyside, GB, L39 2BZ	1 November 2023	(4428247)
TRENDLER, Richard Sidney	572 Riverside Island Marina, Fen Bank, Isleham, Ely, Cambridgeshire, CB7 5SL formerly of 42 Harwood Court, Colville Estate, London, N1 5NF. 3 July 2022	Porter Dodson LLP, 15 High Street, Wellington Somerset, TA21 8QR (Victor Alan Trendler.)	1 November 2023	(4428158)
TURNER, Jerrold Beavis	3 Beacon View, Embsay, Skipton, North Yorkshire, BD23 6SA. 28 April 2023	Walker Foster, 3 High Street, Skipton, North Yorkshire, BD23 1AA (Maxine Susan Heppenstall.)	1 November 2023	(4427366)
VAN DER WAAG, Joy Monica	Rosedale Court, 6 Homestead Close, Rayleigh, Essex, SS6 8FE. 7 January 2023	Jefferies Essex LLP, 3rd Floor, Dencora Court, Tylers Avenue, Southend-on-Sea, Essex SS1 2BB (Patrick Carl Van Der Waag.)	1 November 2023	(4428963)
VANDERVELDE, Martin Brian	Kingsmead Nursing Home, Prospect Place, Old Town, Swindon previously of 4 Lakeside The Lawns Swindon, Wiltshire, SN3 1QE. 12 April 2023	RWK Goodman LLP, 3 Newbridge Square, Swindon, Wiltshire, SN1 1BY	1 November 2023	(4429163)
VAUGHAN, Neville Nigel	13 Travis Grove, Bletchley Milton Keynes, MK3 6EJ. 21 July 2022	Dominic Mills & Co, 11 Grove Place Bedford, MK40 3JJ ( Lesley Gillian Power, Lindsay Jane Shepherd and David Nigel Vaughan.)	1 November 2023	(4427445)
WAIGHT, Mr Alan John	24 Nordik Gardens, Hedge End, Southampton, SO30 0LQ. Retired. 17 November 2022	Leslie Waight, 13 Silver Glades, YATELEY, GU46 6DN.	1 November 2023	(4430893)
WALECKA, Nancy	1 Peasborough View, Burley in Wharfedale, Ilkley, LS29 7RD. 31 May 2023	Ison Harrison Solicitors, 65 Victoria Road, Guiseley, Leeds, LS20 8DQ (Dominic Peter Stewart Mackenzie & Jonathan Leslie Mark Wearing.)	1 November 2023	(4425924)
WATERS, Andrew Neil	1 Mill Gardens, East Wittering West Sussex, PO20 8PR. 13 July 2023	The Owen Kenny Partnership Limited, Unit 3, Chichester Business Park, City F, Tangmere, Chichester, West Sussex, GB, PO20 2FT (Mark Andrew Waters, Lucy Oughton.)	1 November 2023	(4428152)
WEBB, David Edward	25 Manchester Road Ipswich Suffolk, IP2 9PS. 12 June 2023	Kerseys Solicitors LLP, 32 Lloyds Avenue Ipswich, Suffolk IP1 3HD (Ref: LNM/WEB192/0001) (Jane Elizabeth Riley, Matthew Edward Clemence.)	1 November 2023	(4426788)
WEBBER, Mr Albert Thomas	18 Long Sands, Shoeburyness, SOUTHEND-ON-SEA, SS3 9PN. Pensioner. 11 February 2023	Richard James Sanders, 16 Avon Way, Shoeburyness, SOUTHEND- ON-SEA, SS3 9DZ.	1 November 2023	(4426155)

Name of Deceased (Surname first)	Address, description and date of death of Deceased	Names addresses and descriptions of Persons to whom notices of claims are to be given and names, in parentheses, of Personal Representatives	of Persons to whom notices of notice of claims to be claims are to be given and names, given in parentheses, of Personal	
WHEELDON, Mrs Doris	1 Sycamore Rise, MACCLESFIELD, SK11 8LN. 9 March 2023	Glyn Goodchild, The London Gazette (35376), PO Box 3584, Norwich, NR7 7WD.	1 November 2023	(4430899)
WHITE, Mr William John	34 Whitehall Road, GRAYS, RM17 5NX. 23 January 2023	Thomas Smith, The London Gazette (35380), PO Box 3584, Norwich, NR7 7WD.	1 November 2023	(4431098)
WILLIAMS, Miss Sian Katherine	52 Pryor Road, KETTERING, NN15 7FD. Care Worker. 4 May 2023	Kevin Williams, The London Gazette (35377), PO Box 3584, Norwich, NR7 7WD.	1 November 2023	(4430904)
WILLIS, Thomas Noel	658 Galleywood Road, Chelmsford, Essex, CM2 8BY. 19 February 2023	Kew Law, 1/3 Gate Street, Maldon, CM9 5QW (Ref: ISJ WIL604/1)	1 November 2023	(4426984)
WINSTON, Richard Barrington	40 Lowden Chippenham Wiltshire, SN15 2BJ. 7 July 2023	Forrester Sylvester Mackett, 61 St Mary Street, Chippenham Wiltshire, SN15 3JH	1 November 2023	(4428314)
WOOD, Martin John	Home Farm, Starbotton, Skipton, North Yorkshire BD23 5HY also of Flat 19, Compass Court 39 Shad Thames London, SE1 2NJ. 13 March 2023	Walker Foster, 3 High Street, Skipton, North Yorkshire, BD23 1AA (Clover Jane Mehrlich.)	1 November 2023	(4428265)
WOOLLEY, Gordon	42 Higher Downs, Knutsford, WA16 8AS. 27 January 2023	Weightmans LLP, 100 Old Hall Street, Liverpool, L3 9QJ	1 November 2023	(4428421)
WORMALD, Sheila Mary	12 Woodside Grassington, Skipton North Yorkshire, BD23 5NE. 22 October 2022	Walker Foster, 3 High Street, Skipton North Yorkshire, BD23 1AA (Maxine Susan Heppenstall.)	1 November 2023	(4426376)
YOUNG, Patricia Anne	Cragdale House, Victoria Street, Settle, BD24 9HD. 7 June 2022	Walker Foster Solicitors, First Floor, 7 Victoria Avenue, Harrogate, HG1 1EQ (Robert Young Maxine Susan Heppenstall Stuart William Rowland .)	1 November 2023	(4427431)
YOUNG, David John	14 North Meadow, Peasedown St John, Bath, BA2 8PS. 20 July 2022	RWK Goodman, 5-6 Northumberland Buildings, Queen Square, GB, BA1 2JE (Nichola Joy Walker .)	1 November 2023	(4428155)



# Place a deceased estates notice online

Place a notice in both a local newspaper and online in The Gazette in one easy step, and protect the executor from unknown creditors and beneficiaries. Simply create an account or login to your existing Gazette account and complete the online notice placement form.

Benefits include:

- A cost-effective service
- A quick and easy process
- PO Box forwarding to retain anonymity
- The comfort that you are ensuring due diligence for your client
- A notice which is prominently recorded in the UK's official public record, easily accessible online and nationwide
- Providing the executor with peace of mind knowing they have taken sufficient steps to find any unknown creditors or beneficiaries



To place a notice visit www.thegazette.co.uk/wills-and-probate/place-a-deceased-estates-notice



0266

### Terms and conditions relating to submission of notices

The Gazette (which includes the London, Belfast and Edinburgh Gazette) is an official public record and the United Kingdom's longest continuously published newspaper. It is managed by The National Archives (a non-ministerial government department) under a concessionary contract with The Stationery Office Limited ("TSO" or the "Publisher", as defined below). Any capitalised terms referred to in these terms and conditions relating to submission of notices are defined below.

By placing a Notice in The Gazette you are consenting to put official information permanently on the public record and in the public domain, online (in The Gazette website or via The Gazette mobile app), in print, and via a data service (rather than by having to search for notices on The Gazette website, customers can either create a pdf of the Notices that they are interested in, or subscribe to an electronic version of The Gazette (in full or in part) which is provided as a data service).

These terms should be read in conjunction with:

1 The Publisher's privacy policy

2 The Publisher's policies relating to submission of notice; and

3 Royal Mail general terms and conditions (applicable to Notices Placers utilising the Forwarding Service)

which (as amended from time to time) together govern the submission of Notices.

Notice Placers, as defined below, may place a Notice in The Gazette either because there is a statutory requirement to do so, or to do so voluntarily to put information in The Gazette in order to create an official record of fact. All Notice Placers must have the authority to place the notice that they submit for publishing. TSO, as the Publisher, is required to verify the authority of Notice Placers who place Notices and has the authority to refuse to publish Notices from Notice Placers whose authority cannot be effectively verified.

Notices received for publication usually fall under the following broad headings:

Church, Companies, Environment and Infrastructure, Health and Medicine, Honours and awards, Money, Parliament and Assemblies, People, Royal Family and State. Further information can be found at <u>www.thegazette.co.uk</u>.

These terms and conditions ("Terms and Conditions") govern submission of Notices (as defined below) to The Gazette. By submitting Notices, howsoever communicated, whether at the website <u>www.thegazette.co.uk</u> (the "Website") or by email, post and/or facsimile, the Notice Placer (as defined below) agrees to be bound by these Terms and Conditions. Where the Notice Placer is acting as an agent or as a representative of a principal, the Notice Placer warrants that the principal agrees to be bound by these Terms and Conditions.

The Publisher reserves the right to modify these Terms and Conditions at any time. Such modifications shall be effective immediately upon publication. By submitting Notices to The Gazette after the Publisher has published such modifications, the Notice Placer, including any principal, agrees to be bound by the revised Terms and Conditions.

For the avoidance of doubt the Royal Mail's terms and conditions above will be read subject to the terms and conditions of this Agreement and the Publisher's own terms referenced above will take precedence. The Publisher is not liable to the Notice Placer for the availability, access and/or any accuracy of any information placed on any third-party website.

### 1 Definitions

1.1 In these Terms and Conditions:

"Authorised Scale of Charges" means the scale of charges set out at in the printed copy of the Gazette or at <u>www.thegazette.co.uk/place-notice/pricing</u>, as modified from time to time;

"Charges" means the payment due for the acceptance of a Notice by the Publisher payable by the Notice Placer as set out in the Authorised Scale of Charges;

**"Forwarding Service"** means the postal service provided indirectly via The Royal Mail, in order to use The Gazette's postal box for correspondence in order to prevent a personal address from being publicly and permanently available on the official public record;

**"Local Newspaper Notice"** means any notice placed in a local newspaper other than The Gazette;

"Notice" means all advertisements and state, public, legal or other

notices (without limitation) submitted for potential publication in The Gazette by the Notice Placer, save in respect of any Local Newspaper Notice, to which other terms may apply where indicated in these Terms and Conditions;

"Notice Placer" means any agency, company, firm, organisation or person who has requested to place a Notice in The Gazette, whether acting on their own account or as agent or representative of a principal;

**"Publisher"** means The Stationery Office Limited and or TSO, with registered company number 03049649, acting in accordance with the concessionary contract awarded by The National Archives.

- "Royal Mail" means the Royal Mail Group Limited.
- 1.2 the singular includes the plural and vice-versa; and

1.3 any reference to any legislative provision shall be deemed to include any subsequent re-enactment or amending provision.

2 By submitting a Notice to the Publisher, the Notice Placer agrees to be bound by these Terms and Conditions which, unless stated otherwise in these Terms and Conditions, represent the entire terms agreed between the parties in relation to the publication of Notices in The Gazette and which every Notice shall be subject to. For the avoidance of doubt, these Terms and Conditions shall prevail over any other terms or conditions (whether or not inconsistent with these Terms and Conditions) contained or referred to in any correspondence or documentation submitted by the Notice Placer or implied by custom, practice or course of dealing which the parties agree shall not apply, unless otherwise expressly agreed in writing by the Publisher.

3 The Publisher reserves the right, to be exercised at its sole and absolute discretion, to make reasonable efforts to verify the validity of the Notice Placer.

4 The Publisher may, at its sole and absolute discretion edit the Notice, subject to the following restrictions:

4.1 the sense of the Notice submitted by the Notice Placer will not be altered;

4.2 Notices shall be edited for house style only, not for content;

4.3 Notices can be edited to remove obvious duplications of information;

4.4 Notices can be edited to re-position material for style;

4.5 any additions, amendments or deletions required in order to include the minimum necessary information set out in any Notice guidelines shall be confirmed with the Notice Placer; and

4.6 subject to clause 5 below, no amendments to the text (other than those made as a consequence of 4(i) - (v) above) shall be made without confirmation from the Notice Placer.

For the avoidance of doubt, the Notice Placer agrees and accepts that, subject to the limited rights to edit any Notice referred to above, it is the Notice Placer that shall be solely responsible for the content of any Notice, including its validity and accuracy and that the Publisher shall not be responsible for, nor shall have any liability in respect of such content in any way whatsoever.

5 The Notice Placer accepts that it submits a Notice entirely at its own risk and that the Publisher shall have the sole and absolute discretion whether to accept a Notice for publication; whether to publish it (including after acceptance); the timing of any publication of a Notice; or whether to remove or withdraw the Notice after publication, such decision to be final and without liability on the part of the Publisher. The Notice Placer must satisfy itself as to the legal, statutory and/or procedural requirements and accuracy relating to any Notice and, for the avoidance of doubt, the Publisher shall have the sole and absolute discretion to refuse to publish or withdraw from publication (if already published) any Notices where the content of the Notice, in the Publisher's sole and absolute opinion, may not comply with any such requirements. In instances where publication has not yet taken place, the Publisher shall use all reasonable endeavours to notify the Notice Placer of any action required to remedy any deficiency and publication shall not take place until the Publisher is reasonably satisfied that such action has been taken by the Notice Placer. Where publication has taken place the Notice placer will be sent communication with the proposed remedy which may include, but is not limited to, removal, reinsertion, retraction or substitution notice. The Notice Placer agrees and accepts that the mutual obligations and undertakings under this Agreement are sufficient consideration for the enforceability of these terms and conditions which the Notice Placer agrees are fair and reasonable.

6 Save for any liability that cannot be excluded or restricted by law, The National Archives or the Publisher's (including any successor organisations, affiliates, officers, directors, agents, subcontractors and/or employees) total aggregate liability whether arising from the acts and/or omissions of The National Archives or the Publisher arising out of or made in connection with any Notice or otherwise for any and all liabilities, losses, damages, expenses, costs (including all interest, penalties, legal costs and other professional costs and/or expenses), whether in contract (including under an indemnity or warranty), tort (including negligence), misrepresentation, equity, breach of statutory duty, strict liability or otherwise incurred shall be limited to one hundred and fifty per cent (150%) of the value of the Charge paid for such Notice under these Terms and Conditions.

6A Other than as set out in clause 6 above, neither the Publisher, nor The National Archives, shall have any liability in respect of any act and/or omission of the Notice Placer and/or any third party or in respect of any Notice submitted by any Notice Placer for potential publication in The Gazette, which the Notice Placer warrants and accepts is solely its responsibility.

7 For the avoidance of doubt, subject to clause 6 above and save for the payment of the Charges, in no circumstances shall either party be liable for any losses including, without limitation, loss of revenues, profits, contracts, business or savings or anticipated savings, any loss of goodwill or reputation, or any special, indirect or consequential damages (however arising, including negligence).

8 Where the Publisher is responsible for a failure to publish, or has published a Notice in error or with an error, the Publisher shall, at no charge to the Notice Placer, either publish the Notice at the next suitable opportunity, or in the event of an error, remove the Notice or publish a reinsertion, retraction or substitution Notice as appropriate at the next suitable opportunity. Both parties agree (including on behalf of any principal, if applicable) that this shall be the sole remedy of the Notice Placer (including any principal, if applicable) and the full extent of the limit of the Publisher's liability in these circumstances.

9 In the event that the Publisher believes, in its sole opinion, a Notice Placer is submitting Notices in bad faith, is in breach of clause 11 below, or has dealings with Notice Placers who are in breach of these Terms and Conditions or has breached such Terms and Conditions previously, the Publisher may require further verification of information to be provided by the Notice Placer and may, at its sole and absolute discretion, delay publication of those Notices until it is satisfied that the Notice it has received is based on authentic information.

10 The location of the Notice in The Gazette shall be at the discretion of the Publisher. For the avoidance of doubt, the Notice shall be published in the house style of The Gazette.

11 The Notice Placer warrants and undertakes to the Publisher:

11.1 that it has the (legal) right, power and authority to submit the Notice;

11.2 the Notice is true and accurate in all respects and does not mislead or contain potentially fraudulent information;

11.3 the Notice is submitted in good faith, does not contravene any law (statutory or otherwise), nor is it in any way illegal, defamatory or an infringement of any other party's rights or of any applicable advertising regulations, guidelines or codes of practice, nor is it subject to any court order prohibiting such publication.

12 To the extent permissible by law the Publisher excludes all warranties, conditions or other terms, whether implied by statute or otherwise, relating to the placing of any Notices.

The Notice Placer agrees and accepts that the Publisher is reliant upon the services of the Royal Mail in order to deliver the Forwarding Service and accordingly any failure and/or inability of the Royal Mail to deliver the Forwarding Service shall in so far as reasonably possible not put the Publisher to be in breach of this Agreement unless the Publisher was the sole contributory to the breach and/or negligent on its part to properly supervise the Royal Mail in relation to the Services. 13 The Notice Placer agrees to fully indemnify (as a debt) and hold the Publisher and The National Archives or any successor organisation, including any affiliates, officers, directors, agents, subcontractors and employees harmless from all liabilities, costs, expenses, damages and losses (including, without limitation) any direct, indirect, consequential and/or special losses and/or damage, loss of profit, loss of reputation and/or goodwill and all interest, penalties and legal costs (calculated on a full indemnity basis) and all other professional costs and/or expenses (including legal costs) suffered or incurred (including negligence) in respect of any matter arising out of, in connection with or relating to any Notice (including, without limitation, clause 11 of the Terms and Conditions) and

including (without limitation) in respect of any claim and/or demand (including threatened and/or potential claims or demands) made by any third party which may constitute a breach and/or potential breach by the Notice Placer (or their principal) of these Terms and Conditions or of any law and/or any of the rights of a third party. The Publisher shall consult with the Notice Placer as to the way in which such applicable claims, demands or potential claims or demands are handled but the Publisher shall retain the sole, absolute and final decision on all aspects of any matter arising from the aforementioned indemnity, including the choice of instructing legal representatives, steps taken in or related litigation and/or decisions to settle the case. The Notice Placer shall use best endeavours to provide, at its own expense, such co-operation and assistance as the Publisher may reasonably request including in respect of any principal (if applicable) and including, without limitation, the provision of and/or access to witnesses, access to premises and delivery up of documents and/or any evidence, including supporting any associated litigation and/or dispute resolution process.

14 The Notice Placer shall promptly notify the Publisher in writing of any actual, threatened or suspected claim made by a third party or parties against the Notice Placer and/or the Publisher in relation to a Notice. The Publisher reserves the right, following a claim or threatened claim, to immediately remove the Notice which is the subject of the complaint from the website at www.thegazette.co.uk and all other websites controlled by the Publisher containing the Notice, as well as from any other medium in which the Notice has been placed that is controlled by The Gazette, where possible. The Publisher may (at its sole and absolute discretion) require the Notice Placer to amend the Notice at its own cost before it agrees to republish the Notice if it is capable of rectification to avoid the claim, threatened or suspected claim. Any reinstatement of the Notice shall be at the sole and absolute discretion of the Publisher, whose decision in respect of such matter shall be final.

In exceptional circumstances, for example if the Notice was found by the Publisher to have been submitted by an unauthorised Notice Placer, the Publisher (upon receiving written approval from The National Archives) will remove a Notice in its entirety from www.thegazette.co.uk, but will retain the Notice identification online and include explanatory text saying why and when the Notice was removed.

15 The Notice Placer acknowledges that the Publisher may re-use Notices and/or allow third parties to re-use Notices accepted for publication in The Gazette for ongoing Gazette-related activity – for example to create a company profile page on The Gazette website and where you have also opted for a newspaper advertisement as well as a Gazette notice – and hereby assigns to the Publisher for and on behalf of the Crown, all rights, including but not limited to, copyright and/or other such intellectual property rights (as applicable) in all Notices, and warrants that any such activity in respect of any Notice (including any activity in the preparation of such Notice for publication in The Gazette) by the Publisher and/or third parties does not and will not infringe any legal right of the Notice Placer or any third party. For the avoidance of doubt, all Notices and any content therein shall be Crown copyright and may be subject to the Open Government Licence (or any variation thereof).

16 The Notice Placer accepts that the purpose of The Gazette is to disseminate information of interest to the public as widely as possible in the public interest. The Notice Placer accepts that the information contained in the Notices published in The Gazette may be used for additional purposes by the Publisher or third parties after publication as stated above and that such use may be beyond the control of The Gazette. In such instances, the Notice Placer agrees that the Publisher shall have no liability whatsoever in respect of such use by the Publisher or third parties. The Gazette will at all times act with confidentiality, discretion and adhere to any legislative requirements.

17 The Notice Placer acknowledges and agrees that the publication of any Notice is subject to any court order and/or direction of the court or such other regulatory and/or enforcement authorities including the Information Commissioner's Office, the police, the Financial Conduct Authority (and such other related regulatory organisations), the Solicitors Regulation Authority and such other authorities as may be applicable (without limitation) and that the Publisher may delay, refuse to publish or withdraw from publication if it has received evidence to that effect and may not publish such notice until it has received written evidence from the court (as the Publisher may reasonably require from time to time) that demonstrates that any previous order and/or direction has been withdrawn and/or is no longer applicable (as the Publisher may reasonably require from time to time) and/or, subject to any statutory and/or applicable laws, The Gazette may share information and/or data related to the Notice and/or the Notice Placer's account related to such authorities and the Notice Placer hereby consents to such disclosure(s).

18 In respect of any Local Newspaper Notice, this clause 18 shall apply. For the avoidance of doubt, all other terms of the Terms and Conditions shall apply to Local Newspaper Notices only to the extent that they do not conflict with the terms set out below. In the event of any conflict, the terms set out in this clause 18 shall prevail:

18.1 The Local Newspaper Notice may be placed in a local newspaper by any subcontractor and/or a third party organisation at the Publisher's sole and absolute discretion and the Notice Placer hereby consents to such use (including any activity that is ancillary and/or reasonably necessary to such use). For the avoidance of doubt, this may include the processing of personal data in accordance with the EU General Data Protection Regulation (Regulation 2016/679), and any legislation which implements, amends, re-enacts or replaces it in England and Wales, Northern Ireland or Scotland ("GDPR"), by the Publisher, any subcontractor and/or third party organisation, together with the local newspaper and related organisations;

18.2 The placement of a Local Newspaper Notice shall be upon the standard terms and conditions of the local newspaper in question in addition to these Terms and Conditions. The Notice Placer expressly agrees to such local newspaper terms and by submitting a Local Newspaper Notice to The Gazette, expressly consents to the Publisher, its subcontractors and/or any applicable third party organisation agreeing to such terms on behalf of the Notice Placer;

18.3 To the extent that such local newspaper and the applicable terms allow, where the Publisher, any subcontractor, any third party acting on behalf of the Publisher and/or the local newspaper is responsible for any error, the Publisher shall arrange for the local newspaper to publish the corrected Local Newspaper Notice at no additional cost to the Notice Placer. Both parties agree (including on behalf of any principal, if applicable) that this shall be the sole remedy of the Notice Placer (including any principal, if applicable) and the full extent of the limit of liability in these circumstances;

In the event that a corrected Local Newspaper Notice is not published for whatever reason, the total aggregate liability of the Publisher and The National Archives, whether direct or indirect, and including (without limitation) all liabilities, losses, damages, expenses, costs (including all interest, penalties, legal costs and/or other professional costs and/or expenses) suffered or incurred, howsoever arising (including negligence), whether arising from the acts and/or omissions of the Publisher, The National Archives and/or the Notice Placer and/or any third party (including, without limitation, any principal of the Notice Placer) or arising out of or made in connection with the Notice or otherwise shall be limited to the value of the Local Newspaper Notice placed through The Gazette except that nothing in these Terms and Conditions shall limit or exclude any liability for fraudulent misrepresentation, or for death or personal injury resulting from the Publisher's or The National Archives' negligence or the negligence of the their agents, subcontractors and/or employees or third parties acting on behalf of the Publisher.

19 In respect of the use of the Forwarding Service, which is subject to Clause 12 (above), The Gazette will replace the Notice Placer or executor's address with The Gazette's postal box address in the Notice, and Local Newspaper Notice if applicable. All correspondence received will be sent via Royal Mail from The Gazette to the Notice Placer or executor (if different).

Please be aware that correspondence received by The Gazette's postal box (subject to the final condition of the mail and delivery of the same), will using reasonable endeavours be opened, scanned, and securely stored. The scans will be retained for as long as the service remains (paid for and) in place and for a period of 6 months beyond that time, so that any copies of claims can be requested and communicated to the executor (for example in circumstances where Royal Mail has failed to deliver a claim) and in using this service you are consenting to the provisions of this clause; The Gazette will at all times act with confidentiality, discretion and adhere to any legislative requirements.

The Forwarding Service will remain in place for a period of 10 months from publication of the Notice in The Gazette, or 10 months from publication of the Notice in a Local Newspaper, in line with the Inheritance (Provision for Family and Dependants) Act 1975. After the 10 months period has lapsed any correspondence received will be posted to the sender where possible and or securely destroyed, and the Notice Placer or executor's name and address details will be removed from the Forwarding Service.

20 The Notice Placer accepts that the Charges may be amended from time to time and will be payable at the rate in force at the time of invoicing unless otherwise agreed by the Publisher in writing. The Charges must be paid in full by the Notice Placer in advance of publication unless other requirements of the Publisher in respect of the payment of such Charges (as determined from time to time) are notified to the Notice Placer.

21 Please note that TSO uses Stripe on its eCommerce sites to process credit or debit card transactions. Goods that are out of stock are charged for immediately and despatched when the goods are in stock. By placing an order you agree to this process.

22 If the Notice Placer wishes to make a complaint, all such complaints shall be submitted in writing to customer.services@thegazette.co.uk

23 Save in respect of The National Archives (or any successor organisation), a person who is not a party to these Terms and Conditions has no right under the Contracts (Rights of Third Parties) Act 1999 to enforce any term of these Terms and Conditions but this does not affect any right or remedy of a party specified in these Terms and Conditions or which exists or is available apart from that Act.

24 These Terms and Conditions and all other express terms of the contract shall be governed and construed in accordance with the laws of England and Wales and the parties hereby submit to the exclusive jurisdiction of the English courts.

All communications on the business of The London Gazette should be addressed to The Gazette, PO Box 3584, Norwich NR7 7WD Telephone: +44 (0)333 200 2434 Email: Iondon@thegazette.co.uk



AUTHORISED SCALE OF CHARGES From 1 January 2023		Public sector placing mandatory notices or state notices		All other advertisers	
	All charges are exclusive of VAT at the prevailing rate, currently $20\%$	XML, webform, Gazette template	Other	XML, webform, Gazette template	Other
	No VAT is payable on printed copies	Ex VAT	Ex VAT	Ex VAT	Ex VAT
	Corporate and Personal Insolvency Notices	£0.00	£24.60	£80.00	£109.20
	(2 - 5 Related Companies/Individuals charged at double the single rate)	£0.00	£49.20	£160.00	£218.40
1	(6 - 10 Related Companies charged at treble the single rate)	£0.00	£73.80	£240.00	£327.60
	[Pursuant to the Insolvency Act 1986, the Insolvency (England and Wales) Rules 2016, Companies (Forms) (Amendment) Regulations 1987 and any subsequent amending legislation]				
2	Deceased Estates Notices Pursuant to s.27 Trustee Act 1925			£80.00	£109.20
	All other Notices - charged by event	£0.00	£24.60	£80.00	£109.20
3	(2 - 5 Related events will be charged at double the single rate)	£0.00	£49.20	£160.00	£218.40
	(6 - 10 Related events will be charged at treble the single rate)	£0.00	£73.80	£240.00	£327.60
	If you are unsure how to price your notice or your notice contains more than 40 events please contact london@thegazette.co.uk				
4	Offline proofing		£44.50		£49.75
5	Late advertisements - accepted after 11.30am, two days prior to publication		£44.50		£49.75
6	Withdrawal of Notices - after 11.30 am, two days prior to publication		£24.60	£80.00	£109.20
7	Other services				
	A brand, logo, map, signature image	£63.45	£63.45	£72.55	£72.55
	Forwarding service for Deceased Estates	£63.45	£63.45	£72.55	£72.55
	Newspaper placement for Deceased Estates (webform and template only)	£220.00		£220.00	
	Redaction of information within a published notice	£216.40	£216.40	£240.70	£240.70
	Reinsertion of notice	£24.60	£24.60	£80.00	£109.20

• A single edition of the printed copy is available to notice placers for £8.00 and non-notice placers for £16.00 (VAT exempt)

• An annual subscription to the printed copy is available to notice placers for £1,984.00 and non-notice placers for £3,967.50(VAT exempt)

• An annual subscription to the pdf copy is available to all customers for £727.50 (VAT exempt)

• A commemorative edition of this issue, or any past issue of The Gazette, is available for £75.00 (VAT exempt)

• Take advantage of The Gazette's research service: £30.00 for an individual/company name

For more information or to purchase a subscription or a commemorative edition, please telephone +44 (0)333 200 2434 or email customer.services@thegazette.co.uk, or visit www.thegazette.co.uk/shop

For more information and pricing for our data service please telephone +44 (0)1603 985949 or email data@thegazette.co.uk



Published by TSO (The Stationery Office), a Williams Lea company, and available from:

### Online www.tsoshop.co.uk

### Mail, Telephone & E-mail

TSO PO BOX 29, Norwich, NR3 1GN Telephone orders/General enquiries: +44 (0)333 202 5070 E-mail: customer.services@tso.co.uk Textphone: +44 (0)333 202 5077



Published and printed in the UK by The Stationery Office Limited under the authority and superintendence of Jeff James, Controller of His Majesty's Stationery Office and King's Printer of Acts of Parliament